# **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION **WASHINGTON, DC 20549**

	FO	RM 10-Q
(Mark one)		
X		JRSUANT TO SECTION 13 OR 15 (d) ES EXCHANGE ACT OF 1934
	For the quarterly period ended:	March 31, 2009
		Or
0		JRSUANT TO SECTION 13 OR 15 (d) ES EXCHANGE ACT OF 1934
For the transition period from:		to
Commission File Number:		001-6064
	ALEXA	NDER'S, INC.
	(Exact name of registr	rant as specified in its charter)
	Delaware	51-0100517
(State or other jurisd	iction of incorporation or organization)	(I.R.S. Employer Identification Number)
210 Route	4 East, Paramus, New Jersey	07652
(Address o	of principal executive offices)	(Zip Code)
	(212	2) 587-8541
	(Registrant's telephone	e number, including area code)

# N/A

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. X Yes O No

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (Section 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). O Yes O No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

O Large Accelerated Filer

X Accelerated Filer

O Non-Accelerated Filer (Do not check if smaller reporting company)

O Smaller Reporting Company

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). O Yes X No

As of March 31, 2009, there were 5,105,936 shares of common stock, par value \$1 per share, outstanding.

# ALEXANDER'S, INC.

# INDEX

Page Number PART I. **Financial Information** Financial Statements: Item 1. Consolidated Balance Sheets (unaudited) as of March 31, 2009 and December 31, 2008 Consolidated Statement of Operations (unaudited) for the Three Months Ended March 31, 2009 and 2008 Consolidated Statements of Changes in Equity (unaudited) for the Three Months Ended March 31, 2009 and 2008 Consolidated Statements of Cash Flows (unaudited) for the Three Months Ended March 31, 2009 and 2008 6 Notes to Consolidated Financial Statements (unaudited) Report of Independent Registered Public Accounting Firm 12 Item 2. Management's Discussion and Analysis of 13 Financial Condition and Results of Operations Item 3. Quantitative and Qualitative Disclosures about Market Risk 20 Controls and Procedures 20 Item 4. PART II. **Other Information** Item 1. Legal Proceedings 21 Item 1A. Risk Factors 21 Item 2. Unregistered Sales of Equity Securities and Use of Proceeds 21 **Defaults Upon Senior Securities** 21 Item 3. Item 4. Submission of Matters to a Vote of Security Holders 21 Item 5. Other Information 21 Item 6. **Exhibits** 21 22 Signatures Exhibit Index 23

# ALEXANDER'S, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(Amounts in thousands, except share and per share amounts)

			cember 31, 2008	
ASSETS				
Real estate, at cost:				
Land	\$	74,974	\$	74,974
Buildings, leaseholds and leasehold improvements		595,556		598,114
Construction in progress		317,556		294,887
Total		988,086		967,975
Accumulated depreciation and amortization		(115,899)		(114,235
Real estate, net		872,187		853,740
Cash and cash equivalents		426,265		515,940
Restricted cash		88,988		5,057
Accounts receivable, net of allowance for doubtful accounts of \$1,845 and \$1,357, respectively		3,249		6,580
Receivable arising from the straight-lining of rents		139,696		137,117
Deferred lease and other property costs, net (including unamortized leasing fees to Vornado of \$37,599 and \$38,698, respectively)		60,072		61,525
Deferred debt issuance costs, net of accumulated amortization of \$12,882 and \$13,120, respectively		12,367		12,910
Other assets		10,131		10,699
TOTAL ASSETS	\$	1,612,955	\$	1,603,568
Debt Accounts payable and accrued expenses Amounts due to Vornado Liability for income taxes and other Liability for stock appreciation rights TOTAL LIABILITIES	\$ 	1,230,737 60,941 44,130 49,460 — 1,385,268	\$ 	1,221,255 51,192 44,086 48,826 57,458 1,422,817
COMMITMENTS AND CONTINGENCIES				
STOCKHOLDERS' EQUITY				
Preferred stock: \$1.00 par value per share; authorized, 3,000,000 shares; issued and outstanding, none		_		_
Common stock: \$1.00 par value per share; authorized, 10,000,000 shares; issued,		E 170		E 170
5,173,450 shares; outstanding 5,105,936 shares and 5,091,590 shares, respectively  Additional capital		5,173 31,501		5,173
Retained earnings		189,785		30,647 143,731
		226,459		179,551
Treasury stock: 67,514 and 81,860 shares, at cost		(375)		179,551 (455
Total Alexander's equity		226,084	_	
Noncontrolling interest in consolidated subsidiary		•		179,096
Total Equity		1,603	_	1,655
• •	ф	227,687	φ	180,751
OTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$	1,612,955	\$	1,603,5

See notes to consolidated financial statements.

# ALEXANDER'S, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED)

(Amounts in thousands, except per share amounts)

	-	Three Months Ended March 31,					
		2009	2	2008			
REVENUES							
Property rentals	\$	36,197	\$	36,033			
Expense reimbursements		16,893		15,733			
Total revenues		53,090	_	51,766			
EXPENSES							
Operating (including fees to Vornado of \$1,307 and \$1,203, respectively)		19,035		17,667			
General and administrative (including a reversal of previously recognized stock appreciation rights ("SARs") expense of \$34,275 in 2009, an accrual of SARs expense of \$625 in 2008,							
and management fees to Vornado of \$540 in each period)		(31,684)		1,846			
Depreciation and amortization		5,717	_	5,601			
Total expenses		(6,932)	_	25,114			
OPERATING INCOME		60,022		26,652			
Interest and other income, net		964		4,416			
Interest and debt expense		(14,896)		(15,681)			
Income before income taxes		46,090		15,387			
Income tax expense of taxable REIT subsidiary		(88)		(502)			
Net income		46,002		14,885			
Less: Net loss attributable to the noncontrolling interest		(52)		(267)			
Net income attributable to Alexander's	\$	46,054	\$	15,152			
Net income per common share – basic	\$ <u></u>	9.04	\$	3.00			
Net income per common share – diluted	\$	9.03	\$	2.98			

See notes to consolidated financial statements.

# ALEXANDER'S, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CHANGES IN EQUITY (UNAUDITED) (Amounts in thousands)

		nmon tock		Additional Capital		Retained Earnings		Treasury Stock	A	lexander's Equity	No	oncontrolling Interest		Total Equity
Balance, December 31, 2007	\$	5,173	\$	27,636	\$	103,014	\$	(720)	\$	135,103	\$	2,323	\$	137,426
Net income (loss)		_		_		15,152		_		15,152		(267)		14,885
Common shares issued under option plan		_		130		_		11		141		_		141
Balance, March 31, 2008	\$	5,173	\$	27,766	\$	118,166	\$	(709)	\$	150,396	\$	2,056	\$	152,452
	_		_		٠		_		_		_			
Balance, December 31, 2008	\$	5,173	\$	30,647	\$	143,731	\$	(455)	\$	179,096	\$	1,655	\$	180,751
Net income (loss)		_		_		46,054		_		46,054		(52)		46,002
Common shares issued under option plan			_	854				80		934	_	<u> </u>	_	934
Balance, March 31, 2009	\$	5,173	\$	31,501	\$	189,785	\$	(375)	\$	226,084	\$	1,603	\$	227,687

# ALEXANDER'S, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

(Amounts in thousands)

**Three Months Ended** March 31, 2009 2008 CASH FLOWS FROM OPERATING ACTIVITIES Net income \$ 46,002 14,885 \$ Adjustments to reconcile net income to net cash provided by operating activities: (34,275)625 Liability for stock appreciation rights Depreciation and amortization (including amortization of debt issuance costs) 6,411 6,263 Straight-lining of rental income (2,579)(3,076)Other non-cash adjustments 1,884 Change in operating assets and liabilities: Accounts payable and accrued expenses 6,904 3,715 Accounts receivable, net 3,331 1,605 FIN 48 income tax liability 649 634 Other assets 244 (527)Amounts due to Vornado 44 631 Payment for stock appreciation rights (22,838)Other liabilities (15)(17)Net cash provided by operating activities 5,762 24,738 **CASH FLOWS FROM INVESTING ACTIVITIES** Construction in progress and real estate additions (21,771)(27,239)(83,931)Restricted cash (3,531)Net cash used in investing activities (105,702)(30,770)**CASH FLOWS FROM FINANCING ACTIVITIES** 91,632 Proceeds from borrowings 25,875 Exercise of stock options 934 141 Debt repayments (82,150)(3,685)Debt issuance costs (151)Net cash provided by financing activities 10,265 22,331 Net (decrease) increase in cash and cash equivalents (89,675)16,299 Cash and cash equivalents at beginning of period 515,940 560,231 Cash and cash equivalents at end of period 426,265 576,530 SUPPLEMENTAL DISCLOSURE OF CASH FLOW INFORMATION Cash payments for interest (of which \$2,270 and \$2,380 have been capitalized) 16,848

See notes to consolidated financial statements.

Cash payments for income taxes

15,056

57

702

# ALEXANDER'S, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### 1. Organization

Alexander's, Inc. (NYSE: ALX) is a real estate investment trust ("REIT"), incorporated in Delaware, engaged in leasing, managing, developing and redeveloping its properties. All references to "we," "us," "our," or "Company" refer to Alexander's, Inc. and its consolidated subsidiaries. We are managed by, and our properties are leased and developed by, Vornado Realty Trust ("Vornado") (NYSE: VNO).

# 2. Basis of Presentation

The accompanying consolidated financial statements are unaudited. In our opinion, all adjustments (which include only normal recurring adjustments) necessary to present fairly the financial position, results of operations and cash flows have been made. Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted in accordance with Article 10 of Regulation S-X and the instructions to Form 10-Q. These consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in our Annual Report on Form 10-K for the year ended December 31, 2008, as filed with the Securities and Exchange Commission. The results of operations for the three months ended March 31, 2009 are not necessarily indicative of the operating results for the full year.

The accompanying consolidated financial statements include our accounts and that of our consolidated subsidiaries. All significant intercompany amounts have been eliminated. The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting periods. Actual results could differ from those estimates.

Certain prior year balances have been reclassified in order to conform to current year presentation as a result of the adoption of Financial Accounting Standards Board ("FASB") Statement No. 160, Noncontrolling Interests in Consolidated Financial Statements – An Amendment of ARB No. 51 ("SFAS 160").

We currently operate in one business segment.

# 3. Recently Issued Accounting Literature

In December 2007, the FASB issued Statement No. 141R, *Business Combinations* ("SFAS 141R"). SFAS 141R broadens the guidance of SFAS 141, extending its applicability to all transactions and other events in which one entity obtains control over one or more other businesses. It also broadens the fair value measurement and recognition of assets acquired, liabilities assumed, and interests transferred as a result of business combinations; and acquisition related costs will generally be expensed rather than included as part of the basis of the acquisition. SFAS 141R expands required disclosures to improve the ability to evaluate the nature and financial effects of business combinations. SFAS 141R became effective for all transactions entered into, on or after January 1, 2009. The adoption of SFAS 141R on January 1 2009, did not have a material effect on our consolidated financial statements.

In December 2007, the FASB issued SFAS 160. SFAS 160 requires a noncontrolling interest in a subsidiary to be reported as equity and the amount of consolidated net income specifically attributable to the noncontrolling interest to be identified in the consolidated financial statements. SFAS 160 also calls for consistency in the manner of reporting changes in the parent's ownership interest and requires fair value measurement of any noncontrolling equity investment retained in a deconsolidation. SFAS 160 became effective on January 1, 2009. The adoption of SFAS 160 on January 1, 2009, resulted in (i) the reclassification of our minority interest in consolidated subsidiary," a component of permanent equity on our consolidated balance sheets, and (ii) the reclassification of minority interest expense to "net income attributable to the noncontrolling interest" on our consolidated statements of operations.

# ALEXANDER'S, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### 4. Relationship with Vornado

At March 31, 2009, Vornado owned 32.4% of our outstanding common stock. We are managed by, and our properties are leased and developed by, Vornado, pursuant to the agreements described below, which expire in March of each year and are automatically renewable.

#### Management and Development Agreements

We pay Vornado an annual management fee equal to the sum of (i) \$3,000,000, (ii) 3% of gross income from the Kings Plaza Regional Shopping Center, (iii) \$0.50 per square foot of the tenant-occupied office and retail space at 731 Lexington Avenue and (iv) \$234,000, escalating at 3% per annum, for managing the common area of 731 Lexington Avenue.

In addition, Vornado is entitled to a development fee of 6% of development costs, as defined, with minimum guaranteed fees of \$750,000 per annum. The development fee for the Rego Park II project (see note 5) is estimated to be approximately \$17,500,000, of which \$2,809,000 has been paid as of March 31, 2009. The balance is due on substantial completion of the construction.

# Leasing Agreements

Vornado also provides us with leasing services for a fee of 3% of rent for the first ten years of a lease term, 2% of rent for the eleventh through the twentieth year of a lease term, and 1% of rent for the twenty-first through thirtieth year of a lease term, subject to the payment of rents by tenants. In the event third-party real estate brokers are used, the fees to Vornado increase by 1% and Vornado is responsible for the fees to the third-party real estate brokers. Vornado is also entitled to a commission upon the sale of any of our assets equal to 3% of gross proceeds, as defined, for asset sales less than \$50,000,000 and 1% of gross proceeds, as defined, for asset sales of \$50,000,000 or more. The total of these amounts is payable in annual installments in an amount not to exceed \$4,000,000, with interest on the unpaid balance at one-year LIBOR plus 1.0% (3.02% at March 31, 2009).

# Other Agreements

We have also entered into agreements with Building Maintenance Services, a wholly owned subsidiary of Vornado, to supervise cleaning, engineering and security services at our Lexington Avenue and Kings Plaza properties for an annual fee of the cost for such services plus 6%.

The following is a summary of fees to Vornado that were incurred under the agreements discussed above.

(Amounts in thousands)	Three Months Ended March 31,					
		2009				
Company management fees	\$	750	\$	750		
Development fees		2,998		1,263		
Leasing fees		484		633		
Property management fees and payments for cleaning, engineering						
and security services		1,097		993		
	\$_	5,329	\$	3,639		

At March 31, 2009, we owed Vornado \$30,063,000 for leasing fees, \$12,539,000 for the earned portion of the Rego II development fee discussed above, and \$1,528,000 for management, property management and cleaning fees.

# ALEXANDER'S, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (CONTINUED) (UNAUDITED)

# 5. Rego Park II Project

We own approximately 6.6 acres of land adjacent to our Rego Park I property in Queens, New York, which comprises the entire square block bounded by the Horace Harding Service Road (of the Long Island Expressway), 97<sup>th</sup> Street, 62<sup>nd</sup> Drive and Junction Boulevard. The development at Rego Park II consists of a 600,000 square foot shopping center on four levels and a parking deck containing approximately 1,400 spaces. Construction is expected to be substantially completed by the end of this year and estimated to cost approximately \$410,000,000, of which \$316,198,000 has been expended as of March 31, 2009. The shopping center will be anchored by a 134,000 square foot Century 21 department store, a 138,000 square foot Home Depot and a 132,000 square foot Kohl's.

There can be no assurance that this project will be completed, completed on time, or completed for the budgeted amount.

# 6. 731 Lexington Avenue

On March 25, 2009, Citibank N.A. completed the assignment of its lease aggregating 176,000 square feet to Bloomberg L.P. Bloomberg L.P. now occupies 99% of the office space at the property.

# 7. Flushing

In February 2009, we sub-leased the Flushing property to a developer for the remainder of our ground lease term.

#### 8. Debt

The following is a summary of our outstanding debt.

		Interest Rate at	Balance at				
(Amounts in thousands)	Maturity	March 31, 2009		arch 31, 2009	December 31, 2008		
First mortgage, secured by the office space at the Lexington Avenue property	Feb. 2014	5.33%	\$	370,960	\$	373,637	
First mortgage, secured by the retail space at the Lexington Avenue property <sup>(1)</sup>	Jul. 2015	4.93%		320,000		320,000	
First mortgage, secured by the Kings Plaza Regional Shopping Center	Jun. 2011	7.46%		198,449		199,537	
\$350,000 construction loan, secured by the Rego Park II Shopping Center	Dec. 2010	1.70%		195,082		181,695	
First mortgage, secured by the Rego Park I Shopping Center	Mar. 2012	0.75%		78,246(2)		78,386	
First mortgage, secured by the Paramus property	Oct. 2011	5.92%		68,000		68,000	
			\$	1,230,737	\$	1,221,255	

- (1) In the event of a substantial casualty, as defined, up to \$75,000 of this loan may become recourse to us.
- (2) On March 10, 2009, we repaid the \$78,246 outstanding balance of the Rego Park I mortgage loan which was scheduled to mature in June 2009 and simultaneously completed a refinancing in the same amount. The new loan bears interest at 75 basis points, is secured by the property and is 100% cash collateralized. The proceeds of the new loan were placed in a non-interest bearing restricted mortgage escrow account. The loan is pre-payable without penalty after June 2009.

# ALEXANDER'S, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (CONTINUED) (UNAUDITED)

# 9. FIN 48 Income Tax Liability

As of March 31, 2009 and December 31, 2008, we had \$48,517,000 and \$47,868,000, respectively, of unrecognized tax benefits that if recognized, would result in non-cash income and reduce our effective tax rate. These amounts, which include \$10,537,000 and \$9,888,000 of accrued interest as of March 31, 2009 and December 31, 2008, respectively, are included as a component of "liability for income taxes and other" on our consolidated balance sheets.

We recognize interest related to the unrecognized tax benefits in "interest and debt expense" in our consolidated statements of operations. In the three months ended March 31, 2009 and 2008, we recognized \$649,000 and \$634,000, respectively, of interest related to unrecognized tax benefits.

As of March 31, 2009, TRS tax returns for the years 2003-2008 and REIT tax returns for the years 2006-2008 remain open to examination by the major taxing jurisdictions to which we are subject.

# 10. Stock Appreciation Rights

Stock appreciation rights ("SARs") are granted at 100% of the market price of our common stock on the date of grant. Compensation expense for each SAR is measured by the excess of stock price at the current balance sheet date over the stock price at the previous balance sheet date. If the stock price is lower at the current balance sheet date, previously recognized expense is reversed but not below zero. On March 2, 2009, Steven Roth, the Chairman of our Board of Directors and Chief Executive Officer and Michael Fascitelli, our President, each exercised 150,000 SARs, which were scheduled to expire on March 4, 2009. Mr. Roth and Mr. Fascitelli each received gross proceeds of \$11,419,000. As a result of the March 2, 2009 exercises, we reversed \$34,275,000 of previously recognized SARs compensation expense in the three months ended March 31, 2009. In the three months ended March 31, 2008, we had accrued \$625,000 for SARs expense based on the March 31, 2008 closing stock price of \$354.50 at March 31, 2008, compared to \$353.25 at December 31, 2007. As of March 31, 2009, there are no SARs outstanding.

# 11. Commitments and Contingencies

#### Insurance

We carry commercial liability with limits of \$200,000,000 per location and all risk property insurance for (i) fire, (ii) flood, (iii) rental loss, (iv) extended coverage, and (v) "acts of terrorism," as defined in the Terrorism Risk Insurance Program Reauthorization Act ("TRIPRA") of 2007, with respect to our assets, with limits of \$1.7 billion per occurrence, for all of our properties. To the extent that we incur losses in excess of our insurance coverage, these losses would be borne by us and could be material.

Our debt instruments, consisting of mortgage loans secured by our properties (which are generally non-recourse to us), contain customary covenants requiring us to maintain insurance. Although we believe that we have adequate insurance coverage for the purposes of these agreements, we may not be able to obtain an equivalent amount of coverage at reasonable costs in the future. Further, if lenders insist on greater coverage than we are able to obtain, it could adversely affect our ability to finance and/or refinance our properties.

# **Environmental Remediation**

In July 2006, we discovered an oil spill at our Kings Plaza Regional Shopping Center. We have notified the New York State Department of Environmental Conservation ("NYSDEC") about the spill and have developed a remediation plan. The NYSDEC has approved a portion of the remediation plan and clean up is ongoing. The estimated costs associated with the clean up will aggregate approximately \$2,500,000. We have paid \$500,000 of such amount and the remainder is covered under our insurance policy.

# ALEXANDER'S, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (CONTINUED) (UNAUDITED)

# 11. Commitments and Contingencies - continued

# Flushing Property

In the fourth quarter of 2003, we recognized \$1,289,000 of income representing a non-refundable deposit of \$1,875,000, net of \$586,000 of costs associated with the transaction, from a party that had agreed to purchase this property, as such party had not met its obligations under a May 30, 2002 purchase contract. On September 10, 2002, November 7, 2002, and July 8, 2004, we received letters from the party demanding return of the deposit. On December 28, 2005, the party filed a complaint against us in the Supreme Court of the State of New York alleging that we failed to honor the terms and conditions of the agreement. The complaint seeks specific performance and, if specific performance is denied, it seeks the return of the deposit plus interest and \$50,000 in costs. In our opinion, after consultation with legal counsel, we do not believe the party is entitled to either specific performance or a return of the deposit and we are defending against the action. Accordingly, we have not recorded a loss contingency for this matter.

# **Paramus**

In 2001 we leased 30.3 acres of land located in Paramus, New Jersey to IKEA Property, Inc. The lease has a 40-year term with a purchase option in 2021 for \$75,000,000. We have a \$68,000,000 interest only, non-recourse mortgage loan on the property from a third party lender. The fixed interest rate on the debt is 5.92% with interest payable monthly until maturity in October 2011. The annual triple-net rent is the sum of \$700,000 plus the amount of debt service on the mortgage loan. If the purchase option is exercised, we will receive net cash proceeds of approximately \$7,000,000 and recognize a gain on sale of land of approximately \$62,000,000. If the purchase option is not exercised, the triple-net rent for the last 20 years must include the debt service sufficient to fully amortize \$68,000,000 over the remaining 20-year lease term.

# Letters of Credit

At March 31, 2009, we had approximately \$7,998,000 of standby letters of credit that were issued and outstanding.

# Other

There are various other legal actions against us in the ordinary course of business. In our opinion, the outcome of such matters in the aggregate will not have a material effect on our financial condition, results of operations or cash flows.

# 12. Income Per Share

The following table sets forth the computation of basic and diluted income per share, including a reconciliation of net income and the number of shares used in computing basic and diluted earning per share. Basic earnings per share is determined using the weighted average shares of common stock outstanding during the period. Diluted earnings per share is determined using the weighted average shares of common stock outstanding during the period and assumes all potentially dilutive securities were converted into common shares at the earliest date possible.

	Three Months Ende March 31,			
(Amounts in thousands, except share and per share amounts)	20	2009 2008		
Net income attributable to common stockholders – Basic and Diluted	\$ <u> </u>	46,054	\$	15,152
Weighted average shares outstanding – Basic		5,097,232		5,044,675
Dilutive effect of stock options		5,594		48,298
Weighted average shares outstanding – Diluted		5,102,826		5,092,973
Net income per common share – Basic	\$	9.04	\$	3.00
Net income per common share – Diluted	\$	9.03	\$	2.98

#### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of Alexander's, Inc. Paramus, New Jersey

We have reviewed the accompanying consolidated balance sheet of Alexander's, Inc. and subsidiaries (the "Company") as of March 31, 2009, and the related consolidated statements of operations, changes in equity and cash flows for the three-month periods ended March 31, 2009 and 2008. These interim financial statements are the responsibility of the Company's management.

We conducted our reviews in accordance with the standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with standards of the Public Company Accounting Oversight Board (United States), the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our reviews, we are not aware of any material modifications that should be made to such consolidated interim financial statements for them to be in conformity with accounting principles generally accepted in the United States of America.

As discussed in Notes 2 and 3 to the consolidated financial statements, in 2009 the Company changed its method of accounting for minority interest to conform to the FASB Statement No. 160, *Noncontrolling Interest in Consolidated Financial Statements*, and, retrospectively, adjusted the 2008 consolidated financial statements for the changes.

We have previously audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet of Alexander's, Inc. and subsidiaries as of December 31, 2008, and the related consolidated statements of operations, stockholders' equity, and cash flows for the year then ended prior to retrospective adjustments for the adoption of FASB Statement No. 160, *Noncontrolling Interests in Consolidated Financial Statements*, (not presented herein); and in our report dated February 23, 2009, we expressed an unqualified opinion on those consolidated financial statements. In our opinion, the information set forth in the accompanying consolidated balance sheet as of December 31, 2008 is fairly stated, in all material respects, in relation to the consolidated balance sheet from which it has been derived.

/s/ DELOITTE & TOUCHE LLP

Parsippany, New Jersey May 4, 2009

# ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Certain statements contained herein constitute forward-looking statements as such term is defined in Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Forward-looking statements are not guarantees of future performance. They involve risks, uncertainties and assumptions. Our future results, financial condition, results of operations and business may differ materially from those expressed in these forward-looking statements. You can find many of these statements by looking for words such as "approximates," "believes," "expects," "anticipates," "estimates," "intends," "plans," "would," "may" or other similar expressions in this Quarterly Report on Form 10-Q. We also note the following forward-looking statements: in the case of our development project, the estimated completion date, estimated project costs and costs to complete. These forward-looking statements represent our intentions, plans, expectations and beliefs and are subject to numerous assumptions, risks and uncertainties. Many of the factors that will determine these items are beyond our ability to control or predict. For a further discussion of factors that could materially affect the outcome of our forward-looking statements, see "Item 1A - Risk Factors" in our Annual Report on Form 10-K. For these statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995. You are cautioned not to place undue reliance on the forward-looking statements, which speak only as of the date of this Quarterly Report on Form 10-Q or the date of any document incorporated by reference. All subsequent written and oral forward-looking statements attributable to us or any person acting on our behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. We do not undertake any obligation to release publicly, any revisions to our forward-looking statements to ref

Management's Discussion and Analysis of Financial Condition and Results of Operations includes a discussion of our consolidated financial statements for the three months ended March 31, 2009 and 2008. The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting periods. Actual results could differ from those estimates.

#### Overview

Alexander's, Inc. (NYSE: ALX) is a real estate investment trust ("REIT"), incorporated in Delaware, engaged in leasing, managing, developing and redeveloping its properties. All references to "we," "us," "our," or "Company" refer to Alexander's, Inc. and its consolidated subsidiaries. We are managed by, and our properties are leased and developed by, Vornado Realty Trust ("Vornado"). We have seven properties in the greater New York City metropolitan area including the 731 Lexington Avenue property, a 1.3 million square foot multi-use building in Manhattan, and the Kings Plaza Regional Shopping Center located in Brooklyn.

We compete with a large number of real estate property owners and developers, some of which may be willing to accept lower returns on their investments. Principal factors of competition are rents charged, attractiveness of the location, the quality of the property and breadth and quality of the services provided. Our success depends upon, among other factors, trends of national and local economies, the financial condition and operating results of current and prospective tenants and customers, availability and cost of capital, construction and renovation costs, taxes, governmental regulations, legislation and population trends.

In the second half of 2007 the residential mortgage and capital markets began showing signs of stress, primarily in the form of escalating default rates on sub-prime mortgages, declining home values and increasing inventory nationwide. In 2008, the "credit crisis" spread to the broader commercial credit and financial markets resulting in illiquidity and volatility in the bond and equity markets. These trends have continued in the first quarter of 2009. We are currently in an economic recession which has negatively affected substantially all businesses, including ours. During the past year, real estate transactions have diminished significantly and capitalization rates have risen. Our real estate portfolio may continue to be affected by declining demand for office and retail space due to bankruptcies, layoffs, downsizing, cost cutting as well as general economic conditions, which would result in lower occupancy rates and effective rents and a corresponding decrease in net income, funds from operations and cash flow. During the quarter ended March 31, 2009, the occupancy rate at Kings Plaza decreased 2.6%. In addition, our allowance for doubtful accounts increased during the first quarter, primarily related to Circuit City, which filed for bankruptcy in 2008 and rejected its lease at Rego Park I in March 2009. It is not possible for us to quantify the impact of the above trends, which may persist for the remainder of 2009 and beyond, on our future financial results.

# Stock Appreciation Rights

Stock appreciation rights ("SARs") are granted at 100% of the market price of our common stock on the date of grant. Compensation expense for each SAR is measured by the excess of stock price at the current balance sheet date over the stock price at the previous balance sheet date. If the stock price is lower at the current balance sheet date, previously recognized expense is reversed but not below zero. On March 2, 2009, Steven Roth, the Chairman of our Board of Directors and Chief Executive Officer and Michael Fascitelli, our President, each exercised 150,000 SARs, which were scheduled to expire on March 4, 2009. Mr. Roth and Mr. Fascitelli each received gross proceeds of \$11,419,000. As a result of the March 2, 2009 exercises, we reversed \$34,275,000 of previously recognized SARs compensation expense in the three months ended March 31, 2009. In the three months ended March 31, 2008, we had accrued \$625,000 for SARs expense based on the March 31, 2008 closing stock price of \$354.50 at March 31, 2008, compared to \$353.25 at December 31, 2007. As of March 31, 2009, there are no SARs outstanding.

# Flushing

In February 2009, we sub-leased the Flushing property to a developer for the remainder of our ground lease term.

# Rego Park I

On March 10, 2009, we repaid the \$78,246,000 outstanding balance of the Rego Park I mortgage loan which was scheduled to mature in June 2009 and simultaneously completed a refinancing in the same amount. The new loan bears interest at 75 basis points, is secured by the property and is 100% cash collateralized. The proceeds of the new loan were placed in a non-interest bearing restricted mortgage escrow account. The loan is pre-payable without penalty after June 2009.

# 731 Lexington Avenue

On March 25, 2009, Citibank N.A. completed the assignment of its lease aggregating 176,000 square feet to Bloomberg L.P. Bloomberg L.P. now occupies 99% of the office space at the property.

#### **Critical Accounting Policies**

A summary of our critical accounting policies is included in our Annual Report on Form 10-K for the year ended December 31, 2008 in "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Note 2 – Summary of Significant Accounting Policies" to the consolidated financial statements included therein. There have been no significant changes to those policies during 2009.

# **Recently Issued Accounting Literature**

In December 2007, the Financial Accounting Standard Board ("FASB") issued Statement No. 141R, *Business Combinations* ("SFAS 141R"). SFAS 141R broadens the guidance of SFAS 141, extending its applicability to all transactions and other events in which one entity obtains control over one or more other businesses. It also broadens the fair value measurement and recognition of assets acquired, liabilities assumed, and interests transferred as a result of business combinations; and acquisition related costs will generally be expensed rather than included as part of the basis of the acquisition. SFAS 141R expands required disclosures to improve the ability to evaluate the nature and financial effects of business combinations. SFAS 141R became effective for all transactions entered into, on or after January 1, 2009. The adoption of SFAS 141R on January 1 2009, did not have a material effect on our consolidated financial statements.

In December 2007, the FASB issued Statement No. 160, *Noncontrolling Interest in Consolidated Financial Statements – An Amendment of ARB No.* 51 ("SFAS 160"). SFAS 160 requires a noncontrolling interest in a subsidiary to be reported as equity and the amount of consolidated net income specifically attributable to the noncontrolling interest to be identified in the consolidated financial statements. SFAS 160 also calls for consistency in the manner of reporting changes in the parent's ownership interest and requires fair value measurement of any noncontrolling equity investment retained in a deconsolidation. SFAS 160 became effective on January 1, 2009. The adoption of SFAS 160 on January 1, 2009, resulted in (i) the reclassification of our minority interest in consolidated subsidiary, a component of permanent equity on our consolidated balance sheet and (ii) the reclassification of minority interest expense to net income attributable to the noncontrolling interest on our consolidated statements of operations.

# **Significant Tenants**

Bloomberg L.P. accounted for approximately 34% and 32% of our consolidated revenues in each of the three months ended March 31, 2009 and 2008, respectively. No other tenant accounted for more than 10% of our consolidated revenues.

# Results of Operations for the Three Months Ended March 31, 2009 and 2008

Net income attributable to common stockholders for the quarter ended March 31, 2009 was \$46,054,000, compared to \$15,152,000 for the quarter ended March 31, 2008. Net income attributable to common stockholders for the quarter ended March 31, 2009 includes \$34,275,000 for the reversal of a portion of previously recognized SARs compensation expense, compared to \$625,000 for an accrual of SARs compensation expense, in the quarter ended March 31, 2008.

# **Property Rentals**

Property rentals were \$36,197,000 in the quarter ended March 31, 2009, compared to \$36,033,000 in the prior year's quarter, an increase of \$164,000.

# **Expense Reimbursements**

Tenant expense reimbursements were \$16,893,000 in the quarter ended March 31, 2009, compared to \$15,733,000 in the prior year's quarter, an increase of \$1,160,000. This increase was primarily due to reimbursements for higher real estate taxes.

# **Operating Expenses**

Operating expenses were \$19,035,000 in the quarter ended March 31, 2009, compared to \$17,667,000 in the prior year's quarter, an increase of \$1,368,000. This increase was primarily due to higher real estate taxes.

# General and Administrative Expenses

Excluding \$34,275,000 for the reversal of previously recognized SARs compensation expense in the quarter ended March 31, 2009, and \$625,000 an accrual of SARs compensation expense in the quarter ended March 31, 2008, general and administrative expenses increased by \$1,370,000 from the prior year's quarter. This increase is primarily due to the write-off of previously capitalized costs at our Flushing property.

# Interest and Other Income, net

Interest and other income, net was \$964,000 in the quarter ended March 31, 2009, compared to \$4,416,000 in the prior year's quarter, a decrease of \$3,452,000. This decrease resulted primarily from lower average yields on investments, from 3.3% in the quarter ended March 31, 2008 to 0.5% in the quarter ended March 31, 2009.

# Interest and Debt Expense

Interest and debt expense was \$14,896,000 in the quarter ended March 31, 2009, compared to \$15,681,000 in the prior year's quarter, a decrease of \$785,000. This decrease was primarily due to the repayment of the Rego Park I mortgage loan on March 10, 2009, and a decrease in interest on leasing commissions due to Vornado as a result of a lower rate in the current period.

#### Net Loss Attributable to the Noncontrolling Interest

Net loss attributable to the noncontrolling interest represents our venture partner's 75% pro rata share of net loss in our consolidated partially owned entity, the Kings Plaza energy plant joint venture.

# Income Tax Expense of the Taxable REIT Subsidiary

Income tax expense was \$88,000 in the quarter ended March 31, 2009, compared to \$502,000 in the prior year's quarter, a decrease of \$414,000. The decrease was due to the liquidation of 731 Residential LLC, our taxable REIT subsidiary, in September 2008.

# **Liquidity and Capital Resources**

We anticipate that cash from operations, together with existing cash balances, will be adequate to fund our business operations, recurring capital expenditures, and debt amortization over the next twelve months.

# Rego Park II Development Project

We own approximately 6.6 acres of land adjacent to our Rego Park I property in Queens, New York, which comprises the entire square block bounded by the Horace Harding Service Road (of the Long Island Expressway), 97<sup>th</sup> Street, 62<sup>nd</sup> Drive and Junction Boulevard. The development at Rego Park II consists of a 600,000 square foot shopping center on four levels and a parking deck containing approximately 1,400 spaces. Construction is expected to be substantially completed by the end of this year and estimated to cost approximately \$410,000,000, of which \$316,198,000 has been expended as of March 31, 2009. As of March 31, 2009, \$195,082,000 was drawn on the \$350,000,000 construction loan. The loan has an interest rate of LIBOR plus 1.20% (1.70% at March 31, 2009), and matures in December 2010, with a one-year extension option. The shopping center will be anchored by a 134,000 square foot Century 21 department store, a 138,000 square foot Home Depot and a 132,000 square foot Kohl's.

There can be no assurance that this project will be completed, completed on time, or completed for the budgeted amount.

#### <u>Insurance</u>

We carry commercial liability with limits of \$200,000,000 per location and all risk property insurance for (i) fire, (ii) flood, (iii) rental loss, (iv) extended coverage, and (v) "acts of terrorism," as defined in the Terrorism Risk Insurance Program Reauthorization Act ("TRIPRA") of 2007, with respect to our assets, with limits of \$1.7 billion per occurrence for all of our properties. To the extent that we incur losses in excess of our insurance coverage, these losses would be borne by us and could be material.

Our debt instruments, consisting of mortgage loans secured by our properties (which are generally non-recourse to us), contain customary covenants requiring us to maintain insurance. Although we believe that we have adequate insurance coverage for the purposes of these agreements, we may not be able to obtain an equivalent amount of coverage at reasonable costs in the future. Further, if lenders insist on greater coverage than we are able to obtain, it could adversely affect our ability to finance and/or refinance our properties.

#### **Cash Flows**

Rental income from our properties is our principal source of operating cash flow. Our property rental income is dependent on a number of factors including the occupancy level and rental rates of our properties, as well as our tenants' ability to pay their rents. Our properties provide us with a relatively consistent stream of cash flow that enables us to pay our operating expenses, non-development capital improvements and interest expense. Other sources of liquidity to fund our cash requirements include our existing cash, proceeds from debt financings, including mortgage or construction loans secured by our properties and proceeds from asset sales.

# Three Months Ended March 31, 2009

Cash and cash equivalents were \$426,265,000 at March 31, 2009, compared to \$515,940,000 at December 31, 2008, a decrease of \$89,675,000. This decrease resulted from \$105,702,000 of net cash used in investing activities, partially offset by \$10,265,000 of net cash provided by financing activities and \$5,762,000 of net cash provided by operating activities.

Net cash provided by operating activities of \$5,762,000 was comprised of net income of \$46,002,000, partially offset by adjustments for non-cash items of \$28,559,000 and the net change in operating assets and liabilities of \$11,681,000. The adjustments for non-cash items were comprised of (i) a reversal of the liability for SARs compensation expense of \$34,275,000 and (ii) straight-lining of rental income of \$2,579,000, partially offset by (iii) depreciation and amortization of \$6,411,000 and (iv) other non-cash adjustments of \$1,884,000, primarily due to a \$1,407,000 write-off of previously capitalized costs at our Flushing property. The net change in operating assets and liabilities of \$11,681,000 included a \$22,838,000 payment for SARs compensation expense.

Net cash used in investing activities of \$105,702,000 was primarily comprised of restricted cash of \$83,931,000, primarily related to the fully cash-collateralized mortgage at Rego Park I, and capital expenditures of \$21,771,000, primarily related to the development of our Rego Park II project.

Net cash provided by financing activities of \$10,265,000 was primarily comprised of borrowings under the construction loan to fund expenditures at our Rego Park II project. Financing activities also include the \$78,246,000 refinancing of the Rego Park I mortgage loan.

# Three Months Ended March 31, 2008

Cash and cash equivalents were \$576,530,000 at March 31, 2008, compared to \$560,231,000 at December 31, 2007, an increase of \$16,299,000. This increase resulted from \$24,738,000 of net cash provided by operating activities and \$22,331,000 of net cash provided by financing activities, partially offset by \$30,770,000 of net cash used in investing activities.

Net cash provided by operating activities of \$24,738,000 was comprised of (i) net income of \$14,885,000, (ii) the net change in operating assets and liabilities of \$6,041,000 and (iii) adjustments for non-cash items of \$3,812,000. The adjustments for non-cash items were primarily comprised of depreciation and amortization of \$6,263,000, partially offset by \$3,076,000 for the straight-lining of rental income.

Net cash used in investing activities of \$30,770,000 was primarily comprised of capital expenditures of \$27,239,000, primarily related to the development of our Rego Park II project.

Net cash provided by financing activities of \$22,331,000 was primarily comprised of \$25,875,000 of borrowings under the construction loan to fund expenditures at our Rego Park II project, partially offset by repayments of borrowings of \$3,685,000.

# Funds from Operations ("FFO")

FFO is computed in accordance with the definition adopted by the Board of Governors of the National Association of Real Estate Investment Trusts ("NAREIT"). NAREIT defines FFO as net income or loss determined in accordance with Generally Accepted Accounting Principles ("GAAP"), excluding extraordinary items as defined under GAAP and gains or losses from sales of previously depreciated operating real estate assets, plus specified non-cash items, such as real estate asset depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures.

FFO and FFO per diluted share are used by management, investors and industry analysts as supplemental measures of operating performance of equity REITs. FFO and FFO per diluted share should be evaluated along with GAAP net income and income per diluted share (the most directly comparable GAAP measures), as well as cash flow from operating activities, investing activities and financing activities, in evaluating the operating performance of equity REITs. Management believes that FFO and FFO per diluted share are helpful to investors as supplemental performance measures because these measures exclude the effect of depreciation, amortization and gains or losses from sales of real estate, all of which are based on historical costs which implicitly assumes that the value of real estate diminishes predictably over time. Since real estate values instead have historically risen or fallen with market conditions, these non-GAAP measures can facilitate comparisons of operating performance between periods and among other equity REITs.

FFO does not represent cash generated from operating activities in accordance with GAAP and is not necessarily indicative of cash available to fund cash needs as disclosed in our Statements of Cash Flows. FFO should not be considered as an alternative to net income as an indicator of the Company's operating performance or as an alternative to cash flows as a measure of liquidity.

# FFO attributable to common stockholders for the Quarters Ended March 31, 2009 and 2008

FFO attributable to common stockholders for the quarter ended March 31, 2009 was \$51,643,000, or \$10.12 per diluted share, compared to \$20,353,000, or \$4.00 per diluted share, for the quarter ended March 31, 2008. FFO attributable to common stockholders for the quarter ended March 31, 2009 includes \$34,275,000, or \$6.72 per diluted share, for the reversal of previously recognized SARs compensation expense, compared to \$625,000, or \$0.12 per diluted share, for an accrual of SARs compensation expense in the quarter ended March 31, 2008.

The following table reconciles our net income to FFO:

	Three Months Ended March 31,						
(Amounts in thousands, except share and per share amounts)				2008			
Net income attributable to Alexander's	\$	46,054	\$	15,152			
Depreciation and amortization of real property		5,589		5,201			
FFO attributable to common stockholders	\$	51,643	\$	20,353			
FFO attributable to common stockholders per diluted share	\$	10.12	\$	4.00			
Weighted average shares used in computing diluted FFO per share		5,102,826		5,092,973			

# Item 3. Quantitative and Qualitative Disclosures About Market Risk

We have exposure to fluctuations in interest rates, which are sensitive to many factors that are beyond our control. Our exposure to a change in interest rates is summarized in the table below.

(Amounts in thousands, except per share amounts)	]	Balance as of March 31, 2009	Weighted-Average Interest Rate	Cha	et of 1% ange in e Rates
Variable	\$	239,212	1.94%	\$	2,392
Fixed Rate		1,035,655	5.31%		_
	\$	1,274,867		\$	2,392
	_				
Total effect on diluted earnings per share				\$	0.47

The fair value of our debt, estimated by discounting the future contractual cash flows of our existing debt using the current rates available to borrowers with similar credit ratings for the remaining terms of such debt, was less than the aggregate carrying amount by approximately \$148,423,000 at March 31, 2009.

# Item 4. Controls and Procedures

- (a) Disclosure Controls and Procedures Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended) as of the end of the period covered by this Quarterly Report on Form 10-Q. Based on such evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that, as of the end of such period, our disclosure controls and procedures are effective.
- (b) Internal Control Over Financial Reporting There have not been any changes in our internal control over financial reporting during the fiscal quarter to which this Quarterly Report on Form 10-Q relates that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

# PART II. OTHER INFORMATION

# Item 1. Legal Proceedings

We are from time to time involved in legal actions arising in the ordinary course of business. In our opinion, after consultation with our legal counsel, the outcome of such matters in the aggregate will not have a material effect on our financial condition, results of operations or cash flows.

# Item 1A. Risk Factors

There have been no material changes in our "Risk Factors" as previously disclosed in our Annual Report on Form 10-K for the year ended December 31, 2008.

# Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

# Item 3. Defaults Upon Senior Securities

None.

# Item 4. Submission of Matters to a Vote of Security Holders

None.

# Item 5. Other Information

None.

# Item 6. Exhibits

(a) Exhibits required by Item 601 of Regulation S-K are filed herewith and are listed in the attached Exhibit Index.

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# ALEXANDER'S, INC.

(Registrant)

Date: May 4, 2009

By: /s/ Joseph Macnow

Joseph Macnow, Executive Vice President and Chief Financial Officer (duly authorized officer and principal financial and accounting officer)

#### EXHIBIT INDEX

# Exhibit No. Amended and Restated Certificate of Incorporation. Incorporated herein by reference from Exhibit 3.1 to the registrant's 3.1 Registration Statement on Form S-3 filed on September 20, 1995 By-laws, as amended. Incorporated herein by reference from Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the 3.2 quarter ended June 30, 2000 Real Estate Retention Agreement dated as of July 20, 1992, between Vornado Realty Trust and Keen Realty Consultants, Inc., each 10.1 as special real estate consultants, and the Company. Incorporated herein by reference from Exhibit 10(i)(O) to the registrant's Annual Report on Form 10-K for the fiscal year ended July 25, 1992 10.2 Extension Agreement to the Real Estate Retention Agreement, dated as of February 6, 1995, between the Company and Vornado Realty Trust. Incorporated herein by reference from Exhibit 10(i)(G)(2) to the registrant's Annual Report Form 10-K for the year ended December 31, 1994 Registrant's Omnibus Stock Plan, as amended, dated May 28, 1997. Incorporated herein by reference from Exhibit 10 to the 10.3 registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 1997, filed on August 13, 1997 Agreement of Lease dated as of April 30, 2001 between Seven Thirty One Limited Partnership, landlord, and Bloomberg L.P., 10.4 tenant. Incorporated herein by reference from Exhibit 10(v) B to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2001, filed on August 2, 2001 10.5 Amended and Restated Consolidated Mortgage and Security Agreement dated as of May 31, 2001 among Alexander's Kings Plaza LLC as mortgagor, Alexander's of King LLC as mortgagor and Kings Parking LLC as mortgagor, collectively borrower, to Morgan Guaranty Trust Company of New York, as mortgagee. Incorporated herein by reference from Exhibit 10(v) A1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2001, filed on August 2, 2001 Amended, Restated and Consolidated Promissory Note, dated as of May 31, 2001 by and between Alexander's Kings Plaza LLC, 10.6 Alexander's of Kings LLC, and Kings Parking LLC collectively borrower, and Morgan Guaranty Trust Company of New York, lender. Incorporated herein by reference from Exhibit 10(v) A2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2001, filed on August 2, 2001 Cash Management Agreement dated as of May 31, 2001 by and between Alexander's Kings Plaza LLC, Alexander's of Kings 10.7 LLC, and Kings Parking LLC collectively borrower, and Morgan Guaranty Trust Company of New York, lender. Incorporated herein by reference from Exhibit 10(v) A3 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2001, filed on August 2, 2001

Incorporated by reference.

<sup>\*\*</sup> Management contract or compensatory agreement

10.8	-	Note modification and Severance Agreement dated as of November 26, 2001, between Alexander's Kings Plaza LLC, Alexander's of Kings LLC, and Kings Parking LLC collectively borrower and JP Morgan Chase Bank of New York, lender. Incorporated herein by reference from Exhibit $10(v)(A)(4)$ to the registrant's Annual Report on Form 10-K for the year ended December 31, 2001, filed on March 13, 2002	

- 10.9 Loan Agreement dated as of October 2, 2001 by and between ALX of Paramus LLC as borrower, and SVENSKA HANDELSBANKEN AB (publ), as lender. Incorporated herein by reference from Exhibit 10(v)(C)(1) to the registrant's Annual Report on Form 10-K for the year ended December 31, 2001, filed on March 13, 2002
- 10.10 Mortgage, Security Agreement and Fixture Financing Statement dated as of October 2, 2001 by and between ALX of Paramus LLC as \*borrower, and SVENSKA HANDELSBANKEN AB (publ), as lender. Incorporated herein by reference from Exhibit 10(v)(C)(2) to the registrant's Annual Report on Form 10-K for the year ended December 31, 2001, filed on March 13, 2002
- 10.11 Environmental undertaking letter dated as of October 2, 2001 by and between ALX of Paramus LLC, as borrower, and SVENSKA \* HANDELSBANKEN AB (publ), as lender. Incorporated herein by reference from Exhibit 10(v)(C)(3) to the registrant's Annual Report on Form 10-K for the fiscal year ended December 31, 2001, filed on March 13, 2002
- 10.12 Lease dated as of October 2, 2001 by and between ALX of Paramus LLC, as Landlord, and IKEA Property, Inc. as Tenant. Incorporated \* herein by reference from Exhibit 10(v)(C)(4) to the registrant's Annual Report on Form 10-K for the year ended December 31, 2001, filed on March 13, 2002
- 10.13 First Amendment to Real Estate Retention Agreement, dated as of July 3, 2002, by and between Alexander's, Inc. and Vornado Realty,

  L.P. Incorporated herein by reference from Exhibit 10(i)(E)(3) to the registrant's Quarterly Report on Form 10-Q for the quarter ended

  June 30, 2002, filed on August 7, 2002
- 10.14 59<sup>th</sup> Street Real Estate Retention Agreement, dated as of July 3, 2002, by and between Vornado Realty, L.P., 731 Residential LLC and 731 Commercial LLC. Incorporated herein by reference from Exhibit 10(i)(E)(4) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2002
- Amended and Restated Management and Development Agreement, dated as of July 3, 2002, by and between Alexander's, Inc., the subsidiaries party thereto and Vornado Management Corp. Incorporated herein by reference from Exhibit 10(i)(F)(1) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2002
- Kings Plaza Management Agreement, dated as of May 31, 2001, by and between Alexander's Kings Plaza LLC and Vornado Management Corp. Incorporated herein by reference from Exhibit 10(i)(F)(3) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2002
- Limited Liability Company Operating Agreement of 731 Residential LLC, dated as of July 3, 2002, among 731 Residential Holding LLC, as the sole member, Domenic A. Borriello, as an Independent Manager and Kim Lutthang, as an Independent Manager. Incorporated herein by reference from Exhibit 10(i)(A)(1) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2002
  - \* Incorporated by reference.

10.18	-	Limited Liability Company Operating Agreement of 731 Commercial LLC, dated as of July 3, 2002, among 731 Commercial Holding LLC, as the sole member, Domenic A. Borriello, as an Independent Manager and Kim Lutthang, as an Independent Manager. Incorporated herein by reference from Exhibit 10(i)(A)(2) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2003	
		on August 7, 2002	

- 10.19 Reimbursement Agreement, dated as of July 3, 2002, by and between Alexander's, Inc., 731 Commercial LLC, 731 Residential LLC and Vornado Realty, L.P. Incorporated herein by reference from Exhibit 10(i)(C)(8) to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, filed on August 7, 2002
- 10.20 First Amendment of Lease, dated as of April 19, 2002, between Seven Thirty One Limited Partnership, landlord and Bloomberg L.P., \* tenant. Incorporated herein by reference from Exhibit 10(v)(B)(2) to the registrant's Quarterly Report on Form 10-Q for the fiscal quarter ended June 30, 2002, filed on August 7, 2002
- 10.21 Loan and Security Agreement, dated as of February 13, 2004, between 731 Office One LLC, as Borrower and German American Capital \* Corporation, as Lender. Incorporated herein by reference from Exhibit 10.20 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- 10.22 Amended, Restated and Consolidated Mortgage, Security Agreement, Financing Statement and Assignment of Leases, Rent and Security

  Deposits by and between 731 Office One LLC as Borrower and German American Capital Corporation as Lender, dated as of February 13,
  2004. Incorporated herein by reference from Exhibit 10.21 to the registrant's Annual Report on Form 10-K for the year ended December
  31, 2003, filed on March 2, 2004
- Amended, Restated and Consolidated Note, dated as of February 13, 2004, by 731 Office One LLC in favor of German American Capital \* Corporation. Incorporated herein by reference from Exhibit 10.22 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- 10.24 Assignment of Leases, Rents and Security Deposits from 731 Office One LLC to German American Capital Corporation, dated as of \*February 13, 2004. Incorporated herein by reference from Exhibit 10.23 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Account and Control Agreement, dated as of February 13, 2004, by and among German American Capital Corporation as Lender, and 731
   Office One LLC as Borrower, and JP Morgan Chase as Cash Management Bank. Incorporated herein by reference from Exhibit 10.24 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Manager's Consent and Subordination of Management Agreement dated February 13, 2004 by 731 Office One LLC and Alexander's Management LLC and German American Capital Corporation. Incorporated herein by reference from Exhibit 10.25 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Note Exchange Agreement dated as of February 13, 2004 by and between 731 Office One LLC and German American Capital \*
   Corporation. Incorporated herein by reference from Exhibit 10.26 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
  - \* Incorporated by reference.

- 10.28 Promissory Note A-1 dated as of February 13, 2004 and 731 Office One LLC in favor of German American Capital Corporation. \* Incorporated herein by reference from Exhibit 10.27 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- 10.29 Promissory Note A-2 dated as of February 13, 2004 and 731 Office One LLC in favor of German American Capital Corporation. \* Incorporated herein by reference from Exhibit 10.28 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Promissory Note A-3 dated as of February 13, 2004 and 731 Office One LLC in favor of German American Capital Corporation. \*
  Incorporated herein by reference from Exhibit 10.29 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Promissory Note A-4 dated as of February 13, 2004, and 731 Office One LLC in favor of German American Capital Corporation. \* Incorporated herein by reference from Exhibit 10.30 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Promissory Note A-X dated as of February 13, 2004, and 731 Office One LLC in favor of German American Capital Corporation. \* Incorporated herein by reference from Exhibit 10.31 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Promissory Note B dated as of February 13, 2004, and 731 Office One LLC in favor of German American Capital Corporation.

  Incorporated herein by reference from Exhibit 10.32 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- Guaranty of Recourse Obligations dated as of February 13, 2004, by Alexander's, Inc. to and for the benefit of German American Capital
  Corporation. Incorporated herein by reference from Exhibit 10.33 to the registrant's Annual Report on Form 10-K for the year ended
  December 31, 2003, filed on March 2, 2004
- Environmental Indemnity dated as of February 13, 2004, by Alexander's, Inc. and 731 Office One LLC for the benefit of German \*
   American Capital Corporation. Incorporated herein by reference from Exhibit 10.34 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2003, filed on March 2, 2004
- 10.36 Loan Agreement dated as of July 6, 2005, between 731 Retail One LLC, as Borrower and Archon Financial, as Lender. Incorporated herein by reference from Exhibit 10.1 to the registrant's Current Report on Form 8-K, filed on July 12, 2005
- 10.37 \*\* Form of Stock Option Agreement between the Company and certain employees. Incorporated herein by reference from Exhibit 10.61 to \* the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2005, filed on October 27, 2005
- 10.38 \*\* Form of Restricted Stock Option Agreement between the Company and certain employees. Incorporated herein by reference from Exhibit \* 10.62 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2005, filed on October 27, 2005
  - \* Incorporated by reference.
  - \*\* Management contract or compensatory agreement

- 10.39 \*\* Registrant's 2006 Omnibus Stock Plan dated April 4, 2006. Incorporated herein by reference from Annex B to Schedule 14A, filed by the registrant on April 28, 2006
- Second Amendment to Real Estate Retention Agreement, dated as of January 1, 2007, by and between Alexander's, Inc. and Vornado Realty L.P. Incorporated herein by reference from Exhibit 10.64 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2006, filed on February 26, 2007
- Amendment to 59<sup>th</sup> Street Real Estate Retention agreement, dated as of January 1, 2007, by and among Vornado Realty L.P., 731 Retail
  One LLC, 731 Restaurant LLC, 731 Office One LLC and 731 Office Two LLC. Incorporated herein by reference from Exhibit 10.65 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2006, filed on February 26, 2007
- Building Loan Agreement, dated as of December 21, 2007, among Alexander's of Rego Park II, Inc., as Borrower, PB Capital Corporation, as Lender, Norddeutsche Landesbank Girozentrale, New York Branch, as Lender, Wells Fargo Bank, National Association, as Lender, Landesbank Baden-Wurttemberg, New York Branch, as Lender, Bank of Ireland, Connecticut Branch, as Lender, PB Capital Corporation, as Administrative Agent, PB Capital Corporation and Norddeutsche Landesbank Girozentrale, New York Branch, as Co-Arrangers. Incorporated herein by reference from Exhibit 10.1 to the registrant's Current Report on Form 8-K, filed on December 28, 2007
- Project Loan Agreement, dated as of December 21, 2007, among Alexander's of Rego Park II, Inc., as Borrower, PB Capital Corporation, as Lender, Norddeutsche Landesbank Girozentrale, New York Branch, as Lender, Wells Fargo Bank, National Association, as Lender, Landesbank Baden-Wurttemberg, New York Branch, as Lender, Bank of Ireland, Connecticut Branch, as Lender, PB Capital Corporation, as Administrative Agent, PB Capital Corporation and Norddeutsche Landesbank Girozentrale, New York Branch, as Co-Arrangers. Incorporated herein by reference from Exhibit 10.2 to the registrant's Current Report on Form 8-K, filed on December 28, 2007
- Series I Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement, dated as of December 21, 2007, from \* Alexander's of Rego Park II, Inc., as Mortgagor, to PB Capital Corporation, as Administrative Agent for the Lenders. Incorporated herein by reference from Exhibit 10.3 to the registrant's Current Report on Form 8-K, filed on December 28, 2007
- Series II Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement, dated as of December 21, 2007, from \* Alexander's of Rego Park II, Inc., as Mortgagor, to PB Capital Corporation, as Administrative Agent for the Lenders. Incorporated herein by reference from Exhibit 10.4 to the registrant's Current Report on Form 8-K, filed on December 28, 2007
- Series I Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement, dated as of December 21, 2007, from \* Alexander's of Rego Park II, Inc., as Mortgagor, to PB Capital Corporation, as Administrative Agent for the Lenders. Incorporated herein by reference from Exhibit 10.5 to the registrants Current Report on Form 8-K, filed on December 31, 2007
- Series II Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement, dated as of December 21, 2007, from \* Alexander's of Rego Park II, Inc., as Mortgagor, to PB Capital Corporation, as Administrative Agent for the Lenders. Incorporated herein by reference from Exhibit 10.6 to the registrant's Current Report on Form 8-K, filed on December 28, 2007
  - \* Incorporated by reference.
  - \*\* Management contract or compensatory agreement

<ul> <li>Guaranty of Completion, dated as of December 21, 2007, executed by Alexander's, Inc. for the benefit of PB Capital Corporation, Administrative Agent for itself and the other Lenders Incorporated herein by reference from Exhibit 10.7 to the registrant's Current Report on Form 8-K, filed on December 28, 2007</li> </ul>	
- Guaranty of Payment, dated as of December 21, 2007, executed by Alexander's, Inc. for the benefit of PB Capital Corporation, Administrative Agent for itself and the other Lenders. Incorporated herein by reference from Exhibit 10.8 to the registrant's Current Report on Form 8-K, filed on December 28, 2007	
10.50 - First Amendment to Amended and Restated Management and Development Agreement, dated as of July 6, 2005, by and betwee Alexander's, Inc., the subsidiaries party thereto and Vornado Management Corp. Incorporated herein by reference from Exhibit 10.52 the registrant's Annual Report on Form 10-K, filed on February 25, 2008.	
<ul> <li>Second Amendment to Amended and Restated Management and Development Agreement, dated as of December 20, 2007, by ar between Alexander's, Inc., the subsidiaries party thereto and Vornado Management Corp. Incorporated herein by reference from Exhibit 10.53 to the registrant's Annual Report on Form 10-K, filed on February 25, 2008</li> </ul>	
<ul> <li>Rego II Management and Development Agreement, dated as of December 20, 2007, by and between Alexander's of Rego Park II, Inc., ar Vornado Realty L.P. Incorporated herein by reference from Exhibit 10.54 to the registrant's Annual Report on Form 10-K, filed of February 25, 2008</li> </ul>	
<ul> <li>Third Amendment to Real Estate Retention Agreement, dated as of December 20, 2007, by and between Alexander's, Inc., and Vornac Realty L.P. Incorporated herein by reference from Exhibit 10.55 to the registrant's Annual Report on Form 10-K, filed on February 2 2008</li> </ul>	
10.54 - Rego II Real Estate Retention Agreement, dated as of December 20, 2007, by and between Alexander's, Inc., and Vornado Realty L. Incorporated herein by reference from Exhibit 10.56 to the registrant's Annual Report on Form 10-K, filed on February 25, 2008	P. *
<ul> <li>Loan Agreement dated as of March 10, 2009 between Alexander's Rego Park Shopping Center Inc., as Borrower and U.S. Bank Nation Association, as Lender.</li> </ul>	al
10.56 - Amended and Restated Mortgage, Security Agreement, Fixture Filing and Assignment of Leases and Rentals by and between Alexander Rego Shopping Center, Inc. as Borrower and U.S. Bank National Association as Lender, dated as of March 10, 2009	's
10.57 - Amended and Restated Promissory Note dated as of March 10, 2009, by Alexander's Rego Shopping Center Inc., in favor of U.S. Bar National Association	k
10.58 - Cash Pledge Agreement dated as of March 10, 2009, executed by Alexander's Rego Shopping Center Inc. to and for the benefit of U. Bank National Association	5.
10.59 - Lease dated as of February 7, 2005, by and between 731 Office One LLC, as Landlord, and Citibank, N.A., as Tenant	
<ul> <li>Assignment and Assumption and Consent Agreement, dated as of March 25, 2009, by and between 731 Office One LLC, as Landlor Citicorp North America, Inc., as Assignor, and Bloomberg L.P., as Assignee</li> </ul>	1,
15.1 - Letter regarding unaudited interim financial information	

- 31.1 Rule 13a-14 (a) Certification of the Chief Executive Officer
- 31.2 Rule 13a-14 (a) Certification of the Chief Financial Officer
- 32.1 Section 1350 Certification of the Chief Executive Officer
- 32.2 Section 1350 Certification of the Chief Financial Officer
  - \* Incorporated by reference.

May 4, 2009

Alexander's, Inc.
Paramus, New Jersey

We have reviewed, in accordance with the standards of the Public Company Accounting Oversight Board (United States), of the unaudited interim financial information of Alexander's, Inc. and subsidiaries for the three-month periods ended March 31, 2009 and 2008, as indicated in our report dated May 4, 2009, (which report included an explanatory paragraph regarding the adoption of FASB Statement No. 160, Noncontrolling Interests in Consolidated Financial Statements); because we did not perform an audit, we expressed no opinion on that information.

We are aware that our reports referred to above, which were included in your Quarterly Reports on Form 10-Q for the quarter ended March 31, 2009 is incorporated by reference in the following registration statements of Alexander's, Inc.:

Registration Statement No. 333-151721 on Form S-8 Amendment No. 1 to Registration Statement No. 33-155727 on Form S-3

We also are aware that the aforementioned report, pursuant to Rule 436(c) under the Securities Act of 1933, is not considered a part of the Registration Statement prepared or certified by an accountant or a report prepared or certified by an accountant within the meaning of Sections 7 and 11 of that Act.

/s/ DELOITTE & TOUCHE LLP Parsippany, New Jersey

# I, Steven Roth, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Alexander's, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure control and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 4, 2009
/s/ Steven Roth
Steven Roth
Chief Executive Officer

# I, Joseph Macnow, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Alexander's, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure control and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 4, 2009

/s/ Joseph Macnow

Joseph Macnow

Executive Vice President and Chief Financial Officer

# PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (Subsection (a) and (b) of Section 1350 of Chapter 63 of Title 18 of the United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of Section 1350 of Chapter 63 of Title 18 of the United States Code), the undersigned officer of Alexander's, Inc. (the "Company"), hereby certifies, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the quarter ended March 31, 2009 (the "Report") of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

May 4, 2009 /s/ Steven Roth

Name: Steven Roth

Title: Chief Executive Officer

# PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (Subsection (a) and (b) of Section 1350 of Chapter 63 of Title 18 of the United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of Section 1350 of Chapter 63 of Title 18 of the United States Code), the undersigned officer of Alexander's, Inc. (the "Company"), hereby certifies, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the quarter ended March 31, 2009 (the "Report") of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

May 4, 2009 /s/ Joseph Macnow

Name: Joseph Macnow
Title: Chief Financial Officer

# LOAN AGREEMENT

by and between

# ALEXANDER'S REGO SHOPPING CENTER, INC.,

a Delaware corporation,

as Borrower

and

# U.S. BANK NATIONAL ASSOCIATION,

a national banking association,

as Bank

Dated as of March 10, 2009

# TABLE OF CONTENTS

A DETICAL ELLI CANA	
ARTICLE I LOAN	6
Section 1.1 Principal	6
Section 1.2 Interest	6
Section 1.3 Prepayment	7
Section 1.4 Bank Losses	8
Section 1.5 Payments	8
Section 1.6 Application of Payments	8
ARTICLE II CONDITIONS OF BORROWING	8
Section 2.1 Pre Closing Requirements	8
Section 2.2 Loan Documents	9
Section 2.3 Opinion of Borrower's Attorneys	10
Section 2.4 Evidence of Recording	10
Section 2.5 Satisfaction of Closing Conditions	10
ARTICLE III ADVANCES OF LOAN PROCEEDS	10
Section 3.1 General	10
Section 3.2 Inspections	11
ARTICLE IV REPRESENTATIONS AND WARRANTIES	11
Section 4.1 Borrower's Formation and Powers	11
Section 4.2 Authority	11
Section 4.3 No Approvals	11
Section 4.4 Legal and Valid Obligations	12
Section 4.5 Litigation	12
Section 4.6 Title to Land	12
Section 4.7 Payment of Taxes	12
Section 4.8 Federal Reserve Regulations	12
Section 4.9 Investment Company Act	13
Section 4.10 Unregistered Securities	13
Section 4.11 Accuracy of Information	13
Section 4.12 Compliance	13
Section 4.13 Anti-Terrorism Regulations	13
Section 4.14 Subsidiaries	14
Section 4.15 Rent Roll	14
Section 4.16 Ownership and Control of Borrower	14
ARTICLE V COVENANTS OF BORROWER	14
Section 5.1 Using Loan Proceeds	14
Section 5.2 Keeping of Records	15
Section 5.3 Maintaining Insurance Coverage	15
Section 5.4 Transferring, Conveying or Encumbering the Project	15
Section 5.5 Reporting Requirements	15
Section 5.6 Maintain Existence	16
Section 5.7 Notice	16
Section 5.8 Merger and Consolidation	16
Section 5.9 Patriot Act	16

16

ARTICLE VI DEFAULTS

Section 6.1 Events of Default	16
Section 6.2 Rights and Remedies	17
RTICLE VII INTEREST, FEES AND EXPENSES	18
Section 7.1 Authorization to Make Loan Advances to Cure Borrower's Defaults	18
RTICLE VIII DEPOSIT ACCOUNT	19
Section 8.1 Interest Yield	19
Section 8.2 Offset Against Mortgage Debt	19
RTICLE IX MISCELLANEOUS	19
Section 9.1 Waiver and Amendment	19
Section 9.2 Expenses and Indemnities	19
Section 9.3 Binding Effect; Waivers; Cumulative Rights and Remedies	21
Section 9.4 Incorporation By Reference	21
Section 9.5 Survival	22
Section 9.6 Governing Law; Waiver of Jury Trial; Jurisdiction	22
Section 9.7 Counterparts	22
Section 9.8 Notices	22
Section 9.9 No Third Party Reliance	23
Section 9.10 Time of the Essence	23
Section 9.11 No Oral Modifications	23
Section 9.12 Captions	23
Section 9.13 Borrower-Bank Relationship	23

## **EXHIBITS**

EXHIBIT A LEGAL DESCRIPTION AND PERMITTED ENCUMBRANCES

EXHIBIT B SCHEDULE OF MORTGAGES AND ASSIGNMENTS

EXHIBIT C RENT ROLL

EXHIBIT D ACCOUNT NUMBERS

## LOAN AGREEMENT

THIS LOAN AGREEMENT (this "**Agreement**") is made and entered into as of March 10, 2009, by and between ALEXANDER'S REGO SHOPPING CENTER, INC., a Delaware corporation (the "**Borrower**"), and U.S. BANK NATIONAL ASSOCIATION, a national banking association ("**Bank**" or "US **Bank**"), as lender.

Borrower has requested that the Bank provide the Loan (as hereinafter defined) to Borrower in the principal sum of up to \$78,245,641.77 for the purpose of re-financing existing indebtedness on the Project (as hereinafter defined).

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth, the Loan to be made by Bank pursuant to this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

## **DEFINITIONS:**

For purposes of this Agreement, the following terms shall have the following respective meanings, unless the context hereof clearly requires otherwise:

"Affiliate": Any other Person directly or indirectly controlling, controlled by or under common control with such Person. A Person shall be deemed to control another Person if the controlling Person owns 10% or more of any class of voting securities (or other ownership interests) of the controlled Person or possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of the controlled Person, whether through ownership of stock, by contract or otherwise.

"Agreement": This Loan Agreement, including any amendments hereof and supplements hereto executed by Borrower and the Bank.

"Amended and Restated Note": That certain Amended and Restated Promissory Note of even date herewith in the original principal amount of the Loan from Borrower payable to the order of Bank.

"Anti-Terrorism Laws": Any laws relating to terrorism or money laundering, including Executive Order No. 13224, the USA Patriot Act, the Laws comprising or implementing the Bank Secrecy Act, and the Law administered by the United States Treasury Department's Office of Foreign Asset Control (as any of the foregoing Laws may from time to time be amended, renewed, extended, or replaced).

"Assignment of Leases": that certain Assignment of Leases and Rents dated the date hereof from Borrower in favor of Bank with respect to the Loan.

"Blocked Person": As that term is defined in Section 4.22 below.

"Borrower": Shall have the meaning assigned said term in the introductory paragraph hereof.

- "Business Day": Any day other than a Saturday, a Sunday, or a legal holiday on which Bank is not open for business.
- "Cash Collateral": Shall have the meaning set forth in the Pledge Agreement.
- "Closing Date": March 10, 2009.
- "Code": The Internal Revenue Code of 1986, as amended.
- "Default": Any event which, with the giving of notice to Borrower or the lapse of time, or both, would constitute an Event of Default.
- "<u>Default Rate</u>": The lesser of four percent (4%) per annum in excess of the Loan Rate or the maximum lawful rate of interest which may be charged, if any.
- "Deposit": Means \$78,245,641.77 cash, in readily available funds, deposited by Borrower initially into the Noninterest-bearing Transaction Account to serve as cash collateral for the Loan.
- "Deposit Account": Means the Noninterest-bearing Transaction Account and/or Interest-bearing Investment Account, as the case may be, in which Funds are on deposit from time to time.
- "<u>Deposit Rate</u>": Shall mean a floating rate per annum equal to the Federal Funds Rate <u>minus</u> twenty-five (25) basis points; provided, that in no event shall the Deposit Rate be less than zero percent (0%).
- "Equipment": All furniture, fixtures, equipment and personal property owned by Borrower and located or to be located in or on, and used in connection with the management, maintenance or operation of, the Land and the Improvements.
  - "Event of Default": An Event of Default specified in Section 6.1 hereof.

"Existing Mortgage": That certain Amended, Restated and Consolidated Mortgage and Security Agreement dated May 12, 1999 by and between Mortgagor and The Chase Manhattan Bank, recorded on June 7, 1999 in Reel 5263, Page 2302, in the Office of the City Register of the City of New York, Queens County, and assigned to State Street Bank and Trust Company, as Trustee for the Registered Holders of Chase Manhattan Bank-First Union National Bank Commercial Mortgage Trust, Commercial Mortgage Pass Though Certificates Series, 1999-1, pursuant to an Assignment of Mortgages dated October 10, 2000 and recorded November 22, 2000 in Reel 5727, Page 0118 in the Office of the City Register of the City of New York, Queens County, which consolidates a certain Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement dated March 29, 1995, recorded on March 30, 1995 in Reel 4097, Page 746 in the Office of the City Register of the City of New York, Queens County and a certain Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement dated March 29, 1995 and recorded on March 30, 1995 in Reel 4097, Page 780 in the Office of the City Register of the City of New York, Queens County to for a single lien in the amount of \$82,000,000,00, which consolidated mortgages and assignments thereof are set forth on Exhibit

<u>B</u> attached hereto, securing the Existing Note, and assigned on or about the date hereof to US Bank.

"Existing Mortgage Assignment": An assignment of the Existing Mortgage in favor of Bank, in recordable form, duly executed by the holder of the Existing Mortgage

"Existing Note": That certain Amended, Restated and Consolidated Promissory Note dated May 12, 1999, in the original principal amount of \$82,000,000 from Borrower originally payable to The Chase Manhattan Bank, and assigned on or about the date hereof to US Bank.

"Existing Note Allonge": An Allonge to the Existing Note payable to the order of Bank, in form reasonably acceptable to Bank, endorsing, assigning and setting over the Existing Note to Bank.

"Existing Survey": That certain ALTA/ACSM Survey of the Land entitled: Rego Park Borough of Queens, County of Queens, State of New York Tax Map Section 12, Block 2084, Lot 101" prepared by Gerald T. O'Buckley Land Surveyors, dated April 20<sup>th</sup>, 1999.

"Federal Funds Rate": For any day, the rate per annum (expressed as a decimal, rounded upwards, if necessary, to the next higher 1/100 of 1%) equal to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published for such day; provided (i) if such day is not a business day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding business day as so published on the next succeeding business day, and (ii) if no such rate is so published on such next succeeding business day, the Federal Funds Rate for such day shall be the average rate charged to Bank on such day on such transactions as determined by Bank; provided, further that in no event shall the Federal Funds Rate, as used herein, be less than zero percent (0%).

"FDIC": The Federal Deposit Insurance Corporation, a body politic.

"Fiscal Year": The period of January 1 of any year through December 31 of such calendar year.

"Funds": Shall have the meaning set forth in Section 2 of the Pledge Agreement.

"GAAP": Generally accepted accounting principles consistently applied and maintained throughout the period indicated and consistent with the audited financial statements delivered to Bank pursuant to Article V. Whenever any accounting term is used herein and is not otherwise defined, it shall be interpreted in accordance with GAAP.

"Good Faith Indemnity": The Good Faith Indemnity and Guaranty Agreement of even date herewith executed by Borrower and Indemnitor, including any amendments thereof and supplements thereto executed by Borrower, Indemnitor and Bank.

"Governmental Authority": means any court, board, agency, commission, office or authority of any nature whatsoever or any governmental unit (federal, state, commonwealth, county, district, municipal, city or otherwise) whether now or hereafter in existence.

"Governmental Requirements": All laws, statutes, codes, ordinances, and governmental rules, regulations and requirements of a Governmental Authority applicable to Borrower, Bank or the Project, including without limitation environmental laws, and the requirements of the Americans with Disabilities Act of 1990, as amended, and all regulations thereunder, and all covenants, agreements, restrictions and encumbrances contained in any instruments, either of record or known to Borrower, at any time in force affecting the Project or any part thereof, including any which may (i) require repairs, modifications or alterations in or to the Project or any part thereof, or (ii) in any way limit the use and enjoyment thereof.

"Improvements": The buildings and improvements presently located on the Land.

"Indemnitor": Alexander's, Inc., a Delaware corporation.

"Indemnity.": The Indemnification Agreement of even date herewith executed by Borrower and Indemnitor, including any amendments thereof and supplements thereto executed by Borrower and Bank.

"Interest-bearing Investment Account": Means a money market account at Bank, mutually acceptable to Borrower and Bank, which shall at all times be a balance sheet liability of Bank and not a custodial account, and having the Account Number set forth on Exhibit D hereto, as amended from time to time; provided that any Interest-bearing Investment Account shall pay interest at the Deposit Rate.

"<u>Land</u>": Certain property commonly known as 96-05 Queens Boulevard, Queens Boulevard, New York, and more specifically described on <u>Exhibit A</u>, attached hereto and made a part hereof by this reference.

"Land Records": The Office of the City Register of the City of New York, Queens County.

"Loan": The loan in the maximum aggregate principal amount of \$78,245,641.77.

"<u>Loan Documents</u>": The documents described in <u>Section 2.2</u> of this Agreement, and any other documents which evidence, secure and/or govern the Loan, including, but not limited to, this Agreement, the Note, the Mortgage, the Pledge Agreement, the Indemnity, the Assignment of Leases, the Good Faith Indemnity and any amendments thereof and supplements thereto executed by Borrower and the parties thereto.

"Loan Party(ies)": Borrower and Indemnitor.

"Loan Rate": A fixed rate per annum equal to seventy-five basis points (.75%); <u>provided</u>, <u>however</u>, that for that portion of the Loan which is equal to the dollar amount of the Funds which are invested in an Interest-bearing Investment Account, from time to time, the Loan Rate shall be a floating rate per annum equal to the Federal Funds Rate, in effect from time to time, plus fifty (50) basis points.

"Material Adverse Occurrence": Any occurrence of whatsoever nature (including, without limitation, any adverse determination in any litigation, arbitration or governmental

investigation or proceeding) which Bank shall determine, acting as a commercially reasonable secured lender, is likely to impair the ability of Borrower or Indemnitor to perform its obligations as and when required under any of the Loan Documents.

"Maturity Date": March 10, 2012.

- "Mortgage": That certain Amended and Restated Mortgage, Security Agreement, Fixture Filing and Assignment of Leases and Rents of even date herewith, executed by Borrower in favor of Bank to secure the Loan, and which amends and restates, in its entirety, the Existing Mortgage, including any amendments or modifications thereof and supplements thereto by Borrower and Bank.
- "Noninterest-bearing Transaction Account": Shall mean the non-interest bearing deposit account, as contemplated by 12 CFR Part 370, maintained by US Bank and having the Account Number set forth on Exhibit D hereto, as amended from time to time.
- "Note": That certain Amended and Restated Promissory Note dated of even date herewith, which amends and restates in its entirety the Existing Note, of even date herewith, executed and delivered by Borrower to Bank to evidence the Loan,
- "Organizational Documents": Means, for any entity, the documents pursuant to which such entity has been formed and by which it is governed, including, in the case of a corporation, its articles of organization or certificate of incorporation and its bylaws; in the case of a partnership, its agreement of partnership and certificate of limited partnership, if applicable; in the case of a limited liability company, its certificate or articles of organization and operating agreement; and in the case of a trust, its declaration of trust.
- "Permitted Encumbrances": The liens, charges and encumbrances on title to the Land listed on Exhibit A hereto, and any others accepted by Bank, in advance of recordation.
- "Permitted Transfers": Shall mean and include: (i) such easements, covenants and restrictions as Borrower determines to be necessary or desirable in connection with the operation of the Project (as to which Bank shall cooperate in entering into reasonable subordination agreements with respect thereto at Borrower's request); (ii) Leases; (iii) so long as the Cash Collateral is on deposit in the Deposit Account, transfers by Indemnitor of equity interests in Borrower to a Person (other than a Blocked Person), provided that Indemnitor retains a degree of control over Borrower's business and affairs; and (iv) transfers of equity interests in Borrower to Affiliates of Borrower or Indemnitor.
- "Person": An individual, corporation, partnership, limited liability company, joint venture, trust or unincorporated organization, or a government or any agency or political subdivision thereof.
- "Pledge Agreement": That certain Cash Pledge Agreement of even date herewith from Borrower to Bank for the benefit of Bank, including any amendments thereof and supplements thereto

"Project": The Land, the Improvements and the Equipment.

- "Protective Advance": All necessary costs and expenses (including attorneys' fees and disbursements) incurred by Bank in order to remedy an Event of Default under the Loan Documents, which Event of Default, by its nature, may interfere with the enforceability or enforcement of the Loan Documents, or otherwise materially impair the payment of the Loan (including, without limitation, foreclosure costs, costs of collection, costs incurred in bankruptcy proceedings and other costs incurred in enforcing any of the Loan Documents.
- "Regulation D": Regulation D (or any substitute regulations) of the Board of Governors of the Federal Reserve System (or any successor thereto), together with all amendments from time to time thereto.
- "Related Party": Any one or more of the following: (a) Indemnitor, or (b) an Affiliate of Borrower or Indemnitor or (c) any of the partners, members or other equity holders of Borrower, Indemnitor or any Affiliate thereof.
- "Security Interest": Any lien, pledge, mortgage, encumbrance, charge or security interest of any kind whatsoever (including, without limitation, the lien or retained security title of a conditional vendor) whether arising under a security instrument or as a matter of law, judicial process or otherwise or the agreement by Borrower or any of its Subsidiaries to grant any lien, security interest or pledge, mortgage or encumber any asset.
- "<u>Subsidiary</u>": Any corporation or other entity of which more than 50% of the outstanding capital stock or interests having ordinary voting power to elect a majority of the board of directors or the board of governors or otherwise to control the activities of such entity (irrespective of whether or not at the time other class or classes of the equity of such entity shall or might have voting power upon the occurrence of any contingency) is at the time directly or indirectly owned by Borrower or by Indemnitor and one or more of their respective Subsidiaries, or by one or more other Subsidiaries.
  - "TAG Program": The Transaction Account Guarantee Program component of the FDIC Temporary Liquidity Guarantee Program, 12 CFR Part 370.
- "<u>USA Patriot Act</u>": The Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

## **LOAN**

- <u>Principal</u>. Simultaneously herewith and subject to the terms and conditions hereof, Bank has made an advance to Borrower in the amount of \$78,245,641.77 as further evidenced by the Note. The entire principal balance of the Note shall mature and be payable at the Maturity Date.
- <u>Interest.</u> Borrower shall pay to Bank interest on the Note computed at the Loan Rate. In the event that the interest and/or charges in the nature of interest, if any, provided for by this Agreement or by any other Loan Document, shall contravene a legal or statutory limitation applicable to the Loan, if any, Borrower shall pay only such amounts as would legally be

permitted; <u>provided</u>, <u>however</u>, that if the defense of usury and all similar defenses are unavailable to Borrower, Borrower shall pay all amounts provided for herein. If, for any reason, amounts in excess of the amounts permitted in the foregoing sentence shall have been paid, received, collected or applied hereunder, whether by reason of acceleration or otherwise, then, and in that event, any such excess amounts shall be applied to principal, unless principal has been fully paid, in which event such excess amount shall be refunded to Borrower.

Interest shall accrue on the Loan from and after the date the Loan is advance by the Bank to Borrower. Interest on the Note computed at the Loan Rate shall be payable, as accrued, on the first day of each calendar month, commencing on the first day of the next calendar month following the calendar month in which the Loan is made hereunder, and all unpaid, accrued interest shall be paid in full at the time the Loan is paid in full. Interest computed at the Loan Rate shall be computed on the basis of a 360 day year, but shall be charged for the actual number of days principal is unpaid. If the Loan has not been repaid on or before the Maturity Date or if an Event of Default occurs pursuant to this Agreement or any other Loan Document, then the entire unpaid balance of the Loan shall (without notice to or demand upon Borrower) become due and payable on said date, together with all unpaid, accrued interest thereon, and with interest computed at the Default Rate from and after that date until the Loan is paid in full. Interest at the Default Rate shall be payable on the first day of each calendar month or on demand, at Bank's option.

In the event that Borrower fails to make any required payment of principal or interest on the Note (other than the balloon payment at the Maturity Date) on or before the tenth (10th) day following the due date thereof, Borrower shall pay to each Bank, in addition to interest at the Default Rate, a late payment charge equal to four percent (4%) of the amount of the overdue payment, for the purpose of reimbursing such Bank for a portion of the expense incident to handling the overdue payment. This late charge shall apply individually to all payments past due and there will be no daily prorated adjustment. This provision shall not be deemed to excuse a late payment or be deemed a waiver of any other rights Bank may have including the right to declare the entire unpaid principal and/or interest immediately due and payable. Borrower agrees that the "late charge" is a provision for liquidated damages and represents a fair and reasonable estimate of the damages Bank will incur by reason of the late payment considering all circumstances known to Borrower and Bank on the date hereof. Borrower further agrees that proof of actual damages will be difficult or impossible.

## <u>Prepayment</u>.

- (a) From the Closing Date to and including June 10, 2009, the unpaid principal balance of the Note may not be prepaid.
- (b) After June 10, 2009, the unpaid principal balance of the Note and accrued interest thereon may be prepaid in full or in part, without premium or penalty (except as provided in Section 1.4 hereof), after at least three (3) Business Days' prior written notice from Borrower to Bank of the date of prepayment, either through (i) the payment by Borrower to Bank of immediately available federal funds (other than the Funds) sufficient to reduce the then outstanding principal balance of the Loan in full or in part, or (ii) so long as no Event of Default shall have occurred and be continuing, application of

all or a portion of the Funds then on deposit in the Deposit Account to the then outstanding principal balance of Loan. In the event that Borrower shall fail to provide such three (3) Business Days' notice when required herein, Bank will charge, and Borrower shall pay, additional interest on the amount prepaid, at the Loan Rate through the date three (3) Business Days after the date of prepayment. No Bank shall be obligated hereunder or under any of the other Loan Documents to re-advance to Borrower any sums prepaid by Borrower, whether prepaid voluntarily or involuntarily. In connection with any prepayment of the Loan in full, Bank shall cooperate in assigning the Note and Mortgage as contemplated by the defeasance clause of the Mortgage, that being the last grammatical paragraph of the Mortgage.

- Bank Losses. Except for a failure caused by Bank's default, Borrower shall indemnify Bank and/or any of the Bank against any loss or expense which Bank may sustain or incur as a consequence of (a) any failure of Borrower to make any payment when due of any amount due hereunder, (b) any prepayment of the Loan, other than as permitted hereunder, or (c) the occurrence of any Event of Default. Bank shall provide to Borrower a statement, signed by an officer of Bank, explaining any such loss or expense and setting forth, if applicable, the computation pursuant to the preceding sentence which, in the absence of manifest error, shall be conclusive and binding on Borrower.
- <u>Payments</u>. All payments and prepayments of principal of, and interest on, the Note and all fees, expenses and other obligations under the Loan Documents payable to Bank shall be made, without deduction, set off, or counterclaim, in immediately available funds not later than 2:00 o'clock p.m., Eastern time on the dates due, to Bank at the office specified by it from time to time, except as otherwise specifically provided in this Agreement. Funds received on any day after 2:00 o'clock p.m., Eastern time shall be deemed to have been received on the next Business Day. Whenever any payment to be made hereunder or on the Note shall be stated to be due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and such extension of time shall be included in the computation of any interest or fees. Borrower authorizes Bank to charge any of Borrower's accounts maintained at Bank for the amount of any payment or prepayment on the Note or other amount owing pursuant to any of the other Loan Documents.
- <u>Application of Payments</u>. All payments received by Bank for application to the principal, interest, fees, costs and expenses due to Bank shall be applied in the following order: (i) first, to any fees, costs and expenses due to Bank hereunder; (ii) second, to any unpaid interest then due to the Bank hereunder; and (iii) third, to the unpaid principal balance of the Note.

#### CONDITIONS OF BORROWING

Bank shall not be required to make Loan until the pre closing requirements, conditions and other requirements set forth below have been completed and fulfilled to the reasonable satisfaction of Bank, at Borrower's sole cost and expense.

• <u>Pre Closing Requirements</u>. On or prior to the date of closing of the Loan, Borrower shall provide to Bank each of the following, in form and substance acceptable to Bank:

- (a) A copy of the Existing Survey
- (b) The Environmental Reports.
- (c) A certificate of liability insurance showing Bank as an Additional Insured, together with an Additional Insured Endorsement, and a certificate of property and casualty insurance indicating that coverage is in place.
- (d) A copy of Borrower's Organizational Documents, certified as true, correct and complete by an officer of Borrower authorized to do so, together with (i) a current certificate of good standing from the jurisdiction in which Borrower was organized (and from the jurisdiction in which the Land is located, if different from the jurisdiction in which Borrower was organized), and (ii) resolutions and/or consents of those parties necessary to authorize the transaction contemplated hereby.
- (e) A copy of Indemnitors Organizational Documents, certified as true, correct and complete by an officer of Borrower authorized to do so, together with (i) a current certificate of good standing from the jurisdiction in which Indemnitor was organized, and (ii) resolutions and/or consents of those parties necessary to authorize the transaction contemplated hereby.
- (f) A flood zone certification from a consultant acceptable to Bank indicating that the Project is not located in a flood plain or any other flood-prone area as designated by any governmental agency; <u>provided</u>, <u>however</u>, that if the Project is so located, Borrower shall provide proof of flood insurance to Bank.
  - (g) Borrower shall have established the Noninterest-bearing Transaction Account and shall have made the Deposit with Bank.
  - (h) The Existing Note Allonge, together with originals or certified copies of the Existing Note.
  - (i) The Existing Mortgage Assignment, together with originals or certified copies of the Existing Mortgage.
- (j) All such other agreements, documents and/or exhibits which may be required, in Bank's judgment, to assure compliance with the requirements of this Agreement.
- <u>Loan Documents</u>. On or prior to the date of closing of the Loan, Borrower shall execute and deliver (or cause to be executed and delivered) to Bank the following documents in quantity, form and substance acceptable to Bank and to its counsel, to evidence and secure the Loan:
  - (a) The Note.
  - (b) The Mortgage.

- (c) A first security interest in all Equipment and in all of Borrower's intangible property relating to the Project, created and evidenced by a security agreement (which is incorporated within the Mortgage) and perfected by appropriate Uniform Commercial Code financing statements.
  - (d) The Assignment of Leases and Rents.
  - (e) The Indemnity executed by Borrower and Indemnitor, pertaining to compliance with Environmental Law.
  - (f) The Pledge Agreement executed by Borrower in favor of the Bank.
  - (g) The Good Faith Indemnity
  - (h) Such other ancillary documents as Bank may reasonably require to evidence and secure the Loan.

Bank may designate which of the Loan Documents are to be filed and/or placed of record, the order of filing and/or recording thereof, and the offices in which the same are to be filed and/or recorded. Borrower shall pay all filing, documentary, recording and/or registration taxes and/or fees, if any, due upon the Loan Documents.

- <u>Opinion of Borrower's Attorneys.</u> Bank shall have received from outside counsel for Borrower and Indemnitor a current written opinion, in form and substance acceptable to Bank, addressed to Bank.
- <u>Evidence of Recording</u>. Bank shall have received from outside counsel to the Borrower evidence, reasonably satisfactory to Bank, that the Existing Mortgage has been duly recorded and that the Existing Mortgage Assignment and the Mortgage have been executed and delivered to the title company for recording, with all recording fees and expenses paid, and that no exceptions title to the Project are shown of record other than Permitted Encumbrances, which may be in the form of a certificate of title or title bring down Chicago Title Insurance Company or another reputable, nationally recognized title insurer, reasonably acceptable to Bank.
- <u>Satisfaction of Closing Conditions</u>. Bank's making of the advance at the closing of the Loan shall constitute its agreement that Borrower has satisfied the above requirements of this <u>Article II</u>.

## ADVANCES OF LOAN PROCEEDS

• <u>General</u>. The Loan will be advanced in a single advance by Bank for the benefit of Borrower in accordance with the terms and conditions set forth in this <u>Article III</u>. All monies advanced by the Bank pursuant to this Agreement shall constitute a loan made to Borrower under this Agreement, evidenced by the Note and secured by the other Loan Documents, and interest shall be computed thereon, as prescribed by this Agreement and the Note, from the date the Loan account is charged with the amount of the advance.

• Inspections. Bank and its representatives shall have access to the Project at all reasonable times and upon reasonable notice and shall have the right to enter the Project and to conduct such inspections thereof as they shall deem reasonably necessary or desirable for the protection of Bank's interests. Bank may retain any consultants deemed necessary or desirable by Bank, at Borrower's expense, to make periodic inspections of the Project and to perform such services as may, from time to time, be required by Bank in connection with the Loan, this Agreement, the other Loan Documents or the Project, at no cost to Borrower, unless an Event of Default shall exist. Notwithstanding the foregoing to the contrary, Bank's rights under this Section 3.2 shall be subject to the rights of tenants under leases of the Project and, in each instance, may be exercised by Bank not more than one (1) time in any twelve (12) month period, unless an Event of Default shall exist.

#### REPRESENTATIONS AND WARRANTIES

Borrower represents, warrants and covenants to Bank and Bank that:

- <u>Borrower's Formation and Powers</u>. Borrower is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, and qualified and authorized to do business in all jurisdictions in which the conduct of its business and affairs requires it to be so qualified. Borrower has all power, authority, permits, consents, authorizations and licenses necessary to carry on its business, to own and operate the Project and to execute, deliver and perform its obligations under this Agreement and the other Loan Documents; all consents necessary to authorize the execution, delivery and performance of this Agreement and the other Loan Documents have been duly adopted and are in full force and effect; and this Agreement and the other Loan Documents have been duly executed and delivered by Borrower, and constitute valid and binding obligations of Borrower, enforceable in accordance with their respective terms.
- Authority. The execution, delivery and performance by Borrower of this Agreement and other Loan Documents to which Borrower is a party have been duly authorized by all necessary action and do not and will not (i) violate any provision of any laws, rule, regulation (including, without limitation, Regulation U of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award presently in effect having applicability to Borrower or of Borrower's Organizational Documents, (ii) result in a breach of or constitute a default under any indenture or loan or credit agreement or any other agreement, lease or instrument to which Borrower is a party or by which it or its properties may be bound or affected, or (iii) will not result in or require the creation or imposition of any Security Interest in any of its properties pursuant to the provisions of any agreement or other document binding upon or applicable to Borrower or any of its properties, except pursuant to the Loan Documents.
- <u>No Approvals.</u> No authorization, consent, approval, license, exemption of or filing or registration with any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or will be necessary to the valid execution, delivery or performance by Borrower of this Agreement, the Note, or any other Loan Documents to which Borrower is a party.

- <u>Legal and Valid Obligations</u>. This Agreement, the Note, the Indemnity, the Good Faith Indemnity, the Cash Pledge and the other Loan Documents to which Borrower is a party constitute the legal, valid and binding obligations of Borrower, enforceable against Borrower in accordance with their respective terms, subject to bankruptcy and insolvency laws and other laws generally affecting the enforceability of creditor's rights generally and subject to limitations on the availability of equitable remedies.
- <u>Litigation</u>. Except as to matters covered by insurance or other matters disclosed in writing to Bank from time to time by Borrower, there are no actions, suits or proceedings (whether or not purportedly on behalf of Borrower) pending or, to the knowledge of Borrower, threatened against Borrower or affecting any of the Project or its other assets (if any), at law or in equity or before any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, which contests the validity or enforceability of this Agreement or any of the other Loan Documents or the transactions contemplated hereby or as a result of which Borrower may become subject to any judgment or liability which if determined adversely to Borrower, would constitute a Material Adverse Occurrence as to Borrower. Borrower is not in default with respect to any final judgment, writ, injunction, decree, rule or regulations of any court, arbitrator or federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign.
- <u>Title to Land</u>. Borrower is the owner, in fee simple, of the Land, subject to no lien, charge, mortgage, deed of trust, restriction or encumbrance, except Permitted Encumbrances. Upon delivery of the Existing Note Allonge and the Existing Mortgage Assignment, Bank shall be the owner and holder of the Existing Mortgage and, upon due recording of the Existing Mortgage Assignment and the Mortgage in the Land Records, Bank shall be the owner and holder of a first-priority mortgage of the Project, subject only to the Permitted Encumbrances. The legal description of the Land set forth on Exhibit A and as contained in the Mortgage is one and the same as the Land shown on the Existing Survey
- Payment of Taxes. There have been filed all federal, state and local tax returns with respect to Borrower and its direct and indirect business operations which are required to be filed. Borrower has paid or caused to be paid to the respective taxing authorities all taxes as shown on such returns or on any assessments received by it to the extent that such taxes have become due. Borrower knows of no proposed material tax assessment against Borrower, and Borrower is not obligated by any other agreement, tax treaty, instrument or otherwise to contribute to the payment of taxes owed by any other person or entity. All material tax liabilities are adequately provided for or reserved against on the books of Borrower.
- <u>Federal Reserve Regulations</u>. No portion of the Loan hereunder will be used to purchase or carry any "margin stock" as defined in Regulation U of the Board of Governors of the Federal Reserve System of the United States or for the purpose of reducing or retiring any indebtedness which was originally incurred to purchase or carry any margin security or for any other purpose which might constitute this transaction a "purpose credit" within the meaning of said Regulation U. No portion of the Loan hereunder will be used for any purpose that violates, or which is inconsistent with, the provisions of Regulation X of the Board of Governors of the Federal Reserve System or any other regulation of said Board of Governors.

- Investment Company Act. Borrower is not an "investment company," or an "affiliated person" of, or a "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended. The making of the Loan, the application of the proceeds and repayment thereof by Borrower and the performance of the transactions contemplated by this Agreement will not violate any provision of said Act, or any rule, regulation or order issued by the Securities and Exchange Commission thereunder.
- <u>Unregistered Securities</u>. Borrower has not: (a) issued any unregistered securities in violation of the registration requirements of Section 5 of the Securities Act of 1933, as amended, or any other law; or (b) violated any rule, regulation or requirement under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, in either case where the effect of such violation would constitute a Material Adverse Occurrence as to Borrower.
- <u>Accuracy of Information</u>. All factual information heretofore or herewith furnished by or on behalf of Borrower to Bank for purposes of or in connection with this Agreement or any transaction contemplated hereby is true and accurate in every material respect on the date as of which such information is dated or certified and no such information contains any material misstatement of fact or omits to state a material fact or any fact necessary to make the statements contained therein not misleading as of such date.

## • <u>Compliance</u>. Borrower:

- (i) is in compliance and conformity with all Governmental Requirements the violation of which, individually or in the aggregate, would constitute a Material Adverse Occurrence as to Borrower; and
- (ii) has not received and do not anticipate the receipt of any order or notice of any violation or claim of violation of any Governmental Requirement which would constitute a Material Adverse Occurrence as to Borrower.

## Anti-Terrorism Regulations.

- (a) <u>General</u>. None of Borrower, Indemnitor or any Affiliate thereof is in violation of any Anti-Terrorism Law or engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.
- (b) <u>Executive Order No. 13224</u>. None of the Loan Parties, or any Affiliate of any Loan Party, or their respective agents acting or benefiting in any capacity in connection with the Loan or other transactions hereunder, is any of the following (each a "**Blocked Person**"):
  - (i) a Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;

- (ii) a Person owned or controlled by, or acting for or on behalf of, any Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;
- (iii) a Person or entity with which Bank is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law;
- (iv) a Person or entity that commits, threatens or conspires to commit or supports "terrorism" as defined in Executive Order No. 13224:
- (v) a Person or entity that is named as a "specially designated national" on the most current list published by the U.S. Treasury Department Office of Foreign Asset Control at its official website or any replacement website or other replacement official publication of such list; or
  - (vi) a person or entity who is affiliated or associated with a person or entity listed above.
- (c) None of Borrower, Indemnitor or any Affiliate thereof, nor any of their agents acting in any capacity in connection with the Loan or other transactions hereunder (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person, or (ii) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224.
- (d) Neither Borrower nor any Affiliate thereof, nor any person owning a direct interest therein, are a "Special Designated National" or "Blocked Person" as those terms are defined in the office of Foreign Asset Control Regulations (31 C.F.R. § 500 et. seq.)
- <u>Subsidiaries</u>. Borrower has no Subsidiaries.
- Rent Roll. The rent roll attached hereto as <u>Exhibit C</u> is true and correct in all material respects.
- <u>Ownership and Control of Borrower</u>. As of the date of this Agreement, Borrower is wholly owned by Alexander's, Inc., a Delaware corporation and a public company.

## **COVENANTS OF BORROWER**

While this Agreement is in effect, and until Bank has been paid in full the principal of and interest on the Loan, Borrower agrees to comply with, observe and keep the following covenants and agreements:

<u>Using Loan Proceeds</u>. Borrower shall use the Loan proceeds to refinance the Existing Mortgage.

- <u>Keeping of Records</u>. Borrower shall set up and maintain accurate and complete books, accounts and records pertaining to the Project in a manner reasonably acceptable to Bank. Borrower will permit representatives of Bank, at reasonable times and upon reasonable notice, to have access to and to inspect and copy such books, records and contracts of Borrower and to inspect the Project and to discuss Borrower's affairs, finances and accounts with any of its principal officers, all at such times and as often as may reasonably be requested, at no cost to Borrower, unless an Event of Default shall exist. Any such inspection by Bank or its representatives shall be for the sole benefit and protection of Bank and Bank, and Bank shall have no obligation to disclose the results thereof to Borrower or to any third party. Notwithstanding the foregoing to the contrary, Bank shall not be entitled to exercise its rights under this <u>Section 5.2</u> more than one (1) time in any twelve (12) month period, unless an Event of Default shall exist.
  - <u>Maintaining Insurance Coverage</u>. Borrower shall maintain insurance as provided in <u>Section 1.11</u> of the Mortgage.
- <u>Transferring, Conveying or Encumbering the Project</u>. Borrower shall not make, or permit to be made, any transfer, conveyance or encumbrance restricted under <u>Section 1.15</u> of the Mortgage.
  - <u>Reporting Requirements</u>. Borrower shall furnish or, as appropriate, cause Indemnitor to furnish to Bank the following:
  - (a) <u>Borrowers' Financial Statements</u>. As soon as available and in any event within 120 days after the end of each calendar year, a balance sheet and related statements of income and cash flow of Borrower, as at the end of and for such Fiscal Year, prepared in accordance with the reconciliation of Bank (as applied consistently with the audited statements, if any, previously delivered to Bank), unqualifiedly certified by an officer of the Borrower. If requested by Bank following an Event of Default under this Agreement, such annual financial statements of Borrower shall be audited by an independent certified public accountant acceptable to Bank and prepared on a GAAP basis (or another accounting basis reasonably acceptable to Bank) consistently applied.
  - (b) <u>Litigation and Other Proceedings</u>. Promptly in writing, notice of (i) all litigation against Borrower or Indemnitor in which the amount sought to be recovered exceeds \$10,000,000, except in cases when the claim is covered by insurance and the insurance company has agreed to assume the defense of the claim and (ii) all proceedings before any governmental or regulatory agency affecting Borrower or Indemnitor which, if adversely determined, would constitute a Material Adverse Occurrence as to such party.
  - (c) <u>Reporting of Certain Additional Matters</u>. Promptly after obtaining knowledge, Borrower shall notify Bank of each of these: (a) condition or event that has resulted in or may result in any Material Adverse Occurrence; (b) transfer of fee title to any material portion of the Project or any direct equity interest in Borrower, including a Permitted Transfer; or (c) insolvency, dissolution, liquidation, or other material adverse event affecting Borrower or Indemnitor.

- <u>Maintain Existence</u>. Borrower shall preserve and maintain, and cause Indemnitor to preserve and maintain, its existence, rights and privileges in the jurisdiction of its organization and qualify and remain qualified in each jurisdiction in which such qualification is necessary in view of its business and operations.
- <u>Notice</u>. Borrower shall give prompt written notice to Bank of any action or proceeding instituted by or against Borrower, in any federal or state court or before or by any commission or other regulatory body, federal, state or local, or any such proceedings threatened against Borrower which, if adversely determined, would be a Material Adverse Occurrence as to Borrower.
- <u>Merger and Consolidation</u>. Borrower shall not merge or consolidate into any Person or permit any other Person to merge into it, or acquire (in a transaction analogous in purpose or effect to a consolidation or merger) all or substantially all of the assets of any other Person, except that Borrower shall be permitted to consolidate or merge with an Affiliate provided Borrower takes such steps as Bank may reasonably require to ensure that Bank's lien against the Project is not thereby impaired.
- Patriot Act. Borrower, Indemnitor and their respective Affiliates and agents shall not (i) conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person; (ii) deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; or (iii) engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Executive Order No. 13224, the USA Patriot Act or any other Anti-Terrorism Law. Borrower shall deliver to Bank any certification or other evidence reasonably requested from time to time by Bank, confirming Borrower's compliance with this Section.

#### **DEFAULTS**

- Events of Default. Any of the following events shall constitute an Event of Default under this Agreement:
- (a) Borrower shall default in any payment of principal or interest due according to the terms hereof or of the Note, and such default shall remain uncured for a period of five (5) days after the payment became due; provided that so long as the Cash Collateral shall remain on deposit with Bank, no Event of Default shall be deemed to have occurred if Borrower cures such default within five (5) days after notice from Bank thereof;
- (b) Borrower shall default in the payment of fees or other amounts payable to Bank hereunder other than as set forth in subsection (a) above, and such default continues unremedied for a period of ten (10) days after notice from Bank to Borrower thereof;

- (c) Borrower shall default in the performance or observance of any agreement, covenant or condition required to be performed or observed by Borrower under the terms of this Agreement, other than a default described elsewhere in this <u>Section 6.1</u>, and such default continues unremedied for a period of thirty (30) days after notice from Bank to Borrower thereof, or if such default is not reasonably capable of cure within such 30 day period, then such cure period shall be extended for so long as Borrower is diligently pursuing such cure;
- (d) Any representation or warranty made by Borrower in this Agreement or by Borrower or Indemnitor in any of the other Loan Documents, or in any certificate or document furnished under the terms of this Agreement or in connection with the Loan, shall be untrue or incomplete in any material respect when made, if same would be a Material Adverse Occurrence;
- (e) Borrower shall be in default under any term, covenant or condition of any of the Note or of any of the other Loan Documents, other than a default described elsewhere in this Section 6.1, after the expiration of 30 days' notice provided by Bank, or if such default is not reasonably capable of cure without such 30 day period, then such cure period shall be extended for so long as Borrower is diligently pursuing such cure;
- (f) Any of Borrower or Indemnitor shall commit an act of bankruptcy; or shall apply for, consent to or permit the appointment of a receiver, custodian, trustee or liquidator for it or any of its property or assets; or shall generally fail to, or admit in writing its inability to, pay its debts as they mature; or shall make a general assignment for the benefit of creditors or shall be adjudicated bankrupt or insolvent; or shall take other similar action for the benefit or protection of its creditors; or shall give notice to any governmental body of insolvency of pending insolvency or suspension of operations; or shall file a voluntary petition in bankruptcy or a petition or an answer seeking reorganization or an arrangement with creditors, or to take advantage of any bankruptcy, reorganization, insolvency, readjustment of debt, rearrangement, dissolution, liquidation or other similar debtor relief law or statute; or shall file an answer admitting the material allegations of a petition filed against it in any proceeding under any such law or statute; or shall be dissolved, liquidated, terminated or merged; or shall effect a plan or other arrangement with creditors; or a trustee, receiver, liquidator or custodian shall be appointed for it or for any of its property or assets and shall not be discharged within sixty (60) days after the date of his appointment; or a petition in involuntary bankruptcy or similar proceedings is filed against it and is not dismissed within sixty (60) days after the date of its filing;
- (g) A default occurs in the performance of Borrower's obligations in any of <u>Section 5.4</u> ("Transferring, Conveying or Encumbering the Project") hereof;
- <u>Rights and Remedies</u>. Upon the occurrence of an Event of Default, unless such Event of Default is subsequently waived in writing by Bank, Bank may exercise any or all of the following rights and remedies, consecutively or simultaneously, and in any order:

- (a) declare the entire unpaid principal balance of the Note to be immediately due and payable, together with accrued and unpaid interest on the Loan, without notice to or demand on Borrower;
  - (b) draw upon the Cash Collateral in accordance with the terms of the Pledge Agreement;
- (c) exercise any or all remedies specified herein and in the other Loan Documents, including (without limiting the generality of the foregoing) the right to foreclose the Mortgage, and/or any other remedies which it may have therefor at law, in equity or under statute; and/or
- (d) cure the Event of Default on behalf of Borrower, and, in doing so, enter upon the Project, and expend such sums as it may deem desirable, including reasonable attorneys' fees, all of which shall be deemed to be advances hereunder, even though causing the Loan to exceed the face amount of the Note, shall bear interest at the Default Rate provided herein and shall be payable by Borrower on demand.

In addition to the other remedies set forth herein and in the other Loan Documents, Borrower hereby irrevocably authorizes each Bank, at any time while an Event of Default continues, to set off any sum due to or incurred by Bank against all deposits and credits of Borrower with, and any and all claims of Borrower against, such Bank (subject, however, to any limitations on the rights of Borrower with respect to any impounds). Such right shall exist whether or not Bank shall have made any demand hereunder or under any other Loan Document, whether or not said sums, or any part thereof, or deposits and credits held for the account of Borrower is or are matured or unmatured, and regardless of the existence or adequacy of any collateral, guaranty or any other security, right or remedy available to Bank. Bank agrees that, as promptly as is reasonably possible after the exercise of any such setoff right, it shall notify Borrower of its exercise of such setoff right; provided, however, that the failure of Bank to provide such notice shall not affect the validity of the exercise of such setoff rights. Nothing in this Agreement shall be deemed a waiver or prohibition of or restriction on Bank to all rights of banker's lien, setoff and counterclaim available pursuant to law.

## INTEREST, FEES AND EXPENSES

• Authorization to Make Loan Advances to Cure Borrower's Defaults. If an Event of Default shall occur, Bank may (but shall not be required to) perform any of such covenants and agreements with respect to which Borrower is in Default and of which Bank has notified Borrower. Any amounts expended by Bank in so doing and any amounts expended by Bank in connection therewith shall constitute a Loan and be added to the outstanding principal amount of the Loan, and the Bank shall make the applicable Loans to fund any such disbursements. The authorization hereby granted is irrevocable and no prior notice to or further direction or authorization from Borrower is necessary for Bank to make such disbursements.

#### DEPOSIT ACCOUNT

- Interest Yield. If and when all or any portion of the Funds is in the Interest-bearing Investment Account, to the extent that for any period the calculation of interest earned based upon the Federal Funds Rate in calculating the Deposit Rate differs from the calculation of interest payable on the Loan based upon the Federal Funds Rate in calculating the Loan Rate, and such calculation results in the Bank netting more than seventy-five (75) basis points (.75%), Bank shall, from time to time, adjust the Loan Rate or the Deposit Rate, such that that the economic benefit to Bank is substantially the same as if such Funds were on deposit in the Noninterest-bearing Transaction Account, a seventy-five (75) basis points (.75%) net to the Bank.
- Offset Against Mortgage Debt. If, (i) all or any portion of the Funds are not guaranteed, in full, by the FDIC, either due to either the expiration of the TAG Program, the election by US Bank, to the extent permitted by law, to opt-out of the TAG Program, the election by Borrower to transfer the Funds from the Noninterest-bearing Transaction Account to the Interest-bearing Investment Account pursuant to the Pledge Agreement, or otherwise and (ii) there is a failure of, liquidation of, or other closing or winding up of the affairs of, US Bank, resulting in the loss of some or all of the Funds, Borrower may, in any such instance, at its option, be entitled to assert an offset against the Mortgage debt, equal, dollar for dollar, to the amount of Funds in the Deposit Account, and Bank shall thereupon deliver such assignments or satisfactions and releases as Borrower may request in connection with the foregoing. Upon the effective exercise of any such offset as to all or any portion of any such lost Funds, Borrower shall be deemed to have waived and relinquished any further claim, right, entitlement or remedy with respect to such Funds.

## **MISCELLANEOUS**

• <u>Waiver and Amendment</u>. No failure on the part of Bank or the holder of the Note to exercise and no delay in exercising any power or right hereunder or under any other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise of any power or right preclude any other or further exercise thereof or the exercise of any other power or right. The remedies herein and in any other instrument, document or agreement delivered or to be delivered to Bank hereunder or in connection herewith are cumulative and not exclusive of any remedies provided by law. No notice to or demand on either party hereunder not required hereunder or under the Note or any other Loan Document shall in any event entitle such party to any other or further notice or demand in similar or other circumstances or constitute a waiver of the right of Bank, the holder of the Note or Borrower to any other or further action in any circumstances without notice or demand.

No amendment, waiver or consent shall affect the rights or duties of Bank under this Agreement or any other Loan Document unless it is in writing and signed by Bank.

<u>Expenses and Indemnities</u>.

- (a) <u>Loan Documents</u>. Borrower shall also pay all reasonable costs and expenses of Bank and Borrower in connection with the Project, the preparation and review of the Loan Documents and the making, closing, amendment, and/or repayment of the Loan, including but not limited to the reasonable fees of Bank's attorneys, the fees of Bank's Consultants, disbursement expenses, and all other costs and expenses payable to third parties incurred by Bank or Borrower in connection with the Loan. Such costs and expenses shall be so paid by Borrower whether or not the Loan is fully advanced or disbursed. Borrower agrees to pay and reimburse Bank and each Bank upon demand for all reasonable expenses paid or incurred by Bank or such Bank (including reasonable fees and expenses of legal counsel) in connection with the collection and enforcement of the Loan Documents. Borrower agrees to pay, and save each Bank (including Bank) harmless from all liability for, any mortgage registration, mortgage recording, transfer, recording, stamp, like tax or other charge due to any governmental entity, which may be payable with respect to the execution or delivery of the Loan Documents.
- (b) General Indemnity. In consideration of the Loan, Borrower further agrees to indemnify and defend Bank and its directors, officers, agents and employees (the "Indemnified Parties") from, and hold each of them harmless against, any and all losses, liabilities, claims, damages, deficiencies, interest, judgments, costs or expenses incurred by them or any of them, including, but without limitation, amounts paid in settlement, court costs, and reasonable fees and disbursements of counsel incurred in connection with any investigation, litigation or other proceeding, arising out of or by reason of any investigation, litigation or other proceeding brought or threatened, arising out of or by reason of their execution of any Loan Document and the transaction contemplated thereby, including, but not limited to, any use effected or proposed to be effected by Borrower of the proceeds of the Loan, but excluding any such losses, liabilities, claims, damages or expenses incurred by reason of the gross negligence or willful misconduct of the relevant Indemnified Party ore resulting from the matters set forth in Section 8.2 hereof. Any Indemnified Party seeking indemnification under this Section will notify Borrower of any event requiring indemnification within thirty (30) Business Days following such Indemnified Party's receipt of notice of commencement of any action or proceeding, or such Indemnified Party's obtaining knowledge of the occurrence of any other event, giving rise to a claim for indemnification hereunder. Borrower will be entitled (but not obligated) to assume the defense or settlement of any such action or proceeding or to participate in any negotiations to settle or otherwise resolve any claim using counsel of its choice; provided that:
  - (i) Borrower notifies such Indemnified Party in writing that Borrower will indemnify such Indemnified Party from and against the relevant claim;
    - (ii) such counsel is reasonably satisfactory to such Indemnified Party;
    - (iii) such claim involves only money damages and does not seek an injunction or other equitable relief;
  - (iv) if such Indemnified Party is Bank, settlement of, or an adverse judgment with respect to, such claim is not, in the good faith judgment of such

Indemnified Party, likely to establish a precedential custom or practice materially adverse to the continuing business interests of such Indemnified Party;

- (v) Borrower conducts the defense of such claim actively and diligently;
- (vi) no conflict of interest has arisen which would prevent counsel for Borrower from also representing such Indemnified Party because the defendants in any action include both such Indemnified Party and Borrower; and
- (vii) Borrower will not consent to the entry of any judgment or enter into any settlement with respect to such claim without the prior written consent of such Indemnified Party (not to be withheld unreasonably).

So long as Borrower has assumed the defense of such claim and is conducting such defense in accordance with the foregoing, such Indemnified Party: (x) may retain separate co-counsel at its sole cost and expense and participate in the defense of such claim; (y) will not consent to the entry of any judgment or enter into any settlement with respect to such claim without the prior written consent of Borrower with respect to such claim (not to be withheld unreasonably).

If Borrower fails to assume such defense or, after doing so, Borrower fails to satisfy any of the above conditions to Borrower's defense, such Indemnified Party (and its counsel) may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, such claim in any manner it may reasonably deem appropriate (and such Indemnified Party need not consult with, or obtain any consent from, any Borrower in connection therewith) and Borrower will reimburse such Indemnified Party promptly and periodically for the costs of defending against such claim (including reasonable attorneys' fees and expenses) and Borrower will remain responsible for any loss which such Indemnified Party may suffer resulting from, arising out of, relating to, in the nature of, or caused by such claim to the fullest extent provided for and required by this Agreement.

- <u>Binding Effect; Waivers; Cumulative Rights and Remedies</u>. The provisions of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective heirs, executors, administrators, personal representatives, legal representatives, successors and assigns. No delay on the part of Bank in exercising any right, remedy, power or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder constitute such a waiver or exhaust the same, all of which shall be continuing. The rights and remedies of Bank specified in this Agreement shall be in addition to, and not exclusive of, any other rights and remedies which Bank would otherwise have at law, in equity or by statute, and all such rights and remedies, together with Bank's rights and remedies under the other Loan Documents, are cumulative and may be exercised individually, concurrently, successively and in any order.
- <u>Incorporation By Reference</u>. Borrower agrees that until this Agreement is terminated by the repayment to Bank of all principal and interest due and owing on the Note and

other sums due and owing pursuant to the other Loan Documents, the Note and the other Loan Documents shall be made subject to all the terms, covenants, conditions, obligations, stipulations and agreements contained in this Agreement to the same extent and effect as if fully set forth in and made a part of the Note and the other Loan Documents. In the event of a conflict between any of the Loan Documents and the provisions of this Agreement, this Agreement shall be controlling.

- <u>Survival</u>. All agreements, representations and warranties made in this Agreement shall survive the execution of this Agreement (but in the case of representations and warranties, shall survive as of the date made), the making of the Loan by Bank, and the execution of the other Loan Documents, and shall continue until Bank receive payment in full of all Indebtedness of Borrower incurred under this Agreement and under the other Loan Documents.
- Governing Law; Waiver of Jury Trial; Jurisdiction. IN ALL RESPECTS, INCLUDING, WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, THIS AGREEMENT AND THE OBLIGATIONS ARISING HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, APPLICABLE TO CONTRACTS MADE AND PERFORMED IN SUCH STATE (WITHOUT REGARD TO PRINCIPLES OF CONFLICT LAWS) AND ANY APPLICABLE LAW OF THE UNITED STATES OF AMERICA.

BORROWER AND BANK EACH HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION RELATING TO THE LOAN AND/OR THE LOAN DOCUMENTS. AT THE OPTION OF BANK, THIS AGREEMENT, THE Note AND THE OTHER LOAN DOCUMENTS MAY BE ENFORCED IN THE UNITED STATES DISTRICT COURT FOR THE DISTRICT IN WHICH THE PROJECT LIES; BORROWER CONSENTS TO THE JURISDICTION AND VENUE OF ANY SUCH COURT AND WAIVES ANY ARGUMENT THAT VENUE IN SUCH FORUMS IS NOT CONVENIENT. IN THE EVENT AN ACTION IS COMMENCED IN ANOTHER JURISDICTION OR VENUE UNDER ANY TORT OR CONTRACT THEORY ARISING DIRECTLY OR INDIRECTLY FROM THE RELATIONSHIP CREATED BY THIS AGREEMENT, BANK OR BANKS AT THEIR OPTION SHALL BE ENTITLED TO HAVE THE CASE TRANSFERRED TO ONE OF THE JURISDICTIONS AND VENUES ABOVE DESCRIBED, OR IF SUCH TRANSFER CANNOT BE ACCOMPLISHED UNDER APPLICABLE LAW, TO HAVE SUCH CASE DISMISSED WITHOUT PREJUDICE.

- <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, all of which shall constitute a single Agreement.
- Notices. All notices, demands, requests, consents, approvals or other communications (any of the foregoing, a "Notice") required, permitted or desired to be given hereunder shall be in writing and shall be sent delivered by (a) registered or certified mail, postage prepaid, return receipt requested, (b) Federal Express, Airborne or another reputable overnight courier, or (c) delivered by hand by commercial courier service, addressed to the party to be so notified at its address set forth opposite its signature, below, or to such other address as such party may hereafter specify in accordance with the provisions of this Section 9.7. Any

Notice shall be deemed to have been received: (a) three (3) days after the date such Notice is mailed, unless not actually received then (b) on the date of delivery by hand (or refusal to accept such delivery) if delivered during business hours on a Business Day (otherwise on the next Business Day), and/or (c) on the next Business Day if sent by an overnight commercial courier. Notices shall be deemed effective if delivered by counsel to either party, as if given directly by such party.

Any party may change the address to which any such Notice is to be delivered by furnishing ten (10) days prior written notice of such change to the other parties in accordance with the provisions of this Section 9.8. Notices shall be deemed to have been given on the date as set forth above, even if there is an inability to actually deliver any such Notice because of a changed address of which no Notice was given, or there is a rejection or refusal to accept any Notice offered for delivery.

- <u>No Third Party Reliance</u>. No third party shall be entitled to rely upon this Agreement or to have any of the benefits of Bank's interest hereunder, unless such third party is an express assignee of all or a portion of Bank's interest hereunder.
- <u>Time of the Essence</u>. Time is of the essence hereof with respect to the dates, terms and conditions of this Agreement, subject, however, to the notice, grace and/or cure periods set forth herein.
- <u>No Oral Modifications</u>. No modification or waiver of any provision of this Agreement shall be effective unless set forth in writing and signed by the parties hereto.
- <u>Captions</u>. The headings or captions of the Articles and Sections set forth herein are for convenience only, are not a part of this Agreement and are not to be considered in interpreting this Agreement.
- <u>Borrower-Bank Relationship</u>. The relationship between Borrower and Bank created hereby and by the other Loan Documents shall be that of a borrower and a Bank only, and in no event shall Bank and Bank be deemed to be a partner of, or a joint venturer with, Borrower.

[Signature page follows]

BORROWER:	ALEXANDER'S REGO SHOPPING CENTER, INC.
Address:	
c/o Vornado Realty Trust	Ву:
210 Route 4 East	Name:
Paramus, New Jersey 07652	Title:
Attn: Chief Financial Officer	
With a copy to,	
c/o Vornado Realty Trust	
888 Seventh Avenue	
New York, New York 10019	
Attn: Exec. VP, Capital Markets	
With a copy to the same address,	
Attn: Exec. VP, Retail	
With a copy to:	
Winston & Strawn LLP	
200 Park Avenue	
New York NY 10166-4193	

Attn: Patricia M. Dineen, Esq.

[Signature page to Loan Agreement]

## BANK AND A BANK:

Address:

U.S. Bank National Association One Post Office Square, 29<sup>th</sup> Floor Boston, Massachusetts 02109 Attention: Real Estate Banking Division

With a copy to:

U.S. Bank National Association 1650 Tysons Boulevard, Suite 1580 McLean, Virginia 22102

Attention: Real Estate Banking Division

With a copy to: Halloran & Sage LLP One Goodwin Square 225 Asylum Street Hartford, Connecticut 06103 Attention: James P. Maher, Esq.

U.S. BANK NATIONAL ASSOCIATION a national banking
association

By:	
<i>5</i> -	David Heller, Vice President

[Signature page to Loan Agreement]

## EXHIBIT A

## Legal Description

#### Exhibit A

## LEGAL DESCRIPTION

ALL that certain plot, piece or parcel of land, situate, lying and being in the Borough and County of QUERNS, City and State of NEW YORK, bounded and described as follows:

BEGINNING at the corner formed by the intersection of the easterly side of JUNCTION BOULEVARD (80 feet wide) and the southerly side of 62ND DRIVE, formerly URGULART STREET, as said Boulevard and Drive are shown on the Topographical Map of the City of New York for the Borough of Queens;

running thence easterly along the southerly side of 62ND DRIVE, 456.35 feet to the corner formed by the intersection of the said 62ND DRIVE and the westerly side of 97TH STREET (60 feet wide), as said 97TH STREET was shown on the Topographical Map of the City of New York for the Borough of Queens, prior to the Adoption on December 20, 1951, of Alteration Map No. 3530;

thence southerly along the westerly side of said 97TH STRBET, 529.99 to the northwesterly side of 63RD ROAD (80 feet wide) as said 63RD ROAD is shown on the Topographical Map of the City of New York for the Borough of Queens;

thence southwesterly along the northwesterly side of 63RD ROAD, 406.22 feet to the corner formed by the intersection of the said northwesterly side of 63RD ROAD and the northeasterly side of QUBENS BOULEVARD (200 feet wide) as shown on the Final Topographical Map of the City of New York;

thence northwesterly along the northeasterly side of QUEENS BOULEVARD, 86 feet to the corner formed by the intersection of the said northeasterly side of QUEENS BOULEVARD and the easterly side of said JUNCTION BOULEVARD;

thence northerly along the easterly side of JUNCTION BOULEVARD, 549.82 feet to the corner at the point or place of BEGINNING.

FOR INFORMATION ONLY: BLOCK: 2084 LOT: 101

-LEGAL DESCRIPTION-

SADAL

## Permitted Encumbrances

- 1. Survey made by Gerald T. O'Buckley dated April 20, 1999 shows premises improved by a 2 story and basement brick store with 2 story and basement brick addition and masonry garage and the following:
  - a. Projections on 62nd Drive: lights up to 2.2 feet and stand pipe up to 0.4 feet.
  - b. Projections on 97th Street: signs up to 1.5 feet; lights up to 0.3 feet; stand pipe up to 0.4 feet.
  - c. Projections on 63rd Road: lights up to 1.0 foot and signs up to 1.0 foot.
  - d. Projections on Junction Boulevard: lights up to 1.0 foot; copings up to 0.75 feet; signs up to 1.0 foot and stand pipes up to 0.4 feet.

Subject to any changes that would be shown on an accurate survey made since April 20, 1999.

- 2. Covenants and Restrictions recorded in Liber 2690 Page 430.
- 3. Covenants and Restrictions recorded in Liber 2664 Page 420.
- 4. Covenants and Restrictions recorded in Liber 2689 Page 115.
- 5. Covenants and Restrictions recorded in Liber 2666 Page 180.
- 6. Covenants and Restrictions recorded in Liber 2686 Page 284,
- 7. Covenants and Restrictions recorded in Liber 2664 Page 432.
- 8. Covenants and Restrictions recorded in Liber 2664 Page 441.
- 9. Covenants and Restrictions recorded in Liber 2664 Page 442.
- 10. Covenants and Restrictions recorded in Liber 2664 Page 455.
- 11. Covenants and Restrictions recorded in Liber 2664 Page 457.
- 12. Covenants and Restrictions recorded in Liber 2664 Page 467.
- 13. Covenants and Restrictions recorded in Liber 2664 Page 468.
- 14. Covenants and Restrictions recorded in Liber 2666 Page 195.
- 15. Covenants and Restrictions recorded in Liber 2668 Page 419.

- 16. Covenants and Restrictions recorded in Liber 2668 Page 422.
- 17. Covenants and Restrictions recorded in Liber 2668 Page 425.
- 18. Covenants and Restrictions recorded in Liber 2825 Page 5.
- 19. Declaration recorded in Liber 3624 Page 89.
- 20. Consent and Authorization for the Construction, Maintenance and Operation of a Subway Staircase set forth in Liber 3448 Page 454.
- 21. Easement to the BROOKLYN UNION GAS COMPANY recorded in Liber 6805 Page 1. (Affects the roadbed of 62nd Drive on the south side between Junction Boulevard and 99th Street).
- 22. Water Main Easement to the CITY OF NEW YORK recorded in Liber 7442 Page 456. (Affects 97th Street which includes the easterly 10 feet of Parcel).
- 23. Water Main Easement to the CITY OF NEW YORK recorded in Liber 7442 Page 460. (Affects 62nd Drive)
- 24. Terms, Covenants and Conditions of the Indenture recorded in Liber 6297 Page 149, as amended by Liber 6297 Page 159 and by Liber 6800 Page 84. Pertain to the Elimination of Streets.
- 25. Declaration of Restrictions recorded in Liber 6696 Page 40, as amended by Agreement recorded in Liber 7744 Page 339.
- 26. Agreement for Sewage Pumping Station recorded in Liber 6696 Page 45.
- 27. Terms, Covenants and Conditions of the Declaration of Covenants and Restrictions recorded in Reel 926 Page 1291, as amended by Agreement in Reel 2341 Page 1794.
- 28. Terms, Covenants and Conditions of the Easement Agreement recorded in Reel 926 Page 1282.
- 29. Terms, Covenants and Conditions of the Agreement recorded in Reel 932 Page 207.
- 30. Terms, Covenants and Conditions of the Subway Entrance Agreement in Reel 2342 Page 2288.
- 31. Easement for Parking (Block 2080, Lot 1) to ALEXANDER'S INC., dated 3/29/1995, recorded 3/30/1995 in Reel 4097 Page 818.
- 32. Waiver of Legal Grade in Reel 4165 Page 2009.

 Terms, Covenants, Conditions and Provisions of the Lease dated 3/1/1995 between ALEXANDER'S, INC. (landlord) and MARSHALLS OF RICHFIELD, MN, INC., (tenant), a Memorandum of Lease having been recorded in Reel 4110 Page 751.

With Regard Thereto:

Subordination, Non-disturbance and Attornment Agreement made by and between UNION BANK OF SWITZERLAND (New York Branch), ALEXANDER'S INC. and MARSHALLS OF RICHFIELD, MN., INC. dated 3/29/95, recorded 12/2/96 in Reel 4474 Page 1385.

- 34. Terms, Covenants, Conditions and Provisions of the Lease dated 12/1/1992 between ALEXANDER'S INC., landlord and the CALDOR CORPORATION, tenant, a Notice of Lease having been recorded on 12/9/1992 in Reel 3458 Page 680.
- 35. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between the CALDOR CORPORATION and UNION BANK OF SWITZERLAND (New York Branch) as agent, dated March 29, 1995 and recorded March 30, 1995 in Reel 4097 Page 866.
- 36. Terms, Covenants, Conditions and Agreement of Lease from ALEXANDER'S, INC. (landlord) and SEARS ROEBUCK AND CO. (tenant) dated March 14, 1994 and recorded April 29, 1994 in Reel 3860 Page 638.
- 37. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between SEARS ROEBUCK AND CO. and UNION BANK OF SWITZERLAND (New York Branch) as Agent dated March 29, 1995 and recorded March 30, 1995 in Reel 4097 Page 880, as assigned to THE CHASE MANHATTAN BANK by Reel 5263 Page 2297
- 38. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between MARSHALLS OF RICHFIELD, MN, INC. and UNION BANK OF SWITZERLAND (New York Branch) as Agent dated March 23, 1995 and recorded March 30, 1995 in Reel 4097 Page 891, as assigned to THE CHASE MANHATTAN BANK by Reel 5263 Page 2292.
- 39. Terms, Covenants, Conditions and Provisions of a Subordination Agreement dated March 25, 1995, recorded March 30, 1995 in Reel 4097 Page 848.
- 40. Final Certificate of Eligibility recorded 12/11/02 in Reel 6687 Page 1660.
- 41. Tax Lien Certificate recorded in Reel 4359 Page 546, as assigned to THE CITY OF NEW YORK by Tax Lien Assignment dated 12/27/96, recorded 1/3/97 in Reel 4497 Page 420.
- 42. On 5/21/96, the CITY OF NEW YORK DEPARTMENT OF FINANCE conducted a Tax

Lien Sale. All Charges identified as Paid through Tax Lien Sale continue to be liens subject to the Lien Sale Provisions.

43. Terms, Provisions, Covenants and Agreements as contained in the Lease made by and between ALEXANDER'S OF REGO PARK, INC. and BED BATH & BEYOND, INC., a Memorandum of which is dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1951.

With Regard Thereto;

- a) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and BED BATH & BEYOND, INC. dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1957, as assigned to THE CHASE MANHATTAN BANK by assignment dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2287.
- 44. Terms, Provisions, Covenants and Agreements as contained in the unrecorded Lease made by and between ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 3/1/97.

With Regard Thereto:

- a) Agreement made by and between THE CHASE MANHATTAN BANK, ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2425.
- b) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 3/1/97, recorded 6/12/97 in Reel 4606 Page 1356.
- 45. Terms, Provisions, Covenants and Agreements as contained in the Lease made by and between ALEXANDER'S OF REGO PARK, INC. and CIRCUIT CITY STORES, INC., a Memorandum of which is dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1990.

With Regard Thereto:

- a) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and CIRCUIT CITY STORES, INC. dated 12/12/96, recorded 4/16/99 in Reel 5194 Page 834, as assigned to THE CHASE MANHATTAN BANK by assignment dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2282.
- 46. Easement Agreement made between ALEXANDER'S OF REGO PARK II, INC. and ALEXANDER'S REGO SHOPPING CENTER INC. dated 1/6/06, recorded 2/8/06 in CRFN 2006000077479.

With Regard Thereto:

- a) Amended and Restated Easement Agreement made between ALEXANDER'S OF REGO PARK II, INC. and ALEXANDER'S REGO SHOPPING CENTER INC. dated 12/21/07, recorded 2/14/08 in CRFN 2008000062504.
- 47. 10 foot Easement along the easterly portion of described premises as shown on Tax Map.
- 48. Judgment\*:

Creditor:

Debtor: ALEXANDERS REGO SHOPPING CT

9605 QUEENS BLVD, REGO PARK 11374 0113 CRIMINAL COURT OF THE CITY OF NEW YORK

125-01 QUEENS BLVD, QUEENS, 11415

Amount: \$500.00 Docketed: 9/10/04 Perfected: 9/9/04

Attorney: DISTRICT ATTORNEY OF QUEENS COUNTY

125-01 QUEENS BLVD, QUEENS 11415

49. Environmental Control Board Lien filed as of 1/31/09\*:

ALEXANDER REGO PARK CENTER INC 96-05 QUEENS BOULEVARD, QUEENS, NY 11374 Violation #:038160749L - Docket Date: 1/09 - Amount: \$350.00

50. Any water meter and sewer rent charges.

\*Alexander's Rego Shopping Center, Inc. agrees that it will address these matters in the ordinary course.

#### **EXHIBIT B**

## Schedule of Mortgages and Assignments

1. **Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement** dated March 29, 1995 in the principal amount of \$38,739,611.00 by Alexander's, Inc., a Delaware Corporation to Union Bank of Switzerland, recorded on March 30, 1995 in Reel 4097, Page 746 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

Assignment of Mortgage dated May 12, 1999 by UBS AG, Stamford Branch (successor to Union Bank of Switzerland) to The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2270 in the Office of the City Register of the City of New York, Queens County.

2. **Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement** dated March 29, 1995 in the principal amount of \$46,260,389.00 by Alexander's, Inc., a Delaware Corporation to Union Bank of Switzerland, recorded on March 30, 1995 in Reel 4097, Page 780 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

**Assignment, Assumption and Modification Agreement** dated April 19, 1995 among Alexander's, Inc., a Delaware corporation, Alexander's of Rego Park, Inc., a Delaware corporation and Union Bank of Switzerland, recorded on April 20, 1995 in Reel 4110, Page 739 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

Assignment of Mortgage dated May 12, 1999 by UBS AG, Stamford Branch (successor to Union Bank of Switzerland) to The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2276 in the Office of the City Register of the City of New York, Queens County.

**Amended, Restated and Consolidated Mortgage and Security Agreement** dated May 12, 1999 by and between Alexander's Rego Shopping Center, Inc., a Delaware corporation and The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2302 in the Office of the City Register of the City of New York, Queens County. Consolidates Mortgages 1 and 2 to form a single lien in the amount of \$82,000,000.00.

Assignment of Mortgages dated October 10, 2000 by The Chase Manhattan Bank, a New York banking corporation to State Street Bank and Trust Company, as Trustee for the Registered Holders of Chase Manhattan Bank-First Union National Bank Commercial Mortgage Trust, Commercial Mortgage Pass Though Certificates Series, 1999-1, recorded November 22, 2000 in Reel 5727, Page 0118 in the Office of the City Register of the City of New York, Queens County.

EXHIBIT C

Rent Roll

## EXHIBIT D

## Account Numbers

Account Type	Account Name	Account Number
Non-Interest Bearing Transaction Account	Alexander's Rego Shopping Center Inc.	1-047-9040-8389

Borrower agrees that, in the event the Account Number shall change or Bank shall need to establish an additional account, Bank shall be entitled to amend this Exhibit D upon written notice to Borrower, and any such amended Exhibit D shall be deemed to be a part of this Agreement.

# ALEXANDER'S REGO SHOPPING CENTER, INC., a Delaware corporation

Mortgagor

and

U.S. BANK NATIONAL ASSOCIATION
a national banking association,
as Bank

(the "Mortgagee")

## AMENDED AND RESTATED MORTGAGE, SECURITY AGREEMENT, FIXTURE FILING AND ASSIGNMENT OF LEASES AND RENTS

Dated as of March 10, 2009

· \_\_\_\_\_\_

This instrument affects real and personal property commonly known as 96-05 Queens Boulevard, Queens, New York, having a tax map designation of Block 2084, Lot 101 in the County of Queens.

RECORD AND RETURN TO:

Halloran & Sage LLP One Goodwin Square 225 Asylum St. Hartford, CT 06103 Attention: James P. Maher, Esq.

# AMENDED AND RESTATED MORTGAGE, SECURITY AGREEMENT, FIXTURE FILING AND ASSIGNMENT OF LEASES AND RENTS

THIS AMENDED AND RESTATED MORTGAGE, SECURITY AGREEMENT, FIXTURE FILING AND ASSIGNMENT OF LEASES AND RENTS (this "Mortgage") made as of March 10, 2009 by ALEXANDER'S REGO SHOPPING CENTER, INC., a Delaware corporation having an office and a mailing address at c/o Vornado Realty Trust, 888 Seventh Avenue, New York, New York 10019 (Mortgagor"), in favor of U.S. BANK NATIONAL ASSOCIATION, a national banking association having a place of business and a mailing address at One Post Office Square, 29<sup>th</sup> Floor, Boston, Massachusetts 02109 ("Mortgagee").

## WITNESSETH:

WHEREAS, Mortgagor is the owner of fee title to a parcel of land situated in the Borough of Queens, County of Queens, State of New York, being more particularly described in <u>Schedule A</u> attached hereto and incorporated herein, together with the building, structures and improvements situated thereon, being commonly known as 96-05 Queens Boulevard; and

WHEREAS, Mortgagee is the holder of that certain Amended, Restated and Consolidated Mortgage and Security Agreement dated May 12, 1999 by and between Mortgagor and The Chase Manhattan Bank, recorded on June 7, 1999 in Reel 5263, Page 2302, in the Office of the City Register of the City of New York, Queens County, and assigned to State Street Bank and Trust Company, as Trustee for the Registered Holders of Chase Manhattan Bank-First Union National Bank Commercial Mortgage Trust, Commercial Mortgage Pass Though Certificates Series, 1999-1, pursuant to an Assignment of Mortgages dated October 10, 2000 and recorded November 22, 2000 in Reel 5727, Page 0118 in the Office of the City Register of the City of New York, Queens County, and assigned on or about the date hereof to Mortgagee (the "Existing Mortgage"); and

WHEREAS, the Existing Mortgage consolidates a certain Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement dated March 29, 1995, recorded on March 30, 1995 in Reel 4097, Page 746 in the Office of the City Register of the City of New York, Queens County and a certain Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement dated March 29, 1995 and recorded on March 30, 1995 in Reel 4097, Page 780 in the Office of the City Register of the City of New York, Queens County, which consolidated mortgages and assignments thereof are set forth on Exhibit A hereto, to for a single lien in the amount of \$82,000,000.00; and

WHEREAS, Mortgagee is the owner and holder of the notes secured by the Existing Mortgage (collectively, the "Existing Note"); and

WHEREAS, Mortgagor acknowledges and certifies that, as of the date hereof, there is due and owing on the Existing Note, and secured by the Existing Mortgage, the unpaid principal amount of \$78,245,641.77, interest having been paid in full through the date hereof;

WHEREAS, Mortgagor acknowledges and certifies that, as of the date hereof, there are no defenses or offsets to the Existing Note or the Existing Mortgage; and

WHEREAS, it has been agreed by and between Mortgagor and Mortgagee that the Existing Note shall be amended and restated in its entirety pursuant to the terms of the Note (as hereinafter defined) and that the Existing Mortgage shall be amended and restated in its entirety as hereinafter set forth.

NOW. THEREFORE. TO SECURE THE PAYMENT when and as due of the sum of SEVENTY EIGHT MILLION TWO HUNDRED FORTY FIVE THOUSAND SIX HUNDRED FORTY ONE DOLLARS AND 77/100 DOLLARS (\$78,245,641.77) or so much thereof as may be advanced (the "Loan") pursuant to the terms of that certain Loan Agreement of even date herewith between Mortgagor and Mortgagee (as the same may be amended, restated or supplemented from time to time, the "Loan Agreement"), together with interest thereon to be paid according to the terms of the Loan Agreement and the Note, together with all other sums recoverable by Mortgagee under the terms of the Loan Documents (as hereinafter defined), including, without limitation, interest, default interest, late charges, prepayment premiums and any sums advanced by Mortgagee in accordance with the Loan Documents to protect or preserve the Mortgaged Property, together with all existing and future liabilities of Mortgagor to Mortgagee under the Loan Documents, together with all existing and future liabilities of Mortgagor to Mortgagee, if any, under any instrument, document or agreement which now or hereafter evidences an interest rate hedge agreement, if any (said indebtedness and interest due under the Note and all other sums due hereunder, under the Note and the other Loan Documents, or under any interest rate hedge agreement are herein being hereinafter collectively referred to as the "Indebtedness"); and to secure the full and prompt performance of each obligation of Mortgagor contained herein or in the Loan Documents, Mortgagor has mortgaged, given, granted, bargained, sold, alienated, enfeoffed, conveyed, confirmed, warranted, pledged, assigned, and hypothecated and by these presents does hereby mortgage, give, grant, bargain, sell, alien, enfeoff, convey, confirm, warrant, pledge, assign and hypothecate unto Mortgagee the following:

### THE MORTGAGED PROPERTY

- (A) All of Mortgagor's right, title and interest in and to the following described premises situated in the Borough of Queens, County of Queens, State of New York, to wit: The land described in detail in <u>Schedule A</u> (the "Land") which is attached hereto and incorporated herein and made a part of this document for all purposes.
- (B) **TOGETHER WITH** all of Mortgagor's right, title and interest in and to (1) all the buildings, structures and improvements of every nature whatsoever now or hereafter situated on the Land, and (2) all building materials, supplies and other property stored at or delivered to the Land or any other location for incorporation into any of the aforesaid buildings, structures and improvements, and (3) all fixtures, machinery, appliances, equipment, furniture and personal property of every nature whatsoever located in or on, or attached to, and used or intended to be used in connection with or with the operation of, the Land, any such buildings, structures or other improvements, or in connection with any construction being conducted or which may be conducted thereon (including, but not limited to, any and all such appliances;

furniture; carpeting; floor coverings; draperies; window coverings; furnishings; fences; partitions; generators; doors; windows; millwork; overhead doors; screens; storm windows and doors; locks; hardware; shades; awnings; motors; engines; boilers; tanks; water heaters; pumps; furnaces; heat registers; radiators; thermostats; plumbing; sinks; water closets; basins; faucets; elevators; switchboards; cleaning, vacuum and sprinkler systems; fire extinguishing apparatus and equipment; water tanks; lighting, heating, ventilating, air conditioning and air cooling systems, facilities, units, apparatus and equipment; incinerating, communicating and refrigerating equipment; water, gas, telephone and electric supply fixtures, machinery, ducts, piping, wiring, conduits, appurtenances and equipment; alarm and security systems; electronic intercommunication system; maintenance and cleaning equipment and supplies; lumber, insulation, structural steel, roofing, flooring, concrete panels, cement, blocks, bricks, stone, paint, nails, screws, and all other construction and building materials (including, but not limited to, those acquired by Mortgagor and stored in warehouses with Mortgagee's approval) and parking lot lighting), and all extensions, additions, improvements, betterments, renewals, substitutions and replacements to any of the foregoing, and all right, title and interest of Mortgagor in and to such personal property and fixtures, whether as owner, chattel lessee or otherwise, it being understood and agreed that all such personal property and fixtures are and at all times shall be part and parcel of the real property encumbered hereby and appropriated to the use thereof and, whether or not affixed or annexed to such real property, to the fullest extent permitted by law, are and at all times shall be conclusively deemed to be fixtures, a part of the real property and a part of the security provided hereby (all of the property described in this paragraph (B) being hereinafter collectively called the "Improvements").

- (C) **TOGETHER WITH** all estate, right, title and interest of Mortgagor, of whatever character, whether now owned or hereafter acquired, in and to (1) all streets, roads and public places, open or proposed, in front of or adjoining the Land, and the land lying in the bed of such streets, roads and public places, and (2) all other sidewalks, alleys, ways, passages, water courses, strips and gores of land adjoining or used or intended to be used in connection with any of the property described in paragraphs (A) and (B) hereof, or any part of such property.
- (D) **TOGETHER WITH** all easements, rights-of-way and rights of use or passage, public or private, and all estates, interests, benefits, powers, rights (including, without limitation, any and all lateral support, drainage, slope, sewer, water, air, mineral, oil, gas and subsurface rights), privileges, licenses, profits, royalties, tenements, hereditaments, reversions and subreversions, remainders and subremainders and appurtenances whatsoever in any way belonging, relating or appertaining to any of the property described in paragraphs (A), (B) and (C) hereof, or any part of such property, or which hereafter shall in any way belong, relate or be appurtenant thereto, whether now owned or hereafter acquired by Mortgagor.
- (E) **TOGETHER WITH,** but subject to Borrower's right hereby given to collect and use same so long as no Event of Default has occurred and is continuing, (1) any and all judgments, settlements, claims, awards, insurance proceeds and other proceeds and compensation, and interest thereon (collectively, "**Compensation**"), hereafter made or to be made or hereafter payable in connection with any casualty or other damage to the property described in paragraphs (A), (B), (C) and (D) hereof, any part of such property or any rights appurtenant thereto, or in connection with any condemnation proceedings affecting such property or rights or any taking under power of eminent domain (or any conveyance in lieu of or under

threat of any such taking) of such property or rights, including, without limitation, any and all Compensation for change of grade of streets or any other injury to or decrease in the value of such property or rights, (2) any and all proceeds of any sales, assignments or other dispositions of such property or rights, (3) any and all refunds of insurance premiums, taxes, assessments, water charges, sewer rents or other impositions in respect of such property or rights, (4) all accounts receivable, contract rights, general intangibles, permits, licenses, approvals, actions and rights in action, including, without limitation, all rights to insurance proceeds and unearned or refunded insurance premiums arising from or relating to (a) any such property or rights, and (b) all rights, benefits and privileges of Mortgagor from time to time existing under or with respect to any and all contracts and agreements for planning or design of the Improvements, construction of the Improvements or supply of materials in connection with such construction (including but not limited to all soil tests, reports, appraisals, feasibility studies and engineering reports relating to the Land or the Improvements), and (5) all proceeds, products, replacements, additions, substitutions, renewals, accessions, accretions and relictions of and to such property or rights.

(F) **TOGETHER WITH** all rents, royalties, issues, profits, revenues, income and other benefits to which Mortgagor may now or hereafter be entitled from the property described in paragraphs (A), (B), (C), (D) and (E) hereof or any part of such property, to be applied against the Indebtedness and other sums secured hereby; <u>provided, however</u>, that permission is hereby given to Mortgagor, so long as no Event of Default (as hereinafter defined) shall have occurred and be continuing, to collect and use such rents, royalties, issues, profits, revenues, income and other benefits as they become due and payable, but not in advance thereof. The foregoing assignment shall be fully operative without any further action on the part of either party, and specifically Mortgagee shall be entitled, at its option upon the occurrence of an Event of Default, to all such rents, royalties, issues, profits, revenues, income and other benefits whether or not Mortgagee takes possession of such property. Upon the occurrence of an Event of Default, the permission hereby given to Mortgagor to collect such rents, royalties, issues, profits, revenues, income and other benefits shall terminate. Neither the exercise of any rights under this paragraph by Mortgagee nor the application of any such rents, royalties, issues, profits, revenues, income or other benefits to the Indebtedness and other sums secured hereby, shall cure or waive any Event of Default or notice of any Event of Default hereunder or invalidate any act done pursuant hereto or to any such notice, but shall be cumulative of all other rights and remedies.

The foregoing provisions hereof shall constitute an absolute and present assignment of the rents, royalties, issues, profits, revenues, income and other benefits of or from the property described in paragraphs (A), (B), (C), (D), (E) and (F) hereof, or any part of such property, subject, however, to the conditional permission given to Mortgagor to collect and use such rents, royalties, issues, profits, revenues, income and other benefits as hereinabove provided; and the existence or exercise of such right of Mortgagor shall not operate to subordinate this assignment to any subsequent assignment, in whole or in part, by Mortgagor, and any such subsequent assignment by Mortgagor shall be subject to the rights of Mortgagee hereunder.

(G) **TOGETHER WITH** all right, title and interest of Mortgagor in and to any and all leases, subleases, occupancy, purchase and sale or similar agreements (collectively, "leases") now or hereafter on or affecting the property described in paragraphs (A), (B), (C), and

(D) hereof, or any part of such property, together with all security therefor (including, without limitation, any and all right, title and interest of Mortgagor in and to property of any tenant or other party under any such lease, and all cash or security deposits, advance rentals and deposits or payments of a similar nature thereunder) and together with all moneys payable thereunder, and all books and records which reflect payments made under the leases and all security therefor, subject, however, to the conditional permission hereinabove given to Mortgagor to collect the rents, income and other benefits arising under any such lease. Mortgagee shall have and is hereby granted the right, at any time and from time to time after the occurrence and during the continuance of an Event of Default, to notify any lessee, sublessee, licensee, occupant or purchaser of the rights of Mortgagee as provided by this paragraph.

(H) **TOGETHER WITH** any and all further or greater estate, right, title, interest, claim and demand whatsoever of Mortgagor, whether now owned or hereafter acquired, in or to any of the property described in the foregoing paragraphs or any rights or interests appurtenant thereto.

All of the property described in paragraphs (A), (B), (C), (D), (E), (F), (G) and (H) above, and each item of property therein described, is herein referred to as the "Mortgaged Property." Notwithstanding any contrary provision, the Mortgaged Property shall not be deemed to include the collateral under the Pledge Agreement (as defined in the Loan Agreement).

**TO HAVE AND TO HOLD** the Mortgaged Property, with the privileges and appurtenances thereof, unto Mortgagee, its successors and assigns forever, to its and their own proper use and behoof.

**AND ALSO,** Mortgagor, for itself and its successors and assigns, covenants with and warrants to, Mortgagee, its successors and assigns, that at and until the ensealing of these presents, Mortgagor is well seized of the Land, those Improvements constituting real property and all other real property intended to be encumbered by this Mortgage as a good indefeasible estate in fee simple, that Mortgagor has good and absolute title to all other Mortgaged Property, that Mortgagor has good right to give, grant, bargain, sell, assign and confirm the Mortgaged Property in manner and form as is above written, and that the Mortgaged Property is free and clear of all liens, encumbrances and exceptions to title whatsoever (except as set forth in **Schedule B** attached hereto and made a part hereof (hereinafter referred to as "**Permitted Encumbrances**"));

**AND FURTHERMORE,** Mortgagor does by these presents bind itself, its administrators and its successors and assigns forever to **WARRANT AND DEFEND** the Mortgaged Property to Mortgagee, its successors and assigns, against all claims and demands whatsoever, except for the Permitted Encumbrances.

### THE CONDITION OF THIS MORTGAGE IS SUCH that:

**WHEREAS**, this Mortgage secures, inter alia, present and future advances, if any, made by Mortgagee pursuant to the Loan Documents, up to a maximum principal amount of the Loan, together with all interest thereon. Notwithstanding anything contained herein to the contrary, the maximum amount of principal indebtedness secured by this Mortgage at the time of execution

hereof or which any contingency may become secured by this Mortgage, at any time hereafter is \$78,245,641.77 plus (a) taxes, charges or assessments which may be imposed by law upon the Mortgaged Property; (b) premiums on insurance policies covering the Mortgaged Property; (c) expenses incurred in upholding the lien of this Mortgage, including but not limited to (1) the expenses of any litigation to prosecute or defend the rights and lien created by this Mortgage, (2) any amount, cost or charges to which Mortgagee becomes subrogated, upon payment, whether under recognized principles of law or equity, or under express statutory authority, and (3) interest at the Default Rate.

**WHEREAS,** Mortgagor represents and warrants that it has full power and authority to execute and deliver the Loan Agreement, the Note, this Mortgage and all other documents and instruments required of it by Mortgagee in connection with the making or giving of the Loan; and

**AND WHEREAS,** Mortgagor covenants and agrees with Mortgagee as follows:

### **ARTICLE 1.**

# **GENERAL COVENANTS AND WARRANTIES**

- 1.1 **Performance of Mortgagor's Obligations**. Mortgagor shall duly, punctually and fully pay, do and perform all obligations and things on its part to be paid, done or performed under the Note, under this Mortgage, under the Loan Agreement and under any other instrument which refers to or secures the Note and under any instrument, document or agreement which now or hereafter evidences a interest rate hedge agreement, if any. Mortgagor shall promptly pay to Mortgagee when due all Indebtedness. Time is of the essence hereof subject to the notice, grace and cure periods provided in the Loan Documents.
  - 1.2 **Representations and Warranties by Mortgagor**. Mortgagor represents and warrants to Mortgagee, as follows:
- (a) Mortgagor is a corporation duly organized, existing and in good standing under the laws of the State of Delaware and is duly qualified and is in good standing in all other jurisdictions (including New York) where the character and nature of its business requires such qualification.
- (b) Mortgagor is the lawful owner of and has good and marketable fee simple absolute title to the Mortgaged Property, subject to the Permitted Encumbrances. Mortgagor has good right and lawful authority to grant, bargain, sell, convey, warrant, assign and confirm the same as provided herein; and the Mortgaged Property is free and clear of all mortgages, liens, pledges, security interests, charges and encumbrances, excepting only Permitted Encumbrances. Mortgagor warrants and will defend the title to the Mortgaged Property against all claims and demands whatsoever, except Permitted Encumbrances and Permitted Transfers.
- (c) There is no provision in any indenture, contract or agreement, to which Mortgagor is a party or by which it is bound, or any law, statute, ordinance, governmental rule, regulation or restriction, or any order of any court or administrative agency, to which Mortgagor is subject or by which Mortgagor is bound, which prohibits the execution and delivery by Mortgagor of this Mortgage, the Note, the Loan Agreement or any other instruments which refer

to, evidence, supplement, guarantee, secure or govern the Note or the Loan (as each of the same may be extended, renewed, modified, substituted or replaced, hereinafter collectively referred to as the "Loan Documents" and individually referred to as a "Loan Document"), or the performance or observance by Mortgagor of any of the terms, covenants or conditions of this Mortgage, the Note, the Loan Agreement or any other Loan Document.

- (d) Execution and delivery of this Mortgage, the Note, the Loan Agreement and the other Loan Documents, by and on behalf of Mortgagor, have been duly and validly authorized, and this Mortgage, the Note, the Loan Agreement and said other instruments have been duly and validly executed and delivered, by and on behalf of Mortgagor, and are valid, binding and enforceable obligations of Mortgagor in accordance with their terms.
- (e) There are no actions, suits, proceedings or investigations pending or, to the knowledge of Mortgagor, threatened against Mortgagor, any parent of Mortgagor or subsidiary of Mortgagor (hereinafter collectively referred to as "Affiliates"), or the Mortgaged Property in any court or before any federal, state, municipal or other governmental agency other than actions covered by insurance or actions which, if adversely determined, would not result in a Material Adverse Occurrence (as defined in the Loan Agreement), and neither Mortgagor nor any Affiliate is in default with respect to any order of any court or governmental agency.
  - (f) The Premises are neither agricultural property, property in agricultural use, nor the homestead of Mortgagor.
- (g) All applicable building and zoning laws, rules and regulations, occupational safety and health, energy and environmental laws, ordinances and regulations affecting the Mortgaged Property permit the use and occupancy thereof for mixed-use commercial purposes and have been complied with, in all material respects.
- (h) To Mortgagor's knowledge, except as shown in the Environmental Report and other than usage in the ordinary course of business of Mortgagor or its tenants and in each instance in compliance with Environmental Laws, (i) no dangerous, toxic or hazardous pollutants, contaminants, chemicals, wastes, materials or substances, as defined in or governed by the provisions of any federal, state or local law, statute, code, ordinance, regulation, requirement or rule relating thereto (hereinafter collectively called "Environmental Regulations"), and also including urea-formaldehyde, polychlorinated biphenyls, asbestos, asbestos containing materials, nuclear fuel or waste, radioactive materials, explosives, carcinogens and petroleum products, or any other waste, material, substance, pollutant or contaminant which would subject the owner of the Premises to any damages, penalties or liabilities under any applicable Environmental Regulation (hereinafter collectively called "Hazardous Substances") are now stored, located, generated, produced, processed, treated, transported, incorporated, discharged, emitted, released, deposited or disposed of in, upon, under, over or from the Mortgaged Property; (ii) no threat exists of a discharge, release or emission of a Hazardous Substance upon or from the Mortgaged Property into the environment; (iii) the Premises have not ever been used for a mine, a landfill, a dump or other disposal facility, industrial or manufacturing purposes, or a gasoline service station; (iv) no underground storage tank is now located in the Premises; (v) no violation of any Environmental Regulation now exists in, upon, under, over or from the Mortgaged Property, no notice of any such violation or any

alleged violation thereof has been issued or given by any governmental entity or agency which remains outstanding, and there is not now any investigation or report involving the Mortgaged Property by any governmental entity or agency which in any way relates to Hazardous Substances; (vi) no person, party or private or governmental agency or entity has given any outstanding notice of or asserted any claim, cause of action, penalty, cost or demand for payment or compensation, whether or not involving any injury or threatened injury to human health, the environment or natural resources, resulting or allegedly resulting from any activity or event described in (i) above; (vii) there are not now any actions, suits, proceedings or damage settlements relating in any way to Hazardous Substances, in, upon, under, over or from the Mortgaged Property; (viii) the Mortgaged Property is not listed in the United States Environmental Protection Agency's National Priorities List of Hazardous Waste Sites or any other list of Hazardous Substance sites maintained by any federal, state or local governmental agency; and (ix) the Mortgaged Property is not subject to any lien or claim for lien or threat of a lien in favor of any governmental entity or agency as a result of any release or threatened release of any Hazardous Substance.

1.3 <u>Maintenance of Existence</u>. Mortgagor agrees to maintain its existence as a (i) corporation under the laws of the State of Delaware and (ii) as a foreign business corporation under the laws of the State of New York and not to dissolve, liquidate, wind-up, consolidate or merge during the term hereof, without the prior written consent of Mortgagee.

## 1.4 Further Assurances; Security Agreement.

- (a) Mortgagor will procure, do, execute, acknowledge and deliver each and every further act, deed, conveyance, transfer, document and assurance necessary or proper for the carrying out more effectively of the purpose of this Mortgage and, without limiting the foregoing, for granting, bargaining, selling, conveying, warranting, assigning and confirming unto Mortgagee all of the Mortgaged Property, or property intended so to be, whether now owned or hereafter acquired by Mortgagor, including, without limitation, the preparation, execution and filing of any documents, such as financing statements and continuation statements, deemed advisable by Mortgagee for perfecting and maintaining its lien on the Mortgaged Property.
- (b) This Mortgage shall further constitute and be deemed to be a Security Agreement under the Uniform Commercial Code, now in force and as hereafter amended, and Mortgagor hereby grants to Mortgagee a first and only, present and continuing security interest in any Property, fixtures, equipment, leases, rents, issues, income, profits, personal property, instruments, general intangibles, accounts, contract rights and claims included within or related to the Mortgaged Property, and in all deposits made pursuant to **Section 1.10** hereof and all insurance policies and unearned premiums prepaid thereon, insurance proceeds, and awards, payments or consideration for the taking of the Mortgaged Property, or any portion thereof, by condemnation or exercise of the power of eminent domain, or from any sale in lieu or in anticipation thereof, assigned by Mortgagor to Mortgagee hereunder, to the extent that a security interest may be granted therein under the terms of the Uniform Commercial Code. Mortgagor agrees to supply Mortgagee with an inventory of all such property in a form acceptable to Mortgagee, from time to time, upon receipt of a written request therefor from Mortgagee.

- (c) Mortgagor hereby irrevocably authorizes Mortgagee at any time and from time to time to file in any filing office any initial financing statements and amendments thereto that (a) indicate the collateral as the Mortgaged Property as defined in this Mortgage and (b) contain any other information required by part 5 of Article 9 of the Uniform Commercial Code of any such filing office for the sufficiency or filing office acceptance of any initial financing statement or amendment, including whether Mortgagor is an organization, the type of organization and any organizational identification number issued to Mortgagor. Mortgagor agrees to provide any such information to Mortgagee promptly upon request. Mortgagor also ratifies its authorization for Mortgagee to have filed in any filing office in any Uniform Commercial Code jurisdiction any like initial financing statements or amendments thereto relating to the lien and security interest created by this Mortgage if filed prior to the date hereof. Mortgagor shall pay to Mortgagee, from time to time, within five (5) Business Days after demand, any and all costs and expenses incurred by Mortgagee in connection with the filing of any such initial financing statements and amendments, including attorneys' fees and all disbursements. Such costs and expenses shall bear interest at the Default Rate from the date due, as set forth in the preceding sentence until the date repaid by Mortgagor, and such costs and expenses, together with such interest, shall be part of the obligations and shall be secured by this Mortgage.
- (d) Mortgagor shall notify the Mortgagee promptly of a change in Mortgagor's chief executive office or principal place of business. Mortgagor shall not change its name, jurisdiction of organization or its type of organization without promptly notifying Mortgagee, in writing, of such change.
- 1.5 <u>Uniform Commercial Code Remedies</u>. Mortgagee shall have all the rights, remedies and recourses with respect to the personal property, fixtures, the Leases (as defined in **Section 1.16** hereof) and rents afforded to a "**Secured Party**" by the Uniform Commercial Code in addition to, and not in limitation of, the other rights, remedies and recourses afforded by law, equity and the Loan Documents.

# 1.6 [Intentionally Omitted].

- 1.7 <u>Foreclosure of Security Interest</u>. If an Event of Default shall occur and remain outstanding, Mortgagee may elect, in addition to exercising any and all other rights, remedies and recourses set forth in Section 3.2, to proceed in the manner set forth in the Uniform Commercial Code, relating to the procedure to be followed when a security agreement covers both real and personal property.
- 1.8 **Defined Terms**. Terms defined in the Uniform Commercial Code and not otherwise defined in this Mortgage shall have the same meanings for purposes hereof as are set forth in the Uniform Commercial Code. In the event that a term is used in Article 9 of the Uniform Commercial Code and also in another Article of the Uniform Commercial Code, the term used in **Sections 1.4 through 1.7** is that used in Article 9 of the Uniform Commercial Code. The term "control" has the meaning given in § 9-104, 9-105, 9-106, 9-107, as applicable.
- 1.9 <u>Maintenance of Property</u>. Mortgagor will not remove or demolish any Improvements, except as permitted under applicable law, or change its legal use in any manner

that would cause the Mortgaged Property to be in violation of applicable law without the prior written consent of Mortgagee, provided, however, that the foregoing shall not prevent Mortgagor from performing alterations or improvements to the Mortgaged Property that are determined by Mortgagor to be necessary or desirable.

- 1.10 **Impositions and Liens**. (a) Mortgagor shall, before any penalty or interest attaches thereto because of delinquency in payment, pay and discharge, or cause to be paid and discharged, all taxes, assessments, sewer rents, water rates, levies and governmental charges imposed upon or against the Mortgaged Property or upon or against the Note or the Indebtedness or upon or against the interest of Mortgagee in the Mortgaged Property or in the Note or the Loan Agreement or the Indebtedness (hereinafter referred to as "Impositions") and will thereafter deliver the paid receipts therefor to Mortgagee within ten (10) days after request therefor by Mortgagee. In the event of any legislative enactment or judicial decision after the date of this Mortgage, imposing upon Mortgagee the obligation to pay any such Imposition, or deducting the lien of this Mortgage from the value of the Mortgaged Property for the purpose of taxation, or changing in any way the laws now in force for the taxation of mortgages or debts secured thereby, or the manner, operation or method of collection of any such Imposition, so as to affect the interests of Mortgagee, then, and in such event, Mortgagor shall bear and promptly pay the full amount of such Imposition or any substituted tax; provided, however, that, if for any reason payment thereof by Mortgagor would be unlawful or unenforceable, or if payment thereof by Mortgagor would constitute usury or would render the Loan or the Indebtedness wholly or partially usurious under any of the terms or provisions of the Note or of this Mortgage, or otherwise, Mortgagee may declare the whole sum secured by this Mortgage, with interest thereon, to be immediately due and payable. Mortgagor shall not suffer to exist and shall promptly pay and discharge any mechanic's, statutory or other lien or encumbrance on the Mortgaged Property or any part thereof (hereinafter referred to as "Liens"), except for Permitted Encumbrances.
- (b) Notwithstanding the foregoing, Mortgagor shall not be in default hereunder in respect to the payment of any Impositions or Liens which Mortgagor shall be required by any provision hereof to pay, so long as Mortgagor shall, in good faith, in compliance with all applicable statutes, and with all possible promptness, diligently contest the same, and Mortgagor may postpone or defer payment of a portion of said Impositions or Liens, if, but only if, permitted by statute, and if neither the Mortgaged Property, nor any portion thereof, would, by reason of such postponement or deferment, be in danger of being forfeited or lost. Upon a final adjudication of any such contest, and, in any event, at least thirty (30) days prior to the date on which the interest of Mortgagee in the Mortgaged Property would otherwise forfeit by reason of the nonpayment of any such Imposition or Lien, Mortgagor shall pay the amount thereof then due, including any penalties and interest thereon.
- (c) As soon as reasonably practicable following the recording of this Mortgage, Mortgagor shall deliver or cause to be delivered to Mortgagee a title report showing the recordation of the Existing Mortgage Assignment and this Mortgage. In the event any lien, encumbrance or other matter objectionable to Mortgagee is disclosed by such title report, other than the Permitted Encumbrances, Mortgagor shall bond over or otherwise have removed any such lien or encumbrance within thirty (30) days of the imposition thereof, to the satisfaction of

Mortgagee, unless Mortgagor is contesting such lien or encumbrance in accordance with **Section 1.10(b)** hereof.

- 1.11 <u>Insurance; Casualty or Other Loss</u>. Mortgagor will keep the Improvements on the Mortgaged Property insured against loss by fire or other casualty, and shall maintain liability insurance, with Mortgagee named as an additional insured, under policies of insurance, with coverages, as reasonably determined by Mortgagor to be consistent with the types and amount of insurance maintained by reasonably prudent owners of property similar to the Mortgaged Property in the Greater New York Metropolitan Area and available at commercially reasonable rates. Mortgagor shall timely pay the premiums on all such insurance. In the event of casualty or other loss, Mortgagor shall immediately give written notice thereof to Mortgagee. All proceeds of casualty insurance shall be paid to Mortgagor and applied, after application to Mortgagor's costs of collection to payment of any costs of restoration and repair incurred by Mortgagor or in such other manner as Mortgagor shall determine.
  - 1.12 [Intentionally Omitted].
  - 1.13 [Intentionally Omitted].
- 1.14 Cure of Defaults by Mortgagee. If an Event of Default occurs and Mortgagor shall fail to observe, comply with or perform any of the terms, covenants and conditions herein with respect to the procuring and delivery of insurance, the payment of Impositions, ground rent or Liens, the keeping of the Mortgaged Property in repair, the performance of Mortgagor's obligations under any other term, covenant or condition contained herein, in the Note, in the Loan Agreement or in any other Loan Document, Mortgagee may itself, following reasonable notice to and cure periods for Mortgagor, observe, comply with or perform the same, may make such advances to observe, comply with or perform the same as Mortgagee shall deem appropriate, and may enter the Mortgaged Property for the purpose of observing, complying with or performing any such term, covenant or condition. Mortgagee may expend such sums, including attorneys' fees (prior to trial, at trial and on appeal), to sustain the lien of this Mortgage or its priority, or to protect or enforce, or to obtain the right to enforce, its rights, powers and remedies hereunder, including the payment of any prior liens, ground rent, claims and encumbrances, other than Permitted Encumbrances which are not in default, or to protect the Mortgaged Property, as it may deem desirable. Mortgagor agrees to repay all sums so advanced or expended upon demand, with interest thereon at the Default Rate from the date of advancement or expenditure, and all sums so advanced or expended, with interest, shall be secured hereby, but no such advance or expenditure shall be deemed to relieve Mortgagor from any default hereunder. Mortgagee shall not be bound to inquire into the validity of any Imposition or Lien which Mortgagor fails to pay as and when required hereby and which Mortgagor does not contest in strict accordance with the terms hereof.
- 1.15 <u>Sale, Transfer or Encumbrance</u>. Except for Permitted Encumbrances and Permitted Transfers (as defined in the Loan Agreement), without the prior written consent of Mortgagee, Mortgagor shall not voluntarily or involuntarily agree to, cause, suffer or permit any sale, conveyance, mortgage, grant, lien, encumbrance, security interest, pledge, assignment or transfer of: (a) the Mortgaged Property or any part or portion thereof, or any interest of Mortgagor, legal or equitable, in the Mortgaged Property, except as created by this Mortgage and

the other documents which secure the Note; or (b) any direct ownership interest, in Mortgagor; provided, however, this subsection shall not be applicable to the extent liens are being contested in accordance herewith. If Mortgagor fails to comply with the provisions of this <u>Section 1.15</u>, Mortgagee may, at its election, declare the entire Indebtedness to be immediately due and payable, without notice to Mortgagor (which notice Mortgagor hereby expressly waives), and upon such declaration the entire Indebtedness shall be immediately due and payable.

No transfer, conveyance, lease, sale or other disposition shall relieve Mortgagor from personal liability for its obligations hereunder or under the Note, whether or not the transferee assumes this Mortgage. Mortgagee may, without notice to Mortgagor, deal with any successor owner of all or any portion of the Mortgaged Property in the same manner as with Mortgagor, without in any way discharging the liability of Mortgagor hereunder or under the Note.

- 1.16 <u>Assignment of Leases and Rents</u>. (a) As a source of future repayment of the Indebtedness, Mortgagor hereby absolutely and presently assigns to Mortgagee all Rents (hereinafter defined) and hereby collaterally assigns to Mortgagee all of Mortgagor's rights in and interest as landlord under the Leases (subject to the appointment of Mortgagor as agent for Mortgagee and the license granted to Mortgagor as set forth below); together with the immediate and continuing right to collect and receive all of the Rents now due or which may become due or to which Mortgagor may now or shall hereafter (including the period of redemption, if any) become entitled or may demand or claim, arising or issuing from or out of the Leases or from or out of the Mortgaged Property or any part thereof. It is the intention hereby to establish a choate, present, absolute transfer and assignment of the Rents now due to or which may become due to Mortgagee and not merely the granting of a security interest.
- (b) Notwithstanding the foregoing, Mortgagee hereby (i) appoints Mortgagor as the agent of Mortgagee to collect and receive all of the Rents and apply them as set forth herein and (ii) grants to Mortgagor a license to carry out any and all of the terms and conditions of the Leases and to manage the Mortgaged Property, as hereinafter provided, such appointment and license terminable at the sole election of Mortgagee upon the occurrence of an Event of Default.
- (c) As used herein: (i) "Lease" means each existing or future lease, sublease (to the extent of Mortgagor's rights thereunder) or other agreement under the terms of which any person has or acquires any right to occupy or use the Mortgaged Property, or any part thereof, or interest therein, and each existing or future guaranty of payment or performance thereunder, and all extensions, renewals, modifications and replacements of each such lease, sublease, agreement or guaranty; and (ii) "Rents" means all of the rents, revenue, income, profits and proceeds derived and to be derived from the Mortgaged Property or arising from the use or enjoyment of any portion thereof or from any Lease, including, but not limited to, liquidated damages following default under any such Lease, all proceeds payable under any policy of insurance covering loss of rents resulting from untenantability caused by damage to any part of the Mortgaged Property, all of Mortgagor's rights to recover monetary amounts from any tenant in bankruptcy including, without limitation, rights of recovery for use and occupancy and damage claims arising out of Lease defaults, including rejections, under any applicable law, together with any sums of money that may now or at any time hereafter be or become due and payable to

Mortgagor by virtue of any and all royalties, overriding royalties, bonuses, delay rentals and any other amount of any kind or character arising under any and all present and all future oil, gas, mineral and mining leases covering the Mortgaged Property or any part thereof, and all proceeds and other amounts paid or owing to Mortgagor under or pursuant to any and all contracts and bonds relating to the improvement, construction or renovation of the Mortgaged Property.

# 1.17 [Intentionally Omitted].

1.18 Hazardous Substances. Except for substances normally used for maintenance or operation of retail and commercial projects which are used, stored, and disposed of in accordance with all applicable Environmental Regulations, Mortgagor (i) shall not store, locate, generate, produce, process, treat, transport, incorporate, discharge, emit, release, deposit or dispose of any Hazardous Substance in, upon, under, over or from the Mortgaged Property, (ii) shall not permit any Hazardous Substance to be stored, located, generated, produced, processed, treated, transported, incorporated, discharged, emitted, released, deposited, disposed of or to escape therein, thereupon, thereupon, thereupon or therefrom, (iii) shall cause all Hazardous Substances found thereon to be properly removed therefrom and properly disposed of in accordance with all applicable Environmental Regulations, (iv) shall not install or permit to be installed any underground storage tank therein or thereunder, and (v) shall comply with all Environmental Regulations which are applicable to the Mortgaged Property, Mortgagor shall indemnify, defend and hold harmless Mortgagee, its directors, officers, employees, agents, contractors, licensees, invitees, successors and assigns and any affiliate of Mortgagee which acquires title to the Mortgaged Property at a foreclosure or other sale or takes title to the Mortgaged Property pursuant to a deed in lieu of foreclosure under the provisions of this Mortgage (hereinafter collectively referred to as the "Indemnified Parties") against, shall hold Indemnified Parties harmless from, and shall reimburse the Indemnified Parties for, any and all claims, demands, judgments, penalties, liabilities, costs, damages and expenses, including court costs and reasonable attorneys' fees directly or indirectly incurred by the Indemnified Parties (prior to trial, at trial and on appeal) in any action, administrative proceeding or negotiations against or involving any of the Indemnified Parties, resulting from any breach of the foregoing covenants, from the incorrectness or untruthfulness of any warranty or representation set forth in **Subsection 1.2(i)** hereof, or from the discovery of any Hazardous Substance in, upon, under or over, or emanating from, the Mortgaged Property, whether or not Mortgagor is responsible therefor, or whether or not it was placed, located, deposited or released by Mortgagor, it being the intent of Mortgagor and Mortgagee that the Indemnified Parties shall have no liability or responsibility (x) for damage or injury to human health, the environment or natural resources caused by, for abatement and/or clean-up of, or (y) otherwise with respect to, Hazardous Substances by virtue of the interest of Mortgagee in the Mortgaged Property created hereby or as the result of Mortgagee exercising any of its rights or remedies with respect thereto hereunder, including, but not limited to, becoming the owner thereof by foreclosure or conveyance in lieu of foreclosure. The foregoing representations, warranties and covenants of **Subsection 1.2(i)** and of this **Section 1.18** shall be deemed continuing covenants, representations and warranties for the benefit of the Indemnified Parties, and shall survive the satisfaction or release of this Mortgage, any foreclosure of this Mortgage and/or any acquisition of title to the Mortgaged Property or any part thereof by Mortgagee, or anyone claiming by, through or under Mortgagee by deed in lieu of foreclosure or otherwise. Any amounts covered by the foregoing indemnification shall bear interest from the date incurred at the Default Rate, shall be payable on demand, and shall be

secured hereby. In no event shall Mortgagor be obligated by the provisions hereof to indemnify any Indemnified Party for matters, liabilities or expenses arising from the gross negligence or willful misconduct of such Indemnified Party.

1.19 **Indemnity**. Mortgagor agrees to indemnify, protect, hold harmless and defend Mortgagee from and against any and all losses, liabilities, suits, actions, obligations, fines, damages, judgments, penalties, claims, causes of action, charges, costs and expenses (including attorneys' fees, disbursements and court costs prior to trial, at trial and on appeal) which may be imposed on, incurred or paid by, or asserted against Mortgagee by reason or on account of, or in connection with, (i) any willful misconduct of Mortgagor or any default or Event of Default hereunder, (ii) any construction, reconstruction or alteration of the Mortgaged Property, (iii) any negligence of Mortgagor or any negligence or willful misconduct of any lessee of the Mortgaged Property or any part thereof, or any of their respective agents, contractors, subcontractors, servants, directors, officers, employees, licensees or invitees, or (iv) any accident, injury, death or damage to any person or property occurring in, on or about the Mortgaged Property or any street, drive, sidewalk, curb or passageway adjacent thereto, except to the extent that the same results directly from the willful gross negligence or misconduct of Mortgagee. Any amount payable to Mortgagee under this **Section 1.19** shall be due and payable upon demand therefor and receipt by Mortgagor of a statement from Mortgagee setting forth in reasonable detail the amount claimed and the basis therefor. Mortgagor's obligations under this **Section 1.19** shall survive the repayment or any other satisfaction of the Note and shall not be affected by the absence or unavailability of insurance covering the same or by the failure or refusal of any insurance carrier to perform any obligation on its part under any such policy of insurance. If any claim, action or proceeding is made or brought against Mortgagee which is subject to the indemnity set forth in this Section 1.19, Mortgagor shall resist or defend against the same, in its own name or, if necessary, in the name of Mortgagee, by attorneys for Mortgagor's insurance carrier (if the same is covered by insurance) approved by Mortgagee or otherwise by attorneys retained by Mortgagor and approved by Mortgagee. Notwithstanding the foregoing, Mortgagee, in its discretion, if it disapproves of the attorneys provided by Mortgagor or Mortgagor's insurance carrier, may engage its own attorneys to resist or defend, or to assist therein, and Mortgagor shall pay, or, on demand, shall reimburse Mortgagee for the payment of, all fees and disbursements of said attorneys.

# ARTICLE 2. TAKING OF PROPERTY

2.1 Condemnation or Sale in Lieu Thereof. In case of a taking of or damage to all or any part of the Mortgaged Property as a result of, or a sale thereof in lieu of or in anticipation of, the exercise of the power of condemnation or eminent domain, or the commencement of any proceedings or negotiations which might result in such a taking, damage or sale, Mortgagor shall promptly give Mortgagee written notice thereof, generally describing the nature of such taking, damage, sale, proceedings or negotiations and the nature and extent of the taking, damage or sale which has resulted or might result therefrom, as the case may be, together with a copy of each and every document relating thereto received by Mortgagor. Should any of the Mortgaged Property be taken or damaged by exercise of the power of condemnation or eminent domain, or be sold by private sale in lieu or in anticipation thereof, any award, payment or other consideration for the property so taken, damaged or sold shall, be applied first to the payment of

all costs and expenses incurred by Mortgagor in obtaining and preserving such award, payment or consideration, and second, to the restoration or repair of the Mortgaged Property or in such other manner as Mortgagor determines, without affecting the lien of this Mortgage or the obligations of Mortgagor hereunder.

# ARTICLE 3. DEFAULT AND REMEDIES THEREFOR

- 3.1 **Events of Default**. If any one or more of the following events (herein referred to as "**Events of Default**") shall occur an event which would constitute an "Event of Default" under the Loan Agreement.
- 3.2 Mortgagee's Power of Enforcement; Power of Sale. After the occurrence of any Event of Default, Mortgagee, at its option, may proceed by any appropriate action or proceeding to (i) enforce payment of the Indebtedness pursuant to the Loan Documents, (ii) enforce performance of any term of this Mortgage or any of the other Loan Documents, (iii) enforce any other rights of Mortgagee with respect to the Indebtedness or the Mortgaged Property, (iv) foreclose this Mortgage and sell the Mortgaged Property, as an entirety or in separate portions, pursuant to the judgment, order or decree of any court of competent jurisdiction, (v) to the extent permitted by law, pursue the partial foreclosure of this Mortgage for any part of the Indebtedness then due and payable, subject to the continuing encumbrance of this Mortgage as security for the balance of the Indebtedness not then due, and (vi) pursue any other right, power or remedy available to Mortgagee at law or in equity. Mortgagee may pursue any and all such actions or proceedings, at Mortgagee's option, either with or without entry or taking possession and whether or not the Indebtedness or any part thereof shall have been declared to be immediately due and payable or shall otherwise be due. Mortgagee may pursue any and all such actions or proceedings without prejudice to Mortgagee's right thereafter to foreclose this Mortgage or to bring any other action or proceeding to enforce Mortgagee's rights, powers and remedies with respect to the Indebtedness or the Mortgaged Property, whether or not the basis for any such subsequent action or proceeding shall be a default or Event of Default existing at the time such earlier action or proceeding was commenced.

Mortgagee may, either with or without entry or taking possession of the Mortgaged Property as provided in this Agreement or otherwise, personally or by its agents or attorneys, and without prejudice to the right to bring an action for foreclosure of the Mortgage, sell the Mortgaged Property or any part thereof pursuant to any procedures provided by applicable law, thereto), and all estate, right, title, interest, claim and demand therein, and right of redemption thereof, at one or more sales as an entity or in parcels, and at such time and place upon such terms and after such notice thereof as may be required or permitted by applicable law.

All notices hereunder or under any applicable law pertaining hereto shall be in writing and shall be deemed sufficiently given or served for all purposes when delivered by personal service or courier service, and shall be deemed given on the date when signed for or, if refused, when refused by the person designated as an agent for receipt of service. For purposes hereof, notices may be given by the parties hereto or by their attorneys identified herein.

3.3 Mortgagee's Right To Enter and Take Possession.

- (a) While any Event of Default exists, Mortgagor, upon receipt of written notification by Mortgagee, shall forthwith surrender and deliver to Mortgagee the actual possession of the Mortgaged Property or any part thereof designated by Mortgagee, subject to the rights of tenants. To the extent permitted by law, Mortgagee may enter and take possession of all or any part of the Mortgaged Property and may exclude Mortgagor and its officers, employees, agents, contractors, attorneys and other representatives therefrom, and Mortgagee may have joint access with Mortgagor to the books, papers and accounts of Mortgagor and of any manager of the Mortgaged Property. On the first day of each month occurring after any such entry into possession, or after the appointment of any receiver as provided below, Mortgagor shall pay to Mortgagee or to such receiver, as the case may be, in advance, a use and occupancy charge equal to the fair and reasonable rental value for such month of any leasable portion of the Mortgaged Property which shall then be in the possession of or occupied by Mortgagor. If Mortgagor shall fail to make any such payment as provided above, then, upon request by Mortgagee, Mortgagor shall vacate, deliver and surrender possession of such part of the Mortgaged Property to Mortgagee or to such receiver, as the case may be, and, to the extent permitted by law, Mortgagor may be evicted or dispossessed by summary proceedings or otherwise.
- (b) While an Event of Default exists, Mortgagor for any reason shall fail to surrender or deliver to Mortgagee the Mortgaged Property or any part thereof designated by Mortgagee as provided above, Mortgagee may obtain a judgment or decree conferring on Mortgagee the right to immediate possession and requiring Mortgagor to deliver to Mortgagee immediate possession of the Mortgaged Property or such part thereof. Mortgagor hereby specifically and irrevocably consents to the entry of any such judgment or decree. Upon request by Mortgagee, Mortgagor shall pay to Mortgagee, or to any other person that Mortgagee may designate, all reasonable costs, expenses and liabilities (including, without limitation, reasonable attorneys' fees) incurred by Mortgagee in connection with any such failure to surrender or deliver possession and in connection with any such judgment or decree, together with interest thereon at the Default Rate from the date incurred by Mortgagee until the date so paid to, or as directed by, Mortgagee.
- (c) After any such entry into possession, Mortgagee, in Mortgagor's name or otherwise, may hold, store, use, operate, manage and control the Mortgaged Property, or the part thereof as to which Mortgagee shall have entered into possession, and may conduct the business thereof. In doing so, Mortgagee from time to time may:
  - (i) make all necessary and proper maintenance, repairs, renewals, replacements, alterations, additions, betterments and improvements thereto and thereon and purchase or otherwise acquire additional fixtures, personal property and other property of the type encumbered by this Mortgage;
  - (ii) insure the Mortgaged Property or keep the Mortgaged Property insured;
  - (iii) manage and operate the Mortgaged Property, or such part thereof, and exercise all the rights and powers of Mortgagor with respect to the Mortgaged

Property and the management and operation of the Mortgaged Property (including, without limitation, the right to enter into leases, to cancel, enforce or modify leases, to evict tenants by summary proceedings or otherwise and to take other appropriate steps to enforce leases);

- (iv) enter into agreements with others to exercise the powers herein granted to Mortgagee, all as Mortgagee from time to time may determine; and
- (v) collect and receive all the rents, royalties, issues, profits, revenues, income and other benefits of and from all or any part of the Mortgaged Property, including those past due as well as those accruing thereafter, and apply the moneys so received, in such priority as Mortgagee may determine, to (1) the payment of interest, principal and other amounts due and payable in respect of the Indebtedness or otherwise payable pursuant to any of the Loan Documents, (2) the deposits payable under this Mortgage for Impositions, (3) the cost of insurance, Impositions and other expenses of operating, maintaining, repairing and improving all or any part of the Mortgaged Property, including, without limitation, reasonable renting commissions and rental collecting commissions paid to any agent of Mortgagee or of any receiver, (4) the reasonable compensation, expenses and disbursements of the agents, contractors, attorneys and other representatives of Mortgagee, and (5) amounts advanced for any purpose recognized under this paragraph (c) or otherwise permitted by law or agreement.
- (d) Mortgagee shall be liable to account only for rents, security deposits, royalties, issues, profits, revenues, income and benefits actually received by Mortgagee while in possession of the Mortgaged Property. In the event of any foreclosure, Mortgagee may remain in possession of all or any part of the Mortgaged Property until the foreclosure sale and thereafter during any period of redemption. In the absence of any foreclosure, Mortgagee may remain in possession of all or any part of the Mortgaged Property as long as there exists an Event of Default. The same right of taking possession shall exist during the continuance of any subsequent Event of Default. Mortgagee shall incur no liability for, nor shall Mortgagor assert any claim or set off as a result of, any acts or omissions of Mortgagee, or its officers, employees, agents, contractors, attorneys or other representatives, while in possession of all or any part of the Mortgaged Property (except for damages directly caused by Mortgagee's own gross negligence or intentional wrongful acts), all such liabilities, claims and rights of set off being hereby expressly waived by Mortgagor.
- (e) Upon request by Mortgagee, Mortgagor shall pay to Mortgagee, or to any other person that Mortgagee may designate, all reasonable costs, expenses and liabilities (including, without limitation, reasonable attorneys' fees) incurred by Mortgagee in connection with the management, operation, use, control and maintenance of all or any part of the Mortgaged Property, except to the extent such costs, expenses and liabilities shall have been paid out of collections from the Mortgaged Property as provided above, together with interest thereon at the Default Rate from the date incurred by Mortgagee until the date so paid to, or as directed by, Mortgagee.

# 3.4 Appointment of Receiver.

- (a) After the occurrence of any Event of Default, Mortgagee, to the extent permitted by law and without regard to the value, adequacy or occupancy of the Mortgaged Property, shall be entitled as a matter of right, if it so elects, to the appointment of a receiver to enter upon and take possession of the Mortgaged Property and to collect all rents, royalties, issues, profits, revenues, income and other benefits of and from the Mortgaged Property and apply the same as the court may direct or otherwise as may be permitted by law. Mortgagor hereby specifically and irrevocably consents to such appointment. The receiver shall be entitled to hold, store, use, operate, manage and control the Mortgaged Property and conduct the business thereof as would Mortgagee pursuant to the immediately preceding Section and shall have all rights and powers permitted under the laws of the State of New York and such other rights and powers as the court making such appointment shall confer. The receiver shall be liable to account only for rents, royalties, issues, profits, revenues, income and other benefits actually received by such receiver. Notwithstanding the appointment of any receiver or other custodian, Mortgagee, as pledgee or depository, shall be entitled to the possession and control of any cash, deposits or instruments held by Mortgagee at the time of such appointment or payable or deliverable to Mortgagee from time to time under the terms of this Mortgage or any of the other Loan Documents.
- (b) Upon request by Mortgagee, Mortgagor shall pay to Mortgagee, or to any other person that Mortgagee may designate, or to any such receiver, all reasonable costs, expenses and liabilities (including, without limitation, reasonable attorneys' fees, receivers' fees and the fees of any manager retained by such receiver) incurred by Mortgagee or by such receiver in connection with the appointment of such receiver and the exercise of the rights and powers of such receiver, except to the extent such costs, expenses and liabilities shall have been paid out of collections from the Mortgaged Property as provided in the immediately preceding Section, together with interest thereon at the Default Rate from the date incurred by Mortgagee or by such receiver until the date so paid to, or as directed by, Mortgagee or to such receiver.
- 3.5 Waiver of Certain Rights. Mortgagor agrees, to the extent permitted by law, that neither Mortgagor nor any person at any time claiming through or under Mortgagor shall set up, claim or seek to take advantage of any appraisement, valuation, stay, notice of election to accelerate, mature or declare due the Indebtedness, extension, redemption or moratorium laws, election of remedies, or any exemption from execution or sale, now or hereafter in force, in order to prevent or hinder the foreclosure of this Mortgage after the occurrence of any Event of Default, the final and absolute sale of all or any part of the Mortgaged Property or the final and absolute putting into possession thereof, immediately after any such sale, of the purchaser or purchasers at such sale or the enforcement of any other rights or remedies of Mortgagee under this Mortgage or any of the other Loan Documents. Mortgagor, for itself and for all who may at any time claim through or under Mortgagor or who hereafter may otherwise acquire any interest in or title to all or any part of the Mortgaged Property, hereby waives, to the extent permitted by law, all benefit of any such law or laws, any and all rights of redemption from sale pursuant to any judgment, order or decree of foreclosure of this Mortgage, and any and all right to have the assets constituting the Mortgaged Property marshalled upon any foreclosure or other enforcement of this Mortgage, and any and all right to a jury trial in connection with any enforcement of Mortgagee's rights under any of the Loan Documents and any other cause or proceeding arising out of or otherwise relating to the Loan or the transaction of which this

Mortgage is a part. Mortgagee or any court having jurisdiction to foreclose this Mortgage may sell the Mortgaged Property in part or as an entirety. Mortgagee shall not be required to accept any part or parts of the Mortgaged Property in satisfaction of all or any part of the Indebtedness. Mortgagee shall not be required to accept any apportionment of the Indebtedness to or among any part or parts of the Mortgaged Property. If any law now in force of which Mortgagor might take advantage despite this Section shall hereafter be repealed or cease to be in force, such law shall not thereafter be deemed to preclude the application of this Section.

- 3.6 <u>Leases</u>. Any foreclosure of this Mortgage and any other transfer of title to the Mortgaged Property in extinguishment of all or any part of the Indebtedness may, at Mortgagee's option, be subject to the rights of any tenants of all or any part of the Mortgaged Property, and any failure to make any such tenants parties defendant to any foreclosure proceedings or to foreclose or otherwise terminate their rights will not be, nor be asserted by Mortgagor to be, a defense to any such foreclosure proceedings or to any proceedings seeking collection of all or any part of the Indebtedness, including, without limitation, any deficiency remaining unpaid after the completion of any such foreclosure, any sale in connection therewith or any other transfer in extinguishment of all or any part of the Indebtedness.
- 3.7 <u>Suits To Protect Mortgaged Property</u>. Mortgagee is hereby irrevocably authorized, at Mortgagee's option, to institute and maintain any and all suits and proceedings as Mortgagee may deem advisable after an Event of Default (a) to prevent any impairment of the Mortgaged Property or the security of this Mortgage by any unlawful acts or omissions, (b) to prevent the occurrence or continuance of any violation of this Mortgage or any of the other Loan Documents, (c) to foreclose this Mortgage (after the occurrence of an Event of Default), and (d) to preserve and protect its interest in the Mortgaged Property.
- 3.8 **Proofs of Claim**. In the case of any receivership, insolvency, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial case or proceeding affecting Mortgagor, Mortgagee, to the extent permitted by law, is hereby irrevocably authorized, at Mortgagee's option, to file such proofs of claim and other documents as may be necessary or advisable in order to have its claims allowed in such case or proceeding for the entire Indebtedness, at the date of the institution of such case or proceeding, and for any additional amounts that may become due and payable under any of the Loan Documents after such date.
- 3.9 <u>Application of Moneys by Mortgagee</u>. Any moneys collected or received by Mortgagee in connection with the enforcement of its rights or remedies following any Event of Default shall be applied, in such priority as Mortgagee may determine, to the payment of reasonable compensation, expenses and disbursements of the agents, contractors, attorneys and other representatives of Mortgagee, to the payment of all or any part of the Indebtedness or for any other purpose authorized by any of the Loan Documents or by law.

### 3.10 No Waiver.

(a) No delay or omission of Mortgagee to insist upon strict performance of any obligations of Mortgagor or any Guarantor under or in connection with this Mortgage or any of the other Loan Documents or to exercise any right, power or remedy available after the

occurrence of any Event of Default shall waive, exhaust or impair any such obligation or any such right, power or remedy, nor shall any such delay or omission be construed to waive any such Event of Default or to constitute acquiescence therein. Notwithstanding any such delay or omission, Mortgagee thereafter shall have the right, from time to time and as often as may be deemed expedient by Mortgagee, to insist upon and enforce strict performance of any and all obligations of Mortgagor and any Guarantor under or in connection with this Mortgage or any of the other Loan Documents. Every right, power and remedy given to Mortgagee may be exercised from time to time and as often as may be deemed expedient by Mortgagee.

- (b) No waiver of any Event of Default shall extend to or affect any subsequent Event of Default or any other Event of Default then existing, nor shall any such waiver impair any rights, powers or remedies consequent upon any Event of Default. After the occurrence of any Event of Default (whether or not the Indebtedness shall have been declared to be due and payable immediately), Mortgagee may accept payments of amounts owing in respect of the Indebtedness, and no such acceptance shall waive any such Event of Default or result in any Indebtedness which shall have been declared to be due and payable no longer being due and payable.
- 3.11 **Remedies Cumulative**. No right, power or remedy conferred upon or reserved to Mortgagee or to any receiver by any of the Loan Documents, by law or by any court, is or shall be exclusive of any other right, power or remedy, but each and every such right, power and remedy shall be cumulative and concurrent and shall be in addition to each and every other right, power and remedy given under any of the Loan Documents or now or hereafter existing at law, in equity or by statute.
- 3.12 <u>Purchase by Mortgagee</u>. Upon any foreclosure sale, Mortgagee may bid for and purchase all or any part of the Mortgaged Property and, upon compliance with the terms of sale, may hold, retain, possess and dispose of such property in its own absolute right without further accountability. Upon any foreclosure sale, Mortgagee may, if permitted by law, and after allowing for costs and expenses of the sale, compensation and other charges, in paying the purchase price, apply all or any part of the Indebtedness in lieu of cash, to the amount which shall, upon distribution of the net proceeds of such sale, be payable in connection therewith.
- 3.13 <u>Discontinuance of Proceedings</u>. If Mortgagee shall have proceeded to enforce any right or remedy under this Mortgage by foreclosure, entry or otherwise and such proceedings shall have been discontinued or abandoned for any reason, or if such proceedings shall have resulted in a final determination adverse to Mortgagee, then and in every such case, to the extent permitted by law and any determination of such proceedings, Mortgagor and Mortgagee shall be restored to their former positions and rights hereunder, and all rights, powers and remedies of Mortgagee shall continue as if no such proceedings had occurred or had been taken.
- 3.14 **Additional Security and Guaranty**. If Mortgagee at any time holds additional security for, or any guaranty of, all or any part of the Indebtedness, Mortgagee may, to the extent permitted by law, foreclose such security or otherwise enforce Mortgagee's rights with respect to, or realize upon, such security or such guaranty (as the case may be), at Mortgagee's option, either before or concurrently with or after a foreclosure or other enforcement of this Mortgage, without being deemed to have made an election thereby or to have accepted the benefits of such

guaranty, the security of this Mortgage or such additional security (or the proceeds of such security) in full settlement of the Indebtedness and Mortgagee's rights with respect thereto. Any judgment, order or decree with respect to the Note or with respect to any such guaranty or security, whether rendered in the State of New York or elsewhere, shall not in any manner affect the security of this Mortgage, and any deficiency or other debt represented by said judgment, order or decree shall, to the extent permitted by law, be secured by this Mortgage to the same extent that the Indebtedness was secured hereby prior to the rendering of such judgment.

3.15 **<u>Default Rate</u>**. After the occurrence of any Event of Default until cured, the principal portion of the Indebtedness shall, at Mortgagee's option, bear interest at the Default Rate.

# ARTICLE 4. MISCELLANEOUS

- 4.1 <u>Binding Effect; Survival; Number; Gender</u>. Whenever any of the parties hereto is referred to, such reference shall be deemed to include and apply to the successors and assigns of such party, subject to the provisions of <u>Section 1.15</u> hereof; and all covenants, promises and agreements by or on behalf of Mortgagor in this Mortgage contained shall bind Mortgagor and also its successors and assigns and shall inure to the benefit of Mortgagee and its successors and assigns, whether elsewhere herein so expressed or not. All representations and warranties contained herein or otherwise heretofore made by Mortgagor or any Guarantor to Mortgagee shall survive the execution and delivery hereof but are made of the date hereof. The singular of all terms used herein shall include the plural, the plural shall include the singular, and the use of any gender herein shall include all other genders, where the context so requires or permits.
- 4.2 **Severability**. The unenforceability or invalidity of any provision or provisions of this Mortgage as to any persons or circumstances shall not render that provision nor any other provision or provisions herein contained unenforceable or invalid as to any other persons or circumstances, and all provisions hereof, in all other respects, shall remain valid and enforceable. Mortgagee shall be subrogated for further security to the lien, whether or not released of record, of any and all encumbrances paid out of the proceeds of the Note or the Loan Agreement or out of any advances made by Mortgagee hereunder.
- 4.3 **Notices**. All notices, demands, requests, consents, approvals or other communications (any of the foregoing, a "**Notice**") required, permitted or desired to be given hereunder shall be in writing and shall be sent delivered by (a) registered or certified mail, postage prepaid, return receipt requested, (b) Federal Express, Airborne or another reputable overnight courier, or (c) delivered by hand by commercial courier service, addressed to the party to be so notified at its address set forth below, or to such other address as such party may hereafter specify in accordance with the provisions of this <u>Section 4.3</u>. Any Notice shall be deemed to have been received: (a) three (3) days after the date such Notice is mailed, unless not actually received then (b) on the date of delivery by hand (or refusal to accept such delivery) if delivered during business hours on a Business Day (otherwise on the next Business Day), and/or (c) on the next Business Day if sent by an overnight commercial courier. Notices shall be deemed effective if delivered by counsel to either party, as if given directly by such party. Notices shall be addressed as follows:

If to Mortgagor:

Alexander's Rego Shopping Center, Inc.

c/o Vornado Realty Trust

210 Route 4 East

Paramus, New Jersey 07652 Attention: Chief Financial Officer

With a copy to,

c/o Vornado Realty Trust 888 Seventh Avenue

New York, New York 10019 Attn: Exec. VP, Capital Markets

With a copy to the same address, Attn: Exec. VP, Retail

with a copy to: Winston & Strawn LLP 200 Park Avenue New York, NY 10166-4193 Attention:

Patricia M. Dineen, Esq.

If to Mortgagee: U.S. Bank National Association One Post Office Square, 29<sup>th</sup> Floor Boston,

Massachusetts 12109 Attention: Real Estate Banking Group

with a copy to: U.S. Bank National Association 1650 Tysons Blvd., Suite 1580 McLean, VA 22102

Attn.: Real Estate Banking Division

with a copy to: Halloran & Sage LLP One Goodwin Square 225 Asylum Street Hartford,

Connecticut 06103 Attn: James P. Maher, Esq.

Any party may change the address to which any such Notice is to be delivered by furnishing ten (10) days prior written notice of such change to the other parties in accordance with the provisions of this <u>Section 4.3</u>. Notices shall be deemed to have been given on the date as set forth above, even if there is an inability to actually deliver any such Notice because of a changed address of which no Notice was given, or there is a rejection or refusal to accept any Notice offered for delivery. All notices required under Section 34-25-10(b) and Section 34-25-11 of the General Laws of New York, as amended, shall be accepted by Mortgagee at the address stated above.

4.4 <u>Waiver of Trial by Jury</u>. MORTGAGOR AND MORTGAGEE, FOR THEMSELVES AND THEIR RESPECTIVE SUCCESSORS AND ASSIGNS, HEREBY

(a) AGREE THAT NEITHER OF THEM SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, ACTION, PROCEEDING, COUNTERCLAIM OR OTHER LITIGATION PROCEDURE BASED UPON OR ARISING OUT OF OR OTHERWISE RELATING TO THE INDEBTEDNESS, THIS MORTGAGE, ANY OF THE OTHER LOAN DOCUMENTS, ANY RELATED INSTRUMENT OR AGREEMENT, ANY COLLATERAL FOR ALL OR ANY PART OF THE INDEBTEDNESS, OR THE DEALINGS OR RELATIONSHIP BETWEEN OR AMONG MORTGAGOR, MORTGAGEE AND ANY GUARANTOR (OR ANY OF THEM) IN CONNECTION THEREWITH, (b) IRREVOCABLY WAIVE ANY AND ALL RIGHT TO ANY SUCH JURY TRIAL, AND (c) AGREE THAT NEITHER OF THEM SHALL SEEK TO CONSOLIDATE ANY SUCH LAWSUIT, ACTION, PROCEEDING, COUNTERCLAIM OR OTHER LITIGATION PROCEDURE AS TO WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER LAWSUIT, ACTION, PROCEEDING, COUNTERCLAIM OR OTHER LITIGATION PROCEDURE AS TO WHICH A JURY TRIAL CANNOT OR HAS NOT BEEN WAIVED. THIS SECTION HAS BEEN FULLY DISCUSSED BY MORTGAGOR AND MORTGAGEE, EACH OF WHOM HAS BEEN REPRESENTED BY COUNSEL. THIS SECTION SHALL NOT BE SUBJECT TO ANY EXCEPTIONS, AND NO SUCH PERSON HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PERSON THAT THIS SECTION WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.

4.5 Acts of Mortgagee. In the event Mortgagee (a) grants any extension of time or forbearance with respect to the payment of any Indebtedness; (b) takes other or additional security for the payment thereof; (c) waives or fails to exercise any right, power or remedy granted herein, in the Note or in any other Loan Document; (d) grants any release, with or without consideration, of the whole or any part of the security for the payment of the Indebtedness or the release of any person, party or entity liable for payment of said indebtedness; and/or (e) amends or modifies in any respect any of the terms and provisions hereof, of the Note (including substitution of another note) or of any other Loan Document; then, and in any such event, such act or omission to act shall not release Mortgagor under any covenant of this Mortgage, of the Note, or of any other Loan Document, nor preclude Mortgagee from exercising any right, power or privilege herein or therein granted or intended to be granted, and shall not in any way impair or affect the lien or priority of this Mortgage. In the event any additional real property, improvements, leases, fixtures or personal property not herein specifically identified shall be or become a part of the Mortgaged Property, then this Mortgage shall immediately attach to and constitute a lien against or security interest in such additional items, as appropriate, without further act or deed of either party hereto.

4.6 **Applicable Law**. The Loan secured by this Mortgage was negotiated in the State of New York and the laws of the State of New York shall govern the construction, interpretation, validity and legal effect of this Mortgage and the rights and duties of the parties hereunder, except the laws of the state in which the Mortgaged Property is located shall govern the procedures for enforcing the rights and remedies of the Mortgagee hereunder. Notwithstanding any provision herein, in the Note or in any other Loan Document, the total liability for payments in the nature of interest hereunder and thereunder shall not exceed interest at the maximum rate permitted by the laws of the State of New York on the Indebtedness, if any, and any amounts paid in excess of said maximum rate shall be applied to the principal balance of such

indebtedness, unless said principal balance has been paid in full, in which event any such excess amounts shall be refunded to Mortgagor. This instrument shall be construed in accordance with its intent and with the fair meaning of its provisions, and without regard to any presumption or other rule requiring construction against the party which caused the same to be drafted.

- 4.7 <u>Counterparts</u>. This Mortgage may be executed simultaneously in two (2) or more identical counterparts, each of which, standing alone, shall be an original, but all of which shall constitute but one (1) agreement.
- 4.8 **Release of Mortgage**. If all of the Indebtedness be paid as the same becomes due and payable and all of the covenants, warranties, undertakings and agreements made in this Mortgage are kept and performed, and all obligations, if any, of Mortgagee for further advances have been terminated, then, and in that event only, all rights under this Mortgage shall terminate (except to the extent expressly provided herein with respect to indemnifications, representations and warranties and other rights which are to continue following the release hereof) and the Mortgaged Property shall become wholly clear of the liens, security interests, conveyances and assignments evidenced hereby, and such liens and security interests shall be released by Mortgagee in due form at Mortgagor's cost or at Mortgagor's option, assigned as provided in the last paragraph of this Mortgage. Without limitation, all provisions herein for indemnity of Mortgagee or Mortgagee shall survive discharge of the Indebtedness and any foreclosure, release or termination of this Mortgage.
- 4.9 <u>Invalidity of Certain Provisions</u>. A determination that any provision of this Mortgage is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Mortgage to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances.
- 4.10 <u>Successors and Assigns</u>. The terms, provisions, covenants and conditions hereof shall be binding upon Mortgagor, and the heirs, devisees, representatives, successors and assigns of Mortgagor, and shall inure to the benefit of Mortgagee and Mortgagee and shall constitute covenants running with the Premises and the leasehold estate. All references in this Mortgage to Mortgagor shall be deemed to include all such heirs, devisees, representatives, successors and assigns of Mortgagor.
- 4.11 **No Partnership**, **Etc**. The relationship between Mortgagee and Mortgagor is solely that of lender and borrower. Mortgagee has no fiduciary or other special relationship with Mortgagor. Nothing contained in the Loan Documents is intended to create any partnership, joint venture, association or special relationship between Mortgagor and Mortgagee or in any way make Mortgagee a co-principal with Mortgagor with reference to the Mortgaged Property. All agreed contractual duties between or among Mortgagee, Mortgagor and Mortgagee are set forth herein and in the other Loan Documents and any additional implied covenants or duties are hereby disclaimed. Any inferences to the contrary of any of the foregoing are hereby expressly negated.
- 4.12 **Capitalized Terms**. Capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Loan Agreement.

- 4.13 **Special Provisions**. Notwithstanding any contrary provisions contained herein: (a) <u>Non-Residential Property</u>. This Mortgage does not cover real property principally improved by one or more structures containing in the aggregate not more than six (6) residential dwelling units having their own separate cooking facilities.
- (b) <u>Security Deposits</u>. In furtherance of and not in limitation of any other provisions of this Mortgage, any pledge or assignment of security deposits is subject to Sections 7-103, 7-105 and 7-106 of the New York General Obligations Law.
- (c) <u>Maximum Principal Indebtedness</u>. The parties hereto intend that this Mortgage shall secure unpaid balances of the indebtedness secured hereby whether incurred by Mortgagor at the date hereof or after this Mortgage is delivered for recordation in the official records of the county in which the Mortgaged Property is located. The maximum principal amount of the indebtedness which is or under any contingency may be secured at the date of execution hereof or at any time thereafter by this Mortgage is \$78,245,641.77.
- (d) Section 254. The clauses and covenants contained herein which are construed by Section 254 of the Real Property Law of the State of New York shall, except as otherwise expressly provided herein, be construed as provided in that Section; the additional clauses and covenants contained herein shall afford rights supplemental to and not exclusive of the rights conferred by the clauses and covenants construed by said Section 254 and shall not impair, modify, alter or defeat such rights notwithstanding that such additional clauses and covenants may relate to the same subject matter or provide for different or additional rights in the same or similar contingencies as the clauses and covenants construed by said Section 254; the clauses and covenants herein which are similar to those contained in said Section 254 but which afford additional rights to Mortgagee or to Mortgagor, shall supersede the clauses and covenants contained in Section 254.
- (e) <u>RPAPL</u>. If an Event of Default shall occur and be continuing, Mortgagee may elect to sell (and, in the case of any default of any purchaser, resell) the Mortgaged Property or any part thereof by exercise of the power of foreclosure or of sale granted to Mortgagee by Articles 13 or 14 of the New York Real Property Actions and Proceedings Law (the "RPAPL"). In such case, Mortgagee may commence a civil action to foreclose this Mortgage pursuant to Article 13 of the RPAPL.
- (f) <u>New York Real Property Law Article 4-A</u>. If this Mortgage shall be deemed to constitute a "mortgage investment" as defined by New York Real Property Law Section 125, then this Mortgage shall and hereby does confer upon Mortgagee the powers, and impose upon Mortgagor the duties, of trustees set forth in New York Real Property Law Section 126.
- (g) <u>Trust Fund</u>. Pursuant to Section 13 of the Lien Law of New York, Mortgagor shall receive the advances secured hereby and shall hold the right to receive such advances as a trust fund to be applied first for the purpose of paying the cost of any improvement and shall apply such advances first to the payment of the cost of any such improvement on the Mortgaged Property before using any part of the total of the same for any other purpose.

Mortgagor will indemnify and hold Mortgagee harmless from and against any loss, liability, cost or expense, including any judgments, attorneys' fees, costs of appeal bonds or printing costs, arising out of or relating to any proceedings instituted by any claimant alleging a violation by Mortgagor of Article 3-A of the New York Lien Law.

- (g) <u>Statement in Accordance with Section 274-a of the New York Real Property Law</u>. Mortgagee shall, within fifteen (15) days after written request, provide Mortgagor with the statement required by Section 274-a of the New York Real Property Law.
- (i) <u>Article 3-A of the Lien Law</u>. Mortgagor agrees that it shall indemnify and hold Mortgagee and its successors and assigns harmless against any loss, liability, cost or expense arising out of or relating to any proceedings instituted by any claimant alleging priority over the lien of this Mortgage or alleging a violation by Mortgagor or Mortgagee of any section of Article 3-A of the New York Lien Law.

NOW THEREFORE, if Mortgagor shall pay or cause to be paid (i) the principal and the premium, if any, and interest on the Note and (ii) all other Indebtedness and sums payable hereunder, and if Mortgagor shall comply with all the terms, conditions and requirements hereof, then this Mortgage shall be null and void and of no further force and effect and shall be released by Mortgagor upon the written request of Mortgagor with all reasonable costs relating to the recordation of such release paid by Mortgagor, otherwise to remain in full force and effect. Notwithstanding the foregoing, if requested by Mortgagor, and upon payment of (i) the principal and the premium, if any, and interest on the Note and (ii) all other Indebtedness and sums payable hereunder, and if Mortgagor has complies with all the terms, conditions and requirements hereof, Mortgagee will assign this Mortgage and deliver the Note with an allonge to such party as Mortgagor shall designate, without recourse, representation or warranty whatsoever except a representation that Mortgagee has not previously assigned, and, if requested by Mortgagor, shall provide an executed termination and release with respect to the other Loan Documents, except to the extent that any obligations under such Loan Documents, by their terms or by applicable law, survive the repayment of the Loan. If any original note is lost or destroyed, Mortgagee shall provide a reasonable lost note affidavit. Mortgagor shall pay Mortgagee's reasonable costs incurred in discharging or assigning this Mortgage, as applicable.

[Remainder of page intentionally left blank; signature page follows.]

Signed and Acknowledged		ALEXANDER'S REGO SHOPPING CENTER, INC.
in the Presence of:		By:
Name:		Name: Its:
STATE OF NEW YORK	: : SS	
COUNTY OF	:	
On the day of March, in the year 2009, before me, the undersigned, personally appeared, the of Alexander's Rego Shopping Center, Inc., personally known to me or proved to me on the basis of satisfactory evidence to be the individual whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his capacity, and that by his signature on the instrument, the individual or the person upon behalf of which the individual acted, executed the instrument.		
		Notary Public My Commission Expires:
[Signature Page to Amended and Restated Mortgage, Security Agreement, Fixture Filing and Assignment of Leases and Rents]		

IN WITNESS WHEREOF, Mortgagor has caused this Mortgage to be executed as of the day and year first above written.

# SCHEDULE A [Legal Description]

Exhibit A

#### LEGAL DESCRIPTION

ALL that certain plot, piece or parcel of land, situate, lying and being in the Borough and County of QUEENS, City and State of NEW YORK, bounded and described as follows:

BEGINNING at the corner formed by the intersection of the easterly side of JUNCTION BOULEVARD (80 feet wide) and the southerly side of 62ND DRIVE, formerly URGUHART STREET, as said Boulevard and Drive are shown on the Topographical Map of the City of New York for the Borough of Queens;

running thence easterly along the southerly side of 62ND DRIVE, 456.35 feet to the corner formed by the intersection of the said 62ND DRIVE and the westerly side of 97TH STREET (60 feet wide), as said 97TH STREET was shown on the Topographical Map of the City of New York for the Borough of Queens, prior to the Adoption on December 20, 1951, of Alteration Map No. 3530;

thence southerly along the westerly side of said 97TH STREET, 529.99 to the northwesterly side of 63RD ROAD (80 feet wide) as said 63RD ROAD is shown on the Topographical Map of the City of New York for the Borough of Queens;

thence southwesterly along the northwesterly side of 63RD ROAD, 406.22 feet to the corner formed by the intersection of the said northwesterly side of 63RD ROAD and the northeasterly side of QUEENS BOULEVARD (200 feet wide) as shown on the Final Topographical Map of the City of New York;

thence northwesterly along the northeasterly side of QUEENS BOULEVARD, 86 feet to the corner formed by the intersection of the said northeasterly side of QUEENS BOULEVARD and the easterly side of said JUNCTION BOULEVARD;

thence northerly along the easterly side of JUNCTION BOULEVARD, 549.82 feet to the corner at the point or place of BEGINNING.

FOR INFORMATION ONLY: BLOCK: 2084 LOT: 101

-LEGAL DESCRIPTION-

LEGAL

### **SCHEDULE B**

### [Permitted Encumbrances]

- 1. Survey made by Gerald T. O'Buckley dated April 20, 1999 shows premises improved by a 2 story and basement brick store with 2 story and basement brick addition and masonry garage and the following:
  - a. Projections on 62nd Drive: lights up to 2.2 feet and stand pipe up to 0.4 feet.
  - b. Projections on 97th Street: signs up to 1.5 feet; lights up to 0.3 feet; stand pipe up to 0.4 feet.
  - c. Projections on 63rd Road: lights up to 1.0 foot and signs up to 1.0 foot.
  - d. Projections on Junction Boulevard: lights up to 1.0 foot; copings up to 0.75 feet; signs up to 1.0 foot and stand pipes up to 0.4 feet.

Subject to any changes that would be shown on an accurate survey made since April 20, 1999.

- 2. Covenants and Restrictions recorded in Liber 2690 Page 430.
- 3. Covenants and Restrictions recorded in Liber 2664 Page 420.
- 4. Covenants and Restrictions recorded in Liber 2689 Page 115.
- 5. Covenants and Restrictions recorded in Liber 2666 Page 180.
- 6. Covenants and Restrictions recorded in Liber 2686 Page 284,
- 7. Covenants and Restrictions recorded in Liber 2664 Page 432.
- 8. Covenants and Restrictions recorded in Liber 2664 Page 441.
- 9. Covenants and Restrictions recorded in Liber 2664 Page 442.
- 10. Covenants and Restrictions recorded in Liber 2664 Page 455.
- 11. Covenants and Restrictions recorded in Liber 2664 Page 457.
- 12. Covenants and Restrictions recorded in Liber 2664 Page 467.
- 13. Covenants and Restrictions recorded in Liber 2664 Page 468.

- 14. Covenants and Restrictions recorded in Liber 2666 Page 195.
- 15. Covenants and Restrictions recorded in Liber 2668 Page 419.
- 16. Covenants and Restrictions recorded in Liber 2668 Page 422.
- 17. Covenants and Restrictions recorded in Liber 2668 Page 425.
- 18. Covenants and Restrictions recorded in Liber 2825 Page 5.
- 19. Declaration recorded in Liber 3624 Page 89.
- 20. Consent and Authorization for the Construction, Maintenance and Operation of a Subway Staircase set forth in Liber 3448 Page 454.
- 21. Easement to the BROOKLYN UNION GAS COMPANY recorded in Liber 6805 Page 1. (Affects the roadbed of 62nd Drive on the south side between Junction Boulevard and 99th Street).
- 22. Water Main Easement to the CITY OF NEW YORK recorded in Liber 7442 Page 456. (Affects 97th Street which includes the easterly 10 feet of Parcel).
- 23. Water Main Easement to the CITY OF NEW YORK recorded in Liber 7442 Page 460. (Affects 62nd Drive)
- 24. Terms, Covenants and Conditions of the Indenture recorded in Liber 6297 Page 149, as amended by Liber 6297 Page 159 and by Liber 6800 Page 84. Pertain to the Elimination of Streets.
- 25. Declaration of Restrictions recorded in Liber 6696 Page 40, as amended by Agreement recorded in Liber 7744 Page 339.
- 26. Agreement for Sewage Pumping Station recorded in Liber 6696 Page 45.
- 27. Terms, Covenants and Conditions of the Declaration of Covenants and Restrictions recorded in Reel 926 Page 1291, as amended by Agreement in Reel 2341 Page 1794.
- 28. Terms, Covenants and Conditions of the Easement Agreement recorded in Reel 926 Page 1282.
- 29. Terms, Covenants and Conditions of the Agreement recorded in Reel 932 Page 207.
- 30. Terms, Covenants and Conditions of the Subway Entrance Agreement in Reel 2342 Page 2288.
- 31. Easement for Parking (Block 2080, Lot 1) to ALEXANDER'S INC., dated 3/29/1995,

- recorded 3/30/1995 in Reel 4097 Page 818.
- 32. Waiver of Legal Grade in Reel 4165 Page 2009.
- 33. Terms, Covenants, Conditions and Provisions of the Lease dated 3/1/1995 between ALEXANDER'S, INC. (landlord) and MARSHALLS OF RICHFIELD, MN, INC., (tenant), a Memorandum of Lease having been recorded in Reel 4110 Page 751.

With Regard Thereto:

Subordination, Non-disturbance and Attornment Agreement made by and between UNION BANK OF SWITZERLAND (New York Branch), ALEXANDER'S INC. and MARSHALLS OF RICHFIELD, MN., INC. dated 3/29/95, recorded 12/2/96 in Reel 4474 Page 1385.

- 34. Terms, Covenants, Conditions and Provisions of the Lease dated 12/1/1992 between ALEXANDER'S INC., landlord and the CALDOR CORPORATION, tenant, a Notice of Lease having been recorded on 12/9/1992 in Reel 3458 Page 680.
- 35. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between the CALDOR CORPORATION and UNION BANK OF SWITZERLAND (New York Branch) as agent, dated March 29, 1995 and recorded March 30, 1995 in Reel 4097 Page 866.
- 36. Terms, Covenants, Conditions and Agreement of Lease from ALEXANDER'S, INC. (landlord) and SEARS ROEBUCK AND CO. (tenant) dated March 14, 1994 and recorded April 29, 1994 in Reel 3860 Page 638.
- 37. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between SEARS ROEBUCK AND CO. and UNION BANK OF SWITZERLAND (New York Branch) as Agent dated March 29, 1995 and recorded March 30, 1995 in Reel 4097 Page 880, as assigned to THE CHASE MANHATTAN BANK by Reel 5263 Page 2297
- 38. Terms, Covenants, Conditions and Provisions of a Subordination, Non-Disturbance and Attornment Agreement between MARSHALLS OF RICHFIELD, MN, INC. and UNION BANK OF SWITZERLAND (New York Branch) as Agent dated March 23, 1995 and recorded March 30, 1995 in Reel 4097 Page 891, as assigned to THE CHASE MANHATTAN BANK by Reel 5263 Page 2292.
- 39. Terms, Covenants, Conditions and Provisions of a Subordination Agreement dated March 25, 1995, recorded March 30, 1995 in Reel 4097 Page 848.
- 40. Final Certificate of Eligibility recorded 12/11/02 in Reel 6687 Page 1660.
- 41. Tax Lien Certificate recorded in Reel 4359 Page 546, as assigned to THE CITY OF NEW

- YORK by Tax Lien Assignment dated 12/27/96, recorded 1/3/97 in Reel 4497 Page 420.
- 42. On 5/21/96, the CITY OF NEW YORK DEPARTMENT OF FINANCE conducted a Tax Lien Sale. All Charges identified as Paid through Tax Lien Sale continue to be liens subject to the Lien Sale Provisions.
- 43. Terms, Provisions, Covenants and Agreements as contained in the Lease made by and between ALEXANDER'S OF REGO PARK, INC. and BED BATH & BEYOND, INC., a Memorandum of which is dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1951.

# With Regard Thereto;

- a) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and BED BATH & BEYOND, INC. dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1957, as assigned to THE CHASE MANHATTAN BANK by assignment dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2287.
- 44. Terms, Provisions, Covenants and Agreements as contained in the unrecorded Lease made by and between ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 3/1/97.

# With Regard Thereto:

- a) Agreement made by and between THE CHASE MANHATTAN BANK, ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2425.
- b) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and THE GAP, INC. dated 3/1/97, recorded 6/12/97 in Reel 4606 Page 1356.
- 45. Terms, Provisions, Covenants and Agreements as contained in the Lease made by and between ALEXANDER'S OF REGO PARK, INC. and CIRCUIT CITY STORES, INC., a Memorandum of which is dated 12/12/96, recorded 9/29/98 in Reel 4986 Page 1990.

# With Regard Thereto:

a) Subordination, Attornment and Non-Disturbance Agreement made by and between UNION BANK OF SWITZERLAND, ALEXANDER'S OF REGO PARK, INC. and CIRCUIT CITY STORES, INC. dated 12/12/96, recorded 4/16/99 in Reel 5194 Page 834, as assigned to THE CHASE MANHATTAN BANK by assignment dated 5/12/99, recorded 6/7/99 in Reel 5263 Page 2282.

46. Easement Agreement made between ALEXANDER'S OF REGO PARK II, INC. and ALEXANDER'S REGO SHOPPING CENTER INC. dated 1/6/06, recorded 2/8/06 in CRFN 2006000077479.

# With Regard Thereto:

- a) Amended and Restated Easement Agreement made between ALEXANDER'S OF REGO PARK II, INC. and ALEXANDER'S REGO SHOPPING CENTER INC. dated 12/21/07, recorded 2/14/08 in CRFN 2008000062504.
- 47. 10 foot Easement along the easterly portion of described premises as shown on Tax Map.
- 48. Judgment\*:

Debtor: ALEXANDERS REGO SHOPPING CT

9605 QUEENS BLVD, REGO PARK 11374 0113 CRIMINAL COURT OF THE CITY OF NEW YORK

125-01 QUEENS BLVD, QUEENS, 11

Amount: \$ 500.00 Docketed: 9/10/04 Perfected: 9/9/04

Creditor:

Attorney: DISTRICT ATTORNEY OF QUEENS COUNTY

125-01 QUEENS BLVD, QUEENS 11415

49 . Environmental Control Board Lien filed as of 1/31/09\*:

# ALEXANDER REGO PARK CENTER INC 96-05 QUEENS BOULEVARD, QUEENS, NY 11374

Violation #:038160749L - Docket Date: 1/09 - Amount: \$350.00

50. Any water meter and sewer rent charges.

\*Alexander's Rego Shopping Center, Inc. agrees that it will address these matters in the ordinary course.

## Exhibit A

## Schedule of Mortgages and Assignments

1. **Building Loan Mortgage, Assignment of Leases and Rents and Security Agreement** dated March 29, 1995 in the principal amount of \$38,739,611.00 by Alexander's, Inc., a Delaware Corporation to Union Bank of Switzerland, recorded on March 30, 1995 in Reel 4097, Page 746 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

**Assignment of Mortgage** dated May 12, 1999 by UBS AG, Stamford Branch (successor to Union Bank of Switzerland) to The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2270 in the Office of the City Register of the City of New York, Queens County.

2. **Project Loan Mortgage, Assignment of Leases and Rents and Security Agreement** dated March 29, 1995 in the principal amount of \$46,260,389.00 by Alexander's, Inc., a Delaware Corporation to Union Bank of Switzerland, recorded on March 30, 1995 in Reel 4097, Page 780 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

**Assignment, Assumption and Modification Agreement** dated April 19, 1995 among Alexander's, Inc., a Delaware corporation, Alexander's of Rego Park, Inc., a Delaware corporation and Union Bank of Switzerland, recorded on April 20, 1995 in Reel 4110, Page 739 in the Office of the City Register of the City of New York, Queens County. CERTIFIED COPY

**Assignment of Mortgage** dated May 12, 1999 by UBS AG, Stamford Branch (successor to Union Bank of Switzerland) to The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2276 in the Office of the City Register of the City of New York, Queens County.

Amended, Restated and Consolidated Mortgage and Security Agreement dated May 12, 1999 by and between Alexander's Rego Shopping Center, Inc., a Delaware corporation and The Chase Manhattan Bank, a New York banking corporation, recorded on June 7, 1999 in Reel 5263, Page 2302 in the Office of the City Register of the City of New York, Queens County. Consolidates Mortgages 1 and 2 to form a single lien in the amount of \$82,000,000.00.

**Assignment of Mortgages** dated October 10, 2000 by The Chase Manhattan Bank, a New York banking corporation to State Street Bank and Trust Company, as Trustee for the Registered Holders of Chase Manhattan Bank-First Union National Bank Commercial Mortgage Trust, Commercial Mortgage Pass Though Certificates Series, 1999-1, recorded November 22, 2000 in Reel 5727, Page 0118 in the Office of the City Register of the City of New York, Queens County.

#### AMENDED AND RESTATED PROMISSORY NOTE

March 10, 2009 Queens, New York

\$78,245,641.77

FOR VALUE RECEIVED, ALEXANDER'S REGO SHOPPING CENTER INC., a Delaware corporation with an office at c/o Vornado Realty Trust, 888 Seventh Avenue, New York, New York 10019 ("Maker"), hereby promises to pay to the order of U.S. BANK NATIONAL ASSOCIATION, a national bank chartered under the laws of the United States of America with an office at One Post Office Square, 29<sup>th</sup> Floor, Boston, Massachusetts 02109 ("Bank"), or any subsequent holder hereof, the principal sum of SEVENTY-EIGHT MILLION TWO HUNDRED FORTY-FIVE THOUSAND SIX HUNDRED FORTY-ONE DOLLARS AND 77/100 (\$78,245,641.77) or so much thereof as may have been advanced to or for the benefit of Maker and remains unpaid from time to time (hereinafter called "Principal Balance"), with interest on the Principal Balance, until paid in full, at the rates per annum hereinafter specified, in coin or currency, which, at the time or times of payment, is legal tender for the payment of public and private debts in the United States of America, all in accordance with the terms hereinafter set forth. All interest payable hereunder shall be computed on the basis of a 360 day year, but shall be charged for the actual number of days principal is unpaid.

Bank is the owner and holder of a certain Promissory Note from Maker dated May 12, 1999, in the original principal amount of \$82,000,000 originally payable to The Chase Manhattan Bank, and endorsed to the order of Bank ("Existing Note"). This Note amends and restates the indebtedness previously evidenced by the Existing Note. Maker intends, and Bank, by its acceptance of this Note, agrees, that the indebtedness previously evidenced by the Existing Note remains outstanding, but such indebtedness shall henceforth be evidenced by this Note, and the terms and conditions concerning Maker's obligation to repay said indebtedness and interest thereon shall be governed by the provisions of this Note and the other documents and instruments evidencing and securing this Note (the "Loan Documents"). Neither the execution, delivery or acceptance of this Note nor any of the terms and provisions set forth herein shall be deemed or construed to effect a novation or to cause all or any part of the indebtedness previously evidenced by the Existing Note, or the liability of any person with respect thereto or any security therefor, to be, or to be deemed to have been, paid, satisfied or discharged.

- 1 .<u>Payment Location</u>. All payments of principal and interest under this Note shall be made in lawful money of the United States of America in immediately available funds to Bank at Bank's office at One Post Office Square, 29th Floor, Boston, Massachusetts 02109, Attention: Real Estate Banking Division, or at such other place as may be designated by Bank to Maker in writing.
- 2. <u>Capitalized Terms</u>. Unless the context otherwise indicates, capitalized terms not otherwise defined herein shall have the meanings provided for such terms in that certain Loan Agreement of even date herewith by and between Maker and Bank (hereinafter as it may be amended, modified or supplemented from time to time called the "Loan Agreement").
  - 3. Payments. This Note shall be payable by Maker to Bank as follows:

- (a) Subject to the provisions of Paragraph 4(b) below, interest accruing in accordance herewith shall be payable, as accrued, on the first day of each calendar month hereafter until the principal of the Loan is paid in full, commencing on the first day of the month next following the month in which the Closing occurs.
- (b) The entire unpaid Principal Balance and all interest accrued thereon, together with all other amounts then due and payable under the other Loan Documents, shall be due and payable in full on the Maturity Date.
- (c) Except as otherwise provided in the succeeding sentence, from and after the date hereof, and until the date on which this Note is paid in full, Maker shall pay interest on the Principal Balance at the Loan Rate. The Loan Agreement provides for interest at a Default Rate and for late payment charges on the conditions set forth therein.
- (d) No payment of interest or other consideration made or agreed to be made by Maker pursuant to this Note or any other instrument referring to or securing this Note shall, at any time, be deemed to have been computed at an interest rate in excess of the maximum rate of interest permissible by law, if any. In the event such payments of interest or other consideration provided for in this Note or any other instrument referring to or securing this Note shall result in payment of an effective rate of interest which, for any period of time, is in excess of the limit of the usury law or any other law applicable to the loan evidenced hereby, all sums in excess of those lawfully collectible as interest for the period in question shall, without further agreement or notice between or by any party or parties hereto, be applied to the Principal Balance immediately upon receipt of such monies by Bank with the same force and effect as though Maker had specifically designated, and Bank had agreed to accept, such extra payments as a principal payment, without premium or penalty. If the Principal Balance has been fully paid, any such excess amount shall be refunded to Maker. This provision shall control over every other obligation of Maker hereunder and under any instrument which secures this Note; and
  - (e) Otherwise, all payments made hereunder shall be applied to amounts due in accordance with the Loan Agreement.
- 4. <u>Prepayments</u>. The Principal Balance and accrued interest thereon may be prepaid in full or in part as provided in <u>Section 1.3</u> of the Loan Agreement and subject to any fees payable under <u>Section 1.4</u> of the Loan Agreement.
- 5. <u>Security</u>. The payment and performance of this Note is secured by the lien of that certain Amended and Restated Mortgage, Security Agreement, Fixture Filing and Assignment of Leases and Rents, of even date herewith (the "**Mortgage**") which constitutes a single first priority lien on Maker's fee interest in certain real property located at 96-05 Queens Boulevard, in the Borough of Queens, County of Queens and State of New York (the "**Property**"), and by other documents (herein collectively referred to as the "**Other Security Documents**"), all of even date herewith; and is further secured by a Cash Pledge Agreement of even date herewith from Maker (the "**Pledge**"). The Mortgage, the Other Security Documents, and Pledge are sometimes herein collectively called "**Security Documents**." Advances of the sums evidenced by this Note are to be made pursuant to the Loan Agreement. All of the agreements, conditions, covenants, warranties, representations, provisions and stipulations made by or imposed upon Maker under the Security Documents are hereby made a part of this Note to the same extent and with the same force and effect as if they were fully set forth herein, and Maker covenants and

agrees to keep and perform the same, or cause them to be kept and performed, strictly in accordance with their terms. This Note is given for business purposes and none of the proceeds of the loan evidenced hereby or this Note will be used for personal, family or household purposes.

Each Maker, co-maker, endorser, surety and guarantor hereby guaranties payment of this Note, and waives demand for payment, presentment for payment, notice of nonpayment, protest, notice of protest, notice of dishonor, notice of intention to accelerate maturity, notice of acceleration of maturity, notice of intent to foreclose on any collateral securing this Note, all other notices as to this Note other than notices expressly required under the Loan Documents, diligence in collection as to each and every payment due hereunder, and all other requirements necessary to charge or hold such person or entity to any obligation hereunder, and agrees that without any notice Bank on behalf of the Bank and the other banks now or hereafter parties to the Loan Agreement (hereinafter called "Other Banks") may take additional security herefor or may release any or all security herefor, or alone or together with any present or future owner or owners of all or any part of the Property or by any Other Security Documents, may from time to time extend, renew, or otherwise modify the date or dates or amount or amounts of payment above recited, or Bank may from time to time release any part or parts of the property and interest subject to the Mortgage or the Other Security Documents from the Mortgage and/or the Other Security Documents, with or without consideration, and that, in any such case, each Maker, co-maker, endorser, surety and guarantor shall continue to be bound hereby and to be liable to pay the unpaid balance of the indebtedness evidenced hereby, as so additionally secured, extended, renewed or modified, and notwithstanding any such release, and further agrees to indemnify Bank and the Bank against and hold Bank and Bank harmless from and pay all costs and expenses of collection, including court costs and attorneys' fees (prior to trial, at trial and on appeal) incurred in collecting the indebtedness evidenced hereby, or in exercising or defending, or obtaining the right to exercise, the rights of Bank and Bank hereunder, under the Loan Agreement or under any Security Document, whether suit be brought or not, and in foreclosure, in bankruptcy, insolvency, arrangement, reorganization and other debtor-relief proceedings, in probate, in other court proceedings, or otherwise, whether or not Bank prevails therein, and all costs and expenses incurred by Bank in protecting or preserving the property and interests which are subject to the Mortgage and/or the Other Security Documents.

6. <u>Default</u>. Time is of the essence hereof, subject, however, to notice, grace and cure periods set forth in the Loan Documents. The occurrence of an Event of Default under the Loan Agreement shall constitute an Event of Default under this Note. From and after the date of occurrence and during the continuance of any such Event of Default, and from and after the Maturity Date, interest shall accrue on the Principal Balance at the Default Rate and shall be payable on the first (1<sup>st</sup>) business day of each calendar month or on demand, at the option of Bank.

Bank shall not by any act, delay, omission or otherwise be deemed to have waived any of its rights or remedies, and no waiver of any kind shall be valid unless in writing and signed by Bank. All rights and remedies of Bank under the terms of this Note, under the terms of the Loan Agreement and/or of any Security Document, and under any statutes or rules of law shall be cumulative and may be exercised successively or concurrently by Bank on the Other Banks' behalf. Maker agrees that the Other Banks shall be entitled to all the rights of a holder in due course of negotiable instruments. Any provision of this Note which may be unenforceable or invalid under any law shall be ineffective to the extent of such unenforceability or invalidity without affecting the enforceability or validity of any other provision hereof.

7. Applicable Law; Jury Trial. THIS NOTE WAS NEGOTIATED, EXECUTED AND DELIVERED IN THE STATE OF NEW YORK, IN ALL RESPECTS, INCLUDING, WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, THIS NOTE AND THE OBLIGATIONS ARISING HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, APPLICABLE TO CONTRACTS MADE AND PERFORMED IN SUCH STATE (WITHOUT REGARD TO PRINCIPLES OF CONFLICT LAWS) AND ANY APPLICABLE LAW OF THE UNITED STATES OF AMERICA. TO THE FULLEST EXTENT PERMITTED BY LAW, MAKER HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVES ANY CLAIM TO ASSERT THAT THE LAW OF ANY OTHER JURISDICTION GOVERNS THIS NOTE.

MAKER AND BANK EACH HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION RELATING TO THE PRINCIPAL BALANCE AND/OR THE SECURITY DOCUMENTS. AT THE OPTION OF BANK, THIS NOTE AND THE OTHER SECURITY DOCUMENTS MAY BE ENFORCED IN THE UNITED STATES DISTRICT COURT FOR THE DISTRICT IN WHICH THE PROPERTY LIES; MAKER CONSENTS TO THE JURISDICTION AND VENUE OF ANY SUCH COURT AND WAIVES ANY ARGUMENT THAT VENUE IN SUCH FORUMS IS NOT CONVENIENT. IN THE EVENT AN ACTION IS COMMENCED IN ANOTHER JURISDICTION OR VENUE UNDER ANY TORT OR CONTRACT THEORY ARISING DIRECTLY OR INDIRECTLY FROM THE RELATIONSHIP CREATED BY THIS NOTE, BANK, AT ITS OPTION, SHALL BE ENTITLED TO HAVE THE CASE TRANSFERRED TO ONE OF THE JURISDICTIONS AND VENUES ABOVE DESCRIBED, OR IF SUCH TRANSFER CANNOT BE ACCOMPLISHED UNDER APPLICABLE LAW, TO HAVE SUCH CASE DISMISSED WITHOUT PREJUDICE.

8. Non-Recourse. Bank shall not enforce the liability and obligation of Maker to perform and observe the obligations contained in this Note and the Mortgage by an action or proceeding wherein an *in personam* money judgment shall be sought against Maker, except that Bank may bring a foreclosure action, an action for specific performance or any other appropriate action or proceeding to enable Bank to enforce and realize upon this Note, the Mortgage, the Pledge, the other Loan Documents, and Maker's interests in the Mortgaged Property, the Cash Collateral, and any other collateral given to Bank pursuant to the Mortgage and the other Loan Documents; provided, however, that, except as specifically provided in this Section, any judgment in any such action or proceeding shall be enforceable against Maker only to the extent of Maker's interest in the Mortgaged Property and in any other collateral given to Bank. Subject to the qualifications set forth below, Bank, by accepting this Note, the Mortgage and the other Loan Documents, agrees that it shall not sue for, seek or demand any deficiency judgment against Maker in any such action or proceeding, under, by reason of or in connection with this Note or the Mortgage. The provisions of this Section shall not, however (a) constitute a waiver, release or impairment of any obligation evidenced or secured by this Note, the Mortgage or the other Loan Documents; (b) impair the right of Bank to name Maker as a party defendant in any action or suit for foreclosure and sale under the Mortgage; (c) affect the validity or enforceability of any guaranty or indemnity made in connection with this Note, the Mortgage, or the other Loan Documents; (d) impair the right of Bank to obtain the appointment of a receiver; (e) impair the enforcement of any other Loan Document; (f) impair the right of Bank to bring suit with respect to fraud or material misrepresentation by Maker or any guarantor in connection with the Loan, this Note, the Mortgage, or the other Loan Documents; or (g) affect the

of the Indemnification Agreement or the Good Faith Indemnity and Guaranty Agreement or limit the liability or recourse of Maker or any other party thereunder.

Nothing set forth in this Note shall be deemed to be a waiver of any right which Bank may have under Section 506(a), 506(b), 1111(b) or any other provisions of the U.S. Bankruptcy Code to file a claim for the full amount of the indebtedness secured by the Mortgage or to require that all collateral shall continue to secure all of the indebtedness owing to Bank in accordance with this Note and the other Loan Documents.

Notwithstanding any other provisions in this Note, the Mortgage or the other Loan Documents, Maker shall be fully liable for and shall indemnify Bank for any or all claims, demands, liabilities, losses, damages, judgments, penalties, costs and expenses (including, without limitation, attorneys' fees) which may be imposed upon, asserted against, or incurred or paid by Bank by reason of, arising out of or attributable or relating to (collectively and inclusive of (A) through (G) hereof):

- (A) the misappropriation of any proceeds from the sale, refinancing or other transfer of any of the collateral for the Loan received and retained by Maker, or any Affiliate of Maker (Maker or any of the foregoing, a "Liable Party") in violation of the terms of any of the Loan Document (to the extent of such proceeds, rents, revenues, profits and/or other benefits) (it being agreed that proceeds shall not be deemed misapplied in any instance unless same are received by Maker and not paid to Bank, in a circumstance in which Bank is expressly entitled to receive same from Maker pursuant to the terms of the Loan Documents);
- (B) any fraud or willful misconduct of any Liable Party with respect to the Project, including, without limitation, any breach of a representation or warranty made by any Liable Party under any Loan Document that was known to be false in any material respect when made;
- (C) misapplication or conversion by any Liable Party of (i) any insurance proceeds paid by reason of any loss, damage or destruction to the Property; (ii) any awards or other amounts received in connection with the condemnation of all or a portion of the Property; or (iii) rents, income, accounts receivable, issues, profits, proceeds, accounts or other amounts received by Maker or any member after an Event of Default under the Loan Documents and so long as such Event of Default is continuing (it being agreed that proceeds shall not be deemed misapplied in any instance unless same are received by Maker and not paid to Bank, in a circumstance in which Bank is expressly entitled to receive same from Maker pursuant to the terms of the Loan Documents);
- (D) intentional damage to or bad faith waste of the Property and damage or loss of value to the Property resulting from gross negligence or willful misconduct of any Liable Party with respect to their obligations to Bank or with respect to the operation of the Property;
- (E) failure to obtain Bank's prior written consent to a transfer or encumbrance of the Property to the extent such consent is required under the Loan Documents;
- (F) failure by Maker to pay taxes, assessments, charges for labor or materials or other charges that result in liens on any portion of the Property that would be superior

to the lien of the Mortgage, which are not bonded over or otherwise disposed of in accordance with the terms of the Mortgage; or

(G) any security deposits, advance deposits or retained rents and profits collected with respect to the Property which are not delivered to Bank upon a foreclosure of the Mortgage or action in lieu thereof

provided; however that the limitations on recourse set forth in this <u>Section 8</u>, will be null and void and completely inapplicable, and this Note shall be full recourse to Maker in the event that:

(A) Maker, any member of Maker, or any member of a member of Maker, shall file a voluntary bankruptcy or insolvency proceeding against Maker, or (2) an involuntary bankruptcy or insolvency proceeding (not brought by Bank) shall be brought against Maker, which is facilitated, conducted or directed by Maker, any member of Maker, any of its principals or affiliates.

The provisions of this Note regarding the limited non-recourse nature of the indebtedness evidenced by this Note shall survive any termination, satisfaction, assignment, entry of a judgment of foreclosure, exercise of power of sale, acceptance by Bank of a deed in lieu of foreclosure or repayment of the sums secured hereby (provided, however, it shall not survive repayment in accordance with the terms of the Loan Documents, and any such post-foreclosure/acquisition shall terminate if there is no claim within twelve (12) months).

[Remainder of page intentionally left blank; signature page to follow]

IN WITNESS WHEREOF, Maker has caused this Note to be duly executed and delivered, under seal, as of the day and year first above set forth.

# ALEXANDER'S REGO SHOPPING CENTER, INC.

	By:		
		Name:	
		Its:	
[Signature p	age to Amended and Restated Note]		

## **CASH PLEDGE AGREEMENT**

THIS CASH PLEDGE AGREEMENT (this "Agreement") is made as of the 10<sup>th</sup> day of March, 2009, by ALEXANDER'S REGO SHOPPING CENTER, INC., a Delaware corporation (the "Borrower"), to and for the benefit of U.S. BANK NATIONAL ASSOCIATION, a national banking association ("Bank").

#### RECITALS:

- A. Pursuant to that certain Loan Agreement (the "**Loan Agreement**") of even date herewith by and between Bank and Borrower, Bank has agreed to make a loan to Borrower in the aggregate principal amount of up to \$78,245,641.77 (the "**Loan**"), which is evidenced by, among other things, that certain Amended and Restated Promissory Note of even date herewith given by Borrower to Bank in the stated principal amount of \$78,245,641.77, as the same may be amended, supplemented, renewed or replaced from time to time (hereinafter called the "**Note**"). Capitalized terms used and not defined herein have the meaning ascribed to them in the Loan Agreement.
- B. As an inducement to the Bank to make the Loan, and as a condition precedent thereto, Borrower has agreed to deposit with Bank \$78,245,641.77 cash, in readily available funds (the "**Deposit**"), to serve as cash collateral for the Obligations (as hereinafter defined) and a source for satisfaction of the Obligations.
- C. Bank is also the holder of a first-priority mortgage of the Property (the "Mortgage") and assignment of leases and rentals of the Property (the "Assignment of Rents"; the Loan Agreement, the Note, the Mortgage, the Assignment of Rents and the other instruments, documents and agreements that evidence and secure the Loan are collectively referred to as the "Loan Documents"). The principal of and all interest on the Loan, all of Borrower's other obligations under the Loan Documents, including without limitation all fees, costs and expenses of Bank incurred in connection with the Loan are hereinafter collectively referred to as the "Obligations."
- D. The execution and delivery of this Agreement and the performance by Borrower of its obligations hereunder is a condition precedent to Banks' making of the Loan.

NOW, THEREFORE, to induce Banks to make the Loan and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound, Borrower hereby covenants and agrees as follows:

#### 1. **Deposit Account**.

(a) Prior to or contemporaneously with the execution of this Agreement, Borrower shall (i) establish with Bank the Deposit Account, subject to the terms of this Agreement and the Loan Agreement, and (ii) deposit the Deposit therein. The Deposit Account and any interest or other income thereon or proceeds thereof (collectively, the "Funds") shall be kept on deposit at the Bank (Bank being sometimes referred to herein as the "Depository"), at its office located in the

1406118v.9

State of Minnesota or the State of New York in an account that is insured in whole or in part by the FDIC. Bank represents to Borrower that Bank's head office is in Minnesota. The Deposit Account has the account numbers set forth in Exhibit A, as amended from time to time. The Deposit Account shall be kept under the sole dominion and control of the Depository. The Deposit Account shall at all time be a balance sheet liability of Bank and shall not be a custodial account. Other than a transfer of Funds pursuant to Section 1(c) below or the application of the Funds in connection with a permitted prepayment of the Loan, Borrower shall have no right to withdraw any amounts from the Deposit Account, including any interest paid thereon, unless and until the Obligations shall have been paid in full.

- (b) Borrower agrees to execute and deliver such additional documents as may be necessary from time to time to establish and maintain each Deposit Account. If any Deposit Account or portion thereof is evidenced by a certificate or instrument, Borrower shall deliver and, if necessary, endorse in any manner necessary to transfer, to the Bank any such certificates or instruments constituting, representing or evidencing any such Deposit Account or portion thereof.
- (c) The Funds shall be maintained in the Noninterest-bearing Transaction Account, provided that, at any time after June 10, 2009, Borrower may, from time to time, by written notice to Bank (a "Transfer Notice"), elect to transfer all or any portion of the Funds from the Noninterest-bearing Transaction Account to the Interest-bearing Investment Account, or vice versa, effective as of the first (1<sup>st</sup>) day of next following month. To be effective, each Transfer Notice must designate the exact dollar amount of Funds to be so transferred and be given no earlier than thirty (30) or later than ten (10) days prior to the proposed effective date of transfer. Following an effective election to transfer Funds, on the first day of the month next following the delivery of a Transfer Notice, the specified amount of the Funds, as the case may be, shall be so transferred by the Depository into an Interest-bearing Investment Account or the Non-interest-bearing Transaction Account, subject to the lien and security interest of Bank hereunder and the other terms of this Agreement and of the Loan Agreement. Borrower and Bank agree to cooperate fully with one another in establishing the Interest-bearing Investment Account and funding of the Funds therein.

#### 2. <u>Treatment of Cash Collateral; Disbursements</u>.

- (a) All interest earnings and other returns on the Funds shall be payable for the account of Borrower and added to the funds on deposit in the Deposit Account. Borrower represents and warrants that it shall be solely responsible for payment of state and federal income taxes in respect of the Funds, and Bank agrees to issue Borrower in the normal course the necessary federal tax reporting statements (e.g., Form 1099) for the interest earned or other returns on the Cash Collateral. Borrower's federal taxpayer identification number is 22-3652541. Bank shall provide Borrower with statements of account with respect to activity in the Deposit Account in accordance with its customary practices. So long as no Default or Event of Default shall have occurred and be continuing, all interest earnings and other returns on the Deposit Account, if any, shall be disbursed by Bank to Borrower monthly. All such disbursements shall be made by Bank in accordance with and pursuant to its standard operating procedures.
  - (b) Bank agrees not to pledge any of its right, title or interest in and to the Deposit Account as collateral or otherwise encumber the Deposit Account.

## 3. Pledge; Remedies.

- (a) For good and valuable consideration, Borrower hereby pledges to Bank and grants Bank a security interest in any and all right, title and interest of Borrower in and to (i) the Deposit Account, (ii) the Funds, (iii) any cash or other property at any time and from time to time receivable or otherwise distributable in respect of, in exchange for, or in substitution for, the Deposit Account or any portion thereof, including interest accruing thereon, (iv) all replacements, substitutions, extensions, renewals and proceeds of any and all of the foregoing, and (v) all rights and privileges of Borrower(s) with respect to any and all of the foregoing (collectively, the "Cash Collateral"), as security for the Obligation. It is the intention of Borrower and Bank that the security interests granted by Borrower to Bank shall be made effective and shall be perfected to the fullest extent possible by Bank.
  - (b) Borrower represents, warrants and covenants as follows:
  - (i) Borrower is a duly formed corporation organized and existing under the laws of the state of Delaware and has full right, power and authority to pledge, assign and grant a security interest in the Cash Collateral;
  - (ii) the execution, delivery and performance of this Agreement, the pledge of the Cash Collateral hereunder and the exercise of the Bank's rights hereunder and/or under applicable law do not and will not violate or contravene the terms of Borrower's charter documents or any agreement, instrument, law, rule, regulation, or judgment binding on Borrower or its properties;
  - (iii) no registration with, or consent or approval of, or other action by or with, any court or governmental body or authority or any other Person is required in connection with the execution, delivery and performance of this Agreement or the exercise of the Bank's rights hereunder;
    - (iv) this Agreement constitutes the legal, valid and binding obligation of Borrower enforceable in accordance with its terms;
  - (v) the security interests granted hereunder to the Bank are and will continue to be (or will be, in the case of Cash Collateral hereafter arising) a valid first lien on, and security interest in the Cash Collateral, superior and prior to the rights of all third Persons, and no filing or other act is required to create and perfect such lien and security interest (excluding the Bank's own actions with respect to the Collateral);
  - (vi) Borrower is not currently insolvent and the transfer and pledge of the Collateral to the Bank does not result in the insolvency of the Borrower;
  - (vii) there is no litigation, arbitration, investigation or proceeding of or before any court, arbitrator or administrative or governmental authority is currently pending or, to the knowledge of Borrower, threatened, that could, if adversely determined, result in a Material Adverse Occurrence;
  - (viii) Borrower has not incurred any indebtedness, or any contingent liability, and will not incur any indebtedness or contingent liability, that materially adversely

affects, or will materially adversely affect, Borrower's ability to perform under this Agreement;

- (ix) Borrower will defend the security interest of Bank in the Cash Collateral against all claims and demands of all persons at any time claiming the same or any interest therein adverse to Bank.
- (x) There are no security interests, liens or encumbrances on all or any of the interest of the Borrower in the Cash Collateral, and there are no deposit account control agreements or other notices, statements or agreements creating or evidencing a security interest or lien naming Borrower or Borrower's interest in any of the Cash Collateral, other than those in favor of Bank.
- (xi) Borrower will not sell or transfer an interest in or grant, permit or suffer to exist any security interest, lien, encumbrance or other interest in any of the Cash Collateral, except as may be granted to Bank.
- (xii) Borrower shall perform, at its sole reasonable cost and expense, any and all actions, and shall pay the amount of all reasonable expenses necessary to obtain, preserve, perfect, defend and enforce the security interest of Bank in any of the Cash Collateral, as provided herein.
- (xiii) Borrower shall not enter into any acknowledgment or agreement that gives any other person or entity except Lender control over, or any other security interest, lien or title in, the Deposit Account.
- (xiv) Borrower shall sign and deliver, and cause to be signed and delivered, at the sole cost of Borrower, any instruments furnished by the Bank, including, without limitation, financing statements and continuation statements, which are necessary or desirable in the reasonable judgment of the Bank to obtain, create, maintain and perfect the security interest hereunder and to enable the Bank to comply with any federal or state law in order to obtain, create or perfect the Bank's interest in the Cash Collateral or to obtain proceeds of the Cash Collateral.
- (xv) Borrower shall notify the Bank immediately of a material adverse change in any matter warranted or represented by Borrower in this Agreement.
- (c) <u>Certain Remedies</u>. In addition to any other rights, powers or remedies the Bank may have under this Agreement, the other Loan Documents, applicable law, or as a secured party under the Uniform Commercial Code, if an Event of Default shall exist, and the Bank shall have accelerated payment of the Loan, the Bank may, in its sole discretion, (i) receive interest and other payments with respect to the Cash Collateral and apply same to the Obligations, (ii) apply the Cash Collateral to the outstanding Loan balance in accordance with the terms of this Agreement and the other Loan Documents, or (iii) continue to hold and maintain the Cash Collateral as additional collateral for the Loan subject to the terms of this Agreement, and, in its sole discretion, transfer any of the Cash Collateral or evidence thereof into its own name or that of its nominee and apply same to the Obligations. If any Event of Default shall exist, and the Bank shall have accelerated payment of the Loan, Bank may, at its option, without demand or notice, setoff or recoup all or any part of the Cash Collateral so received by Bank against all or

any part of the Obligations (whether matured or unmatured), such order and priority as Bank may elect in its discretion.

Borrower agrees that Bank may exercise its remedies as to the Cash Collateral, regardless of whether: (i) Bank has made demand on Borrower; (ii) Bank has taken any other enforcement or other action of any nature against Borrower or the Property; (iii) Bank has pursued any rights which Bank has against any other person who may be liable for any of the Obligations; (iv) Bank holds or has resorted to any security for any of the Obligations; or (v) Bank has invoked any other remedies or rights Bank has available with respect to any of the Obligations, including, without limitation, foreclosure of the Mortgage. The liability of Borrower is unconditional and Borrower agrees that the Cash Collateral may be applied against the Obligations regardless of whether any of the Loan Documents is for any reason invalid or unenforceable.

- (d) No Release or Discharge. No application of all or any part of the Cash Collateral to the Obligations pursuant this Section shall in any way release, satisfy or discharge any remaining unpaid Obligations or any security therefor. No delay or omission of Bank to insist upon strict performance of any obligations of any other party hereto under or in connection with this Agreement or to exercise any right, power or remedy available under or in connection with this Agreement (after the occurrence of any Event of Default or otherwise) shall waive, exhaust or impair any such obligation or any such right, power or remedy, nor shall any such delay or omission be deemed to be a waiver of, or acquiescence in or to, any Event of Default. Notwithstanding any such delay or omission, Bank thereafter shall have the right, from time to time and as often as Bank deems advisable, to insist upon strict performance of any and all obligations of the parties hereto and to exercise any and all rights, powers and remedies available under or in connection with this Agreement.
- (e) <u>Bankruptcy</u>. In the event that Borrower shall file a petition with any bankruptcy court or be the subject of any petition filed under the Bankruptcy Code, Borrower hereby acknowledges and agrees that all such Cash Collateral shall constitute Bank's cash collateral within the meaning of § 363 of the Bankruptcy Code. The Borrower further acknowledges that in such event, Bank does not consent to Borrower's use of such cash collateral, and that Borrower shall have no right to use or apply any such cash collateral other than for application to the Obligations.
- 4. **Power of Attorney.** In addition to the provisions contained elsewhere in this Agreement, to the full extent permitted by law, Borrower hereby irrevocably designates, makes, constitutes and appoints Bank (and all persons designated by Bank) as Borrower's true and lawful attorney (and agent-in-fact) and Bank, or Bank's agent, may, without notice to Borrower from and after an Event of Default and during the continuance thereof, and at such time or times thereafter as Bank or said agent, in its sole discretion, may determine, in Borrower's or Bank's name take such action with respect to the Cash Collateral as Bank determines may be necessary to satisfy payment of the Obligations. This appointment shall be deemed a power coupled with an interest and shall not be terminable as long as the Obligations are outstanding.

# 5 . Certain Waivers.

(a) Borrower waives any and all notice (other than as specifically set forth herein) with respect to: (i) acceptance by Bank of this Agreement or any of the Loan Documents; and (ii) the

provisions of any of the Loan Documents or any other instrument or agreement relating to the Obligations; and (iii) any default in connection with the Obligations.

- (b) Borrower waives any presentment, demand, notice of dishonor or nonpayment, protest, notice of protest and notice of nonpayment in connection with the Obligations other than any notice specifically set forth herein or in the Loan Documents.
- (c) Borrower waives all requirements of law relating to the marshalling of assets, if any, which would be applicable in connection with the enforcement by Bank of its remedies.
  - (d) Borrower waives any right to require that any action be brought against any other person or to require that resort be had to any other security.
- 6. Bank's Rights. Borrower agrees that Bank may from time to time and as many times as Bank, in its sole discretion, deems appropriate, do any of the following without notice to Borrower and without adversely affecting the validity or enforceability of this Agreement or any other agreement, document or instrument given by Borrower to Bank in connection with this Agreement or the Obligations, including, without limitation, any mortgage or other security instrument given by Borrower to secure its Obligations under this Agreement: (i) release, surrender, exchange, compromise or settle the Obligations, or any part thereof; (ii) change, renew or waive the terms of the Obligations, or any part thereof; (iii) waive the terms of any of the Loan Documents or any other note, instrument or agreement relating to the Obligations, such rights in Bank to include without limitation the right (with the Borrower's approval) to increase the amount of the Loan or to change the rate of interest charged to Borrower (in which event the Obligations shall be deemed also to include all interest at such changed rate); (iv) grant any extension or indulgence with respect to the payment or performance of the Obligations or any part thereof; (v) enter into any agreement of forbearance with respect to the Obligations, or any part thereof; (vi) release, surrender, exchange or compromise any security held by Bank for any of the Obligations; (vii) release any person who is a guarantor or surety or who has agreed to purchase the Obligations or any part thereof; (viii) release, surrender, exchange or compromise any security or lien held by Bank for the liabilities of any person who is guarantor or surety for the Obligations or any part thereof; and (ix) settle, release, adjust or compromise any claim of Bank against Borrower or any other person secondarily or otherwise liable, including but not limited to any other guarantors or sureties of the Obligations. Borrower agrees that Bank may do any of the above as it deems necessary or advisable, in it
  - 7. **Borrower's Remedy Upon Bank Failure**. The provisions of <u>Section 8.2</u> of the Loan Agreement are hereby incorporated herein, <u>mutatis mutandis</u>.
- 8. **Rescission**. If at any time all or any part of any payment theretofore applied by Bank to any of the liabilities is or must be rescinded or returned by Bank for any reason whatsoever (including, without limitation, the insolvency, bankruptcy or reorganization of Borrower), such liability shall, for the purposes of this Agreement, to the extent that such payment is or must be rescinded or returned, be deemed to have continued in existence, notwithstanding such application by Bank, and this Agreement shall continue to be effective or be reinstated, as the case may be, as to such liabilities, all as though such application by Bank had not been made.

- 9. <u>Indemnity</u>. Borrower agrees to defend, indemnify and hold Bank and its directors, officers, employees, attorneys, successors and assigns (collectively "Indemnified Parties") harmless from and against any and all claims, losses, liabilities, costs, damages and expenses, including, without limitation, reasonable legal fees (collectively, "Claims"), arising out of or in any way related to this Agreement, excepting only liability arising out of the gross negligence or willful misconduct of an Indemnified Party. IN NO EVENT WILL BANK BE LIABLE FOR ANY INDIRECT DAMAGES, LOST PROFITS, SPECIAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES WHICH ARISE OUT OF OR IN CONNECTION WITH THE SERVICES CONTEMPLATED BY THIS AGREEMENT EVEN IF BANK HAS BEEN INFORMED OF THE POSSIBILITY OF SUCH DAMAGES.
- 10. <u>Legal Process and Insolvency</u>. In the event Bank receives any form of legal process concerning the Deposit Account, including, without limitation, court orders, levies, garnishments, attachments, and writs of execution, or in the event Bank learns of any insolvency proceeding concerning Borrower, including, without limitation, bankruptcy, receivership, and assignment for the benefit of creditors, Bank will respond to such legal process or knowledge of insolvency in the normal course or as required by law.
- 11. <u>Control</u>. The parties agree that this Agreement shall constitute an authenticated record for purposes of § 9-104 of the Uniform Commercial Code and that the Bank shall have "control" of the Cash Collateral for all purposes and as described in § 9-104 of the Uniform Commercial Code.
- 12. **Sealed Instrument**. Borrower recognizes that this Agreement when executed constitutes a sealed instrument and as a result the instrument will be enforceable as such without regard to any statute of limitations which might otherwise be applicable and without regard to consideration.
- 13. <u>Notices</u>. Borrower agrees that all notices, statements, requests, demands and other communications made pursuant to or under this Agreement shall be made in the manner set forth in the Loan Agreement and if sent to Borrower, to each of their respective addresses listed under their signatures, below, and if sent to Bank, to the address set forth in the Loan Agreement.
- 14. Entire Agreement; Modification. This Agreement is the entire agreement between the parties hereto with respect to the subject matter hereof, and supersedes and replaces all prior discussions, representations, communications and agreements (oral or written). This Agreement shall not be modified, supplemented, or terminated, nor any provision hereof waived, except by a written instrument signed by the party against whom enforcement thereof is sought, and then only to the extent expressly set forth in such writing.
- 15. **Binding Effect; Joint and Several Obligations**. This Agreement is binding upon and inures to the benefit of Borrower, Bank and their respective heirs, executors, legal representatives, successors, and assigns, whether by voluntary action of the parties or by operation of law. Borrower may not delegate or transfer his/its obligations under this Agreement. If there is more than one Borrower, each Borrower shall be jointly and severally liable hereunder.
- 16. <u>Unenforceable Provisions</u>. Any provision of this Agreement which is determined by a court of competent jurisdiction or government body to be invalid, unenforceable or illegal shall

be ineffective only to the extent of such determination and shall not affect the validity, enforceability or legality of any other provision, nor shall such determination apply in any circumstance or to any party not controlled by such determination.

- 17. **Due Authorization and Execution**. This Agreement has been duly executed and delivered and constitutes the valid and legally binding obligation of the Borrower, enforceable in accordance with its terms. The execution, delivery and performance of this Agreement by the Borrower will not violate any provisions of law, any order of any court or governmental agency, the charter documents and by-laws or partnership agreement or operating agreement of the Borrower.
- 18. <u>Duplicate Originals; Counterparts</u>. This Agreement may be executed in any number of duplicate originals, and each duplicate original shall be deemed to be an original. This Agreement (and each duplicate original) also may be executed in any number of counterparts, each of which shall be deemed an original and all of which together constitute a fully executed Guaranty even though all signatures do not appear on the same document.
- 19. **Remedies Not Exclusive**. Borrower agrees that the enumeration of Bank's rights and remedies set forth in this Agreement is not intended to be exhaustive and the exercise by Bank of any right or remedy shall not preclude the exercise of any other rights or remedies, all of which shall be cumulative and shall be in addition to any other right or remedy given hereunder or under any other agreement among the parties to the Loan Documents or which may now or hereafter exist at law or in equity or by suit or otherwise.
- 20. **No Waiver**. Borrower agrees that no failure on the part of Bank to exercise any of its rights under this Agreement shall be a waiver of such rights or a waiver of any default by Borrower. Borrower further agrees that each written waiver shall extend only to the specific instance actually recited in such written waiver and shall not impair the rights of Bank in any other respect.
- 21. <u>Costs</u>. Borrower agrees to pay all costs and expenses, including reasonable attorneys' fees, incurred by Bank in enforcing this Agreement against Borrower.
- 22. **No Election of Remedies**. Borrower acknowledges that Bank may, in its sole discretion, elect to enforce this Agreement for the total Obligations or any part thereof against Borrower without any duty or responsibility to pursue any other person or entity and that such an election by Bank shall not be a defense to any action Bank may elect to take against Borrower, provided Bank has provided applicable notice of default if required by the Loan Documents.
- 23. <u>Governing Law</u>. The construction, validity and performance of this Agreement and the obligations arising hereunder shall be governed by, and construed in accordance with, the laws of the State of New York applicable to contracts made and performed in such state (without regard to principles of conflict of laws) and any applicable law of the United States of America. To the fullest extent permitted by law, Borrower hereby unconditionally and irrevocably waives any claim to assert that the law of any other jurisdiction governs this Agreement.
- 24. <u>Jurisdiction</u>. BORROWER HEREBY IRREVOCABLY CONSENTS TO THE JURISDICTION OF THE UNITED STATES DISTRICT COURT FOR THE DISTRICT IN WHICH THE PROPERTY LIES OR THE STATE COURT SITTING IN QUEEN'S COUNTY,

NEW YORK, AND WAIVES PERSONAL SERVICE OF ANY AND ALL PROCESS UPON HIM AND AGREES THAT ALL SUCH SERVICE OF PROCESS BE MADE BY CERTIFIED OR REGISTERED MAIL DIRECTED TO BORROWER AT THE ADDRESS SET FORTH ON THE SIGNATURE PAGE HEREOF AND SERVICE SO MADE SHALL BE DEEMED TO BE COMPLETED UPON ACTUAL RECEIPT THEREOF. BORROWER WAIVES ANY OBJECTION TO JURISDICTION AND VENUE OF ANY ACTION INSTITUTED AGAINST HIM AS PROVIDED HEREIN AND AGREES NOT TO ASSERT ANY DEFENSE BASED ON LACK OF JURISDICTION OR VENUE. BORROWER ACKNOWLEDGES AND AGREES THAT THE VENUE PROVIDED ABOVE IS THE MOST CONVENIENT FORUM FOR BANK, BORROWER AND BORROWER. NOTHING CONTAINED HEREIN SHALL PREVENT BANK FROM BRINGING ANY ACTION, ENFORCING ANY AWARD OR JUDGMENT OR EXERCISING ANY RIGHTS AGAINST ANY PARTY INDIVIDUALLY, AGAINST ANY SECURITY OR AGAINST ANY PROPERTY OF ANY PARTY WITHIN ANY OTHER COUNTY, STATE OR OTHER FOREIGN OR DOMESTIC JURISDICTION.

25. Jury Trial Waiver. BORROWER AND BANK WAIVE THE RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING BASED UPON OR RELATED TO THE SUBJECT MATTER OF THIS AGREEMENT OR ANY OF THE OTHER LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS RELATED TO ANY OF THE LOAN DOCUMENTS. THIS WAIVER IS KNOWINGLY, INTENTIONALLY AND VOLUNTARILY MADE BY EACH PARTY AND EACH PARTY ACKNOWLEDGES THAT NEITHER THE OTHER NOR ANY PERSON ACTING ON BEHALF OF THE OTHER HAS OR HAVE MADE ANY REPRESENTATIONS OF FACT TO INDUCE THIS WAIVER OF TRIAL BY JURY OR IN ANY WAY TO MODIFY OR NULLIFY ITS EFFECT. BORROWER AND BANK EACH FURTHER ACKNOWLEDGES THAT IT HAS BEEN REPRESENTED (OR HAS HAD THE OPPORTUNITY TO BE REPRESENTED) IN THE SIGNING OF THIS AGREEMENT AND IN THE MAKING OF THIS WAIVER BY INDEPENDENT LEGAL COUNSEL, SELECTED OF ITS OWN FREE WILL AND THAT IT HAS HAD THE OPPORTUNITY TO DISCUSS THIS WAIVER WITH COUNSEL.

26. <u>Return of Cash Collateral</u>. Upon full, final and indefeasible payment of all Obligations in cash, all Cash Collateral then remaining, if any, shall be returned to Borrower.

[Remainder of page is blank; signatures appear on next page]

S

1406118v 9

WITNESS:	BORROWER:
	ALEXANDER'S REGO SHOPPING CENTER, INC., a Delaware corporation
By: Name: Title:	By: Name: Title:
	Address: c/o Vornado Realty Trust 210 Route 4 East Paramus, New Jersey 07652 Attn: Chief Financial Officer
	With a copy to, c/o Vornado Realty Trust 888 Seventh Avenue New York, NY 10019 Attn: Exec. VP, Capital Markets
	With a copy to the same address, Attn: Exec. VP, Retail
	With a copy to: Winston & Strawn LLP 200 Park Avenue New York, NY 10166-4193 Attn: Patricia M. Dineen, Esq.
406118v.9	[Signature page to Cash Pledge Agreement]

STATE OF NEW YORK	:	
COLINERY OF	:	SS
COUNTY OF	:	
Alexander's Rego Shopping Center, Inc	., personal cknowled	ne, the undersigned, personally appeared, the of whown to me or proved to me on the basis of satisfactory evidence to be the individual whose name is do to me that he executed the same in his capacity, and that by his signature on the instrument, the individual ed, executed the instrument.
		Notary Public My Commission Expires:
1406118v.9		[Acknowledgement page to Cash Pledge Agreement]

# Exhibit A

## Account Numbers

Account Type	Account Name	Account Number
Non-Interest Bearing Transaction Account	Alexander's Rego Shopping Center Inc.	1-047-9040-8389

Borrower agrees that, in the event the Account Number shall change or Bank shall need to establish an additional account, Bank shall be entitled to amend this Exhibit A upon written notice to Borrower, and any such amended Exhibit A shall be deemed to be a part of this Agreement.

Powered by Proskauer Commercial Leasing System 1.03 © Proskauer Rose LLP 2002-2004 Patent Pending

**LEASE** 

between

731 OFFICE ONE LLC,

Landlord,

and

CITIBANK, N.A.,

Tenant.

731 Lexington Avenue New York, New York 10022

as of January \_\_\_\_\_, 2005

# TABLE OF CONTENTS

Article/Sec	ction	Page
Article 1 Dl	EMISE, TERM, FIXED RENT	2
1.1	Demise	2
1.2	Commencement Date	2
1.3	Rent Commencement Date	2
1.4	Fixed Rent	2
1.5	Payments of Fixed Rent	2
1.6	Certain Definitions	3
Article 2 ES	SCALATION RENT	4
2.1	Operating Expense Definitions	4
2.2	Calculation of Operating Expenses	8
2.3	Operating Expense Payment	11
2.4	Auditing of Operating Expense Statements	13
2.5	Tax Definitions	14
2.6	Tax Payment	16
2.7	Reductions in Taxes	17
Article 3 US	SE	19
3.1	Permitted Use	19
3.2	Limitations	21
3.3	Bloomberg Lease Restrictions	21
3.4	Rules	22
3.5	Fire Stairs	23
3.6	Risers	24
3.7	Tenant's Signs	24
3.8	Promotional Displays	25
3.9	Antennae Roof Rights	26
3.10	Wireless Internet Service	28
3.11	Telecommunications Provider/Telephone Closets	28
3.12	Access to Beacon Court	29
3.13	Lobby Turnstiles	29
3.14	Fitness Center	30
3.15	Floor Loads	31
Article 4 SE	ERVICES	31
4.1	Certain Definitions	31
4.2	Passenger Elevator Service	31
4.3	Freight Elevator Service	32
4.4	Heat, Ventilation and Air-Conditioning	34
4.5	Cleaning	35
4.6	Water	36
4.7	Building Employee Listing	37
4.8	Building Security	37
4.9	Fire System	38
4.10	Gas	39

i

	4.11	Steam	39
	4.12	Operation of Building Systems	39
	4.13	Supplemental HVAC System	40
	4.14	No Other Services	40
	4.15	Labor Harmony	41
Article	e 5 ELECTR	ICITY	41
	5.1	Capacity	41
	5.2	Electricity for the Building	42
	5.3	Submetering	42
	5.4	Termination of Electric Service	43
Article	e 6 INITIAL	CONDITION OF THE PREMISES	44
	6.1	Condition of Premises	44
Article	e 7 ALTERA	TIONS	45
	7.1	General	45
	7.2	Basic Alterations and Material Alterations	46
	7.3	Approval Process	46
	7.4	Performance of Alterations	48
	7.5	Financial Integrity	49
	7.6	Effect on Building	50
	7.7	Time for Performance of Alterations	50
	7.8	Removal of Alterations and Tenant's Property	51
	7.9	Contractors and Supervision	52
	7.10	Landlord's Expenses	53
	7.11	Pantry	53
	7.12	Window Coverings	53
	7.13	Standby Generator System	53
	7.14	Tenant Fund	55
	7.15	Air-Cooled HVAC Installations	57
	7.16	Initial Alterations Access	58
	7.17	Warranties	58
	7.18	Violations and Landlord Defaults; Tenant's Remedies	58
Article	e 8 REPAIR		59
	8.1	Landlord's Repairs	59
	8.2	Tenant's Repairs	60
	8.3	Certain Limitations	61
	8.4	Overtime	61
Article		; LANDLORD'S CHANGES	62
	9.1	Access	62
	9.2	Landlord's Obligation to Minimize Interference; Repair of Damage	63
	9.3	Reserved Areas	63
	9.4	Ducts, Pipes and Conduits	64
	9.5	Keys	64
	9.6	Landlord's Changes	64
	9.7	Tenant's Access	65
	- • •		

Article 10 UNAV	OIDABLE DELAYS AND INTERRUPTION OF SERVICE	66
10.1	Unavoidable Delays	66
10.2	Interruption of Services	66
10.3	Rent Credit	67
Article 11 REQU	JIREMENTS	68
11.1	Tenant's Obligation to Comply with Requirements	68
11.2	Landlord's Obligation to Comply with Requirements	68
11.3	Tenant's Right to Contest Requirements	69
11.4	Requirements During Last Two Years of Term	69
11.5	Certificate of Occupancy	70
Article 12 QUIE	T ENJOYMENT	71
12.1	Quiet Enjoyment	71
Article 13 SUBO	ORDINATION	71
13.1	Subordination	71
13.2	Terms of Nondisturbance Agreements	72
13.3	Attornment	74
13.4	Amendments to this Lease	74
13.5	Tenant's Estoppel Certificate	75
13.6	Landlord's Estoppel Certificate	75
13.7	Rights to Cure Landlord's Default	75
13.8	Zoning Lot Merger Agreement	76
13.9	Existing Mortgages and Existing Superior Leases	76
13.10	Condominium	76
13.11	Multiple Owners of the Office Unit	77
Article 14 INSUI		78
14.1	Tenant's Insurance	78
14.2	Landlord's Insurance	79
14.3	Mutual Waiver of Subrogation	80
14.4	Evidence of Insurance	81
14.5	No Concurrent Insurance	81
14.6	Tenant's Obligation to Comply with Landlord's Fire and Casualty	
	Insurance	82
14.7	Tenant's Insurance Proceeds	82
Article 15 CASU	JALTY	84
15.1	Notice	84
15.2	Landlord's Restoration Obligations	84
15.3	Tenant's Restoration Obligations	85
15.4	Rent Abatement	85
15.5	Landlord's Termination Right	86
15.6	Tenant's Termination Right	87
15.7	Termination Rights at End of Term	88
15.8	No Other Termination Rights	89
Article 16 COND	DEMNATION	89
16.1	Effect of Condemnation	89

	16.2	Condemnation Award	90
	16.3	Temporary Taking	91
Article 17	ASSIGNMENT	Γ AND SUBLETTING	91
	17.1	General Limitations	91
	17.2	Landlord's Expenses	93
	17.3	Recapture Procedure	93
	17.4	Certain Transfer Rights	95
	17.5	Conditional Approval	98
	17.6	Deemed Approval	99
	17.7	Transfer Taxes	99
	17.8	Transfer Profit	99
	17.9	Permitted Transfers	100
	17.10	Special Occupant	102
	17.11	Recognition Agreements	103
Article 18	TENANT'S RI	GHT OF FIRST OFFER TO LEASE	106
	18.1	Right of First Offer	106
	18.2	Option Notice	107
	18.3	Option Procedure	107
	18.4	Certain Limitations	109
	18.5	Lease Provisions Apply	110
	18.6	Delivery	111
	18.7	Special Rules	111
Article 19	RENEWAL		112
	19.1	Renewal Option	112
	19.2	Partial Renewal Space	113
	19.3	Lease Provisions Apply	113
Article 20	FAIR MARKE	T RENT	114
	20.1	Certain Definitions	114
	20.2	Fair Market Rent Assumptions	115
	20.3	Fair Market Procedure	115
Article 21	DEFAULT		117
	21.1	Events of Default	117
	21.2	Termination	118
Article 22	TENANT'S IN	SOLVENCY	118
	22.1	Assignments pursuant to the Bankruptcy Code	118
	22.2	Replacement Lease	119
	22.3	Insolvency Events	120
	22.4	Effect of Stay	121
	22.5	Rental for Bankruptcy Purposes	121
Article 23	REMEDIES A	ND DAMAGES	122
	23.1	Certain Remedies	122
	23.2	No Redemption	122
	23.3	Calculation of Damages	123

Article 24 LANDLORI	D'S EXPENSES AND LATE CHARGES	124
24.1	Landlord's Costs After Event of Default	124
24.2	Interest on Late Payments	124
Article 25 END OF TE	ERM	125
25.1	End of Term	125
25.2	Holdover	125
Article 26 NO WAIVE	IR.	125
26.1	No Surrender	125
26.2	No Waiver by Landlord	126
26.3	No Waiver by Tenant	126
Article 27 JURISDICT	TION	127
27.1	Governing Law	127
27.2	Submission to Jurisdiction	127
27.3	Waiver of Trial by Jury	127
Article 28 NOTICES		127
28.1	Addresses; Manner of Delivery	127
Article 29 BROKERA	GE	129
29.1	Broker	129
Article 30 INDEMNIT	TY	129
30.1	Tenant's Indemnification of the Landlord Indemnitees	129
30.2	Landlord's Indemnification of the Tenant Indemnitees	130
30.3	Indemnification Procedure	131
Article 31 LANDLORI	D'S CONSENTS; ARBITRATION	133
31.1	Certain Limitations	133
31.2	Expedited Arbitration; Tenant's Remedies	133
Article 32 ADDITION	AL PROVISIONS	135
32.1	Tenant's Property Delivered to Building Employees	135
32.2	Not Binding Until Execution	135
32.3	No Third Party Beneficiaries	135
32.4	Extent of Landlord's Liability	135
32.5	Survival	136
32.6	Recording	136
32.7	Entire Agreement	136
32.8	Exhibits	136
32.9	Gender; Plural	136
32.10	Divisibility	136
32.11	Vault Space	136
32.12	Adjacent Excavation	137
32.13	Captions	137
32.14	Parties Bound	137
32.15	Authority	137
32.16	Rent Control	138
32.17	Consequential Damages	138

v

# Index of Defined Terms

Term	Page
29 <sup>th</sup> Floor Option Space	106
32 B-J Hourly Rate	33
Abatement Event	67
Abatement Percentage	67
Access Agreement	29
Affiliate	3
Aggregate Rental	103
Alterations	45
Alterations Notice	46
Antennae	26
Antennae Availability Notice	26
Antennae Commencement Date	27
Antennae Mast	26
Antennae Mast Mechanical Room	24
Antennae Notice	26
Antennae Response Notice	26
Antennae Site	26
Applicable Area	114
Applicable Date	115
Applicable Option Space	107
Applicable Rate	3
Applicable Recapture Period	94
Applicable Terms	103
Appraiser	116
Assessed Valuation	14
Available Antennae Space	26
Average Cost per Kilowatt Hour	42
Bankruptcy Code	118
Base Electrical Capacity	41
Base Operating Expense Year	4
Base Operating Expenses	4
Base Rate	3
Base Rental Amount	115
Base Tax Floor	14
Base Tax Year	14
Base Taxes	14
Basic Alteration	46
Basic Alterations Threshold	46
Basic Sublease Provisions	98
Bloomberg	22
Bloomberg Lease	22
Bloomberg Primary Competitor	21
Brokers	129

1

Building

Building Change	50
Building Hours	31
Building Systems	31
Business Days	3
By-Laws	71
Casualty Statement	87
Citigroup	79
Claim	131
Commencement Date	2
Comparable Space	96
Compliance Challenge	69
Conditional Consent Notice	98
Condominium	1
Condominium Board	71
Condominium Declaration	1
Condominium Documents	71
Condominium Withdrawal	77
Construction Union Personnel	33
Consumer Price Index	3
Control	4
Core Business	142
Day Tank	53
Deficiency	123
Disbursement Request	56
Electricity Additional Rent	42
Escalation Rent	4
Escalation Rent Per Square Foot	68
Event of Default	117
Excluded Amounts	14
Exclusive Elevator	32
Expedited Arbitration Proceeding	134
Expiration Date	2
Exterior Sign	25
Fair Market Rent	114
Final Transaction Documents	98
Fire Stairs	23
Fitness Center	30
Fixed Expiration Date	2
Fixed Rent	2
Governmental Authority	68
Holidays	4
HVAC	31
HVAC Systems	31
Impeding Building Violation	58
Impeding Premises Violation	59

Indemnitee

Indemnitor	133
Initial Alterations	45
Initial Freight Car	33
Insolvency Events	123
Institutional Lender	83
Interference Area	67
Interference Day	67
Landlord	1
Landlord Indemnitees	130
Landlord Liability Claim	130
Landlord Owners	77
Landlord's Award	90
Landlord's Determination	115
Landlord's Liability Policy	79
Landlord's Property Policy	79
Landlord's Work	44
Lessor	72
License Fee	27
Life Safety System Alteration	52
Lobby Sign	24
Long-Term Sublease	94
Mailroom	19
Major Sublease	104
Major Sublease Guarantor	105
Major Sublease Unit	105
Mast Manager	27
Material Alteration	46
Maximum Disbursement Amount	56
Minimum Demise Requirement	4
Minimum Occupancy Requirement	4
Mortgage	72
Mortgagee	72
Nondisturbance Agreement	72
Occupancy Agreement	92
Office Unit	1
Office Unit One	1
Office Unit One Owner	1
Office Units	1
Operating Expense Payment	{
Operating Expense Statement	
Operating Expense Year	
Operating Expenses	
Option	107
Option Cutoff Date	109

Option Notice

Option Response Notice

Option Space	106
Option Space Commencement Date	111
Option Term	107
Overtime Periods	31
Partial Renewal Space	113
Permitted Antennae	26
Permitted Party	92
Person	4
Post-Possession Premises	44
Predecessor Tenant	119
Premises	113
Proceeds	10
Proposed Transfer Terms	98
Prospective Operating Statement	11
Qualified Alteration	52
Qualified Tax Benefits	15
Real Property	15
Recapture Procedure	93
Recapture Space	93
Recapture Termination	94
Recapture Termination Notice	94
Recognition Agreement	105
Recognition Effective Date	105
Removed Space	113
Renewal Notice	113
Renewal Option	112
Renewal Premises	113
Renewal Term	112
Rent Commencement Date	2
Rent Notice	115
Rent Per Square Foot	67
Rentable Area	4
Rental	2
Rental Value	114
Requirements	68
Reserved Areas	63
Residential Unit	1
Risers	24
Rules	22
Scheduled Option Space Commencement Date	107
Second Bite Date	88
Second Bite Notice	88
Secondary Freight Car	33
Secure Areas	64

132

Self Insurance Programs

Settlement

Soft Costs	55
Special Occupant	102
Specialty Alterations	45
Standby Generator System	53
Steam Additional Rent	39
Storage Space	19
Substantial Completion	45
Successor	72
Successor Limitation Items	74
Superior Lease	72
Tax Payment	15
Tax Statement	15
Tax Year	16
Taxes	15
Tenant	1
Tenant Fund	55
Tenant Indemnitees	130
Tenant Liability Claim	129
Tenant Obligor	123
Tenant's Determination	115
Tenant's Liability Policy	78
Tenant's Operating Expense Share	8
Tenant's Property	46
Tenant's Property Policy	78
Tenant's Signs	25
Tenant's Tax Share	16
Term	2
Terrorism Insurance	79
Transfer	91
Transfer Date	93
Transfer Expenses	94
Transfer Inflow	100
Transfer Notice	93
Transfer Outflow	100
Transfer Profit	99
Transferee	94
Transferor	92
Unamortized Alterations Cost	83
Unamortized Tenant Fund Amount	84
Unavoidable Delay	66
Unit One Option Space	107
Usable Area	4
Utility Company	43

Work Access

# S

	EXHIBITS
Exhibit "A"	Premises
Exhibit "3.3-1"	Bloomberg Primary Competitors
Exhibit "3.3-2"	Modification of List of Bloomberg Primary Competitors
Exhibit "3.4-1"	Rules
Exhibit "3.4-2"	Alterations Rules
Exhibit "3.6"	Risers
Exhibit "3.7-1"	Lobby Sign
Exhibit "3.7-2"	Exterior Sign
Exhibit "3.11"	Telephone/Electric Closets
Exhibit "3.12"	Access Agreement
Exhibit "3.13"	Lobby Turnstiles Location
Exhibit "3.14"	Fitness Center Plans and Specifications
Exhibit "3.15"	Floor Loads
Exhibit "4.2-1"	Passenger Elevator Locations
Exhibit "4.2-2"	Passenger Elevator Specifications
Exhibit "4.3"	Freight Elevator
Exhibit "4.4"	HVAC Specifications
Exhibit "4.5"	Cleaning Specifications
Exhibit "4.9"	Class "E" System Connection Points
Exhibit "6.1(B)"	Landlord's Work

Exhibit "7.3(F)" Alterations Approved in Concept Standby Generator System Specifications Exhibit "7.13(A)" Exhibit "7.15" Location of Supplemental HVAC System

Distribution Portion of Buildings Systems in Premises Exhibit "8.2"

Exhibit "13.1" Condominium Non-Disturbance Agreement Exhibit "13.9" Mortgages and Superior Leases
Exhibit "17.11" Recognition Agreement
Exhibit "18.1" Option Space
Exhibit "19.2" Partial Renewal Space

xiii

THIS LEASE, dated as of the day of January, 2005, by and between 731 OFFICE ONE LLC, a Delaware limited liability company, having an
address c/o Alexander's Inc., 888 Seventh Avenue, New York, New York 10019, as landlord, and CITIBANK, N.A., a national banking association, having an
address at One Court Square, 8th Floor, Zone 1, Long Island City, New York 11120, as tenant (the Person that holds the interest of the landlord hereunder at any
particular time being referred to herein as "Landlord"; subject to Section 17.1(G) hereof, the Person that holds the interest of the tenant hereunder at any particular
time being referred to herein as " <u>Tenant</u> ").

## WITNESSETH:

WHEREAS, pursuant to a Declaration of Beacon Court Condominium, dated as of December 4, 2003, the ownership of the fee interests in the land and building (the "Building"; such land and building being referred to herein, collectively, as the "Real Property") described therein was submitted to a condominium form of ownership in accordance with Article 9.1(B) of the New York Real Property Law (the Beacon Court Condominium being referred to herein as the "Condominium"; such declaration, as may be amended from time to time, being referred to herein as the "Condominium Declaration");

WHEREAS, the Building is known by the street address of 731 Lexington Avenue, New York, New York;

WHEREAS, the Condominium is comprised of a retail unit, various residential units (collectively, the "Residential Unit"), and two (2) office units (collectively, the "Office Units");

WHEREAS, the Office Units are comprised of (i) an office unit (the "Office Unit One"), which includes, together with certain other space in the Building, the seventeenth (17<sup>th</sup>) through the twentieth (20<sup>th</sup>) floors of the Building (the owner, from time to time, of the Office Unit One being referred to herein as the "Office Unit One Owner"), and (ii) an additional office unit (the "Office Unit"), which includes, together with certain other space in the Building, the twenty-first (21<sup>st</sup>) through the twenty-ninth (29<sup>th</sup>) floors of the Building; and

WHEREAS, Landlord wishes to demise and let unto Tenant, and Tenant wishes to hire and take from Landlord, on the terms and subject to the conditions set forth herein, the premises as shown on Exhibit "A" attached hereto and made a part hereof on lower level 3, the twenty-first (21st), twenty-second (22nd), twenty-third (23rd), twenty-fourth (24th), twenty-fifth (25th), twenty-sixth (26th), twenty-seventh (27th) and twenty-eighth (28th) floors of the Building (such premises being referred to herein collectively as the "Premises").

NOW, THEREFORE, in consideration of the premises, and other good and valuable consideration, the mutual receipt and legal sufficiency of which the parties hereto hereby acknowledge, Landlord and Tenant hereby agree as follows:

# Article 1 DEMISE, TERM, FIXED RENT

#### 1.1. Demise.

Subject to the terms hereof, Landlord hereby leases to Tenant and Tenant hereby hires from Landlord the Premises for the term to commence on the Commencement Date and to end on the day immediately preceding the date that is ten (10) years after the Rent Commencement Date (the "Fixed Expiration Date"; the Fixed Expiration Date, or such earlier or later date that the term of this Lease expires or otherwise terminates pursuant to the terms hereof (including, without limitation, Article 19 hereof) or pursuant to applicable Requirements, being referred to herein as the "Expiration Date"; the term commencing on the Commencement Date and ending on the Expiration Date being referred to herein as the "Term").

#### 1.2. Commencement Date.

- (A) Landlord shall deliver to Tenant vacant and exclusive possession of the Premises on the date hereof (the "Commencement Date").
- (B) The term "Rental" shall mean, collectively, the Fixed Rent, the Escalation Rent and the additional rent payable by Tenant to Landlord hereunder.

#### 1.3. Rent Commencement Date.

The term "Rent Commencement Date" shall mean December 15, 2005.

#### 1.4. Fixed Rent.

The annual fixed rent for the Premises (the annual fixed rent payable hereunder for the Premises at any particular time being referred to herein as the "Fixed Rent") shall be:

- (1) Thirteen Million Twenty-Four Thousand Dollars and No Cents (\$13,024,000.00) (\$1,085,333.33 per month) for the period commencing on the Rent Commencement Date and ending on the day immediately preceding the date that is five (5) years after the Rent Commencement Date; and
- (2) Thirteen Million Nine Hundred Four Thousand Dollars and No Cents (\$13,904,000.00) (\$1,158,666.67 per month) for the period commencing on the date that is five (5) years after the Rent Commencement Date and ending on the Fixed Expiration Date.

## 1.5. <u>Payments of Fixed Rent.</u>

(A) Tenant shall pay the Fixed Rent in lawful money of the United States of America that is legal tender in payment of all debts and dues, public and private, at the time of payment, in equal monthly installments, in advance, on the first (1st) day of each calendar month during the Term commencing on the Rent Commencement Date, at the office of Landlord or such other place as Landlord may designate from time to time on at least thirty (30) days of

advance notice to Tenant, without any set-off, offset, abatement or deduction whatsoever (except to the extent otherwise expressly set forth herein).

- (B) Tenant shall pay the Fixed Rent when due only by electronic funds transfer to an account designated from time to time by Landlord on at least thirty (30) days of advance notice to Tenant (and shall not be permitted to pay the Fixed Rent by check).
- (C) If the Rent Commencement Date is not the first (1st) day of a calendar month, then (x) the Fixed Rent due hereunder for the calendar month during which the Rent Commencement Date occurs shall be adjusted appropriately based on the number of days in such calendar month, and (y) Tenant shall pay to Landlord such amount (adjusted as aforesaid for such calendar month) on the Rent Commencement Date. If the Expiration Date is not the last day of a calendar month, then the Fixed Rent due hereunder for the calendar month during which the Expiration Date occurs shall be adjusted appropriately based on the number of days in such calendar month.

#### 1.6. <u>Certain Definitions.</u>

- (A) The term "Affiliate" shall mean a Person that (1) Controls, (2) is under the Control of, or (3) is under common Control with, the Person in question.
- (B) The term "Applicable Rate" shall mean, at any particular time, the lesser of (x) three hundred (300) basis points above the Base Rate at such time, and (y) the maximum rate permitted by applicable law at such time.
- (C) The term "Base Rate" shall mean the rate of interest announced publicly from time to time by Citibank, N.A., or its successor, as its "prime lending rate" (or such other term as may be used by Citibank, N.A. (or its successor), from time to time, for the rate presently referred to as its "prime lending rate").
  - (D) The term "Business Days" shall mean all days, excluding Saturdays, Sundays and Holidays.
- (E) The term "Consumer Price Index" shall mean the Consumer Price Index for All Urban Consumers published by the Bureau of Labor Statistics of the United States Department of Labor, All Items (1982-84 = 100), seasonally adjusted, for the most specific area that includes the location of the Building (currently New York Northern New Jersey Long Island, N.Y. NJ CT PA), or any successor index thereto. If the Consumer Price Index is converted to a different standard reference base or otherwise revised, then the determination of adjustments provided for herein shall be made with the use of such conversion factor, formula or table for converting the Consumer Price Index as may be published by the Bureau of Labor Statistics or, if said Bureau does not publish such conversion factor, formula or table, then with the use of such conversion factor, formula or table as may be published by Prentice-Hall, Inc. or any other nationally recognized publisher of similar statistical information. If the Consumer Price Index ceases to be published, and there is no successor thereto, then Landlord and Tenant shall use diligent efforts, in good faith, to agree upon a substitute index for the Consumer Price Index. Either party shall have the right to submit the issue of the designation of such substitute index to an Expedited Arbitration Proceeding.

- (F) The term "Control" shall mean the direct or indirect ownership of more than thirty percent (30%) of the outstanding voting stock of a corporation or other majority equity interest if not a corporation together with the possession of power to direct or cause the direction of the management and policy of such corporation or other entity, whether through the ownership of voting securities, by statute or by contract (it being agreed that if a Person is the largest shareholder of a public corporation and owns or has voting control over not less than twenty-five (25%) of all of the then voting stock of such corporation, then such Person shall be deemed to "Control" such corporation).
  - (G) The term "Escalation Rent" shall mean the Rental payable to Landlord under Article 2 hereof.
- (H) The term "Holidays" shall mean all days (other than Saturdays and Sundays) observed as legal holidays by the New York Stock Exchange.
- (I) The term "Minimum Demise Requirement" shall mean the requirement that this Lease demises at least eighty-eight thousand (88,000) square feet of Rentable Area.
- (J) The term "Minimum Occupancy Requirement" shall mean the requirement that Tenant (or an Affiliate of Tenant) occupies at least eighty-eight thousand (88,000) square feet of Rentable Area that is demised by this Lease for the conduct of business.
- (K) The term "Person" shall mean any natural person or persons or any legal form of association, including, without limitation, a partnership, a limited partnership, a corporation, and a limited liability company.
- (L) The term "Rentable Area" shall mean, with respect to a particular floor area, the rentable square foot area thereof, as determined in accordance with the standards that the parties used to calculate that the rentable square foot area of the Premises is one hundred seventy-six thousand (176,000) square feet in the aggregate, comprised of twenty-two thousand (22,000) square feet with respect to each full floor of the Premises.
- (M) The term "<u>Usable Area</u>" shall mean, with respect to a particular floor area, the usable area thereof, as determined in accordance with The Recommended Method of Floor Measurement of Office Buildings, Effective January 1, 1987, as published by The Real Estate Board of New York, Inc.

## Article 2 ESCALATION RENT

#### 2.1. Operating Expense Definitions.

- (A) The term "Base Operating Expenses" shall mean the Operating Expenses for the Base Operating Expense Year.
- (B) The term "Base Operating Expense Year" shall mean the 2005 calendar year.

(C) The term "Operating Expenses" shall mean, subject to the terms of this Section 2.1 and to Section 2.2(F) hereof, the expenses paid or
incurred by or on behalf of Landlord in insuring, maintaining, repairing, managing and operating the Office Unit (and employing personnel therefor) as
determined in accordance with generally accepted accounting principles, consistently applied. Operating Expenses shall include, without limitation, the common
charges assessed by the Condominium Board and that are fairly allocated to the Office Unit (but only to the extent that the costs and expenses that comprise such
common charges would not have been otherwise excluded from Operating Expenses in accordance with the terms hereof). Operating Expenses shall exclude:

- (1) Taxes,
- (2) Excluded Amounts,
- (3) subject to Section 2.2(F) hereof, payments of interest or principal in respect of Landlord's debt (including, without limitation, any debt that is secured by Mortgages),
- (4) expenses that relate to leasing space in the Office Unit (including, without limitation, the cost of tenant improvements, the cost of rent concessions, advertising expenses, leasing commissions and the cost of lease buy-outs),
  - (5) expenses that Landlord incurs in selling, purchasing or refinancing the Office Unit,
- (6) the cost of any repairs, replacements or improvements to the Office Unit that are required to be capitalized by generally accepted accounting principles (except as otherwise provided in Section 2.2(F) hereof),
  - (7) depreciation expense (subject, however, to Section 2.2(F) hereof),
- (8) the cost of electricity that is furnished to the portions of the Office Unit that Landlord has leased or that Landlord is offering for lease (it being understood that Operating Expenses may include the cost of electricity that is required to operate the Building Systems),
  - (9) salaries and the cost of benefits in either case for personnel above the grade of building manager,
- (10) rent paid or payable under Superior Leases (other than in the nature of additional rent consisting of taxes or operating expenses, but only to the extent that such taxes or operating expenses would not have been otherwise excluded from Taxes or Operating Expenses in accordance with the terms hereof),
- (11) subject to Section 2.2 hereof, any expense for which Landlord is otherwise compensated (other than by virtue of other tenants in the Office Unit making payments to Landlord for Operating Expenses as escalation rental) or for which Landlord receives insurance or condemnation proceeds,

- (12) the cost of providing any level of service that exceeds the level of service that Landlord furnishes to Tenant hereunder,
- (13) the cost of remedying defects in the initial construction of the Building,
- (14) legal fees and disbursements that are paid or incurred in connection with the negotiation of, or disputes arising out of, any lease for space in the Office Unit,
- (15) costs that Landlord incurs in restoring the Building after the occurrence of a fire or other casualty or after a partial condemnation thereof,
  - (16) advertising and promotional costs that are paid or incurred for the Building,
- (17) any fee or expenditure that is paid or payable to any Affiliate of Landlord to the extent that such fee or expenditure exceeds the amount that would be reasonably expected to be paid in the absence of such relationship, including, without limitation, management fees paid (or treated as paid) to any Affiliate of Landlord,
  - (18) interest, penalties and late charges that in either case are paid or incurred as a result of late payments made by Landlord,
  - (19) fees, dues and other contributions paid by or on behalf of Landlord or Landlord's Affiliates to civic or other real estate organizations,
- (20) the cost of any alterations, additions, changes, replacements and improvements that are made solely in order to prepare space for occupancy by a new tenant (including Tenant) and any contribution or concession by Landlord to such tenant in connection therewith,
- (21) the cost of heating, air-conditioning and ventilation during overtime periods for any tenants or occupants of the Unit or the Building (including Tenant),
- (22) the cost of operating and providing freight elevator service and loading dock security during overtime periods for any tenants or occupants of the Unit or the Building (including Tenant),
- (23) costs of acquiring, leasing, insuring, restoring, removing or replacing sculptures and paintings that are deemed to be "fine art" except to the extent required by Requirements (rather than decorative art work customarily found in buildings comparable to the Building),
- (24) payments of any amounts to any person (including Tenant) seeking recovery for negligence or other torts committed by Landlord or the Condominium Board,
  - (25) costs relating to withdrawal liability or unfunded pension liability under the Multi-Employer Pension Plan Act or similar Requirement,

- (26) the cost of installing, operating and maintaining any specialty facility, such as an observatory, lodging, broadcasting facilities, luncheon club, athletic or recreational club, child care facility, auditorium, cafeteria or dining facility, conference center or similar facilities,
- (27) any compensation paid to clerks, attendants or other persons in commercial concessions or any parking facility located in the Office Unit or the Building operated by Landlord or the Condominium Board,
- (28) costs incurred by Landlord or the Condominium Board which result from Landlord's or any tenant's breach of a lease (including this Lease) or Landlord's or the Condominium Board's tortious or negligent conduct,
- (29) the cost of operating the entity that constitutes Landlord or the Condominium Board (as opposed to the costs of operating and maintaining the Office Unit and the Building), including accounting fees, legal fees and any costs incurred by Landlord or the Condominium Board in disputes with (a) Building employees, (b) Persons employed by Landlord or the Condominium Board that are not engaged in Building operations, or (c) any Superior Lessor or Mortgagee,
  - (30) charitable and political contributions,
- (31) costs to comply with any violation of Requirements noted against the Real Property as of the date hereof or insurance requirements in effect on the date hereof, or to correct any condition that would constitute a Landlord misrepresentation under this Lease,
  - (32) fines or penalties (except for fines or penalties relating to minor, common infractions, such as sidewalk violations),
- (33) costs and expenses incurred by Landlord in connection with any obligation of Landlord to indemnify any tenant or occupant of the Office Unit or the Building (including Tenant) pursuant to its lease or otherwise,
- (34) costs incurred with respect to removal or encapsulation or other treatment of hazardous materials (unless such hazardous materials were introduced to the Building by Tenant or anyone claiming by through or under Tenant), but excepting costs of normal and customary testing and monitoring,
- (35) lease payments for rented equipment, the cost of which equipment would constitute a capital expenditure if the equipment were purchased (except to the extent otherwise expressly provided herein),
- (36) common charges of the Condominium, if and to the extent that such common expenses duplicate or are in excess of amounts otherwise includable in Operating Expenses in accordance with this Lease (i.e., the determination of whether and to what extent an item of expense is includable as an Operating Expense in accordance with the terms and conditions of this Lease shall be made without regard to whether the amount of such item is payable by Landlord as part of common charges or directly to another Person), and

(37) costs expressly excluded from Operating Expenses by any other provision of this Lease.

No item of expense shall be counted more than once either as an inclusion in Operating Expenses or as an exclusion from Operating Expenses (including, without limitation, by reason of any potential duplication of functions, if any, performed by employees of Landlord and employees of the Condominium). If (x) pursuant to Section 14.2(A) hereof, Landlord obtains Terrorism Insurance as part of Landlord's Property Policy in any Operating Expense Year subsequent to the Base Operating Expense Year, and (y) Operating Expenses for the Base Operating Expense Year did not include a charge for Terrorism Insurance for the entire Base Operating Expense Year, then Landlord shall have the right to include the costs of obtaining such Terrorism Insurance in any Operating Expense Year subsequent to the Base Operating Expense Year only if, for purposes of calculating the Operating Payment for such Operating Expense Year, the Base Year Operating Expenses are increased to include a charge for Terrorism Insurance (which charge shall be reasonably determined to be the charge that Landlord would have paid for such Terrorism Insurance if Landlord had obtained such Terrorism Insurance during the Base Operating Expense Year) for the entire Base Operating Expense Year.

- (D) The term "Operating Expense Payment" shall mean, with respect to any Operating Expense Year, the product obtained by multiplying (i) the excess (if any) of (A) the Operating Expenses for such Operating Expense Year, over (B) the Base Operating Expenses, by (ii) Tenant's Operating Expense Share.
- (E) The term "Operating Expense Statement" shall mean a statement that shows the Operating Expense Payment for a particular Operating Expense Year.
  - (F) The term "Operating Expense Year" shall mean the Base Operating Expense Year and each subsequent calendar year.
- (G) The term "Tenant's Operating Expense Share" shall mean, subject to the terms hereof, ninety-two and five hundred twenty-seven one-thousandths percent (92.527%).

# 2.2. <u>Calculation of Operating Expenses.</u>

(A) If, during any Operating Expense Year, (i) any rentable space in the Office Units shall be vacant or unoccupied, and/or (ii) the tenant or occupant of any space in the Office Units undertook to perform work or services therein in lieu of having Landlord, the Office Unit One Owner, or the Condominium Board, as the case may be, perform the same and the cost thereof would otherwise have been included in Operating Expenses, then, in any such event, the Operating Expenses for such Operating Expense Year which would vary with the percentages of occupancy of the Office Units or the percentages of tenants or occupants for which work or services are performed by Landlord, the Office Unit One Owner, or the Condominium Board, as the case may be, shall be reasonably adjusted to reflect as closely as possible the variable Operating Expenses that actually would have been incurred if such space in the Office Unit had been occupied or if Landlord, the Office Unit One Owner, or the Condominium Board, as the case may be, had performed such work or services (for example, if, during an Operating Expense Year, fifty percent (50%) of the rentable space in the Office Unit was vacant and as a result

thereof Landlord's actual cost of providing base building cleaning to rentable space in the Office Unit was reduced by forty percent (40%) rather than fifty percent (50%) due to economies of scale or for any other reason, then the line item for base building cleaning provided to the rentable space in the Office Unit for such Operating Expense Year would be increased by the same forty (40%) percent, rather than fifty (50%) percent, to reflect as closely as possible the variable Operating Expenses that actually would have been incurred if such space had been occupied for the entire Operating Expense Year). The provisions of this Section 2.2(A) shall not be applicable to the Base Operating Expense Year except as otherwise provided in Section 2.2(B) hereof.

- (B) Landlord and Tenant acknowledge that the Office Unit will be unoccupied for a substantial portion of the Base Operating Expense Year and that the Building will not be fully occupied for a substantial portion of the Base Operating Expense Year. Accordingly, Landlord and Tenant agree that the Base Operating Expenses shall be recalculated promptly following the expiration of the Operating Expense Year that constitutes calendar year 2006 to an amount equal to the actual Operating Expenses incurred by Landlord for the Operating Expense Year that constitutes calendar year 2006 (as same may be adjusted in accordance with the provisions of Section 2.2(A) hereof), reduced by any identifiable percentage increases in such Operating Expenses, on an item-by-item basis, from the Base Year Operating Expenses to the Operating Expense Year that constitutes calendar year 2006 (for example, if Landlord pays \$105.00 for a particular Operating Expense in the Operating Expense Year that constitutes calendar year 2006 and Landlord's cost for such particular Operating Expense increased by five percent (5%) from the Base Operating Expense Year to the Operating Expense Year that constitutes calendar year 2006, then the amount included in Base Operating Expenses for such particular Operating Expense shall be \$100.00). Tenant shall have the right to audit and challenge the calculation of Base Operating Expenses provided for in this Section 2.2(B) hereof in accordance with the provisions of Section 2.4 hereof. Tenant acknowledges that Landlord may require Tenant to pay Operating Expenses for the Operating Expense Year that constitutes calendar year 2006 pursuant to a Prospective Operating Statement, in which event the Base Operating Expenses that are a part of the calculation of the Operating Payment set forth therein shall be calculated pursuant to Section 2.2(A) hereof, subject to adjustment following the expiration of the Operating Expense Year that constitutes calendar year 2006 in accordance with the terms of this Section 2.2(B).
- (C) Landlord and Tenant acknowledge that the common charges assessed by the Condominium Board are allocated to the Office Unit (and, accordingly, included in Operating Expenses) based upon methods of allocation set forth in the Condominium Documents, and that such allocations may change as the Building develops an operating history to reflect a more fair allocation of such common charges. Accordingly, Landlord and Tenant each agree that if, in calendar year 2006 or calendar year 2007, (i) an allocation (or method of allocation) of a particular common charge changes from the particular allocation (or method of allocation) used in the Base Operating Expense Year (i.e., calendar year 2005), and (ii) such allocation (or method of allocation) impacts Operating Expenses (including Base Operating Expenses), then Base Operating Expenses shall be recalculated to reflect such change in allocation (or method of allocation) as if such allocation (or method of allocation) had originally been used to calculate Base Operating Expenses. Landlord and Tenant each acknowledge that pursuant to Section 2.1(C) hereof, Operating Expenses may include common charges assessed

against the Office Unit only to the extent that they are fairly allocated to the Unit and would not have been otherwise excluded from Operating Expenses in accordance with the terms of this Article 2. Accordingly, notwithstanding the fact that Landlord may be bound by certain allocations and methods of allocation pursuant to the Condominium Documents, Tenant shall have the right to audit and challenge the inclusion of common charges assessed by the Condominium Board in any Operating Expense Statement in accordance with the provisions of Section 2.4 hereof. Tenant shall also have the right to audit and challenge the calculation of Base Operating Expenses provided for in this Section 2.2(C) hereof in accordance with the provisions of Section 2.4 hereof.

- (D) If (i) Landlord includes an expense in Operating Expenses for a particular Operating Expense Year, and (ii) Landlord receives insurance proceeds, warranty proceeds or other proceeds (collectively, "Proceeds") as reimbursement for such expense in a subsequent Operating Expense Year, then Landlord shall deduct such Proceeds from Operating Expenses in the Operating Expense Year during which Landlord receives such Proceeds (without making a corresponding reduction in the Operating Expenses for the Operating Expense Year during which Landlord incurred such expense, except if Landlord incurred such expense in the last full Operating Expense Year that occurs during the Term or the last partial Operating Expense Year that occurs during the Term) (it being understood that such deduction shall not exceed the amount of the expense that Landlord previously included in Operating Expenses); provided, however, that if Landlord receives such Proceeds in any Operating Expense Year prior to the Operating Expense Year that constitutes calendar year 2008 for an expense that Landlord included in Base Operating Expenses, then Landlord shall have the right to deduct such Proceeds (in an amount that does not exceed the applicable expense) from Base Operating Expenses (rather than deducting such Proceeds from Operating Expenses for the Operating Expense Year during which Landlord receives such Proceeds), in which case Tenant shall pay to Landlord the additional Operating Expense Payment for each preceding Operating Expense Year that derives from such reduction in the Base Operating Expenses, on or prior to the thirtieth (30th) day after the date that Landlord gives Tenant an invoice therefore, together with reasonable supporting documentation.
- (E) Landlord shall have the right to include in Operating Expenses, for the electricity supplied to the Building Systems and other common elements of the Building that in either case benefit the Office Unit, an amount equal to the product obtained by multiplying (i) the Average Cost per Kilowatt Hour, by (ii) the number of kilowatt hours of electricity used by the Building Systems and the other common elements of the Building that in either case benefit the Office Unit for the applicable period (as registered on a submeter or submeters, or, at Landlord's option, as determined by a survey prepared by an independent and reputable electrical consultant) (it being understood that such number of kilowatt hours as described in clause (ii) above shall not include the number of kilowatt hours that are attributable to the operation of the Building Systems to the extent that Tenant (or other tenants in the Building) make separate payment to Landlord therefor), including, without limitation, separate payments for the delivery of condenser water, overtime HVAC and overtime freight elevator service.
- (F) If (i) Landlord makes an improvement to the Office Unit or a replacement of equipment at the Office Unit in either case in connection with the maintenance, repair, management or operation thereof, (ii) generally accepted accounting principles require Landlord

to capitalize the cost of such improvement or such replacement, and (iii) such improvement or replacement is made (a) to comply with a Requirement that is first enacted after the date hereof or that the applicable Governmental Authority enforces in a different manner (by more than a *de minimis* extent) after the date hereof, if such Requirement was enacted prior to the date hereof, or (b) for the purpose of saving or reducing Operating Expenses (such as, for example, an improvement that reduces labor costs), then Landlord shall include in Operating Expenses for each Operating Expense Year the amount that amortizes the cost of such improvement or such replacement, together with interest thereon calculated at the Base Rate, in equal annual installments over the useful life of such improvement or such equipment as determined in accordance with generally accepted accounting principles (until the cost of such improvement or such equipment is amortized fully); provided, however, that for any such improvement or replacement that Landlord makes for the purpose of saving or reducing Operating Expenses (and that Landlord does not make to comply with a Requirement as aforesaid), the aforesaid amount that Landlord includes in Operating Expenses for any particular Operating Expense Year shall not exceed the amount of the reduction in other Operating Expenses for such Operating Expense Year that derives from such improvement or such replacement.

# 2.3. <u>Operating Expense Payment.</u>

- (A) Landlord may render to Tenant at any time from and after the Rent Commencement Date an Operating Expense Statement for an Operating Expense Year. Landlord's failure to render an Operating Statement to Tenant during or with respect to any Operating Expense Year shall not prejudice Landlord's right to render an Operating Statement during or with respect to any other Operating Expense Year, and shall not limit or impair Tenant's obligation to pay the Operating Payment for any Operating Expense Year, except that if Landlord renders an Operating Statement to Tenant with respect to any Operating Expense Year more than two (2) years after the last day thereof, then Tenant shall not be obligated to make an Operating Payment in respect of such Operating Expense Year.
- (B) No later than thirty (30) days after the date on which Landlord delivers to Tenant an Operating Statement, Tenant shall pay to Landlord the Operating Payment shown thereon to be due for the Operating Expense Year covered thereby.
- (C) Subject to the terms of this Section 2.3, Landlord shall have the right, from time to time, to give an Operating Statement to Tenant, on a prospective basis, for an Operating Expense Year, based on Landlord's reasonable good faith estimate of Operating Expenses for such Operating Expense Year. In no event may any such estimate of Operating Expenses exceed, by more than eight percent (8%), the actual Operating Expenses for the immediately preceding Operating Expense Year, if any; provided, however, that such estimate may exceed such actual Operating Expenses by more than eight percent (8%) to the extent Landlord has included in such estimate a bona fide liquidated payment (constituting an Operating Expense) which Landlord expects to make during the Operating Expense Year covered thereby (any such Operating Statement that Landlord gives to Tenant on a prospective basis as contemplated by this Section 2.3(C) being referred to herein as a "Prospective Operating Statement"). If Landlord gives to Tenant a Prospective Operating Statement, then Tenant shall pay to Landlord, on a monthly basis, as additional rent, on account of the Operating Payment due hereunder for such Operating Expense Year, an amount equal to Landlord's reasonable good faith estimate of

monthly Operating Expenses as shown on such Prospective Operating Statement. Tenant shall make such monthly payments to Landlord for the period, and in the amounts, as reflected on the applicable Prospective Operating Statement. Tenant shall pay to Landlord the amount stated in any such Prospective Operating Statement on or prior to the thirtieth (30th) day after the date that Landlord gives such Prospective Operating Statement to Tenant.

- (D) If Landlord gives to Tenant a Prospective Operating Statement, then Landlord shall also provide to Tenant, within one hundred eighty (180) days after the last day of the Operating Expense Year covered by such Prospective Operating Statement, an Operating Statement for such Operating Expense Year. If the aggregate amount of the Operating Payment as reflected on such Operating Statement exceeds the aggregate amount of the aforesaid monthly installments that Tenant pays to Landlord pursuant to the Prospective Operating Statement, then Tenant shall pay to Landlord the amount of such excess not later than the thirtieth (30th) day after the date that Landlord gives such Operating Statement to Tenant. If the aggregate amount of the aforesaid monthly installments that Tenant pays to Landlord pursuant to the Prospective Operating Statement exceeds the aggregate amount of the Operating Payment as reflected on such Operating Statement, then Landlord pursuant to the Prospective Operating Statement exceeds the aggregate amount of the Applicable Rate on the amount of such Operating Payment, then Landlord shall also pay to Tenant an amount equal to interest computed at the Applicable Rate on the amount of such excess from the date that the aggregate amount of the monthly payments made by Tenant to Landlord pursuant to the Prospective Operating Statement first exceeded such Operating Payment until the date that Landlord pays such excess to Tenant. Landlord's obligations with respect to any refund payable to Tenant under this Section 2.3(D) shall survive the expiration or earlier termination of this Lease.
- (E) If the Rent Commencement Date occurs later than the first (1st) day of the Operating Expense Year that immediately succeeds the Base Operating Expense Year, then the Operating Expense Payment for the Operating Expense Year during which the Rent Commencement Date occurs shall be an amount equal to the product obtained by multiplying (X) the Operating Expense Payment that would have been due hereunder if the Rent Commencement Date was the first (1st) day of such Operating Expense Year, by (Y) a fraction, the numerator of which is the number of days in the period beginning on the Rent Commencement Date and ending on the last day of such Operating Expense Year, and the denominator of which is three hundred sixty-five (365) (or three hundred sixty-six (366), if such Operating Expense Year is a leap year).
- (F) If the Expiration Date is not the last day of an Operating Expense Year, then the Operating Expense Payment for the Operating Expense Year during which the Expiration Date occurs shall be an amount equal to the product obtained by multiplying (X) the Operating Expense Payment that would have been due hereunder if the Expiration Date was the last day of such Operating Expense Year, by (Y) a fraction, the numerator of which is the number of days in the period beginning on the first (1st) day of such calendar year and ending on the Expiration Date, and the denominator of which is three hundred sixty-five (365) (or three hundred sixty-six (366), if such Operating Expense Year is a leap year).

(G) If the Operating Expenses for the Base Operating Expense Year are redetermined in accordance with the applicable provisions of this Article 2 at any time after the date that Landlord gives an Operating Expense Statement to Tenant for an Operating Expense Year, then Landlord shall give to Tenant a revised Operating Expense Statement that recalculates the Operating Expense Payment for each Operating Expense Year following the Base Operating Expense Year (using the Operating Expenses that reflects such redetermination for the Base Operating Expense Year). If and to the extent that any such revised Operating Expense Statement results in an underpayment by Tenant, Tenant shall pay to Landlord an amount equal to the excess of (i) the Operating Expense Payment as reflected on such revised Operating Expense Statement, within thirty (30) days after Landlord gives such revised Operating Expense Statement to Tenant. If and to the extent that any such revised Operating Expense Statement results in an overpayment by Tenant, Landlord shall pay to Tenant an amount equal to the excess of (i) the Operating Expense Payment as reflected on such revised Operating Expense Statement, over (ii) the Operating Expense Payment as reflected on the prior Operating Expense Statement, within thirty (30) days after Landlord gives such revised Operating Expense Statement to Tenant. Tenant shall have the right to submit to an Expedited Arbitration Proceeding a dispute with respect to the provisions of this Section 2.3(G).

# 2.4. <u>Auditing of Operating Expense Statements.</u>

- (A) Subject to the terms of this Section 2.4(A), any Operating Statement sent to Tenant shall be conclusively binding upon Tenant unless, within one (1) year after such Operating Statement is sent, Tenant gives a notice to Landlord objecting to such Operating Statement and specifying the respects in which such Operating Statement is disputed. Landlord shall disclose to Tenant, in connection with Tenant's review of each Operating Statement, whether any Person to which Landlord has paid amounts that Landlord has included in Operating Expenses constitutes an Affiliate of Landlord. If Tenant gives such notice to Landlord, then Tenant (together with its reputable legal counsel, consultants and/or independent public accountants) may promptly examine Landlord's books and records relating to such Operating Statement to determine the accuracy thereof.
- (B) Tenant recognizes the confidential nature of such books and records and shall deliver to Landlord a confidentiality statement signed by an officer of Tenant (and a partner, principal or officer of Tenant's legal counsel, consultants and/or certified public accountants to which such books and records have been delivered) wherein Tenant and its legal counsel, consultants and/or certified public accountants agree to maintain the information obtained from such examination in strict confidence, except that Tenant and such legal counsel, consultants and/or certified public accountants may reveal such documentation to their agents, representatives and employees who are actively involved in such investigation during the course of its review, or to an arbitrator (provided that such agents and representatives and, if permitted, any such arbitrator, shall at such time deliver to Landlord a confidentiality statement signed by an officer, partner or principal authorized to bind such party) or to a court of competent jurisdiction in the event of a dispute relating to the subject matter contained in whole or in part in such records or as otherwise may be required by any Requirement.

(C) If, after such examination, Tenant still disputes such Operating Statement, then either party may elect to have the decision of the issues determined by an Expedited Arbitration Proceeding (it being understood that the arbitrator in such matter shall be a qualified, disinterested and impartial person (x) who is, and has been for the last ten (10) years, a partner in a reputable accounting firm that is regularly engaged in and familiar with the accounting concepts applicable to the computation of operating expenses for first-class office buildings located in midtown Manhattan, and (y) who has not been employed by either party during the previous three (3) years). If such decision shows that Tenant has overpaid with respect to such Operating Statement by more than three percent (3%), then Landlord shall also reimburse Tenant for the reasonable out-of-pocket costs of Tenant's audit of Landlord's books and records (provided, however, that notwithstanding Tenant's fee arrangements with its auditor, Landlord shall only reimburse Tenant for the actual hours of service provided by such auditor at the then reasonable and customary hourly rate for such services) within thirty (30) days after demand therefor. If such decision shows that Tenant overpaid with respect to any particular item of Operating Expenses by more than five percent (5%), then Tenant may examine Landlord's books and records covering the Operating Expense Years three (3) years prior to such decision, but only in respect of such particular item and only to the extent that Tenant has not theretofore examined Landlord's books and records in respect of such Operating Expense Years. Any examination of Landlord's books and records pursuant to the immediately preceding sentence (and any dispute arising out of such inspection) shall be made (and resolved) in accordance with the terms of this Section 2.4(C). Notwithstanding the giving of such notice by Tenant, and pending the resolution of any such dispute, Tenant shall pay to Landlord when due all undisputed

## 2.5. <u>Tax Definitions.</u>

- (A) The term "Assessed Valuation" shall mean the amount for which the Office Unit is assessed pursuant to applicable provisions of the New York City Charter and of the Administrative Code of The City of New York, in either case for the purpose of calculating all or any portion of the Taxes (it being understood that pursuant to current Requirements, the lower of the transitional assessment and the actual assessment is used for purposes of calculating Taxes for a particular Tax Year).
- (B) The term "Base Taxes" shall mean, subject to Section 2.7(D) hereof, the Taxes for the Base Tax Year, but in no event shall Base Taxes be less than an amount (the "Base Tax Floor") equal to the product of (x) Twenty-Three Million Three Hundred Fourteen Thousand Five Hundred Sixty-Eight Dollars and 00/100 Cents (\$23,314,568.00), and (y) the tax rate in effect for the Base Tax Year.
  - (C) The term "Base Tax Year" shall mean the fiscal year commencing on July 1, 2005 and ending on June 30, 2006.
- (D) The term "Excluded Amounts" shall mean (i) any taxes imposed on Landlord's income, (ii) franchise, estate, inheritance, gains, capital stock, excise, excess profit, occupancy or rent, gift, payroll or stamp taxes imposed on Landlord, (iii) any charges and/or

taxes that are paid by individual tenants, (iv) any transfer or mortgage recording taxes imposed in connection with a transfer of the Building or the Office Unit or any refinancing of either, or (v) any interest, penalties or late charges imposed against Landlord.

- (E) The term "Qualified Tax Benefits" shall mean any exemption or deferral of Taxes that (i) derives from a program a Governmental Authority adopts to promote the improvement, redevelopment or commercial occupancy of real property, (ii) derives from a particular tenant's occupancy (or prospective occupancy) of a portion of the Building, and (iii) Landlord is required by applicable Requirements or the requirements of such program to apply (and that Landlord does actually apply) for on behalf of a particular tenant.
- The term "Taxes" shall mean the aggregate amount of real estate taxes and any general or special assessments that in each case are imposed upon the Office Unit, without taking into account (a) any discount that Landlord receives by virtue of any early payment of Taxes, (b) any penalties or interest that the applicable Governmental Authority imposes for the late payment of such real estate taxes or assessments, (c) any Excluded Amounts, (d) any Qualified Tax Benefits, or (e) the tax exemption benefits allocable to the Office Unit under Section 421-a of the Real Property Tax Law of The State of New York that derive from the construction of a portion of the Building as a Multiple Dwelling (as such term is defined in Section 421-a(1)(c) of the Real Property Tax Law of the State of New York); provided, however, that if, because of any change in the taxation of real estate, any other tax or assessment, however denominated (including, without limitation, any franchise, income, profit, sales, use, occupancy, gross receipts or rental tax), is imposed upon the Office Unit, the owner thereof, or the occupancy, rents or income derived therefrom, in substitution for any of the Taxes, then such other tax or assessment to the extent substituted shall be included in Taxes for purposes hereof (assuming that the Office Unit is Landlord's sole asset and the income therefrom is Landlord's sole income). Taxes shall include, without limitation, (i) any taxes or assessments levied from and after the date of this Lease, in whole or in part, for public benefits to the Office Unit, including, without limitation, any business improvement district taxes and assessments, or (ii) any fee, tax or charge imposed by any Governmental Authority for any vaults or vault spaces that in either case are appurtenant to such Office Unit; provided, however, that Taxes shall not include (i) separate assessments made upon or with respect to any "air" and "development" rights now or hereafter appurtenant to or affecting such Office Unit, or (ii) separate assessments for any sign or billboard on the Building that is leased to a third-party. If any Taxes are payable in installments without interest, premium or penalty, then Landlord shall include in Taxes for any particular Tax Year only the installment of such real estate taxes or assessments that the applicable Governmental Authority requires Landlord to pay (and that Landlord actually pays) during such Tax Year.
- (G) The term "Tax Payment" shall mean, with respect to any Tax Year, the product obtained by multiplying (i) the excess of (A) Taxes for such Tax Year, over (B) the Base Taxes, by (ii) Tenant's Tax Share.
  - (H) The term "Tax Statement" shall mean a statement that shows the Tax Payment for a particular Tax Year.

- (I) The term "<u>Tax Year</u>" shall mean the Base Tax Year and each subsequent period from July 1 through June 30 (or such other period as hereinafter may be duly adopted by the Governmental Authority then imposing Taxes as its fiscal year for real estate tax purposes).
- (J) The term "<u>Tenant's Tax Share</u>" shall mean, subject to the terms hereof, eighty-nine and six hundred ninety-five one-thousandths percent (89.695%).

## 2.6. <u>Tax Payment.</u>

- (A) Subject to the provisions of this Section 2.6, Tenant shall pay to Landlord, as additional rent, the Tax Payment.
- (B) Subject to the provisions of this Section 2.6, Tenant shall pay to Landlord the Tax Payment for a particular Tax Year on or prior to the thirtieth (30<sup>th</sup>) day before the date that the applicable Governmental Authority obligates Landlord to make the corresponding payment of Taxes for such Tax Year. If the applicable Governmental Authority requires Landlord to pay the Taxes for a Tax Year in more than one (1) installment, then Tenant shall pay the Tax Payment to Landlord for such Tax Year in a corresponding number of installments. Tenant shall not be required to make a Tax Payment to Landlord (or to pay an installment thereof to Landlord) earlier than the fifteenth (15<sup>th</sup>) day after the date that Landlord gives Tenant a Tax Statement therefor. If Tenant's obligation to make the Tax Payment hereunder commences on a date that is not the date that the applicable Governmental Authority requires Landlord to make a corresponding payment of Taxes, then Tenant shall pay to Landlord, on such date that Tenant's obligation to make the Tax Payment hereunder commences, the installment of the Tax Payment due hereunder for the corresponding period, which installment shall be apportioned appropriately.
- (C) If the Rent Commencement Date occurs later than the first (1st) day of the Tax Year that immediately succeeds the Base Tax Year, then the Tax Payment for the Tax Year during which the Rent Commencement Date occurs shall be an amount equal to the product obtained by multiplying (X) the Tax Payment that would have been due hereunder if the Rent Commencement Date was the first (1st) day of such Tax Year, by (Y) a fraction, the numerator of which is the number of days in the period beginning on the Rent Commencement Date and ending on the last day of such Tax Year, and the denominator of which is three hundred sixty-five (365) (or three hundred sixty-six (366), if such Tax Year includes the month of February in a leap year).
- (D) If the Expiration Date is not the last day of a Tax Year, then the Tax Payment for the Tax Year during which the Expiration Date occurs shall be an amount equal to the product obtained by multiplying (X) the Tax Payment that would have been due hereunder if the Expiration Date was the last day of such Tax Year, by (Y) a fraction, the numerator of which is the number of days in the period beginning on the first (1st) day of such Tax Year and ending on the Expiration Date, and the denominator of which is three hundred sixty-five (365) (or three hundred sixty-six (366), if such Tax Year includes the month of February in a leap year).
- (E) The Tax Payment shall be computed initially on the basis of the Assessed Valuation in effect on the date that Landlord gives the applicable Tax Statement to Tenant (as the

Taxes may have been settled or finally adjudicated prior to such time) regardless of any then pending application, proceeding or appeal to reduce the Assessed Valuation, but shall be subject to subsequent adjustment as provided in Section 2.7 hereof.

- (F) Tenant shall pay the Tax Payment regardless of whether Tenant is exempt, in whole or part, from the payment of any Taxes by reason of Tenant's diplomatic status or otherwise.
- (G) If Taxes are required to be paid on any date or dates other than as presently required by the Governmental Authority imposing Taxes, then the due date of the installments of the Tax Payment shall be adjusted so that each such installment is due from Tenant to Landlord thirty (30) days prior to the date that the corresponding payment is due to the Governmental Authority.
- (H) Landlord's failure to give to Tenant a Tax Statement for any Tax Year shall not impair Landlord's right to give to Tenant a Tax Statement for any other Tax Year, provided, however, that if Landlord fails to give Tenant a Tax Statement for any Tax Year on or before the date that occurs two (2) years after the last day of such Tax Year to which such Tax Statement applies, then Landlord shall be deemed to have waived the payment of any then unpaid additional rent which would have been due pursuant to such Tax Statement.
- (I) Landlord shall give to Tenant a copy of the relevant tax bill for each Tax Year (to the extent that the applicable Governmental Authority has issued such tax bill to Landlord) together with the Tax Statement for such Tax Year.

## 2.7. Reductions in Taxes.

- (A) Landlord, at Landlord's cost and expense, shall institute and prosecute in good faith tax certiorari proceedings to reduce the Assessed Valuation for each Tax Year occurring during the Term pursuant to and in accordance with applicable Requirements, unless Tenant agrees in writing that Landlord shall not be required to institute any such proceedings with respect to any particular Tax Year (in which event Landlord shall have the option in its discretion to institute any such proceedings). Tenant, at Tenant's cost and expense, shall have the right to have Tenant's tax certiorari attorney or consultants participate in the institution and prosecution of any such proceedings and Landlord shall reasonably take into account the recommendations of Tenant's attorney or consultant in connection with any such proceedings. Landlord shall not enter into a settlement with the applicable Governmental Authority with respect to any such proceeding without obtaining Tenant's prior consent thereto, which consent Tenant shall not unreasonably withhold, condition or delay. Landlord shall not make deliberate efforts in connection with any tax certiorari proceeding instituted and prosecuted by Landlord hereunder (or any settlement in connection therewith) to cause the applicable Governmental Authority to lower the Assessed Valuation for the Base Tax Year in return for agreeing to higher Assessed Valuations in Tax Years occurring after the Base Tax Year. In no event shall any tax certiorari proceeding result in Base Taxes being less than the Base Tax Floor.
- (B) If, after a Tax Statement has been sent to Tenant, an Assessed Valuation that Landlord used to compute the Tax Payment for a Tax Year is reduced, and, as a result

thereof, a refund of Taxes is actually received by, or credited to, Landlord, then Landlord, promptly after Landlord's receipt of such refund (or such refund is credited to Landlord, as the case may be), shall send to Tenant a Tax Statement adjusting the Taxes for such Tax Year and setting forth, based on such adjustment, the portion of such refund for which Tenant is entitled a credit as set forth in this Section 2.7(B). Landlord shall credit the portion of such refund to which Tenant is entitled against the Rental thereafter coming due hereunder. The portion of such refund to which Tenant is entitled shall be limited to the portion of the Taxes, if any, that Tenant had theretofore paid to Landlord on account of the Tax Payment for the Tax Year to which the refund is applicable on the basis of the Assessed Valuation before it had been reduced. The Tax Payment paid by Tenant for such Tax Year (after taking into account such refund) shall be an amount equal to the Tax Payment that Tenant would have paid hereunder if the Assessed Valuation used in computing Taxes for such Tax Year had reflected initially the aforesaid reduction thereof that yielded such refund. If (x) Tenant is entitled to a credit against Rental pursuant to this Section 2.7(B), and (y) the Expiration Date occurs prior to the date that such credit is exhausted, then Landlord shall pay to Tenant the unused portion of such credit on or prior to the thirtieth (30th) day after the Expiration Date (and Landlord's obligation to make such payment shall survive the Expiration Date). If (i) Landlord receives such refund (or a credit therefor) after the Expiration Date, and (ii) Tenant is entitled to a portion thereof as contemplated by this Section 2.7(B), then Landlord shall pay to Tenant an amount equal to Tenant's share of such refund (or such credit) within thirty (30) days after the date that such refund is paid to Landlord (or such refund is credited to Landlord, as the case may be) (and Landlord's obligation to make such payment shall survive the E

- (C) If the Assessed Valuation for the Base Tax Year is reduced at any time after the date that Landlord gives a Tax Statement to Tenant for a Tax Year, then Landlord shall have the right to give to Tenant a revised Tax Statement that recalculates the Tax Payment for a Tax Year (using the Taxes that reflect such reduction in such Assessed Valuation) (it being agreed that the Base Taxes shall in no event be less than the Base Tax Floor). Tenant shall pay to Landlord an amount equal to the excess of (i) the Tax Payment as reflected on such revised Tax Statement, over (ii) the Tax Payment as reflected on the prior Tax Statement, within thirty (30) days after Landlord gives such revised Tax Statement to Tenant.
- (D) If, (I) the Assessed Valuation for (x) the Tax Year commencing on July 1, 2006 and ending on June 30, 2007, or (y) the Tax Year commencing on July 1, 2007 and ending on June 30, 2008 increases, in either case, over the Assessed Valuation for the prior Tax Year, and (II) any portion of any such increase is attributable to physical improvements made to the Office Unit that are identifiable in any documentation (including the tax bill) received by Landlord from the applicable Governmental Authority in respect of such Tax Year, then Landlord shall recalculate the Base Taxes as follows: (i) the Assessed Valuation for the Base Tax Year shall be deemed increased by the amount of any such increase for such physical improvements as aforesaid, and (ii) the Base Taxes shall be recalculated by multiplying such increased Assessed Valuation pursuant to the immediately preceding clause (i) by the tax rate that was in effect for the Base Tax Year, and (iii) such recalculated amount shall be deemed to be the Base Taxes retroactive to the Base Tax Year (subject to the terms of this clause (D)). If the Assessed Valuation for the Base Tax Year is increased as aforesaid, then Landlord shall give to Tenant a revised Tax Statement that recalculates the Tax Payment theretofore paid by Tenant for each Tax Year affected by such recalculation of the Base Taxes (using the recalculated Base

Taxes as aforesaid), and from and after Tenant's receipt of such revised Tax Statement, Tenant shall have the right to credit against its next installments of Rental due hereunder an amount equal to the excess of (i) the applicable Tax Payment theretofor paid by Tenant for the applicable Tax Year prior to the date of such recalculation by Landlord, over (ii) the Tax Payment as reflected on the revised Tax Statement given by Landlord to Tenant. Any recalculation of Base Taxes pursuant to this Section 2.7(D) shall be subject to further recalculation to the extent that the increases for physical improvements reflected in the Assessed Valuation in either of the Tax Years described in this clause (D) are reduced by virtue of a tax certiorari proceeding or settlement (and, accordingly, Landlord shall have the right to give Tenant a further revised Tax Statement for an applicable Tax Year if such physical improvements are reduced as aforesaid). In no event shall the provisions of this Section 2.7 result in Base Taxes being less than the Base Tax Floor.

Article 3 <u>USE</u>

#### 3.1. <u>Permitted Use.</u>

- (A) Subject to Section 3.2 hereof and Section 3.3 hereof, Tenant shall use the Premises (other than the Mailroom and the Storage Space), and Tenant shall cause any other Person claiming by, through or under Tenant to use the Premises, in either case only as general and executive offices and for uses reasonably ancillary or incidental thereto. Landlord acknowledges that Tenant's use of the Premises (or portions thereof) as a trading facility shall not be deemed to violate the provisions of this Section 3.1. Subject to Section 3.2 and Section 3.3 hereof, Tenant shall use and occupy, and shall cause any Person claiming by, through or under Tenant to use and occupy, the portion of the Premises located on lower level 3 of the Building and denoted on Exhibit "A" hereto as the "message center" (the "Mailroom") only for purposes of receiving and distributing mail and packages, and for messenger deliveries in connection with Tenant's conducting business in the Premises and other tenants in the Office Unit conducting business in their premises in the Office Unit, and for no other purpose. Subject to Section 3.2 and 3.3 hereof, Tenant shall use and occupy, and shall cause any Person claiming by, through or under Tenant to use and occupy, the portion of the Premises located on lower level 3 of the Building and denoted on Exhibit "A" attached hereto as the "storage space" (the "Storage Space") for storage purposes, and for no other purpose. Subject to Section 3.2 hereof, such ancillary or incidental uses may include, without limitation, the following:
  - (1) kitchens, dining facilities, pantries and/or vending machines (which may be supplied by any party selected by Tenant);
  - (2) board rooms, conference rooms, meeting rooms, conference centers and facilities and an auditorium;
- (3) one or more data centers for computer and other electronic data processing and business machine operations in connection with the business operations of Tenant and any Permitted Parties;

- (4) duplicating, photographic reproduction and/or offset printing facilities in connection with the business operations of Tenant and any Permitted Parties;
  - (5) mailroom facilities in connection with the business operations of Tenant and any Permitted Parties;
- (6) secured facilities and storage of equipment, records, files and other items in connection with the business operations of Tenant and any Permitted Parties;
  - (7) medical or health facilities exclusively serving Tenant and any Permitted Parties;
  - (8) travel services or agencies exclusively serving Tenant and any Permitted Parties;
  - (9) day care facilities exclusively serving Tenant and any Permitted Parties;
  - (10) an exercise facility for use only by Tenant and any Permitted Parties; and
  - (11) a messenger center for use only by Tenant and any Permitted Parties.
- (B) Tenant, at Tenant's cost and expense, shall have the right to perform Alterations in accordance with the applicable terms and provisions of Article 7 hereof to make available Building Systems to the Storage Space. Tenant, at Tenant's cost and expense, shall have the right to perform Alterations in accordance with the applicable terms and provisions of Article 7 hereof to make available Building Systems to provide solely electricity and HVAC to the Mailroom. Tenant's employees or agents that use the Storage Space and the Mailroom shall have the right to use on a non-exclusive basis the restrooms located on lower level 3 of the Building (including the restroom located in the loading dock area of the Building). Tenant shall have the right to place lockers in the Storage Space for Tenant's security guard stationed behind the lobby desk in the Building and for any other security guard that is posted to the lobby of the Building in accordance with the terms of this Lease.
- (C) Tenant expressly acknowledges that it shall operate in the Mailroom a package and mail center for purposes of distributing and receiving mail and packages and a messenger center on behalf of the tenants or occupants in the Office Unit, subject to the Rules. If Tenant no longer meets the Minimum Demise Requirement, then Landlord shall have the right to terminate this Lease solely with respect to the Mailroom by giving Tenant written notice thereof, in which event this Lease solely with respect to the Mailroom shall terminate on the date that is thirty (30) days after the date that Landlord gives such notice to Tenant (without any reduction in the Rental payable by Tenant hereunder). From and after the date that this Lease terminates with respect to the Mailroom as aforesaid, Landlord shall operate as a service hereunder a center for the distribution of mail and packages, and for messenger deliveries, for the tenants, including Tenant, in the Office Unit.
- (D) Notwithstanding the provisions of Article 4 hereof, Landlord shall have no obligation to clean the Mailroom or the Storage Space, as the case may be, to provide HVAC thereto pursuant to Section 4.4 hereof, or to provide gas, steam, or water thereto.

# 3.2. <u>Limitations.</u>

- (A) Tenant shall not use the Premises or any part thereof, or permit the Premises or any part thereof to be used:
- (1) for the conduct of retail trade on an off-the-street basis or otherwise (including, without limitation, for (x) the sale of pornographic or obscene materials or for any similar purpose or as a "massage parlor", "sex club", "topless bar", or other similar establishment, (y) a facility for the sale of paraphernalia for use with illicit drugs, or (z) an off-track betting parlor).
  - (2) by any Governmental Authority or any other Person having sovereign or diplomatic immunity;
- (3) for a restaurant, bar, or for the sale, storage, preparation, service or consumption of food or beverages in any manner whatsoever, including, without limitation, for the sale of confectionery, soda or other beverages, sandwiches, ice cream or baked goods (except that, as provided in Section 3.1(A)(1) herein, Tenant has the right to store, prepare, and serve food and beverages, by any reasonable means (including, without limitation, by means of customary vending machines), for consumption by Tenant's officers, employees and business guests in the Premises);
- (4) as an employment agency, executive search firm or similar enterprise, labor union, school, or vocational training center (except for the incidental use of portions of the Premises from time to time for presenting seminars and training to the employees of Tenant, any Permitted Party and/or their Affiliates, or their clients, customers, service providers or other people with whom Tenant, any Permitted Party or their Affiliates have a significant business relationship, to the extent that any such use or the manner of such use would not (i) create a volume of pedestrian traffic and use of the elevators in excess of that found in the ordinary course at first-class office buildings in midtown Manhattan, (ii) violate any Requirements, (iii) require an amendment to the Certificate of Occupancy for the Building, or (iv) obligate Landlord to make any physical alterations to any part of the Building (except to the extent that such alterations are limited to the Premises, performed by Tenant at its sole cost and expense, and consented to by Landlord, which consent shall be granted or withheld by Landlord in accordance with the terms and conditions of the Article 7 herein and otherwise performed in accordance with Article 7 hereof); or
- (5) for the business of photographic, multilith or multigraph reproductions of offset printing, except in connection with, either directly or indirectly, Tenant's own business and/or activities.

# 3.3. <u>Bloomberg Lease Restrictions.</u>

Provided that the Bloomberg Lease is still in full force and effect and the provisions of Sections 40.1 through 40.7 of the Bloomberg Lease have not been rendered ineffective pursuant to Section 40.8 of the Bloomberg Lease, the Premises or any portion thereof may not be used or occupied by any Person that is a Bloomberg Primary Competitor. The term "Bloomberg Primary Competitor" shall mean a Person that is included on the list of Bloomberg Primary Competitors

annexed as Exhibit "3.3-1" attached hereto and made a part hereof. Pursuant to the Lease (the "Bloomberg Lease"), dated as of April 30, 2001, between Seven Thirty One Limited Partnership, as landlord, and Bloomberg L.P. ("Bloomberg"), as tenant, the list of Bloomberg Primary Competitors may change only to the extent provided on Exhibit "3.3-2" attached hereto and made a part hereof. Tenant shall not be bound by any change in the list of Bloomberg Primary Competitors set forth as Exhibit "3.3-2" attached hereto unless and until Tenant has received written notice of such change in accordance with Article 28 hereof. Nothing contained in this Section 3.3 shall prevent the Premises or any portion thereof (subject to the other terms and provisions of this Lease), from being used by a Person that is a Bloomberg Primary Competitor if such Person did not constitute a Bloomberg Primary Competitor on the earlier to occur of (x) the date that such Person entered into occupancy of the Premises (or a portion thereof), and (y) the date that such Person entered into an agreement to occupy the Premises (or a portion thereof).

#### 3.4. <u>Rules.</u>

- (A) Subject to the terms of this Section 3.4, Tenant shall comply with, and Tenant shall cause any other Person claiming by, through or under Tenant to comply with, (i) the rules set forth in Exhibit "3.4-1" attached hereto and made a part hereof, (ii) the rules with respect to Alterations set forth in Exhibit "3.4-2" attached hereto and made a part hereof, and (iii) other reasonable rules that Landlord hereafter adopts from time to time on reasonable advance notice to Tenant (such rules that are attached hereto, and such other rules, being collectively referred to herein as the "Rules"), which other rules, in Landlord's reasonable judgment, are necessary for the reputation, safety, care and appearance of the Real Property or the operation or maintenance thereof, and which do not unreasonably affect the conduct of Tenant's business in the Premises or Tenant's use of the Premises. Tenant shall have the right to institute an Expedited Arbitration Proceeding to determine whether any additional Rule hereafter adopted by Landlord complies with this Section 3.4 only by giving written notice thereof to Landlord within thirty (30) days after the date that Landlord gives Tenant written notice of Landlord's adoption of any such additional Rule (it being understood that Tenant shall not have any other right to dispute the compliance with such standards by any such additional Rule hereafter adopted by Landlord). Except as otherwise provided in this Section 3.4, nothing contained in this Lease shall be construed to impose upon Landlord any obligation to enforce the Rules or the terms of any other lease against any other tenant, and Landlord shall not be liable to Tenant for violation thereof by any other tenant. Landlord shall not enforce any Rule against Tenant that Landlord is not then enforcing against all other office tenants in the Office Unit (or in the Office Unit One if the Office Unit One Owner is then Landlord or its Affiliate) and shall not enforce any Rules in a manner discriminatory to Tenant. If a conflict or
- (B) If (i) another occupant in the Building fails to comply with a Rule that is applicable to Tenant under this Lease, and (ii) such failure by such other occupant either (a) violates the terms of a lease between such occupant and its landlord, or (b) violates such other occupant's obligations to the Condominium, (iii) such failure by such other occupant has a material adverse effect on Tenant's ability to conduct business in the Premises, (iv) Tenant is then in compliance with such Rule, and (v) Tenant requested that Landlord enforce Landlord's

rights against such other occupant in respect of such failure (or, if such occupant is (x) an occupant of a unit in the Condominium owned by an Affiliate of Landlord, Tenant requested that Landlord cause such Affiliate to enforce such Affiliate's rights, or (y) an occupant of a unit in the Condominium not owned by Landlord and/or its Affiliates, Tenant requested that Landlord cause the Condominium Board to enforce the Condominium's rights), then Landlord shall use due diligence to enforce such rights as aforesaid (or to cause its Affiliate or the Condominium Board, as the case may be, to enforce such rights as aforesaid) against such other occupant in respect of such failure (it being agreed, however, that Landlord shall have no obligation to institute a proceeding against such other occupant) promptly after Tenant's request to Landlord therefor. Landlord shall consult (or shall cause the Condominium Board to consult) from time to time with Tenant in connection with Landlord's using due diligence (or the Condominium's using due diligence) to enforce the aforesaid rights of Landlord or the Condominium Board against such other occupant as contemplated by this Section 3.4. Tenant agrees that Landlord's costs to enforce the rights described in this Section 3.4 hereof shall constitute an Operating Expense.

# 3.5. <u>Fire Stairs.</u>

Subject to the terms of this Section 3.5, Tenant shall have the right to use (and to permit Permitted Parties to use) at no additional cost (but subject, however, to recoupment to the extent provided in Article 2 hereof) the fire stairs serving the Premises, for purposes of permitting personnel to move among the floors of the Building that comprise the Premises (such fire stairs being referred to herein as the "Fire Stairs"). A Permitted Party shall not have the right to use the Fire Stairs as contemplated by this Section 3.5 to gain access to the street adjacent to the Building (except in the event of an emergency). Tenant shall use (or permit another Permitted Party to use) the Fire Stairs only to the extent permitted by, and in a manner that is consistent with, applicable Requirements. Tenant shall not have the right to use the Fire Stairs in a manner that prevents free passage therein from floors of the Building other than the Premises. Nothing contained in this Section 3.5 diminishes Landlord's right to make installations in the Fire Stairs to limit Tenant's ability to gain access to portions of the Building (other than the Premises) from the Fire Stairs. Tenant shall not have the right to perform any Alterations in the Fire Stairs (except that Tenant shall have the right to install, in accordance with Article 7 hereof and to the extent permitted by Requirements, (x) a security system in the Fire Stairs that seeks to prevent unauthorized persons from entering the Premises from the Fire Stairs, and (y) reasonable finishes in the Fire Stairs (such as floor covering, paint and lighting), in either case at Tenant's expense). After the Commencement Date, Landlord, at Landlord's cost and expense, shall install (x) photo-luminescent paint on the treads and railings located in the Fire Stairs, and (y) two (2) security gates in the Fire Stairs (one such gate to be located in the Fire Stairs directly above the landing to the twenty-ninth (29<sup>th</sup>) floor of the Building and one such gate to be located in the Fire Stairs directly below the landing thereof to the twenty-first (21st) floor of the Building), which work Landlord shall perform in accordance with all applicable Requirements and in a good and workerlike manner. The aforesaid gates shall be locked in an effort to prevent unauthorized access into the Fire Stairs that serve the Premises between the twenty-first (21st) and twenty-ninth (29th) floors of the Building, and shall automatically unlock in the event of an emergency.

# 3.6. <u>Risers.</u>

Subject to the terms of this Section 3.6, Landlord hereby consents to Tenant's installing and maintaining fuel lines, electrical lines, (A) telecommunications and computer lines and cabling, exhaust ducts and flues or other similar lines, cabling, ducts, risers and conduits or any other equipment that may be lawfully installed therein (collectively, the "Risers") in the shaft locations shown on Exhibit "3.6" attached hereto and made a part hereof, all at no additional charge to Tenant. If Tenant licenses the Antennae Site in accordance with Section 3.9 hereof, then Landlord consents to Tenant's installing and maintaining a Riser in a shaft location reasonably designated by Landlord that runs from the mechanical room (the "Antennae Mast Mechanical Room") that supports the Antennae Mast and the Antennae located on the fifty-fourth (54<sup>th</sup>) floor of the Building to the Premises. Landlord shall provide Tenant with access in accordance with good construction practice at reasonable times and upon reasonable advance notice for the installation, operation and maintenance of the Risers, provided that such access shall (i) not unreasonably interfere with or interrupt the operation and maintenance of the Building, and (ii) be upon such other terms reasonably designated by Landlord. Tenant shall install the Risers at Tenant's expense (subject to Tenant's right to use a portion of the Tenant Fund for such expense). Tenant shall perform such installation in accordance with the provisions of this Lease, including, without limitation, the provisions pertaining to the performance of Alterations. If Tenant exercises Tenant's right to install the Risers as contemplated by this Section 3.6, then Tenant, at Tenant's expense, shall maintain the Risers in good condition during the Term. Tenant, upon the Expiration Date, shall not be required to remove the Risers; provided, however, that (i) Landlord reserves the right to require the Risers to be disconnected, capped and sealed at Tenant's cost upon the Expiration Date, and (ii) Landlord shall have the right to require Tenant to remove the Risers upon the Expiration Date to the extent required by any Requirement. Landlord shall use reasonable efforts to provide Tenant with additional shaft locations in the Building (in addition to those set forth on Exhibit "3.6" attached hereto) if Tenant desires to install additional Risers, upon Tenant's written request therefor and subject to there being riser space that is then available in the Building.

## 3.7. <u>Tenant's Signs.</u>

(A) Subject to the terms of this Section 3.7, Tenant shall have the right to erect and maintain one (1) sign that identifies Tenant or its Affiliate or its successor pursuant to Section 17.9 hereof as an occupant of the Building (and for no other purpose) in the lobby of the Building at the location described in, and in accordance with the specifications described in, Exhibit "3.7-1" attached hereto and made a part hereof (the "Lobby Sign"). Tenant shall be permitted to erect and maintain the Lobby Sign only during the period that the Minimum Demise Requirement is satisfied. At any time that Tenant meets the Minimum Demise Requirement, Tenant may elect to have the Lobby Sign identify, in lieu of Tenant (or its Affiliate or successor as aforesaid), a Permitted Party that subleases from Tenant at least eighty-eight thousand (88,000) square feet of Rentable Area of the Premises. Tenant shall have the right to replace the Lobby Sign that is depicted on Exhibit "3.7-1" attached hereto provided that Tenant obtains Landlord's prior written consent thereto (it being agreed that Landlord shall not unreasonably withhold, condition or delay its consent to any such replacement Tenant's Sign if such replacement Tenant Sign is the same size as the Tenant's Sign depicted on Exhibit "3.7-1" attached hereto, and is the same color and made with the same materials as the Tenant's Sign

depicted on Exhibit "3.7-1" attached hereto). If Tenant no longer satisfies the Minimum Demise Requirement, then Tenant shall not have the right to maintain the Lobby Sign that is described on Exhibit "3.7-1" attached hereto (as the Lobby Sign may have theretofore been replaced as aforesaid), provided that Landlord, at Landlord's cost and expense, shall erect a sign in a location in the lobby of the Building designated by Landlord in Landlord's reasonable discretion that identifies Tenant and that will indicate the elevator bank that accesses the Premises (which sign shall be of a size that is proportionate to similar signs erected by Landlord for other tenants or occupants of the Office Unit, based on the relative square footage then leased by Tenant and such other tenants or occupants).

- (B) Tenant shall have the right at any time during the Term to install a sign on the exterior of the Building (the "Exterior Sign"; the Lobby Sign and the Exterior Sign being collectively referred to herein as "Tenant's Signs") on the left hand column of the entrance to the Lexington Avenue lobby of the Building, which column is more particularly described on Exhibit "3.7-2" attached hereto and made a part hereof. The dimensions of the Exterior Sign shall be three (3) feet four (4) inches wide and six (6) inches high. Landlord, in Landlord's sole discretion, shall have the right to determine the exact location of the Exterior Sign on such column, provided that such location determined by Landlord shall be within the designated range depicted on Exhibit "3.7-2" attached hereto, and the colors and materials that will comprise the Exterior Sign. Tenant shall be permitted to erect and maintain the Exterior Sign only during the period that Tenant leases the entire Rentable Area of five (5) or more floors of the Building that comprise the Premises on the date hereof. If (x) the Exterior Sign is removed by virtue of Tenant no longer leasing five (5) or more floors of the Building as aforesaid, (y) Tenant meets the Minimum Demise Requirement, and (z) Landlord permits another tenant or occupant in the Office Unit to install a sign on the aforesaid column, then Tenant shall thereafter have the right to erect a sign in a location on such column designated by Landlord that is similar to such sign installed by such other tenant or occupant. The Exterior Sign may only identify one (1) Person, which may be Citibank, N.A. or Citigroup, Inc. or any successor of Citibank, N.A. or Citigroup, Inc.
- (C) Tenant's installation of Tenant's Signs shall be performed at Tenant's cost in accordance with the provisions set forth in Article 7 hereof. Tenant shall not have the right to erect Tenant's Signs that would create the impression in the mind of a reasonable person that the Building is named for Tenant. Tenant, at Tenant's expense, shall operate, maintain and repair any Tenant's Signs that Tenant erects pursuant to this Section 3.7 in accordance with customary standards for first-class office buildings in the vicinity of the Building and in compliance with all applicable Requirements. Tenant, at Tenant's expense, shall remove Tenant's Signs promptly upon the earlier to occur of (x) the Expiration Date, and (y) the date that Tenant has no further right to erect or maintain Tenant's Signs pursuant to this Section 3.7, and shall repair any damage caused by the installation of Tenant's Signs or such removal.

# 3.8. <u>Promotional Displays.</u>

Tenant shall not have the right to use any window in the Premises for any sign or other display that is designed principally for advertising or promotion.

# 3.9. Antennae Roof Rights.

- (A) Tenant shall have the right to install Antennae on the antennae mast (the "Antennae Mast") installed by Landlord on the roof of the Building for use solely by Tenant (or other Permitted Parties) (and not for resale purposes) on the terms and subject to the conditions set forth in this Section 3.9. At any time during the Term, Tenant shall have the right to install two (2) Antennae on the Antennae Mast (each of which shall be no larger than eighteen (18) inches in diameter) (such two (2) Antennae being referred to herein as the "Permitted Antennae"). Tenant shall give Landlord no less than thirty (30) days prior written notice (the "Antennae Notice") of Tenant's election to install the Permitted Antennae. Tenant shall describe in the Antennae Notice with specificity the type of Antennae Tenant elects to install on the Antennae Mast and the intended use thereof. The term "Antennae" shall mean, collectively, any Tenant antennae or antennaes, together with related equipment, mountings and supports, installed on behalf of Tenant on the Antennae Mast in accordance with this Section 3.9. At any time during the Term Tenant shall have the right to install additional Antennae on the Antennae Mast by giving Landlord written notice thereof, in which notice Tenant shall set forth a description of the additional Antennae that Tenant desires to install and the intended use thereof. If (x) Landlord then has available space on the Antennae Mast to accommodate the additional Antennae described in such notice, and (y) Tenant has theretofore installed the Permitted Antennae, then Tenant shall have the right to install the additional Antennae described in such notice on the terms and conditions set forth in this Section 3.9.
- (B) Promptly after the date that Landlord has sold or licensed for use approximately seventy percent (70%) of the space on the Antennae Mast, Landlord shall give Tenant a notice (the "Antennae Availability Notice") informing Tenant of the amount of space that then remains on the Antennae Mast (such space being referred to herein as the "Available Antennae Space"). Tenant has the right to license all or any portion of the Available Antennae Space by giving notice (the "Antennae Response Notice") thereof to Landlord no later than sixty (60) days after the date Landlord gave the Antennae Availability Notice to Tenant. Time shall be of the essence with respect to Tenant's giving the Antennae Response Notice to Landlord no later than sixty (60) days after the date Landlord gave the Antennae Availability Notice to Tenant. Tenant shall describe with reasonable specificity in the Antennae Response Notice the Antennae that Tenant proposes to install on the Antennae Mast and the intended use of the Antennae that Tenant proposes to install on the Antennae Mast. Landlord has no obligation to give Tenant an Antennae Availability Notice if, at the time approximately seventy percent (70%) of the space on the Antennae Mast has been sold or licensed by Landlord, Tenant no longer meets the Minimum Demise Requirement. If Tenant fails to give Landlord an Antennae Response Notice, then Landlord may thereafter license the remaining space on the Antennae Mast to any Person without having any obligation to notify Tenant thereof; subject, however, to Tenant's right set forth in Section 3.9(A) hereof to install the Permitted Antennae without respect to whether seventy percent (70%) of the space on the Antennae Mast has been sold or licensed by Landlord (i.e., sufficient space for the Permitted Antennae shall be reserved by Landlord for Tenant).
- (C) Landlord shall have the right to reasonably designate the location of the Antennae on the Antennae Mast (the area on the Antennae Mast so reasonably designated by Landlord being referred to herein as the "Antennae Site"). Commencing on the date (the

"Antennae Commencement Date") an Antennae is installed on the Antennae Mast, the amount of space used thereon by any such Antennae shall be licensed by Landlord to Tenant on the terms and conditions contained in this Section 3.9. Landlord shall provide Tenant with riser space for the Antennae pursuant to Section 3.6 hereof. If, on the Antennae Commencement Date, there is electrical capacity available in the Antennae Mast Mechanical Room for Tenant's use in operating the Antennae, then Landlord shall make such electrical capacity available to Tenant for such purpose. Tenant's use of any such electricity shall be measured by submeter, installed by Landlord at Landlord's expense, and computed and billed to Tenant in accordance with the terms and provisions of Section 5.3 hereof. If, however, on the Antennae Commencement Date, there is not sufficient electrical capacity available as aforesaid, then Tenant, at Tenant's cost and expense, shall have the right to run, in accordance with the terms of this Lease, an electrical line from the Premises to the Antennae Mast Mechanical Room for purposes of operating the Antennae. Landlord (or, at Landlord's option, the Mast Manager) shall perform, at Tenant's expense, the work necessary to install the Antennae on the Antennae Mast, including making any electrical connections with respect thereto (other than running an electrical line from the Premises, if necessary, as aforesaid). Except as otherwise expressly set forth in this Section 3.9, the Antennae shall be deemed for all purposes of this Lease to be a Specialty Alteration. The rights granted to Tenant in this Section 3.9 shall not be assignable by Tenant separate and apart from this Lease.

- (D) Landlord may engage a third party (the "Mast Manager") to perform the construction, installation, maintenance or repair of the Antennae on commercially reasonable terms. Tenant shall pay a fee (the "License Fee") to Landlord, as additional rent, for Tenant's right to use the Antennae Site, and for the services of the Mast Manager, in a monthly amount equal to the Fair Market Rent therefor, payable in equal monthly installments in advance during the Term from and after the Antenna Commencement Date. Landlord and Tenant shall determine the Fair Market Rent of the Antennae Site in accordance with the terms of Article 20 hereof. Tenant shall promptly pay when due any tax, license, permit or other fees or charges imposed pursuant to any Requirements relating to the installation, maintenance or use of the Antennae.
- (E) Landlord retains the right to use the area in and around the Antennae Mast for any reasonable purpose whatsoever provided that such use does not unreasonably interfere with the use of the Antennae by Tenant (or the applicable Permitted Party). If, after any Antennae is installed on behalf of Tenant, it is discovered that such Antennae interferes with (i) the reception or transmission of communication signals by or from any antennae, satellite dishes or similar equipment installed on the Antennae Mast or any other location on the roof of the Building in either case prior to the date that the Antennae is installed, or (ii) the operation of the Building Systems, then Landlord shall cause the Mast Manager, on Tenant's behalf and at Tenant's expense, to relocate the Antennae to another area on the Antennae Mast or roof of the Building reasonably designated by Landlord. If such interference still occurs despite such relocation, and, despite Landlord's diligent efforts, there exists no reasonably practicable site on the Antennae Mast or on the roof of the Building for the relocation of the Antennae, then Landlord shall cause the Mast Manager upon prior notice to Tenant and at Tenant's expense, to promptly remove the Antennae from the Antennae Mast or the roof of the Building, as the case may be.

- (F) If any antennae or satellite dish installed on the Antennae Mast after the date that Tenant's Antennae is installed thereon interferes with the reception or transmission of communication signals by or from the Antennae, then Landlord shall cause the Mast Manager to relocate such antennae or satellite dish to another area on the Antennae Mast or roof of the Building. If such interference still occurs despite such relocation and despite Landlord's diligent efforts, then Landlord shall cause the Mast Manager to promptly remove such antennae or satellite dish from the Antennae Mast or the roof of the Building.
- (G) Neither Landlord nor the Mast Manager shall have any obligation to cause the Antennae to comply with Requirements (and, accordingly, Landlord shall not be required to obtain or maintain any required permits or licenses for the Antennae; provided, however, that Landlord at Tenant's cost and expense, shall cooperate reasonably with Tenant in connection with Tenant's obtaining any such permits or licenses). Neither Landlord nor the Mast Manager shall be responsible for any damage that is caused to Tenant or the Antennae by any other tenant or occupants of the Building, or for any interference or disturbance caused to the Antennae by any equipment installed on or in the Building (subject to the provisions of clause (E) of this Section 3.9) except to the extent of the negligence or willful misconduct of Landlord, the Mast Manager, or their employees, agents or contractors. Landlord makes no representation that the Antennae will receive or transmit communication signals without interference or disturbance and Tenant agrees that Landlord shall not be liable to Tenant therefor.
- (H) The license granted to Tenant under this Section 3.9 shall automatically terminate and expire upon the expiration or earlier termination of this Lease.
- (I) The Mast Manager, at Tenant's expense, shall remove the Antennae from the Antennae Site promptly after the date that Tenant's rights under this Section 3.9 lapse or otherwise terminate. Any installment of the License Fee paid by Tenant to Landlord hereunder in respect of any period of time after the date that Tenant so removes the Antennae shall be credited by Landlord against Tenant's next installments of Rental due hereunder.

# 3.10. <u>Wireless Internet Service.</u>

Tenant has the right to install wireless Internet voice and data systems and equipment in the Premises using a provider selected by Tenant. Tenant shall not solicit other occupants of the Building (i.e., outside the Premises) to use wireless Internet service that emanates from the Premises. Tenant shall not permit the signals of Tenant's wireless Internet service (if any) to emanate beyond the Premises in a manner that interferes in any material respect with any Building Systems or with any other occupant's use of other portions of the Building.

# 3.11. <u>Telecommunications Provider/Telephone Closets.</u>

Tenant, at Tenant's cost and expense, shall have the right to use a telecommunications provider selected by Tenant. Tenant agrees that Landlord shall have the right to access (in accordance with the terms of Article 9 hereof) the telephone and electric closets that comprise a portion of the Premises and that are shown on Exhibit "3.11" attached hereto and made a part hereof for purposes of Landlord's accessing Landlord's equipment contained therein (including

fire alarm and security installations and telecommunications risers). Tenant agrees not to use, move, or otherwise disturb Landlord's equipment located in such telephone closets.

## 3.12. Access to Beacon Court.

- (A) Tenant shall have the right to use, in common with the other occupants of the Building, at least one (1) access lane in Beacon Court, solely for the purpose of discharging and picking up passengers. Tenant shall have such right to the use of Beacon Court twenty-four (24) hours per day, seven (7) days per week. Tenant acknowledges that Tenant's right to use Beacon Court as described in this Section 3.12 shall be subject to the Rules. Tenant acknowledges that Tenant has no right to use the lobbies of the Building that open onto Beacon Court (it being understood that Tenant has the right to use only the lobby of the Building that opens onto Lexington Avenue) except to the extent provided in the Access Agreement (the "Access Agreement"), dated as of the date hereof, among Landlord, Bloomberg and Tenant, a copy of which is annexed as Exhibit "3.12" attached hereto and made a part hereof. Landlord shall use due diligence to (and shall use due diligence to cause the Condominium to) keep the access lane of Beacon Court reasonably free of traffic and generally available for Tenant's use. Landlord shall engage, or cause the Condominium to engage, an attendant to assist with traffic flow through to the extent reasonably necessary.
- (B) If Bloomberg fails to provide Tenant with access to the Beacon Court Lobby as aforesaid in violation of the Access Agreement, then Landlord shall use reasonable efforts to cause Bloomberg to comply with the provisions of the Access Agreement, it being agreed, however, that Landlord shall have no obligation to institute a proceeding against Bloomberg in connection therewith. Tenant shall reimburse Landlord, from time to time, for the reasonable out-of-pocket costs and expenses incurred by Landlord in connection with Landlord's using such reasonable efforts to cause Bloomberg to comply with its obligations under the Access Agreement. Landlord shall include in any such request for reimbursement reasonable supporting information therefor. Landlord shall consult from time to time with Tenant in connection with Landlord's using such efforts to cause Bloomberg to comply with the provisions of the Access Agreement as contemplated by this Section 3.12.

## 3.13. <u>Lobby Turnstiles.</u>

(A) Subject to the terms of this Section 3.13, Landlord acknowledges that Tenant shall have the right to install in the lobby of the Building, in the location shown on Exhibit "3.13" attached hereto and made a part hereof, a turnstile system to limit access to the elevator bank that serves the Premises. Tenant shall not have the right to install such turnstile system prior to obtaining Landlord's consent to the specifications therefor, which consent shall not be unreasonably withheld, conditioned or delayed. Tenant shall perform such installation in accordance with the terms of Section 3.13 hereof (including, without limitation, the provisions thereof that require Tenant to obtain Landlord's prior consent to such installation, which consent Landlord shall not unreasonably withhold, condition or delay). If (x) Tenant installs such turnstile system in the lobby of the Building as contemplated by this Section 3.13, and (y) at any time during the Term, this Lease does not demise all of the Rentable Area that is served by such elevator bank, then Tenant shall make reasonable arrangements with Landlord and the occupants of the Rentable Area that is served by such elevator bank to provide Landlord and such

occupants with access to the turnstile system at all times. Landlord and Tenant shall use reasonable efforts to arrange for each tenant that occupies any Rentable Area served by such elevator bank to use a security/identification card that coordinates with Tenant's security system (and which would operate the turnstile system installed by Tenant).

(B) From and after the date that Tenant no longer meets the Minimum Demise Requirement, Landlord, at Landlord's cost and expense, shall have the right to remove the turnstile system installed by Tenant. If Landlord so removes the turnstiles, then Landlord, at Landlord's cost and expense, shall nonetheless be required to provide a security system or procedure reasonably designed to prevent unauthorized access into the passenger elevators that serve the Premises that is comparable to the security systems or procedures used or employed by landlords of comparable first-class buildings located in midtown Manhattan.

## 3.14. <u>Fitness Center.</u>

- (A) Subject to the terms of this Section 3.14, Tenant acknowledges that Landlord intends to construct (or to cause to be constructed) a fitness center (the "Fitness Center") on the twenty-ninth (29<sup>th</sup>) floor of the Building for use by the Residential Unit owners. Landlord shall construct (or cause to be constructed) the Fitness Center in accordance with the plans and specifications therefor in Exhibit "3.14" attached hereto and made a part hereof (it being understood that Landlord shall have the right to alter such plans and specifications to relocate the equipment in the Fitness Center and to alter the configuration of the Fitness Center, provided that such altered plans and specifications provide for substantially the same construction methods that are contemplated by the plans and specifications that are attached hereto as Exhibit "3.14", so that, for example, if (i) the plans and specification that are attached hereto as Exhibit "3.14" provide for noise-dampening installations to accommodate certain fitness equipment, and (ii) the Fitness Center is reconfigured to relocate such fitness equipment, then Landlord shall construct (or cause to be constructed) the Fitness Center in a manner that relocates such noise-dampening installations to reflect the aforesaid relocation of such fitness equipment). If, after Landlord's completion of the Fitness Center, (x) Tenant notifies Landlord that noise or vibrations emanating from the Fitness Center to the portion of the Premises located on the twenty-eighth (28<sup>th</sup>) floor of the Building are disturbing beyond a *de minimis* extent Tenant's employees working in such portion of the Premises, and (y) it is determined that Landlord did not construct (or cause to be constructed) the Fitness Center in substantial accordance with the plans and specifications annexed as Exhibit "3.14" attached hereto, then Landlord, at Landlord's cost and expense and with due diligence, shall perform (or cause to be performed) in the Fitness Center the work necessary to cause the Fitness Cent
- (B) During any period of time that Tenant meets the Minimum Demise Requirement and upon prior notice from Tenant to Landlord, Tenant shall have the right for up to twelve (12) of its employees to use the Fitness Center in accordance with the rules and regulations established therefor pursuant to the Condominium Documents. Each Tenant employee that intends to use the Fitness Center shall enter into a contract with the board of managers of the Residential Unit, which contract shall govern such employee's use of the Fitness

Center (including the fees payable for such use of the Fitness Center, which fees shall be in keeping with the fees charged by owners or operators of fitness centers in first-class office Buildings in midtown Manhattan).

(C) Landlord, at Landlord's cost and expense and in a location reasonably designated by Landlord, shall install a door in the common corridor on the twenty-ninth (29<sup>th</sup>) floor of the Building in an effort to reduce the likelihood of Residential Unit owners who are using the Fitness Center from entering any of the passenger elevators on such floor that service the Premises and gaining unauthorized access to the Premises, which door shall be secured by a card reader system. Landlord shall perform such work in a good workerlike manner with due diligence from and after the Commencement Date.

#### 3.15. Floor Loads.

The floors loads for each floor of the Building that comprises the Premises (or any portion thereof) are described on Exhibit "3.15" attached hereto and made a part hereof.

# Article 4 SERVICES

## 4.1. Certain Definitions.

- (A) The term "Building Hours" shall mean the period from 08:00 am to 08:00 pm on Business Days and the period from 08:00 am to 01:00 pm on Saturdays that are not Holidays, except that with respect to the operation of the freight elevators "Building Hours" shall mean the period from 08:00 am to 05:00 pm on Business Days only.
- (B) The term "Building Systems" shall mean the service systems of the Building, including, without limitation, the mechanical, gas, steam, electrical, sanitary, HVAC, elevator, plumbing, and life-safety systems of the Building (it being understood that the Building Systems shall not include any systems that Tenant installs in the Premises as an Alteration).
  - (C) The term "HVAC" shall mean heat, ventilation and air-conditioning.
  - (D) The term "HVAC Systems" shall mean the Building Systems that provide HVAC.
  - (E) The term "Overtime Periods" shall mean any times that do not constitute Building Hours.

# 4.2. <u>Passenger Elevator Service.</u>

Subject to the terms of Article 10 hereof and this Section 4.2, Landlord shall provide Tenant with passenger elevator service for the Premises using the Building Systems therefor. From and after the Commencement Date, such passenger elevator service shall be provided by the four (4) passenger elevators depicted on Exhibit "4.2-1" attached hereto and made a part hereof. The passenger elevators shall satisfy, at a minimum, the specifications set forth on

Exhibit "4.2-2" attached hereto and made a part hereof. Tenant's use of the passenger elevators shall be in common with other occupants of the Building. Landlord shall not have the right to reprogram the passenger elevators so that they serve floors in the Building other than the twenty-first (21st) through twenty-ninth (29th) floors or reduce, other than to a *de minimis* extent, the level of service provided by the passenger elevators (but in no event below the level of service set forth on Exhibit "4.2-2" attached hereto), provided, however, that if, after the date hereof, a portion of the Premises is leased to Bloomberg (by virtue of a Recapture Termination or otherwise), then, subject to the provisions of the immediately following sentence, Landlord shall have the right to reprogram the passenger elevators so that they serve Bloomberg's main elevator lobby located on the sixth (6<sup>th</sup>) floor of the Building. Solely during the period of time that Tenant meets the Minimum Demise Requirement, Tenant shall have the right to require Landlord to program one passenger elevator to service exclusively (and not any other floors in the Building) the ground floor and the portion of the Premises located on the twenty-seventh (27<sup>th</sup>) and twenty-eighth (28<sup>th</sup>) floors of the Building (the "Exclusive Elevator"). Tenant shall have the use of the passenger elevators that service the Premises at all times, except that during Overtime Periods, Landlord have the right to have only two (2) passenger elevator in service to serve the Premises. Tenant shall use the passenger elevators only for purposes of transporting persons to and from the Premises. If (x) the Exclusive Elevator is in use pursuant to this Section 4.2, (y) Landlord recaptures two (2) or more full floors of the Premises, and (z) Tenant meets the Minimum Demise Requirement, then, at Tenant's option, Landlord shall either (i) reprogram the Exclusive Elevator to serve all of the twenty-first (21<sup>st</sup>) through twenty-ninth (29<sup>th</sup>) floors, or (ii) reprogram two (2) of

# 4.3. <u>Freight Elevator Service.</u>

(A) Subject to the terms of Article 10 hereof and this Section 4.3, Landlord shall provide Tenant with freight elevator service for the Premises using the Building Systems therefor. From and after the Commencement Date, such freight elevator service shall be provided by the freight elevator depicted on Exhibit "4.3" attached hereto and made a part hereof, in addition to the freight elevator in the Building's loading dock located at the grade level in the Building and the appurtenant truck elevator. Tenant's use of the freight elevators shall be in common with other occupants of the Building and shall be on a first-come, first served basis (i.e., no advance scheduling), subject to Tenant's right to use the freight elevators on a reserved basis during Overtime Periods. Landlord shall have the right to prescribe reasonable rules from time to time regarding the rights of the occupants in the Building (including, without limitation, Tenant) to use the freight elevators (governing, for example, the responsibility of occupants of the Building to reserve freight elevator use in advance, particularly for Overtime Periods). Tenant shall use the freight elevators in accordance with applicable Requirements. Subject to Section 4.3(B) hereof, if Tenant uses the freight elevator during Overtime Periods, then Tenant shall pay to Landlord, as additional rent, an amount calculated at the reasonable hourly rates that Landlord charges from time to time therefor, within thirty (30) days after Landlord's giving to Tenant an invoice therefor.

- (B) Landlord and Tenant acknowledge and agree that two (2) freight elevators are required to bring freight in and out of the Premises: (i) the freight elevator located at the loading dock of the Building (the "Initial Freight Car"); and (ii) the freight elevator located on lower level 3 of the Building (the "Secondary Freight Car"; the Initial Freight Car and the Secondary Freight Car being referred to herein collectively as the "freight elevators"). The hourly charge for each of the freight elevators used during Overtime Periods is an amount equal to the product of (x) one hundred forty-five percent (145%), and (y) the hourly cost of a Local 32 B-J elevator operator in effect from time to time (the "32 B-J Hourly Rate"). As of the date of this Lease, the 32B-J Hourly Rate is Fifty-Five Dollars (\$55), which, when multiplied by one hundred forty-five percent (145%), results in an hourly charge for each freight elevator used during Overtime Periods of Seventy-Nine Dollars and 75/100 Cents (\$79.75). Landlord shall have the right to charge Tenant for a particular minimum number of hours of usage of the freight elevator during Overtime Periods to the extent that the applicable union contract or service contract requires Landlord to engage the necessary personnel (including, without limitation, a freight elevator operator and loading dock attendant) for such minimum number of overtime hours (which minimum number of hours as of the date hereof is four (4)). If (x) Tenant requests Landlord to provide Tenant with freight elevator service during Overtime Periods as provided in this Section 4.3, and (y) another tenant in the Building also uses, or other tenants in the Building also use, the applicable freight elevators during such Overtime Period, then Landlord shall allocate equitably the charges described in this Section 4.3 among Tenant and such other tenant or tenants (taking into account the differences in the costs to Tenant as provided in this Section 4.3(B) hereof and the charges to such other tenant or tenants).
- (C) Tenant acknowledges that the labor unions involved in the construction of the Building may require Landlord through the date such unions are no longer actively working in the Building to operate the freight elevators in the Building using union labor personnel ("Construction Union Personnel") other than Local 32 B-J, which is the labor union that currently operates such freight elevators following the completion of Building construction. If the Construction Union Personnel are operating the freight elevators during Business Hours, then, if Tenant uses the freight elevators for construction during Business Hours, Tenant shall pay to Landlord the excess of the hourly rates paid by Landlord for the Construction Union Personnel over the 32 B-J Hourly Rate, in each case without any markup. If (i) Tenant uses the freight elevators during Overtime Periods, and (ii) Construction Union Personnel operate the freight elevators at such time, then Tenant shall pay, subject to Section 4.3(D) hereof, the hourly rates paid by Landlord to the applicable union for the Construction Union Personnel during such Overtime Periods, without any markup, in lieu of the rates set forth in Section 4.3(B) hereof (it being agreed that Landlord shall still have the right to charge Tenant for a minimum number of hours, as aforesaid, to the extent required by the union contract of the union employing the Construction Union Personnel).
- (D) Tenant shall not be required to pay for the first two hundred (200) hours of freight elevator use during Overtime Periods solely in connection with Tenant's use of the freight elevators for its Initial Alterations and Tenant's initial move into the Premises (it being understood that if Tenant uses the Initial Freight Car and the Secondary Freight Car, then Tenant shall be charged for two (2) hours of freight elevator use); provided, however, that if Construction Union Personnel are operating the freight elevators during any such Overtime Periods, then Tenant shall pay the excess of the hourly rates actually paid by Landlord to the

applicable union for the Construction Union Personnel over the 32 B-J Hourly Rate, in each case without any markup. Tenant shall nonetheless use reasonable efforts to schedule and use the freight elevators during non-Overtime Periods in connection with the performance of its Initial Alterations in an effort to minimize the number of hours of freight elevator use during Overtime Periods that would otherwise be required.

(E) Subject to Section 4.15 hereof, Tenant, at Tenant's cost and expense, shall have the right to provide Tenant's own security guard to monitor Tenant's activity solely in the freight elevators and in the aforesaid loading dock and truck elevator areas, it being expressly agreed that any such security guard shall in no event be permitted to operate a freight elevator.

## 4.4. <u>Heat, Ventilation and Air-Conditioning.</u>

- (A) Subject to the terms of Article 10 hereof and this Section 4.4, Landlord shall operate the HVAC System to provide HVAC at the perimeter of the Premises that satisfies, at a minimum, the specifications set forth on Exhibit "4.4" attached hereto and made a part hereof. Landlord shall not be required to make any installations in the Premises to distribute HVAC within the Premises. Landlord shall have no liability to Tenant if Landlord cannot meet the HVAC specifications set forth on Exhibit "4.4" attached hereto if such failure is caused by virtue of Tenant's failure to keep closed the curtains, blinds, shades or screens that Tenant installs on the windows of the Premises in accordance with the terms hereof to the extent reasonably necessary to reduce the interference of direct sunlight with the operation of the HVAC System.
- (B) Landlord shall operate the HVAC Systems at all times during Building Hours without additional charge to Tenant. Landlord shall operate the HVAC System for Tenant's benefit during Overtime Periods if Tenant so advises Landlord not later than 02:00 pm of the Business Day immediately preceding the day on which Tenant requires HVAC during Overtime Periods (it being agreed that Landlord intends to provide to Tenant, at no additional charge, software for use on Tenant's computer system, which enables Tenant to order from its computer system such HVAC during Overtime Periods and to order and monitor certain other services provided in the Building). If Landlord so provides HVAC to the Premises during Overtime Periods (as so requested by Tenant), then Tenant shall pay to Landlord, as additional rent, an amount calculated at the hourly rate of Twenty-Five Dollars and No Cents (\$25.00) for all or any floors for which Tenant orders HVAC during Overtime Periods (i.e., not a per floor charge), which amount shall be adjusted on January 1st of each calendar year occurring during the Term commencing on January 1, 2006 to equal an amount equal to the product obtained by multiplying (I) Twenty-Five Dollars (\$25), by (II) a fraction, the numerator of which is the Consumer Price Index for November of the immediately preceding calendar year, and the denominator of which is the Consumer Price Index on the date hereof, within thirty (30) days after Landlord gives to Tenant an invoice therefor. If (i) Tenant requests HVAC during a particular Overtime Period, and (ii) such HVAC is automatically provided to other leaseable areas in the Building by virtue of the HVAC zoning in the Building, then the aforesaid charge for HVAC shall be appropriately prorated between Tenant and any occupant or occupants of such leaseable area who also ordered HVAC during such Overtime Period (or any portion thereof).

# 4.5. <u>Cleaning.</u>

- (A) Subject to the terms of Article 10 hereof and this Section 4.5, Landlord shall cause the Premises to be cleaned substantially in accordance with the standards set forth in Exhibit "4.5" attached hereto and made a part hereof. Landlord shall not be required to clean the portions of the Premises (if any) that Tenant uses for the storage, preparation, service or consumption of food or beverages. Tenant shall pay to Landlord, as additional rent, the actual costs paid by Landlord to a third party in removing from the Building any of Tenant's refuse and rubbish to the extent exceeding the amount of refuse and rubbish usually generated by a tenant that uses the Premises for ordinary office purposes. Tenant shall make such payments to Landlord not later than the thirtieth (30th) day after the date that Landlord gives to Tenant an invoice therefor from time to time, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon. If the cleaning specifications set forth on Exhibit "4.5" attached hereto are not consistently and materially met by Landlord's cleaning contractor, then Landlord, upon Tenant's request therefor from time to time, shall cause its Building management personnel to meet with Tenant in an effort to develop a solution so that the cleaning specifications are thereafter consistently and materially met.
- (B) Tenant, at Tenant's expense, shall exterminate the portions of the Premises that Tenant uses for the storage, preparation, service or consumption of food against infestation by insects and vermin regularly and, in addition, whenever there is evidence of infestation. Tenant shall engage Persons to perform such exterminating that are approved by Landlord, which approval Landlord shall not unreasonably withhold, condition or delay. Tenant shall cause such Persons to perform such exterminating in a manner that is reasonably satisfactory to Landlord.
- (C) Tenant, at Tenant's expense, shall clean daily all portions of the Premises used for the storage, preparation, service or consumption of food or beverages. Tenant shall not have the right to perform any cleaning services in the Premises using any contractor other than the cleaning contractor that Landlord has engaged from time to time to perform cleaning services in the Building for Landlord; provided, however, that (x) Landlord shall not have the right to require Tenant to use such cleaning contractor unless the rates that such cleaning contractor agrees to charge Tenant for such additional cleaning services are competitive with the amounts charged by other cleaning companies that provide services to comparable buildings for comparable services, and (z) subject to Section 4.15 hereof, Tenant shall have the right to use Tenant's own employees for such additional cleaning services. If such cleaning contractor does not agree to charge Tenant for such additional cleaning services at competitive rates as aforesaid, then Tenant may employ to perform such additional cleaning services another cleaning contractor that Landlord approves, which approval Landlord shall not unreasonably withhold, condition or delay. Subject to Section 4.15 hereof, Landlord and Tenant shall cooperate with one another in ensuring that the cleaning performed by any cleaning contractor selected by Tenant in accordance with this Section 4.5, does not result in labor disharmony with Landlord's cleaning contractor.
- (D) Landlord consents to Tenant using no more than four (4) Tenant employees (or four (4) employees of Tenant's facilities manager) to perform certain services in the Premises for Tenant's senior executives, which services shall be limited to setting up offices

and conference rooms for meetings and the cleaning of same, stocking pantries, serving beverages, and other similar and related services, as long as the provision of such services does not result in any labor disharmony. Tenant agrees that it shall be required to use Local 32 B-J personnel employed by Landlord's cleaning contractor to perform certain handyman type services in the Premises, such as hanging pictures, performing minor carpentry repair work, etc., which services shall be ordered and paid for by Tenant. Landlord shall use reasonable efforts to ensure that Tenant's needs with respect to such services are promptly met taking into account the needs of other occupants of the Building at any given time. If Tenant's needs for handyman services in the Premises are such that Tenant requires a dedicated person or persons, then Tenant can hire one or more of the Local 32 B-J personnel employed by Landlord's cleaning contractor. Landlord shall assist Tenant in selecting any such union personnel and shall perform any testing and background checks to the full extent that is permitted by Local 32 B-J's then current union contract.

- (E) Tenant's cleaning contractors that meet the requirements of this Article 4 shall be permitted without charge to use the janitorial closets located on each floor of the Premises and the passenger and freight elevators in the Building (but may not use the freight elevators during Overtime Periods). Landlord shall use reasonable efforts to cause any cleaning contractor that cleans the Premises (other than cleaning contractors selected by Tenant) to keep the entry doors to the Premises locked when such contractor is cleaning the Premises and to turn off the lights in each portion of the Premises when such contractor has finished cleaning such portion of the Premises (to the extent that the lights in the Premises do not automatically shut-off).
- (F) Tenant shall comply with any refuse disposal program (including, without limitation, any waste recycling program) that Landlord imposes reasonably upon not less than twenty (20) days of prior written notice from Landlord to Tenant or that is required by Requirements.
- (G) Tenant shall not clean any window in the Premises, nor require, permit, suffer or allow any window in the Premises to be cleaned, in either case from the outside in violation of Section 202 of the New York Labor Law, any other Requirement, or the rules of the Board of Standards and Appeals, or of any other board or body having or asserting jurisdiction.

## 4.6. Water.

Landlord shall provide to the lavatories located in the portion of the Premises that is within the core of the Building hot and cold water only for ordinary drinking, cleaning and lavatory purposes at no additional charge to Tenant. Landlord shall also provide, through the Building Systems and a copper supply line that is two (2) inches in diameter, cold water at one (1) connection point at the perimeter of the Premises on each floor of the Building for use by Tenant for its cold water needs outside of the core of the Building (at no additional charge to Tenant) and for use in any cafeteria in the Premises constructed by Tenant (which may be metered pursuant to this Section 4.6 and paid for by Tenant). Landlord shall not be required to make any installations in the Premises to distribute water within the Premises. Tenant, at Tenant's cost and expense and subject to there then being sufficient capacity that will not otherwise adversely affect the other occupants of the twenty-ninth (29<sup>th</sup>) floor of the Building,

shall have the right to tap-in to the water heaters to the extent located in the Premises solely for use in Tenant's private restrooms (it being agreed that any such tap-in shall be deemed an Alteration that shall be governed in accordance with Article 7 hereof). Nothing contained in this Section 4.6 limits the provisions of Article 10 hereof. Landlord, at Tenant's cost and expense, shall have the right to install a meter to measure Tenant's consumption of water used by Tenant for any cafeteria in the Premises at Landlord's cost without any markup. Tenant, at Tenant's cost and expense, shall maintain and repair any such meter. Tenant shall pay at Landlord's cost without any markup for water provided to any such cafeteria as measured by such meter, as additional rent hereunder, on or prior to thirty (30) days after Landlord's rendition of a statement therefor (which statements shall not be given to Tenant more frequently than monthly and shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon).

## 4.7. <u>Building Employee Listing.</u>

Tenant has the right to list the names of Tenant and any Permitted Party and their respective officers, directors and employees on the computerized Building employee listing which shall be located behind the security desk in the lobby of the Building and shall be accessible only by Building personnel. At Landlord's sole cost and expense, Landlord shall reprogram such listing to add or delete names of the executive personnel or Permitted Parties promptly after Tenant's request from time to time, except that Tenant shall not have the right to make any such request more frequently than once in any particular period of thirty (30) days.

## 4.8. <u>Building Security.</u>

- (A) Subject to the terms of this Section 4.8(A), Landlord shall provide security for the Building that is consistent with the security provided by landlords of first-class buildings in midtown Manhattan. Tenant acknowledges that Landlord, in agreeing to arrange for such security, does not ensure the security of the Building, and accordingly, Tenant remains responsible for making the Alterations in the Premises that Tenant considers adequate to provide for Tenant's security.
- (B) Landlord and Tenant shall each give the other prompt notice of any advisories, notices or similar communications received by Landlord or Tenant, as the case may be, from a Governmental Authority specifically in respect of the Building and which relates to the security therefor, unless Landlord or Tenant, as the case may be, is prohibited by such Governmental Authority from disclosing the contents of such advisory, notice, or communication. Tenant shall have the right to designate a representative to act as Tenant's security liaison with the Building management office (or another representative designated by Landlord) regarding security matters affecting the Building and the Premises. Landlord's representative shall reasonably cooperate with Tenant's representative on an ongoing basis during the Term with respect to sharing information about Building security, coordinating fire drills and other related security procedures, and planning for special security needs with respect to particular events or visits by third parties to the Premises (or visits by third parties to the Building that may impact Tenant's security in the Premises). Upon Tenant's written request therefor from time to time, Landlord shall cause its Building personnel to meet with Tenant and its representatives to keep Tenant advised as to the status of the Building's security specifications and the strategies to be employed by Landlord and the Condominium Board in connection

therewith. Landlord agrees to act in good faith in implementing (or causing the Condominium Board to implement) Tenant's reasonable recommendations with respect to Building security and Tenant shall pay Landlord for actual increases in Landlord's security costs by virtue of Landlord implementing a Tenant security recommendation (to the extent such increase is not properly includable in Operating Expenses). During any period of time that Tenant meets the Minimum Demise Requirement, Landlord shall give Tenant written notice of any impending material change in the Building's security system or procedures, which notice shall be given to Tenant promptly after Landlord (or the Condominium) makes its decision to effect such change.

- (C) Tenant, at Tenant's cost and expense, shall have the right to engage one (1) security guard (in addition to the employee posted at the desk in the Building lobby) to patrol the Premises and the lobby of the Building, provided that Landlord has the right to designate the uniform worn by such security guard (so that such guard's uniform is consistent with the uniforms worn by other security guards in the Building). Landlord, at no cost to Landlord, shall cooperate with Tenant in connection with Tenant's operating its security program (for example, in coordinating Tenant's evacuation drills). Landlord shall use reasonable efforts to cause the security cards issued by Tenant for its employees for purposes of such employees accessing the Premises (or particular portions thereof) to also be operable with respect to other security checkpoints in the Building.
- (D) Subject to the terms of this Section 4.8(D), Tenant or a Permitted Party that sublets from Tenant at least eighty-eight Thousand (88,000) square feet of Rentable Area shall have the right to post one (1) of its own employees (together with a telephone and computer terminal or other similarly sized equipment reasonably related to Tenant's posting of such employee) at one of the stations at the lobby desk in the Building (as designated by Landlord), at Tenant's cost, provided that Landlord has the right to designate the uniform worn by such employee (so that such employee's uniform is consistent with the uniforms worn by other Building personnel located at the lobby desk in the Building). Tenant or any such Permitted Party shall have the right to post such employee and install such equipment only during the period that the Minimum Demise Requirement is satisfied.

# 4.9. <u>Fire System.</u>

Landlord, at Landlord's cost and expense, will maintain the Building's core fire alarm system. Landlord will provide Tenant with access to the Building System that provides water for the fire suppression system that exists in the Premises on the Commencement Date or that Tenant installs during the Term as an Alteration in accordance with the terms hereof in either case in amounts and at pressures that satisfy applicable Requirements, which connection points are shown on Exhibit "4.9" attached hereto and made a part hereof. Landlord will provide Tenant with a connection point or connection points to the Building System that constitutes the fire alarm system in the Building to the extent required by applicable Requirements. Nothing contained in this Section 4.9 obligates Landlord to make any installations in the Premises to distribute within the Premises such water for such fire suppression system or to install elements of the fire alarm system within the Premises.

# 4.10. Gas.

Tenant, at Tenant's cost and expense, shall have the right to bring gas service to the Premises by directly contracting with the applicable utility company. Tenant shall pay directly to the utility company the cost of such gas. Tenant, at Tenant's expense, shall make any additional installations (subject the provisions of Article 7 hereof) that are required for Tenant to obtain gas from the utility company and shall comply with all applicable Requirements in connection therewith. Landlord, at no cost to Landlord, shall reasonably cooperate with Tenant in connection with Tenant's obtaining such direct service. Landlord shall not be required to maintain or repair during the Term any installations that exist in the Premises with respect to such gas, provided that the foregoing shall not be deemed to limit Landlord's other obligations set forth in Article 8 hereof.

#### 4.11. Steam.

Subject to this Section 4.11, Tenant shall have the right to tap-in to the low-pressure steam system for the Building at the main steam riser on each floor of the Building where the Premises are located for steam service for the Premises. Tenant shall not have the right to tap-in to such low-pressure steam system if, at such time, the Residential Unit owners in the Building would not have sufficient steam available to them by virtue of Tenant's use of such steam. If Tenant elects to have such access and such access does not adversely affect the Residential Unit owners as aforesaid, then Landlord shall not be required to make any installations in the Premises to distribute such steam within the Premises. Landlord shall not be required to maintain or repair during the Term any installations that exist in the Premises, provided that the foregoing shall not be deemed to limit Landlord's other obligations set forth in Article 8 hereof. Tenant's use of steam shall be measured by a meter installed by Tenant at Tenant's expense (which meter shall measure exclusively the steam to be provided to the Premises in accordance with the provisions of this Section 4.11) and Tenant shall pay to Landlord on a monthly basis, as additional rent ("Steam Additional Rent"), an amount equal to one hundred three percent (103%) of the actual charges that Landlord pays to the applicable utility company in connection with providing such steam to Tenant, as measured by such meter. Tenant shall pay the Steam Additional Rent to Landlord on or prior to the thirtieth (30th) day after the date that Landlord gives to Tenant an invoice therefor, which shall be accompanied by Landlord's calculation of the Steam Additional Rent and Landlord's meter readings upon which such calculations are based. Tenant shall have the right to object to Landlord's calculation of the Steam Additional Rent by giving Landlord notice of any such objection on or prior to the sixtieth (60th) day after the date that Landlord gives Tenant the applicable invoice for the Stea

## 4.12. Operation of Building Systems.

Landlord shall (x) maintain the Office Unit and the Building Systems that serve the Premises, (y) cause the Condominium Board to maintain the Building (including the common areas thereof) to the extent required pursuant to the Condominium Documents, and (z) cause any Person that is Landlord's Affiliate and that owns a condominium unit in the Building to maintain

such unit, in any such case, in conformity with the standards maintained by landlords of first-class buildings in midtown Manhattan. Landlord, upon receipt of notice from Tenant from time to time, shall give Tenant the opportunity to review and comment upon operating procedures for Building Systems from which Tenant benefits, it being expressly agreed that Landlord shall have no obligation to modify its procedures based upon Tenant's comments.

## 4.13. Supplemental HVAC System.

- (A) Subject to the terms of Article 10 hereof and this Section 4.13, Tenant, at Tenant's expense, may tap into the applicable Building System to obtain chilled water for a supplemental air-conditioning system (or systems) that Tenant installs in the Premises in accordance with the provisions of Article 7 hereof. Such chilled water shall be available to Tenant twenty-four (24) hours per day, seven (7) days per week, subject to Landlord's right to interrupt the provision of chilled water to Tenant in accordance with Section 10.2 hereof, and shall be supplied at a temperature that ranges from thirty-nine degrees Fahrenheit to forty-five degrees Fahrenheit. Any such supplemental air-conditioning system that Tenant installs shall not have a capacity of more than two hundred (200) tons in the aggregate. Any installations that are required to connect Tenant's supplemental air-conditioning system to the chilled water pipes shall be made by Tenant (based on the plans and specifications therefor as designed by Tenant and approved by Landlord as contemplated by Article 7 hereof). Tenant shall pay to Landlord, as additional rent:
  - (1) the reasonable, third party, out-of-pocket costs that Landlord incurs in making such connection;
  - (2) a one-time 'tap-in' fee in the amount of Two Hundred-Fifty Dollars and No Cents (\$250.00) per ton of capacity so connected; and
- (3) an annual charge in the amount of Three Hundred Fifty Dollars and No Cents (\$350.00) per ton of capacity of the system so connected (which amount per ton shall be increased on each anniversary of the Commencement Date to reflect the percentage increase, if any, in the Consumer Price Index from the Consumer Price Index that is in effect on the Commencement Date).

Tenant shall pay such amounts to Landlord on or prior to the thirtieth (30th) day after the date that Landlord gives to Tenant an invoice therefor from time to time.

(B) From and after the date that Tenant no longer meets the Minimum Demise Requirement, Tenant shall only be entitled to obtain chilled water to meet a tonnage capacity equal to the product of (x) two hundred (200), and (y) a percentage, the numerator of which is the number of square feet of Rentable Area leased by Tenant at the time that Tenant no longer meets the Minimum Demise Requirement, and the denominator of which is one hundred seventy-six thousand (176,000).

# 4.14. <u>No Other Services.</u>

Landlord shall not be required to provide any services to support Tenant's use and occupancy of the Premises, except to the extent expressly set forth herein.

## 4.15. <u>Labor Harmony.</u>

If (i) Tenant employs, or permits the employment of, any contractor, mechanic or laborer in the Premises, whether in connection with any Alteration or otherwise, (ii) such employment interferes or causes any conflict with other contractors, mechanics or laborers engaged in the maintenance, repair, management or operation of the Building, and (iii) Landlord gives Tenant notice thereof (which notice may be given verbally to the person employed by Tenant with whom Landlord's representative ordinarily discusses matters relating to the Premises), then Tenant shall cause all contractors, mechanics or laborers causing such interference or conflict to leave the Building immediately and shall take such other action as may be reasonably necessary to resolve such conflict.

# Article 5 ELECTRICITY

# 5.1. Capacity.

- (A) Subject to the terms of this Article 5, Landlord shall provide to the electrical closet on the floor of the Building where each portion of the Premises is located, for Tenant's use, eight (8) watts of electrical capacity (demand load) per square foot of Usable Area in the portion of the Premises located on such floor of the Building (exclusive of the electrical capacity that is required to operate the HVAC System) (such electrical capacity, as such electrical capacity may be increased pursuant to Section 5.1(B) hereof, being referred to herein as the "Base Electrical Capacity"). Tenant, during the Term, shall use electricity in the Premises only in such manner that complies with the requirements of the Utility Company. Tenant shall not permit the demand for electricity in the Premises to exceed the Base Electrical Capacity. Tenant, subject to applicable Requirements, shall have the right to distribute excess electrical capacity from one floor of the Premises to another floor or floors that comprise the Premises.
- (B) Subject to the terms of this Section 5.1(B), Tenant, from time to time, shall have the right to request that Landlord increase the electrical capacity set forth in Section 5.1(A) by making available to Tenant excess electrical capacity that is then available in the Building to the extent Tenant has a bonafide need for such additional electrical capacity. If (x) Landlord has a bona-fide need for such available Building electricity, or (y) Landlord has theretofore promised such electricity to another tenant or occupant in the Building that has a bona-fide need for such available Building electricity and, in either case, such electricity is not available to Tenant, then Landlord will so inform Tenant and thereafter will cooperate with Tenant, at no cost to Landlord, to make arrangements with the Utility Company to increase the electrical capacity that the Utility Company makes available to the Building, in which case the Base Electrical Capacity shall increased correspondingly. Tenant shall perform any work that is required in connection with any such increase in electrical capacity in accordance with the terms of Article 7 hereof (as if such work constituted an Alteration). Nothing contained in this Section 5.1(B) expands the Premises or otherwise grants to Tenant rights to use portions of the Building that are not otherwise demised to Tenant hereunder.

## 5.2. <u>Electricity for the Building.</u>

Landlord shall arrange with a Utility Company to provide electricity for the Building. Landlord shall not be liable to Tenant for any failure or defect in the supply or character of electricity furnished to the Building, except to the extent that such failure or defect results from the negligence or willful misconduct of Landlord, its employees, agents or contractors. Landlord shall not be required to make any installations in the Premises to distribute electricity within the Premises. If Landlord elects to solicit bids for electricity to the Building from energy providers other than the Utility Company, then Tenant shall have the right to require Landlord to solicit bids from any reliable energy providers suggested by Tenant, it being understood that Landlord has no obligation whatsoever to award the contract to the energy provider or energy providers suggested by Tenant. Landlord shall pass on to Tenant a proportionate share of any tax savings on the electricity charges for the Building for which Landlord has qualified. Landlord, at no cost to Landlord, shall reasonably cooperate with Tenant in Tenant's efforts to obtain rebates or discounts for electricity from the Utility Company by virtue of Tenant's installation in the Premises of energy-saving fixtures or equipment. Tenant shall receive (either by payment from Landlord to Tenant or by virtue of Tenant's installation in the Premises of any such energy-saving fixtures or equipment. Tenant shall not be entitled to any benefits or rebates granted to another tenant or occupant in the Building by virtue of such tenant or occupant's installation of energy-saving fixtures or equipment.

## 5.3. <u>Submetering.</u>

- (A) Subject to the provisions of this Section 5.3, Landlord shall measure Tenant's consumption of electricity (but not Tenant's demand for electricity) in the Premises using submeters that have heretofore been installed by Landlord, at Landlord's cost and expense. Landlord, at Landlord's cost and expense, shall maintain the submeters in the Premises. If, at any time during the Term, Tenant performs Alterations that require modifications to the aforesaid submeter or submeters that Landlord installs, or that require a supplemental submeter or supplemental submeters, then Tenant shall perform such modification, or the installation of such supplemental submeter or submeters, at Tenant's cost, as part of the applicable Alteration. Upon reasonable oral notice from Tenant from time to time, Landlord shall grant Tenant access to the submeters for purposes of Tenant's verifying Landlord's submeter readings.
- (B) Tenant shall pay to Landlord, as additional rent, an amount (the "Electricity Additional Rent") equal to one hundred three percent (103%) of the product obtained by multiplying (x) the Average Cost per Kilowatt Hour, by (y) the number of kilowatt hours of electricity used in the Premises for the applicable billing period, as registered on the submeter or submeters for the Premises. The term "Average Cost per Kilowatt Hour" shall mean, with respect to any particular period, the quotient obtained by dividing (x) the aggregate charge imposed by the Utility Company for the electricity supplied during such period to the portions of the Building in which the Premises are located as measured by the Utility Company meters that measure the electricity supplied to such portions of the Building from time to time (including, without limitation, energy charges, demand charges, all applicable surcharges, time-of-day charges, fuel adjustment charges, rate adjustment charges, taxes and any other factors used by

the Utility Company in computing its charges to Landlord), by (y) the number of kilowatt hours of electricity used in the Building during such period, as reflected on the electric meter or meters for the Building. The term "<u>Utility Company</u>" shall mean, collectively, the local electrical energy distribution company and the competitive energy provider that provides electric service for the Building.

- (C) Landlord shall give Tenant an invoice for the Electricity Additional Rent on a monthly basis, which invoice shall have annexed thereto a copy of the applicable invoice from the Utility Company and the calculation of the aggregate amount set forth on such invoice. Tenant shall pay the Electricity Additional Rent to Landlord on or prior to the thirtieth (30th) day after the date that Landlord gives to Tenant each such invoice. Tenant shall not have the right to object to Landlord's calculation of the Electricity Additional Rent unless Tenant gives Landlord notice of any such objection on or prior to the sixtieth (60th) day after the date that Landlord gives Tenant the applicable invoice for the Electricity Additional Rent. If Tenant gives Landlord a notice objecting to Landlord's calculation of the Electricity Additional Rent, as aforesaid, then Tenant shall have the right to review Landlord's submeter readings and Landlord's calculation of the Electricity Additional Rent, at Landlord's offices or, at Landlord's option, at the offices of Landlord's managing agent, in either case at reasonable times and on reasonable advance notice to Landlord. Either party shall have the right to submit a dispute regarding the Electricity Additional Rent to an Expedited Arbitration Proceeding.
- (D) Subject to the terms of this Section 5.3(D), Tenant, at any time during the Term, shall have the right to make arrangements to obtain electricity directly from the Utility Company or, at Tenant's option, from any other reputable provider, by giving notice thereof to Landlord. If Tenant makes the election as described in this Section 5.3(D), then (i) Tenant, at Tenant's cost, shall make arrangements to obtain electricity directly from the Utility Company or such other provider, as the case may be, (ii) from and after the date on which such direct electricity is provided to Tenant, Landlord shall not be obligated to furnish electricity to the Premises, and (iii) from and after the date on which such direct electricity is provided to Tenant, Tenant shall not be obligated to pay to Landlord the charges for electricity as described in this Article 5. Tenant shall make any installations that are required to accomplish Tenant's obtaining electricity directly from the Utility Company in accordance with the provisions of Article 7 (as if such installations constituted an Alteration hereunder), it being agreed that Tenant shall have the right to use the electrical facilities that then exist in the Building to obtain such direct electric service (without Landlord having any liability or obligation to Tenant in connection therewith). Nothing contained in this Section 5.3(D) gives Tenant the right to use any portion of the Building (except as otherwise set forth in this Lease). Landlord, at no cost to Landlord, shall reasonably cooperate with Tenant in connection with Tenant's obtaining such direct service.

## 5.4. <u>Termination of Electric Service.</u>

(A) If Landlord is required by any Requirement to discontinue furnishing electricity to the Premises as contemplated hereby, then this Lease shall continue in full force and effect and shall be unaffected thereby, except that from and after the effective date of any such Requirement, (x) Landlord shall not be obligated to furnish electricity to the Premises, and (y) Tenant shall not be obligated to pay to Landlord the charges for electricity as described in this Article 5.

- (B) If Landlord discontinues Landlord's furnishing electricity to the Premises pursuant to a Requirement, then Tenant shall use Tenant's diligent efforts to obtain electricity for the Premises directly from the Utility Company or, at Tenant's option, any other reputable provider. Tenant shall pay directly to the Utility Company the cost of such electricity. Tenant shall have the right to use the electrical facilities that then exist in the Building to obtain such direct electric service (without Landlord having any liability or obligation to Tenant in connection therewith except to the extent of the negligence or willful misconduct of Landlord, its employees, agents or contractors). Nothing contained in this Section 5.4 shall permit Tenant to use electrical capacity in the Building that exceeds the Base Electrical Capacity (as such Base Electrical Capacity may be increased pursuant to this Article 5). Tenant, at Tenant's expense, shall make any additional installations that are required for Tenant to obtain electricity from the Utility Company. Landlord shall reasonably cooperate with Tenant in connection with Tenant's obtaining such direct service. Landlord, at no cost to Landlord, shall reimburse Tenant for fifty percent (50%) of Tenant's reasonable, out-of-pocket costs in obtaining such direct electricity on or prior to thirty (30) days after Tenant's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon (it being agreed that if Landlord fails to so reimburse Tenant on or prior to thirty (30) days after Tenant's rendition of a statement therefor, then Tenant shall have the right to offset the amount of the costs set forth in such statement against the Rental due hereunder).
- (C) Landlord shall not discontinue furnishing electricity to the Premises as contemplated by this Section 5.4 (to the extent permitted by applicable Requirements) until Tenant obtains electric service directly from the Utility Company.

# Article 6 INITIAL CONDITION OF THE PREMISES

#### 6.1. Condition of Premises.

- (A) Subject to Section 8.1 and Section 6.1(B) hereof, (a) Tenant shall accept possession of the Premises in the condition that exists on the Commencement Date "as is," and (b) Landlord shall have no obligation to perform any work or make any installations in order to prepare the Building or the Premises for Tenant's occupancy. Except as expressly set forth herein, Landlord has made no representations or promises with respect to the Building, the Real Property or the Premises.
- (B) With respect to the portion of the Premises that is comprised of the twenty-second (22<sup>nd</sup>) and twenty-third (23<sup>rd</sup>) floors of the Building (such portion of the Premises being referred to herein as the "Post-Possession Premises"), Landlord, at its expense, shall cause the work (the "Landlord's Work") described on Exhibit "6.1(B)" attached hereto to be Substantially Completed therein on or prior to January 15, 2005 (or such later date if Tenant notifies Landlord that Tenant intends to commence its performance of the Initial Alterations in the Post-Possession Premises on a later date, it being agreed that Tenant shall notify Landlord prior to January 15, 2005 of the approximate date upon which Tenant intends to commence the performance of its Initial Alterations in the Post-Possession Premises if on a later date as aforesaid). Landlord shall perform the Landlord's Work in accordance with all applicable

Requirements. Landlord shall perform the Landlord's Work in a good and workerlike manner. If Landlord fails to Substantially Complete the Landlord's Work on or prior to January 15, 2005 (or such later date set forth in Tenant's notice as aforesaid) and such failure actually delays Tenant in the performance of its Initial Alterations in the Post-Possession Premises (or a portion thereof), then, for each day that Tenant is so delayed in the performance of its Initial Alterations in the Post-Possession Premises (or the applicable portion thereof), Tenant shall be entitled a one (1) day abatement of the Rental due hereunder from and after the Rent Commencement Date for the Post-Possession Premises (or the applicable portion thereof) in which Tenant is so delayed in the performance of its Initial Alterations) (it being agreed that if, on any particular day, Tenant is delayed in the performance of its Initial Alterations in the Post-Possession Premises by virtue of a Landlord default and/or an Impeding Building Violation pursuant to Section 7.18 hereof and/or a delay pursuant to this Section 6.1(B) hereof, then Tenant shall nonetheless only be entitled to a one (1) day abatement of Rental by virtue of any such delay on such particular day).

(C) Landlord hereby represents and warrant to Tenant that to the actual knowledge of the employees of Landlord or Landlord's development manager who have participated in the development of the Building, there do not exist in the Premises any hazardous substances to an extent that violates applicable Requirements (including, without limitation, the Comprehensive Environmental Response Compensation and Liability Act).

# Article 7 ALTERATIONS

# 7.1. <u>General.</u>

- (A) Except as otherwise provided in this Article 7, Tenant shall not make any Alterations without Landlord's prior consent.
- (B) The term "Alterations" shall mean alterations, installations, improvements, additions or other physical changes (other than decorations such as painting, wall coverings and floor coverings) in each case in or to the Premises that are made by or on behalf of Tenant or any other Person claiming by, through or under Tenant.
  - (C) The term "Initial Alterations" shall mean the Alterations to prepare the Premises for Tenant's initial occupancy.
- (D) The term "Specialty Alterations" shall mean Alterations that (i) perforate by more than six (6) inches in diameter a floor slab in the Premises, (ii) consist of the installation of a vault or other similar device or system that is intended to secure the Premises or a portion thereof in a manner that exceeds the level of security that a reasonable Person uses for ordinary office space, or (iii) consist of the installation of internal staircases.
- (E) The term "Substantial Completion" or words of similar import shall mean that the applicable work has been substantially completed in accordance with the applicable plans and specifications, if any, it being agreed that (i) such work shall be deemed substantially complete notwithstanding the fact that minor or insubstantial details of construction or demolition, mechanical adjustment or decorative items remain to be performed, and (ii) with

respect to work that is being performed in the Premises, such work shall be deemed substantially complete only if the incomplete elements thereof do not interfere materially with Tenant's use and occupancy of the Premises for the conduct of business.

(F) The term "Tenant's Property" shall mean Tenant's personal property (other than non-movable fixtures), including, without limitation, Tenant's movable fixtures, movable partitions, telephone and computer equipment, furniture, furnishings and movable decorations.

#### 7.2. <u>Basic Alterations and Material Alterations.</u>

- (A) Except as provided in Section 7.2(B) hereof, Tenant shall not be required to obtain Landlord's consent for a proposed Basic Alteration. The term "Basic Alteration" shall mean an Alteration that (i) does not affect the exterior (including the appearance of) the Building, (ii) does not affect adversely any part of the Building other than the Premises other than to a *de minimis* extent, (iii) does not require any alterations, installations, improvements, additions or other physical changes to be performed in or made to any portion of the Building other than the Premises, (iv) does not affect adversely the proper functioning of any Building System other than to a *de minimis* extent, (v) does not affect the structure of the Building, (vi) does not impede Landlord's access to Reserved Areas in any material respect, and (vii) does not violate or render invalid the certificate of occupancy for the Building or any part thereof (any Alteration that does not constitute a Basic Alteration shall be a "Material Alteration").
- (B) Tenant shall be required to obtain Landlord's prior consent to a particular Basic Alteration, which consent shall not be unreasonably withheld, conditioned or delayed if the sum of (X) the "hard" construction cost of such Basic Alteration, and (Y) the "hard" construction cost of any other Basic Alterations performed during the immediately preceding period of twelve (12) months without Landlord's consent as contemplated by this Section 7.2, exceeds the Basic Alterations Threshold. The term "Basic Alterations Threshold" shall mean Two Hundred Fifty Thousand Dollars (\$250,000), except that on each anniversary of the Commencement Date, the Basic Alterations Threshold shall be adjusted to reflect the percentage increase in the Consumer Price Index from the Consumer Price Index that is in effect on the Commencement Date.
- (C) Tenant shall be required to obtain Landlord's consent for a proposed Material Alteration, which consent shall not be unreasonably withheld, conditioned or delayed.
  - (D) Nothing contained in this Section 7.2 limits the provisions of Section 7.12 hereof.

# 7.3. <u>Approval Process.</u>

(A) Tenant shall not perform any Alteration unless Tenant first gives to Landlord a notice thereof (an "Alterations Notice") that (i) refers specifically to this Section 7.3, (ii) includes three (3) copies of the plans and specifications for the proposed Alteration (including, without limitation, layout, architectural, mechanical and structural drawings, to the extent applicable) in CADD format that contain sufficient detail for Landlord and Landlord's consultants to reasonably assess the proposed Alteration, (iii) indicates whether Tenant considers the proposed Alteration to constitute a Basic Alteration, (iv) indicates whether Tenant considers the proposed Alteration not to exceed the Basic Alterations Threshold and whether Tenant

intends to perform the proposed Alteration without Landlord's consent as contemplated by this Article 7, and (v) includes with such notice an estimate certified by Tenant of the "hard" construction cost of performing the proposed Alteration (if Tenant considers the proposed Alteration not to exceed the Basic Alterations Threshold and plans to perform such Alteration without Landlord's consent). Tenant shall not be required to deliver the aforesaid plans and specifications to Landlord as part of an Alterations Notice in respect of (x) Alterations that do not require a building permit pursuant to applicable Requirements, and (y) Alterations for which plans and specifications would not customarily be prepared in accordance with good industry practice, it being agreed, however, that Tenant shall be required to submit plans and specifications for a proposed Alteration if, following Landlord's receipt of an Alterations Notice to which Tenant did not annex plans and specifications based on the immediately preceding clauses (i) and (ii), Landlord can demonstrate a valid purpose for requiring such plans and specifications. Tenant shall have the right to submit plans and specifications for a particular Alteration in phases (i.e., pursuant to separate Alterations Notices), it being agreed that if a subsequent submission of plans and specifications for a particular Alteration affects or relates to a prior submission of plans and specifications for such Alteration (x) such that Landlord would not have consented to the prior submission, or (y) and provides new information with respect to the prior submission such that Landlord now needs to reconsider such prior submission, then Landlord shall have the right to re-review any such prior submission as if it had not been previously reviewed by Landlord.

- (B) If (x) Tenant gives Landlord an Alterations Notice, (y) Tenant, in the Alterations Notice, does not indicate that Tenant plans to perform the applicable Alteration without Landlord's consent, and (z) Landlord fails to respond to the Alterations Notice within ten (10) Business Days after the date that Tenant gives the Alterations Notice to Landlord (or within five (5) Business Days with respect to any resubmission of an Alterations Notice), then Landlord shall be deemed to have consented to such Alteration.
- (C) Landlord shall have the right to object to a proposed Alteration only by giving notice thereof to Tenant, and setting forth in such notice a statement in reasonable detail of the grounds for Landlord's objections.
- (D) Subject to Section 7.2(B) hereof, Landlord shall have the right to (a) disapprove any plans and specifications for a particular Alteration in part, (b) reserve Landlord's approval of items shown on such plans and specifications pending Landlord's review of other plans and specifications that Tenant is otherwise required to provide to Landlord hereunder, and (c) condition Landlord's approval of such plans and specifications upon Tenant's making revisions to the plans and specifications or supplying additional information (which Landlord shall have the right to request only reasonably). Nothing contained in this Section 7.3(D) limits the provisions of Section 7.3(C) hereof.
- (E) Tenant acknowledges that (i) the review of plans or specifications for an Alteration by or on behalf of Landlord, or (ii) the preparation of plans or specifications for an Alteration by Landlord's architect or engineer (or any architect or engineer designated by Landlord), is solely for Landlord's benefit, and, accordingly, Landlord makes no representation or warranty that such plans or specifications comply with any Requirements or are otherwise adequate or correct.

- (F) Subject to the terms of this Section 7.3(F), Landlord hereby approves in concept Tenant's performance of the Alterations described in Exhibit "7.3(F)" attached hereto and made a part hereof. Tenant shall not perform such Alterations without Landlord's prior approval to the extent otherwise provided in this Article 7, it being understood, however, that Landlord, in considering such Alterations, shall not have the right to reject such Alterations solely on the grounds that the nature of such Alteration as described in Exhibit "7.3(F)" attached hereto impedes the value or utility of the Building or is otherwise inconsistent with Tenant's use and occupancy of the Premises as contemplated hereby. The terms of this Article 7 shall otherwise apply to any such Alterations that Tenant performs.
  - (G) Tenant shall not have the right to place blast resistant film on any window located in the Premises.

#### 7.4. <u>Performance of Alterations.</u>

- (A) Tenant, at Tenant's expense, prior to the performance of any Alteration, shall obtain all permits, approvals and certificates required by any Governmental Authorities in connection therewith. Landlord shall have the right to require Tenant to make all filings with Governmental Authorities to obtain such permits, approvals and certificates using an expeditor designated reasonably by Landlord (provided that the charges imposed by such expeditor are commercially reasonable and such expeditor performs in a reasonable and competent manner) or another expeditor selected by Tenant and approved by Landlord, which approval shall not be unreasonably withheld, conditioned or delayed. Upon the request of Tenant, Landlord shall join in and promptly execute any applications for any permits, approvals or certificates required to be obtained by Tenant in connection with a particular Alteration (provided that the applicable Requirement requires Landlord to join in such application) and shall otherwise cooperate with Tenant in connection therewith. Tenant shall have the right to request that Landlord join in any such application as aforesaid prior to Landlord's granting its consent to such Alteration (to the extent Landlord's consent is required for such Alteration), provided, however, that such cooperation shall not be construed as Landlord's granting its consent to such Alteration and if Landlord ultimately does not grant its consent to such Alteration in accordance with this Article 7, then Tenant shall withdraw or cause to be removed from consideration any such permit, approval or application (and Landlord shall have no liability to Tenant as a result thereof). Tenant shall reimburse Landlord for any reasonable and actual out-of-pocket costs, including, without limitation, reasonable attorneys' fees and disbursements, that Landlord incurs and pays to an unrelated third party in so joining in such applications and cooperating with Tenant, within thirty (30) days after the date that Landlord gives to Tenant an invoice ther
- (B) Prior to performing any Alteration, Tenant or Tenant's general contractor shall also furnish to Landlord duplicate original policies of, or, at Tenant's option, certificates of, (1) worker's compensation insurance in amounts not less than the statutory limits (covering all persons to be employed by Tenant, and Tenant's contractors and subcontractors, in connection with such Alteration), and (2) commercial general liability insurance (including property damage and bodily injury coverage), in each case in customary form, and in amounts that are not less than Five Million Dollars (\$5,000,000) with respect to general contractors and One Million Dollars (\$1,000,000) with respect to subcontractors, naming the Landlord Indemnitees as

additional insureds; provided, however, that the aforesaid amounts may be increased to those amounts reasonably required by Landlord, as long as such amounts are consistent with the amounts for similar insurance required by landlords for alterations performed by tenants in first-class buildings located in midtown Manhattan. Tenant or Tenant's general contractor shall have the right to satisfy the insurance requirements contained in this Section 7.4(B) with an umbrella insurance policy if such umbrella insurance policy contains an aggregate per location endorsement that provides the required level of protection for the Premises.

- (C) Upon completion of each Alteration, Tenant, at Tenant's expense, shall (1) obtain certificates of final approval for each Alteration to the extent required by any Governmental Authority, (2) furnish Landlord with copies of such certificates, and (3) give to Landlord copies of the "as-built" plans and specifications in CADD format only for Alterations that affect or connect to Building Systems and, for all other Alterations, shall give to Landlord appropriate record drawings, shop drawings or final marked drawings therefor.
- (D) All Alterations shall be made and performed substantially in accordance with the plans and specifications therefor as approved by Landlord (to the extent such approval of Landlord is required under this Article 7), all Requirements and the Rules. All materials and equipment incorporated in the Premises as a result of any Alterations shall be first-quality.
- (E) Subject to the terms of this Section 7.4(E), Landlord shall deliver to Tenant, in connection with Tenant's applications to the applicable Governmental Authority for a building permit regarding any Alterations, a Form ACP-5, duly executed by an appropriate party and covering the Premises.

## 7.5. <u>Financial Integrity.</u>

(A)

- (1) Tenant shall not permit any materials or equipment that are incorporated as fixtures into the Premises in connection with any Alterations to be subject to any lien, encumbrance, chattel mortgage or title retention or security agreement.
- (2) Tenant shall discharge of record any mechanic's lien that is filed against the Real Property for work done for, or for materials furnished to, Tenant (or any Person claiming by, through or under Tenant) within ten (10) days after Tenant has received notice thereof, at Tenant's expense, by payment or filing the bond required by law or otherwise. Nothing contained in this Section 7.5(A)(2) shall prevent Tenant from challenging the claim made by the Person that filed such mechanic's lien, provided that Tenant discharges such mechanic's lien in accordance herewith. Tenant shall not be required to discharge any lien that derives from any act or omission of Landlord.
- (B) Subject to the terms of this Section 7.5(B) Tenant shall deliver to Landlord (i) waivers of lien from the general contractor and all subcontractors who have performed work or supplied materials in connection with such Alterations, and (ii) a copy of any document executed by an architect that is filed by or on behalf of Tenant with a Governmental Authority in connection with such Alteration, in either case on or prior to sixty (60) days after the Substantial Completion of any Alterations. If, with the use of diligent efforts, Tenant cannot provide

Landlord with either or both of the items described in the immediately preceding clause (i) and clause (ii) within such sixty (60) day period of time, then Tenant shall have such longer period of time to provide Landlord with either or both of such items as long as Tenant uses continuous, diligent efforts to provide Landlord with either or both of such items. Tenant shall not be required to deliver to Landlord any waiver of lien if Tenant is disputing in good faith the payment which would otherwise entitle Tenant to such waiver, provided that (x) Tenant keeps Landlord advised in a timely fashion of the status of such dispute and the basis therefor, and (y) Tenant delivers to Landlord the waiver of lien promptly after the date that the dispute is settled. Nothing contained in this Section 7.5(B), however, shall relieve Tenant from complying with the provisions of Section 7.5(A)(2) hereof.

## 7.6. <u>Effect on Building.</u>

- (A) If (i) as a result of any Alterations, any alterations, installations, improvements, additions or other physical changes are required to be performed in or made to any portion of the Building other than the Premises in order to comply with any Requirements (any such alterations, installations, improvements, additions or changes being referred to herein as a "Building Change"), and (ii) such Building Change would not otherwise have had to be performed or made pursuant to applicable Requirements at such time, then (x) Landlord may perform such Building Change (it being agreed that Landlord shall use commercially reasonable efforts to minimize the cost of such Building Change), and (y) Tenant shall pay to Landlord the reasonable and actual costs paid by Landlord to an unrelated Person (without profit or markup), as additional rent, within thirty (30) days after Landlord gives to Tenant an invoice therefor, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon (it being agreed that to the extent Landlord has actual knowledge of a particular Building Change, Landlord shall give Tenant notice thereof reasonably promptly following obtaining such actual knowledge and prior to Landlord's performing such Building Change shall notify Tenant of Landlord's good faith estimate of the cost of such Building Change).
- (B) If (x) Tenant submits an Alterations Notice to Landlord for a proposed Alteration, and (y) Landlord has actual knowledge of a Building Change that would be required to be performed or made as a result of such proposed Alteration, then Landlord's response to such Alterations Notice shall also notify Tenant of the need for such Building Change, if any, and Landlord's good faith estimate of the cost of such Building Change. If Landlord had actual knowledge of a Building Change that would be required to be performed or made as a result of any proposed Alteration and did not so inform Tenant of such Building Change as aforesaid, then Tenant shall not be required to pay Landlord for such Building Change as otherwise required pursuant to this Section 7.6. Any dispute with respect to whether Landlord had knowledge of a Building Change and did not so inform Tenant of such Building Change may be submitted by either party to an Expedited Arbitration Proceeding.

## 7.7. <u>Time for Performance of Alterations.</u>

If and to the extent that the performance of any Alteration by or on behalf of Tenant, or any other Person claiming by, through or under Tenant, during particular hours interferes with or interrupts the maintenance, repair, or operation of the Building in any material and unreasonable

respect or interferes with or interrupts the use and occupancy of the Building by other tenants in the Building or the use and occupancy of any condominium units by residential owners, as the case may be, in any material and unreasonable respect, then Landlord shall have the right to require Tenant to perform such Alteration during other hours reasonably designated by Landlord (taking into account (x) the business hours of other tenants or occupants in the Building, and (y) the needs of residential unit owners).

#### 7.8. <u>Removal of Alterations and Tenant's Property.</u>

- On or prior to the Expiration Date, Tenant, at Tenant's expense, shall remove Tenant's Property from the Premises, and, at Tenant's option, Tenant also may remove, at Tenant's expense, all Alterations made by or on behalf of Tenant or any other Person claiming by, through or under Tenant; provided, however, in any case, that Tenant shall repair and restore in a good and workerlike manner to good condition any damage to the Premises or the Building caused by such removal. Landlord, upon notice to Tenant given at least one hundred eighty (180) days prior to the Expiration Date, may require Tenant to remove any Specialty Alterations from the Premises, and to repair and restore in a good and workerlike manner to good condition any damage to the Premises or the Building caused by such removal; provided, however, that Landlord shall not have the right to require Tenant to remove any Qualified Alterations. Tenant may elect to cause Landlord to remove any Specialty Alterations set forth in such notice given by Landlord to Tenant by giving Landlord notice thereof no later than sixty (60) days after the date Landlord gives its notice to Tenant. If Tenant so elects to cause Landlord to remove any such Specialty Alterations, then (x) Landlord shall remove such Specialty Alterations from the Premises following the Fixed Expiration Date and repair and restore in a good and workerlike manner to good condition any damage to the Premises or the Building caused by such removal, and (y) Tenant shall reimburse Landlord for the actual, reasonable expenses incurred by Landlord in performing such removal and restoration on or prior to thirty (30) days after Landlord's rendition of an invoice therefor, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon. Landlord shall give any such statement to Tenant no later than eighteen (18) months following the Fixed Expiration Date. Tenant's obligation to pay the amounts set forth on any such invoice shall survive the Fixed Expiration Date. If (x) the Expiration Date is not the Fixed Expiration Date, and (y) Landlord gives a notice to Tenant on or prior to the thirtieth (30th) day after the Expiration Date to the effect that Landlord does not wish to retain a particular Specialty Alteration, then Tenant shall pay to Landlord the reasonable costs incurred by Landlord in so removing such Specialty Alterations, and in so repairing and restoring any such damage to the Building or the Premises, within thirty (30) days after Landlord submits to Tenant an invoice therefor; provided, however, that Landlord shall not have the right to give any such notice to Tenant in respect of Qualified Alterations. Any Alterations that remain in the Premises after the Expiration Date shall be deemed to be the property of Landlord (with the understanding, however, that Tenant shall remain liable to Landlord for any default of Tenant in respect of Tenant's obligations under this Section 7.8(A)).
- (B) Prior to Tenant's performance of a Specialty Alteration, Tenant shall have the right to request (prior to, simultaneously with, or any time after Tenant's submission to Landlord of plans and specifications for such Specialty Alteration, but prior to Tenant's performance of such Special Alteration) that Landlord designate that Tenant shall not be required

to remove (or pay the cost to remove) such Specialty Alteration upon the expiration or earlier termination of the Term. Landlord shall have the right to approve or deny any such request in Landlord's sole discretion. If (i) Tenant makes any such request, and (ii) Landlord either approves such request (which approval shall be set forth in the same notice Landlord gives to Tenant with respect to Landlord's consent to such plans and specifications for such Specialty Alteration), or fails to respond to Tenant's aforesaid request on or prior to the fifteenth (15th) Business Day after the date that Tenant gives such request to Landlord, then Landlord shall not have the right to require Tenant to remove (or pay the cost to remove) such Specialty Alteration upon the expiration or earlier termination of the Term (any such Specialty Alteration which Tenant shall not be required to remove (or to pay the cost of removal) as aforesaid being referred to herein as a "Qualified Alteration").

#### 7.9. <u>Contractors and Supervision.</u>

All Alterations that require Landlord's consent shall be performed only under the supervision of a licensed architect that is approved by Landlord, which approval Landlord shall not unreasonably withhold, condition or delay (it being agreed that if Landlord does not respond to Tenant's notice requesting Landlord's consent to any such architect on or prior to ten (10) Business Days after the date Tenant gives such notice to Landlord, then Landlord shall be deemed to have consented to such architect). Subject to the provisions of this Section 7.9, Tenant shall perform Alterations using, at Tenant's option, either (i) contractors, subcontractors and mechanics that in each case are designated from time to time by Landlord, or (ii) contractors, subcontractors or mechanics that in each case are designated by Tenant and approved by Landlord, which approval Landlord shall not unreasonably withhold, condition or delay (it being agreed that if Landlord does not respond to Tenant's notice requesting Landlord's consent to any such contractor, subcontractor or mechanic on or prior to ten (10) Business Days after the date Tenant gives such notice to Landlord, then Landlord shall be deemed to have consented to such contractor, subcontractor or mechanic). If Tenant is Citibank, N.A. or an Affiliate thereof, however, then Landlord's consent shall only be required with respect to (i) Tenant's general contractor who shall perform any Alterations, (ii) any contractors, subcontractors or mechanics for Alterations that affect or connect to Building Systems, and (iii) for structural Alterations (and for all other Alterations Tenant may use a contractor or subcontractor selected by Tenant in its sole discretion). Solely with respect to a Life Safety System Alteration, Tenant shall engage an engineer designated reasonably by Landlord to design such Alteration, provided that the charges of such designated engineer are commercially reasonable. If such designated engineer is Landlord's engineer or a third-party engineer used by Landlord or the Condominium Board, then Tenant shall not be required to pay Landlord for such engineer pursuant to Section 7.10 hereof. Landlord shall give Tenant notice of such contractors and such engineer designated by Landlord promptly after Tenant's request therefor from time to time. The term "Life Safety System Alteration" shall mean an Alteration that involves or affects the Building System that constitutes the Class "E" system and the sprinkler system (it being understood that if only a portion of a particular Alteration involves or affects such Building Systems, then only such portion of such Alteration shall constitute a Life Safety Systems Alteration for purposes hereof).

## 7.10. <u>Landlord's Expenses.</u>

Subject to the terms of this Section 7.10 hereof, Tenant shall pay to Landlord, from time to time, as additional rent, the reasonable and actual out-of-pocket costs paid by Landlord to an unrelated third party in connection with an Alteration, including, without limitation, the costs that Landlord pays to an unrelated third party architect or engineer to review the plans and specifications for Alterations, and inspecting the progress of such Alterations. Any such costs shall in no event include profit or a markup charged by Landlord. Tenant shall pay such costs to Landlord within thirty (30) days after Landlord gives Tenant an invoice therefor, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon. In no event shall Tenant be required to pay Landlord for any such costs in respect of Landlord's engineer or a third-party engineer engaged by Landlord or the Condominium Board.

#### 7.11. <u>Pantry.</u>

Landlord shall not unreasonably withhold, condition or delay Landlord's approval of an Alteration consisting of the installation of a pantry in the Premises for the purpose of warming food for the officers, employees and business guests of Tenant (but not for use as a public restaurant). Any vending machines that Tenant installs in the Premises only on the twenty-first (21<sup>st</sup>) floor or the twenty-second (22<sup>nd</sup>) floor thereof (and any other similar equipment that requires a water line) shall have a waterproof pan located thereunder, connected to a drain.

## 7.12. <u>Window Coverings.</u>

Tenant shall install on the windows of the Premises only the curtains, blinds, shades or screens that Landlord designates for the Building. As of the date hereof, Landlord's approved blinds are as follows: DFB, solar control, 4100 Series, basket weave 10% open, granite color. Landlord acknowledges that once Tenant installs the curtains, blinds, shades or screens that conform to the requirements of this Section 7.12, then Tenant shall not be required to replace such curtains, blinds, shades or screens to conform to curtains, blinds, shades or screens that Landlord thereafter adopts for the Building.

## 7.13. <u>Standby Generator System.</u>

(A) Subject to the terms of this Section 7.13, Landlord consents to Tenant's connecting Tenant's systems to the standby generator system, together with the Day Tank, that constitutes part of the Building Systems (such standby generator system being referred to herein collectively as the "Standby Generator System"), that is located on the twelfth (12<sup>th</sup>) floor of the Building (which Standby Generator System also has its own two hundred seventy-five (275) gallon day tank (the "Day Tank") also located on the twelfth (12<sup>th</sup>) floor of the Building). Tenant shall perform any such connection of Tenant's systems to the Standby Generator System at Tenant's cost, in accordance with all applicable Requirements, and otherwise in accordance with the terms of this Article 7. Landlord, at Landlord's cost and expense, shall maintain in good condition and repair, replace and test the Standby Generator System in accordance with applicable Requirements and pursuant to the specifications annexed as Exhibit "7.13(A)" attached hereto and made a part hereof (it being agreed that Tenant has the right to have a representative witness any such testing that Landlord performs in connection with the Standby

Generator System); provided, however, that for as long as Tenant has exclusive use of the Standby Generator System in accordance with the terms hereof, Tenant (rather than Landlord), at Tenant's cost and expense, shall perform the maintenance and testing of (but not the repair and replacement of) the Standby Generator System. Landlord shall provide a watch engineer to monitor the operation of the Standby Generator System (and the chillers that are part of Tenant's supplemental HAVC system and that are powered by the Standby Generator System) during applicable periods (i.e., when the Standby Generator System is being tested or when the Standby Generator System is being operated to alleviate a power emergency) without cost to Tenant (subject, however, to Landlord's having the right to include the cost of the watch engineer in Operating Expenses to the extent provided in Section 2.1(C) hereof, with the understanding, however, that Landlord shall assign one of the operating engineers who are ordinarily on duty at the Building to constitute the watch engineer during such applicable periods). If Tenant has exclusive use of the Standby Generator System and fails to perform the maintenance and testing thereof in accordance with the aforesaid specifications, then Tenant, at Tenant's cost and expense, shall be responsible for the repair and replacement of the Standby Generator System (to the extent that such repair or replacement is required by reason of Tenant's failure to so maintain and test the Standby Generator System). Tenant, at Tenant's cost and expense, shall maintain in good condition any connection that Tenant makes to the Standby Generator System. Tenant acknowledges that Landlord shall have no liability to Tenant to the extent that the Standby Generator System fails to provide emergency power to Tenant or that the Standby Generator System damages Tenant's systems (except, in either case, to the extent that such damage results from Landlord's negligence or willful misconduct, it being expressly agreed that in no event shall Landlord be liable for any consequential, indirect or punitive damages in connection therewith). Tenant shall not permit the aggregate connected load of Tenant's systems on the Standby Generator System to exceed Seven Hundred Fifty (750) kilowatts. During any period of time that Tenant meets the Minimum Demise Requirement, Tenant shall be entitled to the exclusive use of the Standby Generator System. Landlord shall provide Tenant with reasonably necessary access to the Standby Generator System for purposes of maintaining, testing and inspecting the Standby Generator System and in connection with any necessary repair of any connections to the Standby Generator System, in either case provided that such access shall (i) not unreasonably interfere with or interrupt the operation and maintenance of the Building, and (ii) be upon such other terms reasonably designated by Landlord. Landlord, at no additional cost to Tenant, shall provide Tenant with a non-separately demised amount of space on the twelfth (12<sup>th</sup>) floor of the Building in proximity to the Standby Emergency Generator for Tenant to maintain electrical equipment in connection with the Standby Emergency Generator, which space shall be reasonably sized to accommodate such equipment. Tenant, at Tenant's cost and expense, shall have the right to connect the Standby Generator System to the base Building's air handlers, which work shall be performed in accordance with the applicable terms and provisions of Article 7 hereof.

(B) Subject to the provisions of Section 7.13(C) hereof, Tenant, during the Term, shall pay to Landlord, in consideration of Landlord's making available to Tenant the capacity of the Standby Generator System, as additional rent, an annual amount equal to Eighty-Eight Thousand Dollars and No Cents (\$88,000.00), which Tenant shall pay to Landlord in monthly installments of Seven Thousand Three Hundred Thirty-Three Dollars and Thirty-Three Cents (\$7,333.33) in the same manner as the Fixed Rent that is due hereunder; provided, however, that on each anniversary of the Commencement Date, the aforesaid annual amount and

the aforesaid monthly amount shall be adjusted to reflect the percentage increase, if any, in the Consumer Price Index that is in effect on the Commencement Date. If Landlord operates the Standby Generator System to provide electric service (at a time that the Utility Company is not providing electric service for the Building), then Tenant shall also pay to Landlord, as additional rent, an amount equal to the product obtained by multiplying (x) the out-of-pocket costs that Landlord incurs for utilities in connection with operating the Standby Generator System, including purchasing fuel therefor), by (y) a fraction, the numerator of which is the aggregate connected load (in kilowatts) of Tenant's systems on the Standby Generator System and the denominator of which is the aggregate capacity (in kilowatts) of the Standby Generator System, within thirty (30) days after Landlord's submission to Tenant of an invoice therefor.

- (C) From and after the date that Tenant does not meet the Minimum Demise Requirement, Landlord shall have the right to proportionately reduce the aggregate connected load of Tenant's systems on the Standby Generator System from seven hundred fifty (750) kilowatts to a lesser number of kilowatts, based upon the number of square feet of Rentable Area demised to Tenant under this Lease at such time (as compared to such number of square feet of Rentable Area on the date hereof). From and after the date of any such reduction, the additional rent set forth in Section 7.13(B) hereof of Eighty-Eight Thousand Dollars (\$88,000) per annum shall instead be an annual amount equal to the product of (I) Eighty-Eight Thousand Dollars (\$88,000), and (II) a percentage, the numerator of which is the aggregate connected load (in kilowatts) of Tenant's systems on the Standby Generator System (after the reduction) and the denominator of which is seven hundred fifty (750) (which amount, in addition to any amounts on account of utilities, shall continue to be paid by Tenant pursuant to Section 7.13(B) hereof). Tenant, at Tenant's cost and expense, shall arrange to reconfigure Tenant's systems that are connected to the Standby Generator System so that Tenant does not exceed the permitted aggregate connected load determined by Landlord as aforesaid.
- (D) Landlord covenants that the Building emergency generator system shall provide a capacity of at least one-quarter (1/4) watt per square foot of Usable Area of the Premises for emergency lighting within the Premises.

#### 7.14. <u>Tenant Fund.</u>

- (A) Subject to the terms of this Section 7.14, Landlord shall pay to Tenant (or at Tenant's option (i) to a third party designated by Tenant or (ii) after Landlord's receipt of Tenant's written authorization therefor, to Tenant's contractor or subcontractor), or on behalf of Tenant an amount equal to Ten Million Five Hundred Sixty Thousand Dollars and No Cents (\$10,560,000.00) (the "Tenant Fund") for the costs that Tenant incurs in performing the Initial Alterations. Tenant may use no more than twenty percent (20%) of the Tenant Fund for costs that Tenant incurs in connection with the Initial Alterations that do not constitute "hard" construction costs, including, without limitation, architect's and engineer's fees, permit fees, expediter's fees and designers' fees in each case relating to the Initial Alterations (such costs which do not constitute the "hard" construction costs of the Initial Alterations being collectively referred to herein as "Soft Costs").
- (B) Tenant may request disbursements of the Tenant Fund only by delivering to Landlord a Disbursement Request. Subject to the terms of this Section 7.14, Landlord shall

disburse a portion of the Tenant Fund to Tenant (or at Tenant's option (i) to a third party designated by Tenant, or (ii) after Landlord's receipt of Tenant's written authorization therefor, to Tenant's contractor or subcontractor) from time to time, within thirty (30) days after the date that Tenant gives to Landlord the applicable Disbursement Request. Tenant shall not be entitled to any disbursements of the Tenant Fund if an Event of Default has occurred and is continuing at a time when Tenant would otherwise be entitled to a disbursement of the Tenant Fund, and (y) Tenant subsequently cures such Event of Default, then Landlord shall pay such disbursement to Tenant within thirty (30) days after the date such Even of Default is cured). If a particular Disbursement Request requests Landlord to disburse more than fifty percent (50%) of the Tenant Fund, then Landlord shall not be required to make the disbursement of the Tenant Fund that is contemplated thereby unless all contractors, subcontractors, materialmen, architects, engineers and other Persons who may file a lien against the Real Property in connection with the performance of the Initial Alterations provide to Landlord waivers of lien concurrently with such disbursement of the Tenant Fund. Landlord shall not be required to make disbursements of the Tenant Fund more frequently than once during any particular calendar month. Tenant shall not have the right to request disbursements of the Tenant Fund in an amount that is greater than the excess of (I) the aggregate amounts that Tenant has theretofore paid or that then remain payable in each case to Tenant's contractors, subcontractors, materialmen, suppliers or consultants, as the case may be, for either (a) materials that have been delivered to the Premises for the Initial Alterations, (b) labor that has been performed in the Premises for the Initial Alterations, or (c) the services from which are derived Soft Costs that have been performed for the Initial Alterations, as the case may be, over (II) the agg

- (C) The term "<u>Disbursement Request</u>" shall mean a request for a disbursement of the Tenant Fund signed by an officer of Tenant, together with:
  - (1) such officer's certification that the amount so requested does not exceed the Maximum Disbursement Amount,
- (2) copies of bills and invoices with respect to the applicable work, materials or services covered by such Disbursement Request (which shall have theretofore been completed, furnished or performed, as the case may be), and
- (3) waivers of lien from all contractors and subcontractors who may file a lien against the Real Property in connection with the performance of the Initial Alterations, and for which previous disbursements of the Tenant Fund has been made (except to the extent Tenant gave such waivers of lien to Landlord in connection with a prior Disbursement Request).
- (D) Landlord makes no representation or warranty that the Tenant Fund is sufficient to pay the cost of the Initial Alterations. Tenant shall pay the amount of any excess of the cost of the Initial Alterations over the Tenant Fund.
- (E) If (i) Tenant has theretofore completed the Initial Alterations, (ii) Tenant has theretofore paid (or paid from the proceeds of the Tenant Fund) the cost of the Initial

Alterations in full, (iii) Tenant then occupies the Premises for the conduct of business, (iv) Tenant has not theretofore given to Landlord Disbursement Requests for the entire Tenant Fund, and (v) an Event of Default has not occurred and is not continuing, then Landlord shall pay to Tenant the portion of the Tenant Fund that remains undisbursed, within thirty (30) days after Tenant gives Landlord a request therefor or, at Tenant's option, Tenant shall have the right to apply such undisbursed portion of the Tenant Fund as a credit against Tenant's future installments of the Fixed Rent and additional rent due hereunder. If (x) an Event of Default has occurred and is continuing at the time when Tenant would otherwise be entitled to a payment of such undisbursed portion of the Tenant Fund (or a credit as aforesaid), and (y) Tenant subsequently cures such Event of Default, then Landlord shall pay such undisbursed portion of the Tenant Fund to Tenant (or Tenant may credit such amount as aforesaid) within thirty (30) days after the day such Event of Default has been cured.

- (F) If (i) Landlord fails to make a disbursement of the Tenant Fund when due, and (ii) such failure continues for more than ten (10) days after the date that Tenant gives Landlord notice thereof, then Tenant shall have the right to offset against the Rental due hereunder the amount that Landlord so fails to disburse to Tenant, together with interest thereon calculated at the Applicable Rate for the period beginning on the date that such disbursement first became due to Tenant, and ending on the date that Tenant uses such credit.
- (G) Landlord and Tenant acknowledge the Initial Alterations shall constitute the property of Landlord to the extent that Tenant performs such Initial Alterations using the proceeds of the Tenant Fund as provided in this Section 7.14, subject, nevertheless, to Tenant's rights to use such Initial Alterations as part of the Premises during the Term in accordance with the terms of this Lease.

#### 7.15. Air-Cooled HVAC Installations.

Tenant shall not have the right to install a supplementary HVAC system for the Premises that requires vents or louvers to be installed on the exterior of the Building, except that Landlord consents to Tenant's installing an air-cooled supplementary HVAC system for the Premises in the location on the thirtieth (30<sup>th</sup>) floor of the Building depicted on Exhibit "7.15" attached hereto and made a part hereof. Tenant's use of any electricity provided to such supplementary HVAC system shall be measured by submeter, installed by Landlord, at Landlord's expense, and computed and billed to Tenant in accordance with the terms and provisions of Section 5.3 hereof. Landlord shall provide Tenant with reasonably necessary access in accordance with good construction practice for the installation, operation and maintenance of such HVAC system, provided that such access shall (i) not unreasonably interfere with or interrupt the operation and maintenance of the Building, and (ii) be upon such other terms reasonably designated by Landlord. Tenant shall install such HVAC system at Tenant's expense. Tenant shall perform such installation in accordance with the provisions of this Lease, including, without limitation, the provisions pertaining to the performance of Alterations. If Tenant exercises Tenant's right to install such HVAC system as contemplated by this Section 7.15, then Tenant, at Tenant's expense, shall maintain such HVAC system in good condition during the Term. Tenant, upon the Expiration Date, shall not be required to remove such HVAC system.

## 7.16. <u>Initial Alterations Access.</u>

Landlord, at no expense to Landlord, shall use reasonable efforts to cooperate with Tenant in connection with Tenant's gaining access to the twentieth (20<sup>th</sup>) floor of the Building in order for Tenant to perform certain of its Initial Alterations in the portion of the Premises located on the twenty-first (21<sup>st</sup>) floor of the Building. As part of such cooperation, Landlord agrees to pursue any rights that Landlord may have under the Bloomberg Lease to gain access to the twentieth (20<sup>th</sup>) floor in order to permit Tenant to perform such Initial Alterations.

#### 7.17. Warranties.

Landlord shall use its commercially reasonable efforts to obtain warranties for items installed in the Building (to the extent that such warranties are generally available for construction projects similar to the Building in midtown Manhattan), and shall comply with the terms of any warranties that Landlord so obtains to the extent that the costs of maintaining, repairing or replacing such item would otherwise be payable by Tenant hereunder as part of Operating Expenses.

#### 7.18. Violations and Landlord Defaults; Tenant's Remedies.

Subject to the terms of this Section 7.18, if Landlord defaults in respect of its obligations under this Lease or if there exists a violation of (A) applicable Requirements at the Building, in either case, that impedes or increases the cost of (i) Tenant's performance of an Alteration, (ii) Tenant's obtaining a permit from the applicable Governmental Authority for any such Alteration, or (iii) Tenant's obtaining an amendment to the certificate of occupancy for the Premises pursuant to Article 11 hereof (any such violation being referred to herein as an "Impeding Building Violation"), then Landlord, at Landlord's expense, shall use diligent efforts to cure such default or to cause the Impeding Building Violation to be removed as promptly as reasonably practicable, in either case, after Tenant gives Landlord notice thereof. Nothing contained in this Section 7.18 shall require Landlord to remove any such Impeding Building Violation to the extent that Section 8.2 hereof requires Tenant to comply therewith. If (i) such a default exists or an Impeding Building Violation exists, (ii) Tenant gives Landlord notice thereof, (iii) Landlord fails to remedy such default or Impeding Building Violation within thirty (30) days after Tenant gives Landlord notice thereof, (iv) Tenant, in such notice, describes with reasonable particularity the out-of-pocket costs that Tenant then reasonably expects to incur to the extent attributable solely to the existence of such default or the Impeding Building Violation, (v) Tenant gives to Landlord from time to time additional notices that describe with reasonable particularity the out-of-pocket costs that Tenant actually incurs to the extent attributable solely to the existence of such default or the Impeding Building Violation, with reasonable promptness after the date that Tenant so incurs such costs, and (vi) Tenant takes reasonable steps to minimize such costs to the extent reasonably practicable, then Landlord shall reimburse Tenant for such costs within thirty (30) days after Tenant's request therefor. Either party shall have the right to submit any dispute between the parties regarding any such costs to be reimbursed by Landlord to Tenant or any abatement of Rental, as the case may be, to an Expedited Arbitration Proceeding. If any such default by Landlord or any Impeding Building Violation actually delays Tenant in the performance of its Initial Alterations in the Premises (or a portion thereof), then, for each day that Tenant is so delayed in the performance of its Initial Alterations in the Premises (or the

applicable portion thereof), Tenant shall be entitled a one (1) day abatement of the Rental due hereunder from and after the Rent Commencement Date for the Premises (or the applicable portion thereof) in which Tenant is so delayed in the performance of its Initial Alterations (it being agreed that if, on any particular day, Tenant is delayed in the performance of its Initial Alterations by virtue of both a default by Landlord and an Impeding Building Violation, then Tenant shall nonetheless be entitled to only a one (1) day abatement of Rental as provided in this Section 7.18(A) hereof by virtue of such delay on such day).

(B) If (I) there exists a violation of applicable Requirements with which Tenant is required to comply pursuant to the terms hereof, and (II) such violation impedes or increases the cost of any work being performed in the Building by or on behalf of Landlord or the Condominium Board, or any approval of a Governmental Authority that is required in connection therewith (any such violation being referred to herein as an "Impeding Premises Violation"), then Tenant, at Tenant's expense, shall cause the Impeding Premises Violation to be removed as promptly as reasonably practicable after Landlord gives Tenant notice thereof. If (i) an Impeding Premises Violation exists, (ii) Landlord gives Tenant notice thereof, (iii) Tenant fails to remedy the Impeding Premises Violation within thirty (30) days after Landlord gives Tenant notice thereof, (iv) Landlord, in such notice, describes with reasonable particularity the out-of-pocket costs that Landlord (or the Condominium Board) then reasonably expects to incur to the extent attributable solely to the existence of the Impeding Premises Violation, (iv) Landlord gives to Tenant from time to time additional notices that describe with reasonable particularity the out-of-pocket costs that Landlord (or such unit owner or the Condominium Board) actually incurs to the extent attributable solely to the existence of the Impeding Premises Violation, with reasonable promptness after the date that Landlord (or such unit owner or the Condominium Board) takes reasonable steps to minimize such costs to the extent reasonably practicable, then Tenant shall reimburse Landlord (or such unit owner or the Condominium Board) for such costs within thirty (30) days after Landlord's request therefor. Either party shall have the right to submit to an Expedited Arbitration Proceeding any dispute between the parties regarding any such costs to be reimbursed by Tenant to Landlord (or such unit owner or the Condominium Board).

#### Article 8 REPAIRS

#### 8.1. <u>Landlord's Repairs.</u>

Subject to the terms of this Article 8 and to Article 15 hereof and Article 16 hereof, Landlord, at its sole cost and expense but subject to recoupment to the extent provided in Article 2 hereof, shall maintain and make all necessary repairs and replacements to (i) the Building Systems that service the Premises, (ii) the structural portions of the Building (including, without limitation, the Premises), (iii) the façade and roof of the Building, (iv) the sidewalks that are adjacent to the Building, (v) the core areas and exterior walls of the Premises, (vi) the windows of the Premises, (vii) the plumbing concealed behind the walls in the core restrooms that are located within the Premises, (viii) the perimeter convector units located in the Premises, (ix) the passenger elevators that service any portion of the Premises, and (x) the public portions and

common areas of the Building (including, without limitation, landscaping and lighting in the public portions of the Real Property, and lobbies and hallways in the public portions of the Building to which Tenant has access in connection with Tenant's use and occupancy of the Premises), in each case in conformity with the standards that are customary for first-class office buildings in midtown Manhattan. Nothing contained in this Section 8.1 shall require Landlord to maintain or repair the systems within the Premises that distribute within the Premises electricity, HVAC or water as described on Exhibit "8.2" attached hereto and made a part hereof.

## 8.2. <u>Tenant's Repairs.</u>

- (A) Subject to the terms of this Article 8 and to Article 15 hereof and Article 16 hereof, Tenant, at Tenant's expense, shall take good care of the Premises (including, without limitation, (i) the fixtures and equipment that are installed in the Premises on the Commencement Date, (ii) the Alterations, (iii) the systems within the Premises that distribute within the Premises electricity, HVAC or water, as more particularly described on Exhibit "8.2" attached hereto and made a part hereof, and (iv) the core restrooms located in the Premises other than the plumbing concealed behind the walls therein. Tenant shall make all repairs to the Premises as and when needed to preserve the Premises in good condition, except for reasonable wear and tear, obsolescence and damage for which Tenant is not responsible pursuant to the provisions of Article 15 hereof. Nothing contained in this Section 8.2(A) shall require Tenant to perform any repairs to the Premises that are Landlord's obligation to perform under Section 8.1 hereof. All repairs made by Tenant as contemplated by this Section 8.2(A) shall be in conformity with the standards that are customary for first-class office buildings in midtown Manhattan. Tenant shall perform such repairs in accordance with the terms of Article 7 hereof.
- (B) Subject to the terms of this Section 8.2(B), if (a) Landlord gives Tenant a notice that Tenant has failed to perform a repair that this Section 8.2 obligates Tenant to perform, and (b) Tenant fails to proceed with reasonable diligence to make such repair within twenty (20) days after the date that Landlord gives such notice to Tenant (or such shorter period that Landlord designates in such notice to the extent reasonably required under the circumstances to alleviate an imminent threat to persons or property), then (i) Landlord may make such repair, and (ii) Tenant shall pay to Landlord, as additional rent, the reasonable expenses thereof, with interest thereon at the Applicable Rate calculated from the date that Landlord incurs such expenses, within thirty (30) days after Landlord gives Tenant an invoice therefor. Nothing contained in this Section 8.2(B) limits the remedies that are available to Landlord after the occurrence of an Event of Default.
- (C) Subject to the terms of this 8.2(C), if (a) Tenant gives Landlord a notice that Landlord has failed to perform a repair or provide any item of work, maintenance, repair or service that this Lease obligates Landlord to perform or provide, and (b) Landlord fails to proceed with reasonable diligence to perform or provide same within twenty (20) days after the date that Tenant gives such notice to Landlord, then (i) Tenant shall have the right to perform any such item of work, maintenance, repair or service to the extent that Tenant can perform such item within the Premises or such item affects any Building System that serves the Premises exclusively (and Tenant shall have access to those portions of the Building outside the Premises that service the Premises in order to do so), and (ii) Landlord shall pay to Tenant an amount equal to the out-of-pocket costs that Tenant reasonably pays to an unrelated third party in

performing such item of work, maintenance, repair or service (without profit or markup), together with interest thereon at the Applicable Rate from the date such costs were incurred through the date of reimbursement. Such amounts shall be paid by Landlord on or prior to the thirtieth (30th) day after the date that Tenant gives to Landlord an invoice therefor, accompanied by reasonably satisfactory documentation of such costs (and Landlord's obligation to make such payment to Tenant shall survive the Expiration Date). If Landlord fails to reimburse Tenant for such costs within such thirty (30) day period, then Tenant shall have the right to credit the amount of such costs (with interest as aforesaid) against the Rental thereafter accruing hereunder. Tenant shall have the right to designate in the aforesaid notice that Tenant gives to Landlord a period that is shorter than the period of twenty (20) days that is described in clause (b) above to the extent reasonably required by the circumstances to alleviate an imminent threat to persons or property or a material disruption of Tenant's business in the Premises. Tenant shall not have the right to perform any such repair as contemplated by this 8.2(C) to the extent that Landlord commences the performance of the applicable repair during the aforesaid period of twenty (20) days (or such shorter period that Tenant designates and Landlord thereafter diligently pursues such repair to completion).

#### 8.3. <u>Certain Limitations.</u>

- (A) Tenant, at Tenant's expense, shall repair in accordance with the terms set forth in Section 8.2 hereof all damage to the Premises, or to any other part of the Building or the Building Systems, in each case to the extent resulting from the negligence or willful misconduct of, or Alterations made by, Tenant or any other Person claiming by, through or under Tenant; provided, however, that Landlord shall have the right to perform any such repair to the extent that such repair affects the structure of the Building or such repair affects any Building System, in which case Tenant shall pay to Landlord an amount equal to the out-of-pocket costs that Landlord reasonably incurs and pays to an unrelated third party in performing such repair, on or prior to the thirtieth (30th) day after the date that Landlord gives to Tenant an invoice therefor, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon. Nothing contained in this Section 8.3(A) limits the provisions of Section 14.3 hereof.
- (B) Landlord, at Landlord's expense, shall repair promptly all damage to the Premises that results from Landlord's negligence or willful misconduct. Nothing contained in this Section 8.3(B) limits the provisions of Section 14.3 hereof.

## 8.4. Overtime.

Subject to the provisions of this Section 8.4, Landlord shall have no obligation to employ contractors or labor at overtime or premium pay rates in connection with (x) Landlord's making repairs as contemplated by this Article 8, or (y) Landlord's performing the work for which Landlord requires a Work Access. If the condition that Landlord is required to repair, or the work for which Landlord requires a Work Access, (i) denies Tenant from having reasonable access to the Premises, (ii) threatens the health or safety of any occupant of the Premises, or (iii) materially interferes with Tenant's ability to conduct its business in the Premises during Tenant's ordinary business hours, then Landlord shall employ contractors or labor at overtime or premium pay rates. Landlord, at Tenant's request, shall also perform (a) any other repair that this

Article 8 requires Landlord to perform, or (b) any other work for which Landlord requires a Work Access, using contractors or labor at overtime or premium pay rates, in which case Tenant shall pay to Landlord, as additional rent, an amount equal to the excess of (x) the reasonable out-of-pocket costs that Landlord incurs and pays to an unrelated third party in performing such repair or such work (using contractors or labor at overtime or premium pay rates), over (y) the costs that Landlord would have incurred in performing such repair or such work without using contractors at overtime or premium pay rates, within thirty (30) days after the date that Landlord gives to Tenant an invoice therefor, which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon (it being understood that if more than one tenant requests that Landlord perform any such repair using contractors or labor at overtime or premium pay rates, then Landlord shall allocate such costs among such tenants equitably). Nothing contained in this Section 8.4 shall be deemed to limit or modify Tenant's right pursuant to Section 10.2 and Section 10.3 hereof.

# Article 9 ACCESS; LANDLORD'S CHANGES

#### 9.1. <u>Access.</u>

Subject to the terms of this Section 9.1, Landlord and Landlord's designees may enter the Premises at reasonable times upon reasonable prior notice to Tenant (which notice may be given verbally to the person employed by Tenant with whom Landlord's representative ordinarily discusses matters relating to the Premises) to (i) examine the Premises, (ii) show the Premises to prospective tenants during the last eighteen (18) months of the Term, (iii) show the Premises to prospective purchasers or master lessees of Landlord's interest in the Real Property, (iv) show the Premises to Mortgagees or Lessors (or prospective Mortgagees or Lessors), (v) gain access to Reserved Areas, (vi) to gain access to Landlord's equipment located in the telephone closets in the Premises, or (vii) make repairs, alterations, improvements, additions or restorations that (I) Landlord is required to make pursuant to the terms of this Lease, or (II) are reasonably necessary in connection with the maintenance, repair, or operation of the Real Property (Landlord's entry upon the Premises to perform such repairs, alterations, improvements, additions or restorations being referred to herein as a "Work Access"). Notwithstanding any provision of this Article 9 to the contrary, Landlord shall be required to perform all repairs, alterations, improvements, additions and restorations as referred to in clause (vii) of the immediately preceding sentence during hours other than Building Hours if the nature of any such repair, alteration, improvement, addition or restoration is such as to customarily be performed by landlords in first-class buildings located in midtown Manhattan during hours other than Building Hours (for example, an electrical shutdown for the repair of a riser). If required in accordance with the provisions of Section 8.4 hereof, Landlord shall perform all repairs, alterations, improvements, additions and restorations as referred to in clause (vii) of the immediately preceding sentence during hours other than Building Hours. Landlord shall not be required to give Tenant advance notice of the entry by Landlord or Landlord's designees into the Premises as contemplated by this Section 9.1 to the extent necessary by reason of the occurrence of an emergency, provided, however, that (i) any such entry shall be subject to any applicable Requirements, (ii) during such entry Landlord or Landlord's designees shall accord reasonable care to Tenant's Property, and (iii) Landlord shall give Tenant notice of such emergency access

as promptly as reasonably practicable thereafter. If entry by Landlord or Landlord's designees into the Premises as contemplated by this Section 9.1 is not the result of the occurrence of an emergency, Landlord shall not enter the Premises for the purposes set forth in this Section 9.1 unless Landlord first gives Tenant a reasonable opportunity to have a representative of Tenant present in the Premises at the time of such entry and for the duration of any Work Access. Landlord, in connection with a Work Access, shall have the right to bring into the Premises, and store in the Premises in a reasonable manner for the duration of the Work Access, the materials and tools that Landlord reasonably requires to perform the applicable repair, alteration, improvement, addition or restoration.

(B) Landlord shall have no liability to Tenant for any loss sustained by Tenant by reason of Landlord's entry upon the Premises; provided, however, that (x) subject to Section 14.3 hereof, Landlord shall remain liable to Tenant for personal injury or property damage that derives from Landlord's negligence or willful misconduct in connection with any such entry upon the Premises, (y) nothing contained in this Section 9.1 limits Tenant's rights to an abatement of Rental after a fire or other casualty as provided herein, and (z) nothing contained in this Section 9.1 limits Tenant's rights to an abatement of Rental as provided in Section 10.3 hereof. Landlord shall not have the right to access the Secure Area as provided in this Section 9.1 except to the extent that (x) Landlord requires such access as a Work Access, or (y) Landlord requires such access to gain entry to a Reserved Area (and Landlord has no other means to access such Reserved Area on a commercially reasonable basis).

# 9.2. <u>Landlord's Obligation to Minimize Interference; Repair of Damage.</u>

Subject to Section 8.4 hereof, Landlord shall use commercially reasonable efforts to minimize interference with Tenant's use of the Premises and the conduct of Tenant's business operation therein in connection with Landlord's accessing the Premises as contemplated by Section 9.1 hereof or any work performed by Landlord as contemplated by Section 9.4 or Section 9.6(B) hereof. Any damage to the Premises or to Tenant's Property resulting from Landlord's exercise of its rights under this Article 9 shall be repaired and the Premises restored to its condition prior to such damage promptly by and at the expense of Landlord (and such expense shall be included in or excluded from Operating Expenses to the extent in accordance with the provisions of Article 2 hereof).

## 9.3. <u>Reserved Areas.</u>

The Premises shall not include (i) the demising walls of the Premises (except for the interior face thereof), (ii) the walls of the Premises that constitute the curtain wall for the Building (except for the interior face thereof), (iii) balconies, terraces and roofs that are adjacent to the Premises, and (iv) space that is used for Building Systems or other purposes associated with the operation, repair, management or maintenance of the Real Property, including, without limitation, shafts, stacks, stairways, chutes, pipes, conduits, ducts, fan rooms, mechanical rooms, plumbing facilities, and service closets (except to the extent expressly provided herein) (the areas described in clauses (iii) and (iv) above being collectively referred to herein as the "Reserved Areas").

## 9.4. <u>Ducts, Pipes and Conduits.</u>

Subject to Section 9.1 hereof, Landlord shall have the right to install, use and maintain ducts, cabling, pipes and conduits in and through the Premises, provided that (a) except to the extent the nature of a particular room would not reasonably require such installations to be concealed (for example, a mechanical room), such ducts, cabling, pipes and conduits are concealed within or above partitioning columns or ceilings, (b) such ducts, cabling, pipes and conduits do not reduce the Usable Area of the Premises by more than a *de minimis* amount, and (c) Landlord installs such ducts, cabling, pipes and conduits in a manner that minimizes, to the extent reasonably practicable, any adverse effect on an Alteration theretofore performed in the Premises.

#### 9.5. <u>Keys.</u>

Tenant shall provide Landlord, from time to time, with the keys to the Premises (or with the appropriate means to access the Premises using Tenant's electronic security systems); provided, however, that Tenant shall not be required to give Landlord the keys for (or such other appropriate means of access for) any portions of the Premises with respect to which Tenant has certain security or confidentiality requirements such that Tenant requires that such portions of the Premises be locked and inaccessible to persons unauthorized by Tenant (herein called the "Secure Areas"), provided that such portions of the Premises shall not exceed twenty-five percent (25%) of the Rentable Area of the Premises in the aggregate. Landlord agrees that Landlord's right of access to the Secure Areas shall be restricted subject to the following conditions: (a) Tenant shall deliver to Landlord floor plans or other descriptions designating the Secure Areas, (b) except in cases of emergency, any access to the Secure Areas by Landlord shall be upon not less than twenty-four (24) hours notice to Tenant (which notice may be oral), and accompanied by a representative of Tenant, whom Tenant agrees to make available, and (c) Landlord shall have no obligation to provide to the Secure Areas cleaning services or any other services or repairs that require access to the Secure Areas unless Tenant shall provide Landlord with such access to the Secure Areas for purposes of providing such cleaning services or other services or repairs.

## 9.6. <u>Landlord's Changes.</u>

- (A) Subject to Section 9.6(B) hereof, Tenant shall have the right to use, in common with the other occupants of the Building, the portions of the Building that Landlord dedicates from time to time as common area for the general use of the occupants of the Building. Tenant acknowledges that subject to Section 3.12 hereof, the lobbies of the Building that front on Beacon Court are not currently for the general use of the occupants of the Building, and, accordingly, Tenant will not have access to such lobbies.
- (B) Subject to Section 4.2 hereof, Landlord, from time to time, shall have the right to change the arrangement or location of the public portions of the Building, including, without limitation, lobbies, entrances, passageways, doors, corridors, stairs and toilets that in each case are not located in the Premises, it being agreed that any change to the Lexington Avenue lobby of the Building shall only be *de minimis* in nature. In addition, but subject to Landlord's rights under Section 4.2 hereof with respect to passenger elevators, Landlord shall not

make any change pursuant to this Section 9.6 to the extent that such change (a) unreasonably reduces, interferes with or deprives Tenant of access to or egress from the Building or the Premises, (b) reduces the Rentable Area of the Premises (except by a *de minimis* amount), (c) otherwise unreasonably affects any other rights of Tenant under this Lease, (d) adversely affects Tenant's Alterations or the use thereof in any material respect, or (e) diminishes the visibility of any Tenant signage on the exterior of the Building or in the lobby of the Building. Landlord shall be required to perform all such work during hours other than Business Hours if the nature of such work is such as to customarily be performed by landlords of first-class buildings in midtown Manhattan during hours other than Building Hours. Except in cases of emergency, Landlord shall give Tenant not less than twenty (20) days of prior notice of any such work which will, or is reasonably likely to, affect the Premises or Tenant's use and enjoyment thereof.

- (C) Except as otherwise provided in Section 4.2 hereof, Landlord, from time to time, shall have the right to change, or reduce the number of, the passenger or freight elevators serving the Premises, provided that such change or reduction does not reduce to a material extent the passenger or freight elevator service standards that the passenger and freight elevators meet on the date hereof.
  - (D) Landlord, from time to time, shall have the right to change the name, number or designation by which the Building is commonly known.
- (E) Landlord shall have the right, from time to time, to close, obstruct or darken the windows of the Premises temporarily to the extent required to comply with a Requirement or to perform repairs, maintenance, alterations, or improvements to the Building. Landlord shall have the right to close, obstruct or darken the windows of the Premises permanently to the extent required to comply with a Requirement that does not become applicable to the Building by virtue of Landlord's performance of elective construction in the Building.
- (F) If, at any time, the windows of the Premises are closed, obstructed or darkened temporarily, as aforesaid, then Landlord shall perform (or cause to be performed) such repairs, maintenance, alterations or improvements, or shall comply with the applicable Requirement (or cause such Requirement to be complied with), in each case with reasonable diligence, and otherwise take such action as may be reasonably necessary to minimize the period during which such windows are temporarily closed, obstructed or darkened (it being understood, however, that subject to Section 8.4 hereof, Landlord shall not be required to perform such repairs, maintenance, alterations or improvements using contractors or labor at overtime or premium pay rates).

## 9.7. <u>Tenant's Access.</u>

Tenant shall have access to the Premises twenty-four (24) hours per day, seven (7) days per week, on all days.

#### Article 10\_ UNAVOIDABLE DELAYS AND INTERRUPTION OF SERVICE

#### 10.1. <u>Unavoidable Delays.</u>

Subject to Section 10.3 hereof, Article 15 hereof and Article 16 hereof, this Lease and the obligation of Tenant to pay Rental hereunder and to perform all of Tenant's other covenants shall not be affected, impaired or excused, and Landlord shall not have any liability to Tenant, to the extent that Landlord is unable to perform Landlord's covenants under this Lease by reason of an Unavoidable Delay. The term "Unavoidable Delay" shall mean any cause beyond Landlord's reasonable control, including, without limitation, strikes, labor troubles, acts of terrorism or the occurrence of an act of God; provided, however, that Landlord's failure to make a payment of money (including any failure to satisfy a lien, judgment or other monetary obligation), or any other event that derives from Landlord's lack of funds, shall not constitute an Unavoidable Delay for purposes hereof. An event shall constitute an Unavoidable Delay only to the extent that the effect thereof cannot be mitigated in a reasonably practicable manner by Landlord on commercially reasonable terms (with the understanding that Landlord's failure to perform work on an overtime basis shall not constitute the basis for concluding that Landlord could have mitigated the effects of an event that otherwise constitutes the basis for an Unavoidable Delay).

#### 10.2. <u>Interruption of Services.</u>

(A) Subject to Section 10.3 hereof, Landlord, from time to time, shall have the right to interrupt or curtail the level of service provided by the Building Systems to the extent reasonably necessary to accommodate the performance of repairs, additions, alterations, replacements or improvements that in Landlord's reasonable judgment are desirable or necessary. Landlord shall coordinate with Tenant any such interruption or curtailment of such service in order to minimize to the extent reasonably practicable any interference with the conduct of Tenant's business in the Premises. Landlord shall interrupt or curtail the level of service provided by the Building Systems only during hours other than Business Hours if otherwise such interruption or curtailment would materially interfere with Tenant's conduct of business in the Premises; provided, however, that the time periods during which Landlord has the right to interrupt or curtail Building Systems service shall not be so limited to the extent (a) reasonably necessary by reason of the occurrence of an emergency, or (b) otherwise required by applicable Requirements. Landlord shall give Tenant at least ten (10) Business Days of advance notice of Landlord's plans to interrupt or curtail Building Systems service as contemplated by this Section 10.3, further provided, however, that Landlord shall not be required to give such advance notice to the extent such interruption or curtailment is required by reason of the occurrence of an emergency (in which case Landlord shall give Tenant such advance notice of such interruption or curtailment that is reasonable under the circumstances). Landlord shall set forth in any such notice Landlord's estimate in good faith of the duration of the interruption or curtailment of Building Systems services as described therein (it being understood Landlord shall in no event be liable to Tenant if the actual duration of such interruption or curtail such Building Systems services).

(B) If Landlord gives Tenant written notice of any aforesaid interruption or curtailment of Building Systems services, Tenant shall have the right by notice given by Tenant to Landlord no later than five (5) Business Days after the date Landlord gave Tenant such notice, to require that Landlord postpone by up to thirty (30) days the particular interruption or curtailment of Building Systems service described in Landlord's notice (it being agreed that Tenant shall set forth in such notice the number of days by which Tenant is electing to require Landlord to postpone such proposed interruption or curtailment). Landlord shall not be required to postpone such proposed interruption or curtailment of Building Systems service if such postponement would (i) adversely affect the applicable Building System by more than a *de minimis* amount, or (ii) threaten the health or safety of any occupant of the Building or otherwise adversely affect to more than a *de minimis* extent the conduct of business in any portion of the Building by another tenant or occupant in the Building.

#### 10.3. Rent Credit.

- (A) The term "Abatement Event" shall mean an event that has a material and adverse effect on Tenant's ability to operate its business in the Premises (or a portion thereof) during Building Hours on a particular Business Day (any such Business Day being referred to herein as an "Interference Day"; the Premises or the portion thereof with respect to which the Abatement Event has a material and adverse effect on Tenant's ability to operate its business therein, as the case may be, being referred to herein as an "Interference Area"). If an Abatement Event occurs, then Tenant shall be entitled to a credit to apply against the Rental otherwise due hereunder in an amount equal to the product obtained by multiplying (I) the Abatement Percentage, by (II) the quotient obtained by dividing (a) the applicable Rent Per Square Foot, by (b) Three Hundred Sixty-Five (365), by (III) the number of square feet of Rentable Area in the Interference Area, by (IV) either (i) the number of Business Days in the period beginning on the sixth (6th) consecutive Business Day that constitutes an Interference Day and ending on the day immediately preceding the first subsequent Business Day that is not an Interference Day, or (ii) the excess of the number of Interference Days in any particular period of twenty (20) consecutive Business Days over five (5), as the case may be. Tenant acknowledges that the Utility Company's failure to provide electricity, gas, steam or other similar service shall not constitute the basis for an Abatement Event. As used herein, the term "Abatement Percentage" shall mean fifty percent (50%), except that if Tenant does not use the Interference Area for the operation of Tenant's business on an Interference Day solely by reason of the occurrence of the applicable Abatement Event, then the Abatement Percentage in respect of such Interference Day shall be one hundred percent (100%). Either party shall have the right to submit a dispute between the parties regarding the Rental credit described in this Section 10.3 to an
  - (B) The term "Rent Per Square Foot" shall mean, at any particular time:
  - (i) with respect to the Premises, the sum of (I) the aggregate Fixed Rent therefor, at any particular time, divided by the number of square feet of Rentable Area thereof (excluding the Mailroom and the Storage Space), and (II) the Escalation Rent Per Square Foot; and

(ii) with respect to any Option Space that comprises the twenty-ninth (29<sup>th</sup>) floor of the Building, the sum of (i) the Fixed Rent therefor (as determined pursuant to Article 18 hereof), at any particular time, divided by the number of square feet of Rentable Area thereof, and (ii) the Escalation Rent Per Square Foot.

The term "Escalation Rent Per Square Foot" shall mean, at any particular time, the quotient obtained by dividing (I) the Escalation Rent payable hereunder at such time for the Premises or the aforesaid Option Space, as the case may be, by (II) the Rentable Area of the Premises or such Option Space, as the case may be.

## Article 11\_ REQUIREMENTS

## 11.1. <u>Tenant's Obligation to Comply with Requirements.</u>

- (A) Subject to the terms of this Article 11, Tenant, at Tenant's expense, shall comply with all Requirements applicable to the Premises, including, without limitation, (i) Requirements that are applicable to the performance of Alterations, (ii) Requirements that become applicable by reason of Alterations having been performed, and (iii) Requirements that are applicable by reason of the specific nature or type of business operated by Tenant (or any other Person claiming by, through or under Tenant) in the Premises. Tenant shall not be required to make any Alteration or repair to the structural components of the Building Systems in either case to comply with any Requirement unless (a) such Alteration or repair is required by reason of Alterations having been performed by Tenant (or another Person claiming by, through or under Tenant), or (b) such Alteration or repair is required by reason of the specific nature of the use of the Premises by Tenant (or such other Person) (as opposed to the use of the Premises for the general purposes otherwise permitted under Section 3.1 hereof).
- (B) The term "Requirements" shall mean, collectively, all present and future laws, rules, orders, ordinances, regulations, statutes, requirements, codes and executive orders of all Governmental Authorities, and of any applicable fire rating bureau, or other body exercising similar functions.
- (C) The term "Governmental Authority" shall mean the United States of America, the State of New York, The City of New York, any political subdivision thereof and any agency, department, commission, board, bureau or instrumentality of any of the foregoing, or any quasi-governmental authority, now existing or hereafter created, having jurisdiction over the Real Property or any portion thereof.

# 11.2. <u>Landlord's Obligation to Comply with Requirements.</u>

Landlord shall comply with all Requirements applicable to the Premises and the Building other than the Requirements with respect to which Tenant is required to comply pursuant to Section 11.1 hereof, subject, however, to Landlord's right to contest in good faith the applicability or legality thereof.

## 11.3. <u>Tenant's Right to Contest Requirements.</u>

Subject to the provisions of this Section 11.3, Tenant, at Tenant's expense, may contest by appropriate proceedings prosecuted diligently and in good faith the legality or applicability of any Requirement affecting the Premises or that require a Building Change (any such proceedings instituted by Tenant being referred to herein as a "Compliance Challenge"). Tenant shall not have the right to institute a Compliance Challenge unless Tenant first gives Landlord notice thereof. Tenant shall not institute any Compliance Challenge if, by reason of Tenant's delaying its compliance with the applicable Requirement or by reason of the Compliance Challenge, (a) Landlord (or any Landlord Indemnitee) may be imprisoned, (b) the Real Property or any part thereof may be condemned or vacated, or (c) the certificate of occupancy for the Premises or the Building may be suspended. If Landlord or any Landlord Indemnitee may be subject to any civil fines or penalties or other criminal penalties or if Landlord or any Landlord Indemnitee may be liable to any third party in either case by reason of Tenant's delaying its compliance with the applicable Requirement or by reason of the Compliance Challenge, then Tenant shall furnish to Landlord a cash deposit or a bond of a surety company reasonably satisfactory to Landlord, in form and substance reasonably satisfactory to Landlord, or another security reasonably satisfactory to Landlord, in an amount equal to one hundred twenty percent (120%) of the sum of (A) the cost of such compliance, (B) the criminal or civil penalties or fines that may accrue by reason of such non-compliance (as reasonably estimated by Landlord), and (C) the amount of such liability to third parties (as reasonably estimated by Landlord); provided, however, that Tenant shall not be required to post such bond if Tenant then satisfies the Net Worth Requirement. If Tenant initiates any Compliance Challenge, then Tenant shall keep Landlord advised regularly as to the status of such proceedings.

#### 11.4. Requirements During Last Two Years of Term.

Tenant shall not be required to comply with any Requirement that is enacted or first enforced during the last twenty-four (24) months of the Term (as the Term may be extended pursuant to Article 19 hereof) that Tenant would otherwise be obligated to comply with pursuant to this Lease, if:

- (1) such Requirement is in respect of the interior of the Premises, is not related to hazardous materials, and is not in respect of the structure of the Building or the structure of the Premises;
- (2) Tenant's failure to comply with such Requirement would not have an adverse effect on Building Systems or on the health or safety of any occupant of or visitor to the Building;
- (3) Tenant's failure to comply with such Requirement would not cause Landlord to be in default in respect of its obligations under this Lease or any other lease or occupancy agreement in the Building;
- (4) Tenant's failure to comply with such Requirement would not (i) subject Landlord (or any Landlord Indemnitee) to the possibility of being imprisoned, or (ii) subject the Real Property or any part thereof to the possibility of being condemned or vacated;

- (5) Tenant would not be obligated to comply with such Requirement if the Alterations (other than in the core restrooms) in the Premises and other items in the Premises that would be required to be replaced in connection with a new buildout of the Premises (for example, sprinkler heads) were to be demolished after the Premises are surrendered to Landlord (whether or not such demolition actually occurs); and
- (6) the failure to comply with such Requirement would not constitute a default under any Mortgage, Superior Lease, or the Condominium Documents or prevent or impair Landlord's ability to obtain an amendment to the certificate of occupancy for the Building or to obtain any permits or approvals from any Governmental Agency with respect to the Building.

#### 11.5. <u>Certificate of Occupancy.</u>

- (A) Landlord covenants that as of the date hereof a temporary certificate of occupancy, issued by the Department of Buildings of The City of New York is in full force and effect for the Building, which permits Tenant to commence its Initial Alterations in the Premises but which does not authorize any Person to use or occupy the Premises. Landlord, at Landlord's cost and expense, shall cause such certificate of occupancy or any permanent certificate of occupancy covering the Premises (or such other certificate as may be required by Requirements from time to time to lawfully occupy the Premises to be renewed from time to time prior to the expiration thereof. Tenant, at Tenant's cost and expense, shall be required to obtain an amendment to the certificate of occupancy to permit the Premises to be used for the purposes described in Article 3 hereof after the time that the Initial Alterations have been completed to the extent required under applicable Requirements to obtain such an amendment. Upon the request of Tenant, Landlord shall join in and promptly execute any applications Tenant submits in connection with obtaining any amendment to the certificate of occupancy pursuant to this Section 11.5 (provided that the applicable Requirement requires Landlord to join in such application) and shall otherwise cooperate with Tenant in connection therewith (it being agreed that Tenant shall reimburse Landlord for all reasonable, out-of-pocket costs incurred by Landlord in connection with such cooperation on or prior to thirty (30) days after Landlord's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon). Nothing contained herein constitutes Landlord's covenant, representation or warranty that the Premises or any part thereof lawfully may be used or occupied for any particular purpose or in any particular manner, and Landlord shall have no liability to Tenant under this Section 11.5(A) to the extent the certificate of occupa
- (B) Tenant shall use the Premises only in a manner that conforms with the certificate of occupancy that is in effect for the Premises. Tenant, at Tenant's cost and expense, shall have the right to amend the certificate of occupancy to permit the Premises to be used for any of the general purposes described in Article 3 hereof.
- (C) Upon the request of Tenant, Landlord shall join in and promptly execute any applications Tenant submits in connection with obtaining any public assembly permit required to be obtained by Tenant in connection with any intended use of the Premises that is

permitted pursuant to Article 3 hereof and shall otherwise cooperate with Tenant in connection therewith (it being agreed that Tenant shall reimburse Landlord for all reasonable, out-of-pocket costs incurred by Landlord in connection with such cooperation on or prior to thirty (30) days after Landlord's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon).

## Article 12\_ QUIET ENJOYMENT

## 12.1. Quiet Enjoyment.

Landlord covenants that Tenant may peaceably and quietly enjoy the Premises for the Term, subject, nevertheless, to the terms and conditions of this Lease.

## Article 13\_ SUBORDINATION

#### 13.1. <u>Subordination.</u>

- (A) This Lease shall be subject and subordinate to the priority of each Superior Lease if the applicable Lessor executes and delivers to Tenant a Nondisturbance Agreement. This Lease shall be subject and subordinate to the lien of each Mortgage if the applicable Mortgage executes and delivers to Tenant a Nondisturbance Agreement. This Lease shall be subject and subordinate to a Condominium Declaration if the Condominium Board executes and delivers to Tenant a Nondisturbance Agreement (with the understanding that Landlord shall not permit any such Condominium Declaration to decrease or otherwise impair Tenant's rights under this Lease, or to increase or otherwise expand Tenant's obligations and liabilities under this Lease, except, in either case, to a *de minimis* extent) in the form annexed as Exhibit "13.1" attached hereto and made a part hereof. Tenant, at Tenant's expense, shall execute and deliver promptly a Nondisturbance Agreement that a Lessor or a Mortgagee proposes to use and that conforms to the terms of this Article 13 and any Nondisturbance Agreement delivered by the Condominium Board in the form attached as Exhibit "13.1" hereto. Landlord agrees to deliver to Tenant as of the date hereof a fully executed Nondisturbance Agreement from each of the Condominium Board, the Lessor under any existing Superior Lease, and the Mortgagee under any existing Mortgage, in any such case in accordance with the terms of this Article 13.
  - (B) The term "By-Laws" shall mean the by-laws of the Condominium.
  - (C) The term "Condominium Board" shall mean the board that governs the business and affairs of the Office Unit.
  - (D) The term "Condominium Documents" shall mean the Declaration and the By-Laws.
  - (E) The term "Lessor" shall mean a lessor under a Superior Lease.

- (F) The term "Mortgage" shall mean any trust indenture or mortgage which now or hereafter encumbers the Real Property, the Building or any Superior Lease and the leasehold interest created thereby or the Office Unit.
  - (G) The term "Mortgagee" shall mean any trustee, mortgagee or holder of a Mortgage.
- (H) The term "Nondisturbance Agreement" shall mean, subject to Section 13.2 hereof, an agreement, in recordable form, between a Lessor, a Mortgagee, or the Condominium Board, as the case may be, and Tenant, to the effect that (i) if there is a foreclosure of the Mortgage, then the successor to Landlord by virtue of the foreclosure will not evict Tenant, disturb Tenant's possession under this Lease, or terminate or disturb Tenant's leasehold estate or rights hereunder, and will recognize Tenant as the direct tenant of such successor to Landlord on the same terms and conditions as are contained in this Lease, (ii) if the Superior Lease terminates, then the Lessor will not evict Tenant, disturb Tenant's possession under the Lease, or terminate or disturb Tenant's leasehold estate or rights hereunder, and will recognize Tenant as the direct tenant of such Lessor on the same terms and conditions as are contained in this Lease, or (iii) if there is a sale of the Premises by virtue of the Condominium Board exercising a power of sale that is granted under the Condominium Declaration, then the successor to Landlord by virtue of such sale will not evict Tenant, disturb Tenant's possession under the Lease, or terminate or disturb Tenant's leasehold estate or rights hereunder, and will recognize Tenant as the direct tenant of such successor on the same terms and conditions as are contained in this Lease.
- (I) The term "Superior Lease" shall mean any lease pursuant to which Landlord now or hereafter obtains or retains its interest in the Real Property or the Building (to the extent that Landlord's interest in the Real Property is a leasehold estate).

#### 13.2. <u>Terms of Nondisturbance Agreements.</u>

Subject to the terms of this Section 13.2, any Nondisturbance Agreement (other than the Nondisturbance Agreement from the Condominium Board, which shall be in the form set forth on Exhibit "13.1" attached hereto), may provide that the Person that succeeds to Landlord by reason of the foreclosure of a Mortgage, the termination of a Superior Lease, or the exercise of the power of sale as set forth in the Condominium Declaration, as the case may be (any such Person being referred to herein as the "Successor") shall not be:

- (A) liable for any act or omission of any prior landlord (including, without limitation, the then defaulting landlord), except to the extent that (i) such act or omission continues after the date that the Successor succeeds to Landlord's interest in the Real Property, and (ii) such act or omission of such prior landlord is of a nature that the Successor can cure by performing a service or making a repair, or
- (B) subject to any defenses or offsets that Tenant has against any prior landlord (including, without limitation, the then defaulting landlord) (except for any offsets that are expressly permitted under this Lease), or

- (C) bound by any payment of Rental that Tenant has made to any prior landlord (including, without limitation, the then defaulting landlord) more than thirty (30) days in advance of the date that such payment is due, except to the extent that such advance payment of Rental is expressly required by this Lease or
- (D) bound by any obligation to make any payment to or on behalf of Tenant (other than the Tenant Fund) to the extent that such obligation accrues prior to the date that the Successor succeeds to Landlord's interest in the Real Property (it being agreed that Tenant's sole remedy for any Successor's failure to pay the Tenant Fund (or a portion thereof) shall be to exercise its right of offset pursuant to Section 7.14 hereof), or
  - (E) bound by any obligation to perform any work or to make improvements to the Premises, except for:
- (1) repairs and maintenance that Landlord is required to perform pursuant to the provisions of this Lease and the need for which continues after the date that the Successor succeeds to Landlord's interest in the Real Property,
- (2) repairs to the Premises that become necessary by reason of a fire or other casualty that occurs from and after the date that the Successor succeeds to Landlord's interest in the Real Property and that Landlord is required to perform pursuant to Article 15 hereof,
- (3) repairs to the Premises that become necessary by reason of a fire or other casualty that occurs prior to the date that the Successor succeeds to Landlord's interest in the Real Property and that Landlord is required to perform pursuant to Article 15 hereof, to the extent that the Successor can make such repairs from the net proceeds of Landlord's Property Policy that are actually made available to the Successor (with the understanding, however, that if (i) a fire or other casualty occurs prior to the date that the Successor succeeds to Landlord's interest in the Real Property, (ii) Landlord is required to repair the resulting damage to the Building pursuant to Article 15 hereof, and (iii) the Successor cannot make such repairs from such net proceeds, then Tenant shall have the right to terminate this Lease by giving notice thereof to the Successor within fifteen (15) days after the date that the Successor gives Tenant notice that the Successor does not intend to perform such repairs),
- (4) repairs to the Premises as a result of a partial condemnation that occurs from and after the date that the Successor succeeds to Landlord's interest in the Real Property and that Landlord is required to perform pursuant to Article 16 hereof, and
- (5) repairs to the Premises as a result of a partial condemnation that occurs prior to the date that the Successor succeeds to Landlord's interest in the Real Property and that Landlord is required to perform pursuant to Article 16 hereof, to the extent that the Successor can make such repairs from the net proceeds of any condemnation award made available to the Successor (with the understanding, however, that if (i) a partial condemnation occurs prior to the date that the Successor succeeds to Landlord's interest in the Real Property, (ii) Landlord is required to make repairs to the Building pursuant to Article 16 hereof by reason of such partial condemnation, and (iii) the Successor cannot make such repairs from such net proceeds, then Tenant shall have the right to terminate this Lease by giving notice thereof to the Successor

within fifteen (15) days after the date that the Successor gives Tenant notice that the Successor does not intend to perform such repairs), or

(F) bound by any amendment or modification of this Lease (other than an amendment or modification expressly provided for in this Lease) made without the consent of the Mortgagee or the Lessor, as the case may be (the aforesaid items in clause (A) through (F) above for which a Successor is not liable being referred to herein as the "Successor Limitation Items").

A Successor that is an Affiliate of the Person that constitutes Landlord shall not have the right to include in a Nondisturbance Agreement the Successor Limitation Items. Each party shall have the right to submit any dispute over whether a Nondisturbance Agreement meets the criteria set forth in this Article 13 hereof to an Expedited Arbitration Proceeding.

#### 13.3. Attornment.

- (A) If, at any time prior to the Expiration Date, a Successor succeeds to Landlord's interest in the Real Property, then Tenant, at the Successor's election, shall attorn, from time to time, to the Successor, in either case upon the then executory terms of this Lease, for the remainder of the Term. If the Successor is not an Affiliate of the Person that constituted Landlord immediately prior to such Successor's obtaining an interest in the Premises, then the Successor shall not have liability for the Successor Limitation Items from and after the date that Tenant so attorns to the Successor.
- (B) The provisions of this Section 13.3 shall apply notwithstanding that, as a matter of law, this Lease terminates upon the termination of any Superior Lease or the foreclosure of a Mortgage. No further instrument shall be required to give effect to Tenant's attorning to a Successor as contemplated by this Section 13.3. Tenant, however, upon demand of any Successor, shall execute, from time to time, instruments, in a recordable form and in a form reasonably satisfactory to the Successor, confirming the foregoing provisions of this Section 13.3. A Mortgagee, a Lessor, or the Condominium Board shall have the right to include such provisions in a Nondisturbance Agreement.

## 13.4. <u>Amendments to this Lease.</u>

Tenant shall execute and deliver, from time to time, amendments to this Lease, promptly after Landlord's request, to the extent that (x) such amendments are reasonably required by a Mortgagee or a Lessor that in either case is not an Affiliate of Landlord (or are reasonably required by a proposed Mortgagee or proposed Lessor that in either case is not an Affiliate of Landlord and that consummates the applicable Mortgage or the applicable Superior Lease contemporaneously with Tenant's execution and delivery of such amendment hereof), and (y) Landlord gives to Tenant reasonable evidence to the effect that such Mortgagee or Lessor requires such amendments; provided, however, that Tenant shall not be required to agree to any such amendments to this Lease that (i) increase Tenant's monetary obligations under this Lease, (ii) adversely affects or diminishes, in either case to by more than a *de minimis* extent, Tenant's rights under this Lease, or (iii) increase Tenant's other obligations under this Lease by more than a *de minimis* extent.

# 13.5. <u>Tenant's Estoppel Certificate.</u>

Tenant, within ten (10) Business Days after Landlord's request from time to time (but not more frequently than three (3) times in any particular period of twelve (12) months), shall deliver to Landlord a written statement executed by Tenant, in form reasonably satisfactory to Landlord, (1) stating that this Lease is then in full force and effect and has not been modified (or if this Lease is not in full force and effect, stating the reasons therefor, or if this Lease is modified, setting forth all modifications), (2) setting forth the date to which the Fixed Rent, the Escalation Rent and other items of Rental have been paid, (3) stating whether, to the actual knowledge of Tenant, Landlord is in default under this Lease, and, if Landlord is in default, setting forth the specific nature of all such defaults, and (4) stating any other matters reasonably requested by Landlord and related to this Lease. Tenant acknowledges that any such statement that Tenant delivers to Landlord pursuant to this Section 13.5 may be relied upon by (x) any purchaser or owner of the Real Property or any interest therein (including, without limitation, any Lessor), or (y) any Mortgagee.

## 13.6. <u>Landlord's Estoppel Certificate.</u>

Landlord, within ten (10) Business Days after Tenant's request from time to time (but not more frequently than three (3) times in any particular period of twelve (12) months), shall deliver to Tenant a written statement executed by Landlord, in form reasonably satisfactory to Tenant, (i) stating that this Lease is then in full force and effect and has not been modified (or if this Lease is not in full force and effect, stating the reasons therefor, or if this Lease is modified, setting forth all modifications), (ii) setting forth the date to which the Fixed Rent, the Escalation Rent and any other items of Rental have been paid, (iii) stating whether, to the actual knowledge of Landlord, Tenant is in default under this Lease, and, if Tenant is in default, setting forth the specific nature of all such defaults, and (iv) stating any other matters reasonably requested by Tenant and related to this Lease. Landlord acknowledges that any statement delivered by Landlord to Tenant pursuant to this Section 13.6 may be relied upon by (w) any assignee of Tenant's interest hereunder, (x) any subtenant of all or any part of the Premises, (y) any Person that acquires Control of Tenant (provided that such assignment, sublease or transfer of Control is accomplished in a manner that complies with the provisions of Article 16 hereof), or (z) any Person that extends credit to Tenant.

## 13.7. Rights to Cure Landlord's Default.

If (x) a Superior Lease or Mortgage exists, (y) the Lessor or Mortgagee is not an Affiliate of Landlord, and (z) Landlord gives Tenant notice thereof, then Tenant shall not seek to terminate this Lease by reason of Landlord's default hereunder until Tenant has given written notice of such default to such Lessor or such Mortgagee in either case at the address that has been furnished to Tenant. If any such Lessor or Mortgagee notifies Tenant, within ten (10) Business Days after the date that such Lessor or Mortgagee receives such notice from Tenant, that such Lessor or Mortgagee intends to remedy such act or omission of Landlord, then Tenant shall not have the right to so terminate this Lease unless such Lessor or Mortgagee fails to remedy such act or omission of Landlord within a reasonable period of time after the date that such Lessor or Mortgagee gives such notice to Tenant (it being understood that such Lessor or Mortgagee shall not have any liability to Tenant for the failure of such Lessor or Mortgagee to so

remedy such act or omission of Landlord during such period). If there is any conflict between the provisions of this Section 13.7 and the terms and provisions of any Nondisturbance Agreement, then the provisions of the Nondisturbance Agreement shall control.

#### 13.8. Zoning Lot Merger Agreement.

Tenant hereby waives irrevocably any rights that Tenant may have in connection with any zoning lot merger or transfer of development rights with respect to the Real Property, including, without limitation, any rights that Tenant may have to be a party to, to contest, or to execute any Declaration of Restrictions (as such term is used in Section 12-10 of the Zoning Resolution of The City of New York effective December 15, 1961, as amended) with respect to the Real Property, which would cause the Premises to be merged with or unmerged from any other zoning lot pursuant to such Zoning Resolution or to any document of a similar nature and purpose. Tenant agrees that this Lease shall be subject and subordinate to any Declaration of Restrictions or any other document of similar nature and purpose now or hereafter affecting the Real Property (it being understood, however, that Landlord shall not permit such Declaration of Restrictions or any such other document to impair Tenant's rights hereunder, or expand Tenant's obligations hereunder, except, in either case, to a *de minimis* extent). In confirmation of such subordination and waiver, Tenant, from time to time and at no out-of-pocket cost to Tenant, shall execute and deliver promptly any certificate or instrument that Landlord reasonably requests.

# 13.9. <u>Existing Mortgages and Existing Superior Leases.</u>

Landlord hereby represents and warrants to Tenant that the only Mortgages and Superior Leases that exist as of the date hereof are as described in Exhibit "13.9" attached hereto and made a part hereof.

#### 13.10. Condominium.

- (A) If there is any inconsistency between the terms and provisions of this Lease and the terms and provisions of the Condominium Documents, the provisions of this Lease shall control. Tenant acknowledges that wherever any provision of this Lease grants certain rights to Landlord, the Condominium Board may be entitled to certain corresponding rights pursuant to the Condominium Documents and that although Landlord and Tenant have each endeavored to include all necessary references to the rights of the Condominium Board throughout this Lease, with respect to any provisions that do not expressly reflect such rights, such rights shall be inferred as the context requires. In no event shall the effect of any such inference decrease or otherwise impair Tenant's rights under this Lease, or increase or otherwise expand Tenant's obligations and liabilities under this Lease, except, in either case, to a *de minimis* extent (it being agreed that any dispute with respect to the foregoing may be submitted by Tenant to an Expedited Arbitration Proceeding).
- (B) If during the Term, the Building shall no longer be owned pursuant to a condominium form of ownership in accordance with Article 9.1(B) of the New York Real Property Law, then this Lease shall remain in full force and effect, Landlord and Tenant shall continue to perform their respective obligations under this Lease, and the following shall apply (it being understood that a change in the form of ownership of the Building from a condominium

form of ownership (a "Condominium Withdrawal") shall not result in an increase in the Tax Payments and the Operating Expense Payments payable by Tenant hereunder (but the foregoing shall not be deemed to render void any specific provision of this Lease otherwise providing for the increase or decrease of such Tax Payments or Operating Payments)):

- (1) Tenant's Operating Expense Share and Tenant's Tax Share shall remain the same;
- (2) The Base Operating Expenses shall not be recalculated as a result of such Condominium Withdrawal (but the foregoing shall not render void any specific provisions of this Lease otherwise providing for the recalculation of Base Operating Expenses);
- (3) The Base Taxes shall not be recalculated as the result of any Condominium Withdrawal (but the foregoing shall not render void any specific provision of this Lease otherwise providing for the recalculation of Base Taxes);
- (4) Operating Expenses shall continue to be only those incurred for the Office Unit (as the Office Unit existed immediately prior to such Condominium Withdrawal); and
- (5) Taxes shall continue to be only those incurred for the Office Unit (as the Office Unit existed immediately prior to such Condominium Withdrawal).
  - (C) Any dispute with respect to this Section 13.10 may be submitted to an Expedited Arbitration Proceeding.

#### 13.11. Multiple Owners of the Office Unit.

Tenant acknowledges that Landlord has the right, at any time and from time to time, to consummate a transaction which results in more than one (1) Person owing the fee estate in the Office Unit or a leasehold estate in the Office Unit that is immediately superior to Tenant. If more than one (1) Person owns the fee estate in the Office Unit or if there is a leasehold estate in the Office Unit that is immediately superior to Tenant's leasehold estate, then, subject to the terms hereof, (w) all of the Persons that own such fee estate or superior leasehold estate shall be jointly and severally liable for the obligations of Landlord hereunder (the Persons that own such fee estate or immediately superior leasehold estate being collectively referred to herein as the "Landlord Owners", (x) the Landlord Owners shall designate one (1) of the Landlord Owners that Tenant has the right to contact from time to time to address day-to-day operation and management issues regarding the Premises (including, without limitation, approvals of Landlord as contemplated by this Lease), (y) each Landlord Owner shall be liable for any Landlord Owner's failure to grant an approval to Tenant under this Lease (in cases where (i) Landlord's consent is not to be unreasonably withheld in accordance with the terms hereof, and (ii) such Landlord Owner unreasonably withholds such consent), and (z) Tenant shall be entitled to rely upon an approval or consent granted by the Landlord Owner designated to address day-to-day issues as provided in clause (x) above. Nothing contained in this Section 13.11 limits the provisions of Section 31.2 hereof or Section 32.4 hereof.

#### Article 14\_ INSURANCE

#### 14.1. <u>Tenant's Insurance.</u>

- (A) Tenant, at Tenant's expense, shall obtain and keep in full force and effect (i) an insurance policy for Tenant's Property and any Alterations made by Tenant (including the Initial Alterations), in either case to the extent insurable under the available standard forms of "all-risk" insurance policies, in an amount equal to one hundred percent (100%) of the replacement value thereof (subject, however, at Tenant's option, to a reasonable deductible) (the insurance policy described in this clause (i) being referred to herein as "Tenant's Property Policy."), and (ii) a policy of commercial general liability and property damage insurance on an occurrence basis, with a broad form contractual liability endorsement (the insurance policy described in this clause (ii) being referred to herein as "Tenant's Liability Policy."). Tenant's Property Policy and Tenant's Liability Policy shall name Tenant as the insured. Landlord and any Landlord Indemnitees whose name and address have been furnished to Tenant shall be named as additional insureds on Tenant's Liability Policy. Landlord shall reasonably cooperate with Tenant and Tenant's insurance companies in the adjustment of any claims for any damage to Tenant's Property and Alterations. Tenant shall reimburse Landlord for any reasonable, actual attorney's fees and expenses incurred by Landlord in so cooperating with Tenant no later than thirty (30) days after Landlord's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon.
- (B) Tenant's Liability Policy shall contain a provision that (a) no act or omission of Tenant shall affect or limit the obligation of the insurer to pay the amount of any loss sustained, and (b) the policy is non-cancelable with respect to the Landlord Indemnitees unless at least thirty (30) days of advance written notice is given to Landlord, except that Tenant's Liability Policy may be cancelable on no less than ten (10) days of advance written notice to Landlord for non-payment of premium. If Tenant receives any notice of cancellation or any other notice from the insurance carrier which may adversely affect the coverage of the insureds under Tenant's Property Policy or Tenant's Liability Policy, then Tenant shall immediately deliver to Landlord a copy of such notice. The minimum amounts of liability under Tenant's Liability Policy shall be a combined single limit with respect to each occurrence in the amount of Ten Million Dollars (\$10,000,000) for injury (or death) to persons and damage to property. Landlord may increase such minimum amounts of liability from time to time to the amounts that are then customarily being required by prudent landlords of first-class buildings in midtown Manhattan from tenants leasing space similar to the Premises. If Tenant disputes any increase in such minimum amounts of liability, then Tenant may submit such dispute to an Expedited Arbitration Proceeding.
- (C) Tenant shall cause Tenant's Liability Policy and Tenant's Property Policy to be issued by reputable and independent insurers that are (x) permitted to do business in the State of New York, and (y) rated in Best's Insurance Guide, or any successor thereto, as having a general policyholder rating of A- and a financial rating of at least XII (it being understood that if such ratings are no longer issued, then such insurer's financial integrity shall conform to the standards that constitute such ratings from Best's Insurance Guide as of the date hereof).

- (D) Tenant has the right to satisfy Tenant's obligation to carry Tenant's Liability Policy with an umbrella insurance policy if such umbrella insurance policy contains an aggregate per location endorsement that provides the required level of protection for the Premises. Tenant has the right to satisfy Tenant's obligation to carry Tenant's Property Policy with a blanket insurance policy if such blanket insurance policy provides, on a per occurrence basis, that a loss that relates to any other location does not impair or reduce the level of protection available for the Premises below the amount required by this Lease.
- (E) Tenant shall have the option, in conjunction with Citigroup Inc. ("Citigroup"), Tenant's ultimate parent corporation (or any subsidiaries or affiliates of Citigroup), to maintain self insurance or to provide or maintain any insurance required under this Lease under blanket insurance policies maintained by Citigroup (or any subsidiaries or affiliates of Citigroup), or to provide or maintain insurance through such alternative risk management programs as Citigroup (or any subsidiaries or affiliates of Citigroup) may provide or participate in from time to time (any such types of insurance programs being referred to herein collectively as "Self Insurance Programs"). Any such Self Insurance Programs shall not operate to decrease the insurance coverage or limits set forth in this Section 14.1. Any such Self Insurance Programs shall be deemed to contain all of the terms and conditions applicable to the requirements for Tenant's insurance contained in this Section 14.1. If Tenant elects to provide its insurance through a Self Insurance Program, then, with respect to any claims which may result from incidents occurring during the Term, such the insurance provided by such Self Insurance Program shall survive the expiration or earlier termination of this Lease to the same extent as the insurance required would survive.

#### 14.2. <u>Landlord's Insurance.</u>

(A) Subject to the terms of this Section 14.2, Landlord shall obtain and keep in full force and effect during the Term: (x) insurance against loss or damage by fire and other casualty to the Building, to the extent insurable on commercially reasonable terms under then available standard forms of "all-risk" insurance policies, in an amount equal to one hundred percent (100%) of the replacement value thereof or, at Landlord's option, in such lesser amount as will avoid co-insurance, and (y) insurance against rental loss deriving from a fire or other casualty in clause (x) hereof in an amount that is reasonably expected to cover thirty-six (36) months of rental loss from the Premises (such insurance described in the immediately preceding clause (x) and clause (y) being collectively referred to herein as "Landlord's Property Policy"). Tenant acknowledges that (i) Landlord's Property Policy may encompass rent insurance, and (ii) the risks that Landlord's Property Policy covers may include, without limitation, fire, terrorism, environmental matters, and flood. Landlord shall have the right to obtain and maintain as part of Landlord's Property Policy or otherwise insurance against loss or damage to the Building (excluding Tenant's Property and Alterations) by terrorism ("Terrorism Insurance") to the extent that such insurance is available and is then customarily carried by prudent owners of first-class buildings in midtown Manhattan. Landlord shall obtain and keep in full force and effect during the Term a valid and enforceable policy ("Landlord's Liability Policy.") of commercial general liability insurance on an occurrence basis, with a broad form contractual liability endorsement. Landlord's Liability Policy and Landlord's Property Policy shall be issued by reputable and independent insurers permitted to do business in the State of New York, and rated in Best's Insurance Guide, or any successor thereto (of if there be none, an organization

having a national reputation) as having a general policyholder of "A" and a financial rating of at least "X". The minimum amount of liability covered by Landlord's Liability Policy shall be a combined single limit with respect to each occurrence in the amount of Fifty Million Dollars (\$50,000,000) for injury (or death) to persons and damage to property, which minimum amount shall be subject to reasonable and customary increases from time to time. Landlord shall name Tenant and Affiliates of Tenant and any Permitted Party that occupy the Premises and of which Tenant gives Landlord notice as an additional insured on Landlord's Liability Policy. Landlord's Liability Policy shall contain a provision that (a) no act or omission of Landlord shall affect or limit the obligation of the insurer to pay the amount of any loss sustained to any Person entitled thereto (other than Landlord or Landlord's Affiliates), and (b) the policy shall be non-cancelable with respect to Tenant unless at least thirty (30) days of prior written notice has been given to Tenant by certified mail, return receipt requested, which notice shall contain the policy number and the names of the insured and additional insureds, except that such thirty (30) day period shall be reduced to ten (10) days in respect of a cancellation that derives from Landlord's failure to pay the premium for such policy when due. Upon receipt by Landlord of any notice of cancellation or any other notice from the issuers of Landlord's Property Policy or Landlord's Liability Policy which may adversely affect the coverage of the insureds under such policy of insurance, Landlord shall immediately deliver to Tenant a copy of such notice.

(B) Landlord shall have the right to provide that the coverage of Landlord's Property Policy and Landlord's Liability Policy be subject to reasonable deductibles. Tenant shall cooperate with Landlord and Landlord's insurance companies in the adjustment of any claims for any damage to the Building. Landlord shall reimburse Tenant for any reasonable, actual attorney's fees and expenses incurred by Tenant in so cooperating with Landlord no later than thirty (30) days after Tenant's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon. Landlord shall not be required to carry insurance on Tenant's Property or the Alterations. Landlord shall not be required to carry insurance against any loss suffered by Tenant due to the interruption of Tenant's business.

# 14.3. <u>Mutual Waiver of Subrogation.</u>

(A) Subject to the provisions of this Section 14.3, Landlord and Tenant shall each obtain (and Landlord shall cause the Condominium Board to obtain) an appropriate clause in, or endorsement on, Landlord's Property Policy or Tenant's Property Policy (as the case may be) or any property insurance policy carried by the Condominium Board pursuant to which the insurance companies waive subrogation or consent to a waiver of right of recovery, with respect to losses covered by Landlord's Property Policy, Tenant's Property Policy, and any property policy covered by the Condominium Board. Landlord and Tenant also agree that, having obtained such clauses or endorsements of waiver of subrogation or consent to a waiver of right of recovery, they shall not make any claim (and Landlord shall cause the Condominium Board not make any claim) against or seek to recover from the Landlord Indemnitees, the Tenant Indemnitees, or the Condominium Board (and the individual members of its board of managers) (as the case may be) for any loss or damage to its property or the property of others resulting from fire or other hazards covered by Landlord's Property Policy, Tenant's Property Policy, or any property policy carried by the Condominium Board (as the case may be).

- (B) If the payment of an additional premium is required for the inclusion of a waiver of subrogation provision as described in Section 14.3(A) hereof, then each aforesaid party shall advise the other parties of the amount of any such additional premiums and such other parties at their own election may, but shall not be obligated to, pay such additional premium. If (x) Tenant is the party that elects to pay such additional premium to include such a waiver in Landlord's Property Policy or any property policy carried by the Condominium Board, and (y) other tenants in the Building make concurrently a similar election, then the aforesaid amount that Tenant is obligated to pay to Landlord or the Condominium Board, as the case may be, on account of such additional premium shall be only the portion thereof that Landlord or the Condominium Board allocates equitably to Tenant. If any party does not elect to pay such additional premium, then the party whose insurer is charging the additional premium shall not be required to obtain such waiver of subrogation provision.
- (C) If any party is unable to obtain the inclusion of such waiver of subrogation provision even with the payment of an additional premium, then such party shall attempt to name the other parties as an additional insured (but not a loss payee) under the applicable insurance policy. If the payment of an additional premium is required for naming the other parties as an additional insured (but not a loss payee), then such party shall advise the other parties of the amount of any such additional premium and the other parties at their own election may, but shall not be obligated to, pay such additional premium. If (x) Tenant is the party that elects to pay such additional premium to name Tenant as an additional insured (but not as loss payee), and (y) other tenants in the Building make concurrently a similar election, then the aforesaid amount that Tenant is obligated to pay to Landlord or the Condominium Board on account of such additional premium shall be only the portion thereof that Landlord or the Condominium Board allocates equitably to Tenant. If such other party does not elect to pay such additional premium or if it is not possible to have the other parties named as additional insureds (but not loss payees), even with the payment of an additional premium, then (in either event) the party whose insurer refuses to include such waiver of subrogation provision shall so notify the other parties and such parties shall not have the obligation to name the other party as an additional insured.

#### 14.4. Evidence of Insurance.

On or prior to the Commencement Date, each party shall deliver to the other party appropriate certificates of insurance required to be carried by the parties pursuant to this Article 14, including evidence of waivers of subrogation and naming of additional insureds in either case as required by Section 14.3 hereof. Each party shall deliver to the other party evidence of each renewal or replacement of a policy at least twenty (20) days prior to the expiration of such policy.

#### 14.5. No Concurrent Insurance.

Tenant shall not obtain any property insurance (under Tenant's Property Policy or otherwise) that covers the property that is covered by Landlord's Property Policy. Landlord shall not obtain any property insurance (under Landlord's Property Policy or otherwise) that covers the property that is covered by Tenant's Property Policy. Landlord shall use reasonable efforts to cause any property insurance obtained by the Condominium Board not to cover the property that is covered by Tenant's Property Policy.

# 14.6. <u>Tenant's Obligation to Comply with Landlord's Fire and Casualty Insurance.</u>

- (A) If (i) Tenant (or any other Person claiming by, through or under Tenant) uses the Premises for any purpose other than general office use and the ancillary uses permitted by Article 3, (ii) the use of the Premises by Tenant (or such other Person) causes the premium for Landlord's Property Policy to exceed the premium that would have otherwise applied therefor if Tenant (or such Person) used the Premises for general office purposes (or such ancillary purposes), and (iii) Landlord notifies Tenant of such fact and Tenant does not as soon as reasonably practicable but in no event more than twenty (20) days thereafter, modify such use so as to prevent the imposition of such increase in premium, then Tenant shall pay to Landlord, as additional rent, an amount equal to such excess, on or prior to the thirtieth (30th) day after the date that Landlord gives to Tenant an invoice therefor (which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon). Nothing contained in this Section 14.6 expands Tenant's rights under Article 3 hereof. Tenant must comply with the requirements of the issuer of Landlord's Property Policy (including, without limitation, any such requirements that such issuer requires as the basis for the premium that such issuer charges Landlord for Landlord's Property Policy, provided that such requirements that the issuer of Landlord's Property Policy imposes are reasonably consistent with the requirements imposed by reputable insurers of comparable properties in The City of New York), if Landlord could not obtain the insurance described in Section 14.2 hereof without Tenant complying with such requirements.
- (B) If (i) Landlord fails to comply with any particular obligation under this Lease, (ii) such failure causes the premium for Tenant's Property Policy to exceed the premium that would have otherwise applied if Landlord had complied with such obligation, and (iii) Tenant notifies Landlord of such fact and Landlord does not as soon as reasonably practicable but in no event more than twenty (20) days thereafter, comply with such obligation so as to prevent the imposition of such increase in premium, then Landlord shall pay to Tenant, as additional rent, an amount equal to such excess, on or prior to the thirtieth (30th) day after the date that Tenant gives to Landlord an invoice therefore (which invoice shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon).

# 14.7. <u>Tenant's Insurance Proceeds.</u>

- (A) If (x) this Lease terminates by reason of the occurrence of a fire or other casualty pursuant to Article 15 hereof, and (y) a Mortgagee that constitutes an Institutional Lender then holds a Mortgage on Landlord's interest in the Premises, then Tenant shall pay the proceeds of Tenant's Property Policy (and any other proceeds to which Tenant is entitled under Section 14.1 hereof) to the extent deriving from such fire or other casualty as follows (it being understood that if Tenant exercises its right to maintain self insurance pursuant to Section 14.1(E) hereof, then Tenant shall pay an amount that equals the proceeds that Tenant would have received if Tenant had obtained and maintained a Tenant's Property Policy in accordance with the provisions of Section 14.1 hereof):
- (1) first, to Landlord (or, at Landlord's option, to such Mortgagee), until the aggregate amount paid to Landlord (or such Mortgagee) from such proceeds equals the excess of (I) the lesser of (x) the amount of the outstanding indebtedness secured by such Mortgage, and

(y) eighty percent (80%) of the fair market value of the Premises on the date that such Mortgagee initially funded the debt secured by such Mortgage, over (II) the proceeds of Landlord's Property Policy that are available to Landlord (or such Mortgagee) by reason of such fire or other casualty for the items Landlord is required to restore pursuant to Section 15.2 hereof (or would have been available to Landlord (or such Mortgagee) if Landlord or the Condominium Board carried Landlord's Property Policy in accordance with the terms hereof); and

- (2) second, to Tenant, until the aggregate amount paid to Tenant from such proceeds equals the sum of (a) the value of any Tenant's Property included in such taking, plus (b) the excess of (I) the Unamortized Alterations Cost, over (II) the Unamortized Tenant Fund Amount; and
  - (3) finally, one-half (1/2) of the remainder to Landlord and one-half (1/2) of the remainder to Tenant.
- (B) Landlord and Tenant shall give the other party reasonable access to the such party's books and records to the extent reasonably necessary to determine the amount due from Tenant to Landlord under this Section 14.7. Either party shall have the right to submit any dispute between the parties regarding the amount due from Tenant to Landlord under this Section 14.7 hereof to an Expedited Arbitration Proceeding.
- (C) The term "Institutional Lender" shall mean a savings bank, a savings and loan association, a commercial bank or trust company (whether acting individually, as a trustee, as a servicing agent or in a fiduciary capacity), a private pension fund, a credit union or credit company, an insurance company, a religious, educational or eleemosynary institution, a federal, state or municipal employee's welfare, benefit, pension or retirement fund, any Governmental Agency or entity insured by a Governmental Agency, any brokerage or investment banking organization (or an Affiliate thereof), whether acting in its own capacity or on behalf of its clients, any real estate mortgage investment conduit or similar investment vehicle), or any combination of Persons that would otherwise constitute Institutional Lenders; provided, however, that (i) if a Person (other than a real estate mortgage investment conduit or similar investment vehicle) shall not constitute an Institutional lender for purposes hereof unless such Person (or such Person and its Affiliates) has net assets of at least Two Hundred Fifty Million Dollars (\$250,000,000), and (ii) an Institutional Lender shall also include any other Person that is generally recognized in the capital markets as an institutional lender from and after the date hereof.
- (D) The term "<u>Unamortized Alterations Cost</u>" shall mean, at any particular time with respect to the Premises or the applicable portion thereof, the aggregate amount theretofore paid by or on behalf of Tenant for Alterations in the Premises or the applicable portion thereof, to the extent that such amount then remains unamortized (assuming that such amount is amortized, in equal monthly installments, using an interest factor equal to the Base Rate, over the period commencing on the date that Tenant makes any such payment for Alterations and ending on the Fixed Expiration Date, except that if the Unamortized Alterations Cost is being determined during the Renewal Term, then such excess shall be deemed to be amortized, in equal monthly installments, using an interest factor equal to the Base Rate, over the

period commencing on the date that Tenant makes any such payment for Alterations and ending on the last day of the Renewal Term).

(E) The term "<u>Unamortized Tenant Fund Amount</u>" shall mean the aggregate amount of the Tenant Fund paid to or on behalf of Tenant, to the extent that such amount then remains unamortized (assuming that such amount is amortized, in equal monthly installments, using an interest factor equal to the Base Rate, over the period commencing on the date that Landlord funded each particular installment of the Tenant Fund amount and in any such case ending on the Fixed Expiration Date).

# Article 15\_ CASUALTY

#### 15.1. <u>Notice.</u>

Tenant shall notify Landlord promptly of any fire or other casualty that occurs in the Premises.

#### 15.2. <u>Landlord's Restoration Obligations.</u>

Subject to the terms of this Section 15.2, Landlord, with reasonable diligence, shall (or shall cause the Condominium Board to) promptly settle any insurance claims and Landlord shall repair the damage to the Premises, the Unit and (w) the part of the Building Systems serving the Premises, (x) the floor and ceiling slabs of the Premises, (y) the exterior walls of the Premises, and (z) any other portions of the Premises that Tenant is not obligated to restore pursuant to Section 15.3 hereof, to the extent caused by fire or other casualty. Landlord shall cause the Condominium Board to repair the damage to the Building or any portions thereof to the extent the Condominium Board is responsible therefor pursuant to the Condominium Documents. Landlord shall commence the performance of such repairs as promptly as reasonably practicable after the occurrence of such fire or other casualty and shall use all reasonable efforts to perform such repairs diligently and in a workmanlike manner and in such a manner as to minimize interference with Tenant's use and occupancy of and conduct of business in the Premises. Upon Tenant's written request, Landlord shall perform any such repairs on an overtime basis and Tenant, following Landlord's rendition of a statement therefor from time to time, shall pay to Landlord the excess of the costs to perform such repairs on an overtime basis over the cost of performing any such repairs during non-overtime hours (it being agreed that any such statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon). Landlord shall not be required to restore Tenant's Property or the Specialty Alterations in the Premises. Landlord shall not be required to commence such restoration until Tenant gives Landlord the notice described in Section 15.1 hereof (unless Landlord otherwise has received actual or constructive notice of the fire or other casualty). Landlord shall have the right to adapt the restoration of the Premises as contemplated by this Section 1

# 15.3. <u>Tenant's Restoration Obligations.</u>

Subject to the terms of this Section 15.3, Tenant shall repair the damage to the Alterations (including any Specialty Alterations) to the extent caused by fire or other casualty with reasonable diligence (or, if Tenant elects to demolish the Alterations in the Premises or any portion thereof as provided herein, shall perform such demolition with reasonable diligence) only from and after the date that Landlord completes its repair obligations, if any, pursuant to Section 15.2 hereof (or to complete its demolition, as the case may be, as provided in this Section 15.3). Tenant, at Tenant's option, shall have the right to be exercised separately with respect to each floor (or portion thereof) that constitutes a portion of the Premises (subject to the provisions of Article 7 hereof) to: (x) repair the damage to and restore the Alterations for such floor (or portion thereof) to an open floor plan including, without limitation, installing ceilings, lighting and floor coverings and any and all Alterations that are required pursuant to applicable Requirements to permit the Premises to be used for office purposes, or (y) to demolish the Alterations located on such floor (or portion thereof). Tenant shall give Landlord notice of Tenant's intention with respect to restoring the Premises (or each applicable portion thereof) (i.e., whether Tenant intends to restore or demolish, as the case may be, as aforesaid, as promptly as reasonably practicable after the date of the fire or other casualty). Tenant shall not be required to perform the work described in this Section 15.3 if this Lease terminates by virtue of a fire or other casualty pursuant to this Article 15. If this Lease does not terminate following a fire or other casualty, then Tenant shall be entitled to retain the proceeds of Tenant's Property Policy or any other policies of insurance carried by Tenant in order to pay for the restoration described in this Section 15.3 hereof. If this Lease terminates following a fire or other casualty, then the disposition of the p

# 15.4. Rent Abatement.

(A) Subject to Section 15.4(C) hereof, if a fire or other casualty renders all of the Premises untenantable, then the Fixed Rent and Escalation Rent due hereunder shall abate until the date that is two hundred seventy (270) days after the date that Landlord completes the restoration described in Section 15.2 hereof. If a fire or other casualty renders part (but not all) of the Premises untenantable, then the Fixed Rent and Escalation Rent due hereunder shall abate in the proportion that (i) the Rentable Area of the portion so rendered untenantable, bears to (ii) the Rentable Area of the Premises immediately prior to such fire or other casualty, until the date that is two hundred seventy (270) days after the date that Landlord completes the restoration described in Section 15.2 hereof. Tenant shall not be entitled to an abatement of Rental pursuant to this Section 15.4 that extends (a) for more than thirty-six (36) calendar months after the date of a particular fire or other casualty, or (b) beyond the date that a Permitted Occupant uses the Premises (or the applicable portion thereof) for the conduct of business or the date that Tenant Substantially Completes any demolition of the Alterations in the Premises (if Tenant elects not to restore the Premises or any portion thereof pursuant to Section 15.4 hereof and, accordingly, not occupy the Premises or such portion thereof for the conduct of business) (it being agreed that Tenant shall not be deemed to be conducting business in the Premises (or the applicable portion thereof) by virtue of Tenant (or its employees or agents) accessing the Premises (x) to secure the Premises or Tenant's Property, (y) to undertake file or data retrieval or removal, or (z) for purposes of testing Tenant's equipment and systems and for such other purposes reasonably

related to the casualty and Tenant's restoration of the Premises (or the applicable portion thereof following a fire or other casualty)). For purposes of this Article 15 the term "untenantable" shall mean that the Premises (or portion thereof) is not accessible or is unusable for the conduct of Tenant's business in a manner which is consistent with Tenant's use of the Premises (or applicable portion thereof) during the thirty (30) day period of time immediately prior to the fire or other casualty in question.

- (B) If this Lease demises the entire Rentable Area on a particular floor of the Building, (i) fifty percent (50%) or more of the Rentable Area on such floor is rendered untenantable by reason of a fire or other casualty, (ii) in Tenant's good faith judgment Tenant cannot use the tenantable portion of such floor for the conduct of business, and (iii) Tenant actually ceases to use for the conduct of its business the entire rentable area on such floor of the Building, then such entire floor shall be deemed to be untenantable for purposes of this Section 15.4. If (i) fifty percent (50%) or more of the Rentable Area of the entire Premises is rendered untenantable by reason of a fire or other casualty, (ii) in Tenant's good faith judgment Tenant cannot use the tenantable portion of the Premises for the conduct of Tenant's business in a manner which is consistent with Tenant's use during the thirty (30) day period prior to such fire or other casualty, and (iii) Tenant actually ceases to use the entire Premises for the conduct of its business, then the entire Premises shall be deemed to be untenantable. Landlord shall give Tenant no less than ten (10) Business Days of prior notice of Landlord's Substantial Completion of its repairs pursuant to Section 15.2 hereof. Either party may submit a dispute with respect to whether the other party has Substantially Completed its restoration obligations or has used reasonable diligence under this Article 15 to Substantially Complete its restoration (or, in the case of Tenant, to demolish the Alterations in the Premises as aforesaid) hereof to an Expedited Arbitration Proceeding.
- (C) If a fire or other casualty occurs in the Premises after the Commencement Date and prior to the Rent Commencement Date, then the aggregate abatement of Fixed Rent and the Escalation Rent to which Tenant is entitled as contemplated by this Section 15.4 hereof (from and after the Rent Commencement Date) shall be an amount equal to the aggregate abatement of Fixed Rent and the Escalation Rent to which Tenant would have been entitled under this Section 15.4 hereof if the Rent Commencement Date had occurred immediately prior to such fire or other casualty.

# 15.5. <u>Landlord's Termination Right.</u>

Landlord shall have the right to terminate this Lease if the Condominium Board does not make the election contemplated by Section 339-ee of the New York Real Property Law to restore the Building after a fire or other casualty (to the extent that such election is required to be made by the Condominium Board), it being agreed that if (x) a fire or other casualty affects all or any portion of the Building, and (y) either (a) Tenant does not have the right to terminate this Lease by reason of such fire or other casualty, or (b) Tenant has theretofore waived Tenant's right to terminate this Lease by reason of such fire or other casualty, then Landlord, in its capacity as owner of the Office Unit shall vote, and Landlord shall cause Affiliates of Landlord that own Condominium units to vote, to restore the Building (rather than to terminate the Condominium). If Landlord terminates this Lease in accordance with this Section 15.5, then (I) the Term shall expire on a date set by Landlord that is not sooner than (i) the tenth (10th) day after the date that

Landlord gives such notice (if all or substantially all of the Premises is rendered untenantable by such fire or other casualty), or (ii) the ninetieth (90th) day after the date that Landlord gives such notice (if less than all or substantially all of the Premises is rendered untenantable by such fire or other casualty), and (II) Tenant, on such date set by Landlord, shall vacate the Premises and surrender the Premises to Landlord in accordance with the terms of this Lease that govern Tenant's obligations upon the expiration or earlier termination of the Term. If, however, Tenant is in occupancy of the Premises on the date Landlord gives Tenant the aforesaid notice, then Tenant shall have the one-time right to postpone the date designated by Landlord in Landlord's aforesaid notice by up to one hundred eighty (180) days, by giving notice to Landlord on or prior to thirty (30) days after the date Landlord gave Tenant Landlord's notice (it being agreed that Tenant shall designate the postponed date in Tenant's notice). Landlord may declare ineffective Tenant's postponement of such date by so notifying Tenant but only if such postponement would interfere by more than a *de minimis* extent with Landlord's schedule for the demolition or rebuilding of the Building or the applicable portion thereof. Upon the termination of this Lease under this Section 15.5, the Rental shall be apportioned and any prepaid portion of the Rental for any period after the date that the abatement of Rental as described in Section 15.4 hereof becomes effective shall be refunded promptly by Landlord to Tenant (and Landlord's obligation to make such refund shall survive the Expiration Date).

# 15.6. <u>Tenant's Termination Right.</u>

Landlord, within forty-five (45) days after the earlier to occur of (x) the date that Tenant gives Landlord notice of the occurrence of a fire or other casualty as contemplated by Section 15.1 hereof, and (y) the date that Landlord otherwise has actual notice of such fire or other casualty, shall give to Tenant a statement prepared by a reputable and independent contractor setting forth such contractor's estimate in good faith as to the time required for Landlord to Substantially Complete the restoration described in Section 15.2 hereof (such statement that Landlord gives to Tenant being referred to herein as the "Casualty Statement"). If (x) Landlord fails to give Tenant a Casualty Statement within the aforesaid time period, (y) Tenant notifies Landlord thereof, and (z) Landlord fails to deliver the Casualty Statement to Tenant on or prior to ten (10) Business Days after the date Landlord receives such notice from Tenant, then Tenant shall have the right to terminate this Lease by giving notice thereof to Landlord no later than ten (10) Business Days after the last day of Landlord's ten (10) Business Day period of time to deliver the Casualty Statement and the provisions of Section 15.6(C) hereof shall control (unless Landlord delivers the Casualty Statement to Tenant within the ten (10) day Business Day period of time after Tenant delivers its notice, in which event Tenant's termination of this Lease shall be deemed ineffective and of no force or effect). Landlord shall not designate a particular Person as the aforesaid contractor unless (i) such Person has at least five (5) years of experience in providing such estimates, and (ii) such Person has not been employed by Landlord or Tenant (or their respective Affiliates) for a period of three (3) years prior to the date of the Casualty Statement; provided, however, that if, in Landlord's reasonable judgment, there exists no such Person that satisfies the provisions of this clause (ii), then Landlord shall have the right to designate a Person that complies only with the provisions of clause (i) above, if such Person is otherwise reputable and Landlord and Tenant receive reasonable assurances from such Person that such Person is preparing the Casualty Statement impartially. Tenant shall have the right to dispute the estimated time period as set forth in the Casualty Statement by giving notice thereof to Landlord not more than fifteen (15) days after the

date that Landlord gives the Casualty Statement to Tenant. Either party shall have the right to submit a dispute between the parties regarding the aforesaid estimated time period to an Expedited Arbitration Proceeding. If the estimated time period (as such time period may be determined by arbitration as aforesaid) exceeds twelve (12) months from the date of the applicable fire or other casualty, then Tenant may elect to terminate this Lease by giving notice to Landlord not later than the thirtieth (30th) day after the date that Landlord gives the Casualty Statement to Tenant. If Tenant makes such election to so terminate this Lease, then the Term shall expire on the thirtieth (30th) day after Tenant gives such notice to Landlord.

- (B) This Lease shall terminate if (i) a fire or other casualty occurs, and, by reason thereof, Landlord has an obligation to perform a restoration as contemplated by Section 15.2 hereof, (ii) Tenant does not exercise Tenant's right to terminate this Lease under Section 15.6(A) hereof in connection with such fire or other casualty (or Tenant does not have the right to terminate this Lease under Section 15.6(A) hereof in connection with such fire or other casualty), (iii) Landlord fails to Substantially Complete the performance of the restoration work that Landlord is required to perform on or prior to the later to occur of (I) twelve (12) months after the date of the applicable fire or other casualty, and (II) the date that is thirty-five (35) days after the last day of the estimated time period set forth in the Casualty Statement (the later of the dates described in clause (I) and clause (II) above being referred to herein as the "Second Bite Date"), (iv) Tenant gives Landlord notice to the effect that this Lease will terminate under this Section 15.6(B) if Landlord fails to Substantially Complete the restoration within thirty-five (35) days after the Second Bite Notice"), and (v) Landlord fails to Substantially Complete the restoration within thirty-five (35) days after the Second Bite Date.
- (C) If the Term terminates as provided in this Section 15.6, then (I) Tenant shall vacate the Premises and surrender the Premises to Landlord on the date of such termination "as is" and otherwise in accordance with the terms of this Lease that govern Tenant's obligations upon the expiration or earlier termination of the Term, (II) any Rental due hereunder shall be apportioned as of the date of such termination, and (III) any portion of the Rental that is then prepaid by Tenant and relates to the period after the date that the abatement of Rental as described in Section 15.4 hereof becomes effective shall be promptly refunded by Landlord to Tenant (with the understanding that Landlord's obligation to make any such refund shall survive such termination of this Lease).

# 15.7. <u>Termination Rights at End of Term.</u>

If the Premises are substantially damaged by a fire or other casualty that occurs during the period of twenty-four (24) months immediately preceding the last day of the Term (as the Term may be renewed pursuant to Article 19 hereof), then either Landlord or Tenant may elect to terminate this Lease by notice given to the other party within thirty (30) days after such fire or other casualty occurs. If either party makes such election, then the Term shall expire on the thirtieth (30th) day after the notice of such election is given, and, accordingly, Tenant, on or prior to such thirtieth (30th) day, shall vacate the Premises and surrender the Premises to Landlord in accordance with the provisions of this Lease that govern Tenant's obligation to deliver vacant and exclusive possession of the Premises to Landlord upon the expiration of the Term. If Landlord terminates the Lease pursuant to this Section 15.7 and at such time Tenant is in

occupancy of the Premises, then Tenant shall have the one-time right to postpone the date designated by Landlord in Landlord's aforesaid notice by up to one hundred fifty (150) days, by giving notice to Landlord on or prior to thirty (30) days after the date Landlord gave Tenant Landlord's notice (it being agreed that Tenant shall designate the postponed date in Tenant's notice). Landlord may declare ineffective Tenant's postponement of such date by so notifying Tenant but only if such postponement would interfere by more than a *de minimis* extent with Landlord's schedule for the demolition or rebuilding of the Building or the applicable portion thereof. Upon the termination of this Lease under this Section 15.7, the Rental shall be apportioned and any prepaid portion of the Rental for any period after the Expiration Date shall be refunded promptly by Landlord to Tenant (and Landlord's obligation to make such refund shall survive the Expiration Date). For purposes of this Section 15.7, the term "substantially damaged" shall mean that: (a) a fire or other casualty precludes Tenant from using more than fifty percent (50%) of the Premises for the conduct of its business, and (b) Tenant's inability to so use the Premises (or the applicable portion thereof) is reasonably expected to continue until at least the earlier to occur of (i) the Fixed Expiration Date, or the last day of the Renewal Term, as the case may be, and (ii) the ninetieth (90th) day after the date that such fire or other casualty occurs.

#### 15.8. <u>No Other Termination Rights.</u>

Tenant shall have no right to cancel this Lease by virtue of a fire or other casualty except to the extent specifically set forth herein. This Article 15 is intended to constitute an "express agreement to the contrary" for purposes of Section 227 of the New York Real Property Law.

# Article 16\_ CONDEMNATION

# 16.1. <u>Effect of Condemnation.</u>

- (A) Subject to the provisions of Section 16.2 hereof, if the entire Real Property, the entire Building or the entire Premises is condemned or otherwise acquired by the exercise of the power of eminent domain, then this Lease shall terminate as of the date that such condemnation or acquisition is consummated.
  - (B) If only a part of the Real Property and not the entire Premises is so acquired or condemned, then:
- (1) except as hereinafter provided in this Section 16.1, this Lease shall remain effective, and, from and after the date that the condemnation or acquisition is consummated, (w) the Fixed Rent shall be reduced in the proportion that the number of square feet of Rentable Area of the part of the Premises so acquired or condemned bears to the total Rentable Area of the Premises immediately prior to such acquisition or condemnation, (x) Tenant's Tax Share shall be redetermined based upon the proportion that the number of square feet of Rentable Area of the Premises that is remaining after such acquisition or condemnation bears to the number of square feet of Rentable Area of the Office Unit that is remaining after such acquisition or condemnation, and (y) Tenant's Operating Expense Share shall be redetermined based upon the proportion that the number of square feet of Rentable Area of the Premises remaining after such acquisition or

condemnation bears to the number of sc	uare feet of Rentable Ar	rea of the Office Unit rema	ining after such aco	uisition or condemnation:

- (2) if more than fifty percent (50%) of the Real Property is affected by such acquisition or condemnation, then on or prior to the sixtieth (60th) day after the date that the condemnation or acquisition is consummated, Landlord shall have the right to terminate this Lease by giving notice to Tenant; provided, however, that if the Premises are unaffected by such acquisition or condemnation, then Landlord shall only have the right to so terminate this Lease if Landlord and/or owners of the other non-residential condominium units terminate leases (including this Lease) for at least seventy-five percent (75%) of the leasable area of the Building (excluding any portion of the Building comprising the residential condominium units or any portion of the Building leased to or occupied by Landlord or Landlord's Affiliates); and
- (3) if (a) the part of the Real Property so acquired or condemned contains more than fifteen percent (15%) of the total area of the Premises immediately prior to such acquisition or condemnation, or (b) by reason of such acquisition or condemnation, Tenant no longer has reasonable means of access to the Premises, then Tenant may elect to terminate this Lease by giving notice to Landlord on or prior to the sixtieth (60th) day after the date that Tenant is given notice of such acquisition or condemnation being consummated.

The Term shall expire on the thirtieth (30th) day after the date that Landlord or Tenant give any such notice to terminate this Lease.

- (C) Upon the termination of this Lease and the Term pursuant to the provisions of this Section 16.1, the Rental shall be apportioned as of the termination date, except that with respect to any portion of the Premises which is the subject of the taking, if earlier, the Rental shall be apportioned as of the date of such acquisition or condemnation, and any prepaid portion of the Rental for any period after such date shall be refunded promptly by Landlord to Tenant (and Landlord's obligation to make such refund shall survive the Expiration Date).
- (D) If a part of the Premises is so acquired or condemned and this Lease and the Term is not terminated pursuant to the foregoing provisions of this Section 16.1, then Landlord, at Landlord's cost and expense, shall restore the part of the Premises that is not so acquired or condemned to a self-contained rental unit (subject to Section 16.2 hereof), exclusive of Alterations and Tenant, at Tenant's cost and expense, shall restore any Alterations.

#### 16.2. <u>Condemnation Award.</u>

(A) Subject to Section 16.3 hereof, (I) Tenant shall be entitled to receive the portion of any award for any acquisition or condemnation of all or any part of the Real Property for the value of (x) any Tenant's Property included in such taking, and (y) any Unamortized Alterations less any Unamortized Tenant Fund Amount, and (II) Landlord shall be entitled to receive the portion of any such award for the value of the Building other than the value of the items described in the immediately preceding clause (I) (such portion being referred to herein as "Landlord's Award"). If (x) this Lease terminates by reason of the occurrence of a condemnation or acquisition pursuant to this Article 16, and (y) a Mortgagee that constitutes an Institutional

Lender then holds a Mortgage on Landlord's interest in the Premises, then Tenant shall pay the proceeds of any condemnation or acquisition award as follows:

- (1) first, to Landlord (or, at Landlord's option, to such Mortgagee), until the aggregate amount paid to Landlord (or such Mortgagee) from such proceeds equals the excess of (I) the lesser of (x) the amount of the outstanding indebtedness secured by such Mortgage, and (y) eighty percent (80%) of the fair market value of the Premises on the date that such Mortgagee initially funded the debt secured by such Mortgage, over (II) the amount of Landlord's Award; and
- (2) second, to Tenant, until the aggregate amount paid to Tenant from such proceeds equals the sum of (a) the value of any Tenant's Property included in such taking, plus (b) the excess of (I) the Unamortized Alterations Cost, over (II) the Unamortized Tenant Fund Amount; and
  - (3) *finally*, the remainder to Landlord.

Landlord and Tenant shall give the other party reasonable access to the such party's books and records to the extent reasonably necessary to determine the amount due from Tenant to Landlord under this Section 14.7. Either party shall have the right to submit any dispute between the parties regarding the amount due from Tenant to Landlord under this Section 14.7 hereof to an Expedited Arbitration Proceeding.

# 16.3. <u>Temporary Taking.</u>

If the whole or any part of the Premises is acquired or condemned temporarily during the Term, then (a) Tenant shall give prompt notice thereof to Landlord, (b) the Term shall not be reduced or affected in any way, (c) Tenant shall continue to pay in full all items of Rental payable by Tenant hereunder without reduction or abatement, and (d) Tenant shall be entitled to receive for itself any award or payments for such use, provided, however, that if the acquisition or condemnation is for a period extending beyond the Term, then such award or payment shall be apportioned equitably between Landlord and Tenant. Tenant, at Tenant's expense, shall make Alterations to restore the Premises to the condition existing prior to any such temporary acquisition or condemnation.

#### Article 17\_ ASSIGNMENT AND SUBLETTING

#### 17.1. General Limitations.

- (A) Except as otherwise provided in and subject to the terms of this Article 17, without the prior consent of Landlord in each instance, Tenant shall not, and Tenant shall not permit any other Permitted Party to, consummate a Transfer. The term "Transfer" shall mean:
- (1) (a) an assignment of a Permitted Party's rights under, or a delegation of such Permitted Party's duties under, the applicable Occupancy Agreement by express assignment or by operation of law or by other means, (b) a mortgage or other encumbrance of such Permitted

Party's interest in the applicable Occupancy Agreement, in whole or in part, (c) a subletting, or further subletting, of the Premises or any part thereof, or (d) the occupancy of the Premises or any part thereof by any Person other than such Permitted Party; and

- (2) any transaction that extends the term of, or grants the Transferee the right to occupy additional space under, an Occupancy Agreement (other than this Lease).
- (B) The term "Occupancy Agreement" shall mean the lease, sublease, license or other agreement pursuant to which a Permitted Party has the right to occupy the Premises (or the applicable portion thereof).
- (C) The term "Permitted Party" shall mean Tenant and any other Person that has the right to occupy the Premises (or any part thereof) in accordance with the terms of this Article 17.
  - (D) The term "Transferor" shall mean the Permitted Party that makes a Transfer.
- (E) Except as otherwise provided in, and subject to the terms of Section 17.9 hereof, the transfer of Control in a Permitted Party, however accomplished, whether in a single transaction or in a series of unrelated or related transactions, shall constitute an assignment of such Permitted Party's interest in the applicable Occupancy Agreement for purposes of this Article 17.
- (F) The consent by Landlord to any Transfer shall not relieve Tenant from its obligation to obtain the prior consent of Landlord to any other Transfer to the extent required by this Lease.
- (G) The assignment by any Person that constitutes Tenant of the tenant's interest under this Lease shall not relieve such Person of the obligations of the tenant under this Lease. Such Person's liability under this Lease shall continue notwithstanding (x) the subsequent release of any other Person that constitutes Tenant from liability under this Lease, (y) any limitation on any such other Person's liability hereunder by virtue of the Bankruptcy Code, or (z) any modification or amendment of this Lease that Landlord consummates with any such other Person that constitutes Tenant subsequently; provided, however, that if such other Person is not an Affiliate of such Person, then any such modification or amendment shall not expand such Person's liability hereunder.
- (H) Notwithstanding anything to the contrary contained herein, Tenant shall not, and Tenant shall not permit any other Permitted Party to, (i) enter into any lease, sublease, license, concession or other agreement for use or occupancy of the Premises or any portion thereof which provides for a rental or other payment for such use or occupancy based in whole or in part on the net income or profits derived by any Person from the property leased, occupied or used, or which would require the payment of any consideration that would not qualify as "rents from real property," as that term is defined in Section 856(d) of the Internal Revenue Code of 1986, as amended, or (ii) permit the Premises, or any portion thereof, to be used or occupied by or for the benefit of any Person that the Office of Foreign Assets Control of the United States

Department of the Treasury has listed on its list of Specially Designated Nationals and Blocked Persons.

(I) If Tenant assigns the tenant's interest under this Lease in violation of the terms of this Article 17, then such assignment shall be void and of no force and effect against Landlord; provided, however, that Landlord (x) may collect an amount equal to the then Rental from the assignee as a fee for such assignee's use and occupancy, and (y) shall apply the net amount collected to the Rental reserved in this Lease. If the Premises or any part thereof are sublet to, occupied by, or used by any Person other than Tenant (regardless of whether such subletting, occupancy or use violates this Article 17), then Landlord (a) after the occurrence of an Event of Default, may collect amounts from the subtenant, user or occupant as a fee for its use and occupancy, and (b) shall apply the net amount collected to the Rental reserved in this Lease. No such assignment, subletting, occupancy or use, with or without Landlord's prior consent, nor any such collection or application of fees for use and occupancy, shall (i) be deemed a waiver by Landlord of any term, covenant or condition of this Lease, (ii) be deemed the acceptance by Landlord of such assignee, subtenant, occupant or user as tenant hereunder, or (iii) relieve Tenant of the obligations of the tenant under this Lease.

#### 17.2. <u>Landlord's Expenses.</u>

Tenant shall reimburse Landlord for any reasonable out-of-pocket costs that Landlord incurs to third-parties in connection with any proposed Transfer, including, without limitation, reasonable attorneys' fees and disbursements and the reasonable costs of making investigations as to the acceptability of the proposed Transferee, within thirty (30) days after Landlord gives to Tenant an invoice therefor.

# 17.3. <u>Recapture Procedure.</u>

- (A) Tenant shall have the right to institute the procedure described in this Section 17.3 (the "Recapture Procedure"") only by giving to Landlord notice thereof (a "Transfer Notice"), which:
  - (1) refers expressly to this Section 17.3 and indicates that such notice constitutes a Transfer Notice,
- (2) sets forth a description of the Premises (or the portion thereof) that is involved in the proposed Transfer (the Premises, or the portion thereof, that is involved in the proposed Transfer being referred to herein as the "Recapture Space"),
  - (3) sets forth the term of the proposed sublease (if the proposed Transfer is a sublease), and
- (4) sets forth the date on which Tenant proposes to consummate the proposed Transfer (such date being referred to herein as the "Transfer Date") (it being understood that, subject to the terms of this Section 17.3(A), the Transfer Date shall be no sooner than one hundred twenty (120) days, and no later than five hundred forty (540) days, after the date that Tenant gives the Transfer Notice to Landlord).

Tenant shall not be required to identify, in the Transfer Notice, the Person to which Tenant intends to make the Transfer (the Person to which a Transfer is made being referred to herein as a "<u>Transferee</u>").

- The term "Transfer Expenses" shall mean the sum of the actual out-of-pocket expenses that the Transferor pays solely in consummating a Transfer, including, without limitation, (i) brokerage commissions, (ii) allowances that the Transferor makes available to the Transferee to fund the cost of Alterations that the Transferee makes to the Premises (or the applicable portion thereof that is involved in the Transfer), (iii) costs that the Transferor pays in making Alterations to prepare the Premises (or the applicable portion thereof that is involved in the Transfer) solely for the Transferee's initial occupancy (to the extent paid therefor), (iv) the amount payable to Landlord under Section 17.2 hereof for such Transfer, (v) reasonable attorneys' fees and disbursements that the Transferor pays in connection with consummating such Transfer, (vi) the unamortized construction costs of Alterations installed by or on behalf of the Transferor after the date hereof in connection with its occupancy of the applicable portion of the Premises, but only to the extent such improvements are used by the Transferee in connection with its initial occupancy of such portion of the Premises, (vii) advertising and marketing expenses directly related to the Transfer, (vi) any transfer, sales or gains taxes paid by the Transferor in connection with such Transfer (including the taxes that Tenant pays pursuant to Section 17.7 hereof), (vii) the rental due under the Transferor's Occupancy Agreement for the period commencing on the day immediately following the date on which Transferor and all persons, claiming by, through or under Transferor have vacated the Premises (or the applicable portion thereof) and ending on the earlier to occur of (a) the day immediately preceding the effective date of the applicable Transfer, or (b) the day immediately preceding the date on which the Transferor or any person claiming by, through or under the Transferor has resumed possession of the Premises or the applicable portion thereof (provided, however, that the period described in this subsection (vii) shall not exceed six (6) months), and (x) if the Transferor has granted a free rent concession to the Transferee, the rental due under the Transferor's Occupancy Agreement for the period commencing on the date that the applicable Transfer is consummated and ending on the day immediately preceding the date that the Transferee is obligated to commence the payment of rental for such Transfer.
- (1) If (x) Tenant gives a Transfer Notice to Landlord, and (y) the Transfer described in the Transfer Notice constitutes (x) a sublease by Tenant for the Recapture Space with respect to which the term thereof can expire later than eighteen (18) months before the Fixed Expiration Date (any sublease being referred to herein as a "Long-Term Sublease"), or (y) an assignment of Tenant's interest under this Lease, then, in any such case, Landlord shall have the right to terminate this Lease with respect to the Recapture Space, on the terms set forth in this Section 17.3, by giving notice thereof (the "Recapture Termination Notice") to Tenant not later than the (x) thirtieth (30th) day after the date that Tenant gives the Transfer Notice to Landlord for Recapture Space consisting of not more than the entire Rentable Area of two (2) floors in the Building (such period being referred to herein as the "Applicable Recapture Period"; and any such termination of this Lease with respect to the Recapture Space being referred to herein as a "Recapture Termination"). The parties acknowledge and agree that Landlord's recapture rights described in this Section 17.3 shall not apply to any proposed sublease which cannot expire later than eighteen (18) months before the

rixeu Expiration Date	of to ally Transfer	mai is proposed to be	consummated b	y a reminieu	raity oulei ulai	i Tellalit.				
(2)	If Landlord give	es Tenant a Recapture	Termination No	tice and the F	Recapture Space	constitutes t	he entire Pre	mises, t	hen the T	Γerm shall

- (2) If Landlord gives Tenant a Recapture Termination Notice and the Recapture Space constitutes the entire Premises, then the Term shall terminate on the Transfer Date. Tenant, on the Transfer Date, shall vacate the Premises and deliver exclusive possession thereof to Landlord in accordance with the terms of this Lease that govern Tenant's obligations upon the expiration or earlier termination of the Term.
- (3) If (x) Landlord gives to Tenant a Recapture Termination Notice, and (y) the Recapture Space does not constitute the entire Premises, then:
- (a) Landlord, at Landlord's expense, shall demise the Recapture Space separately from the remainder of the Premises on or prior to the Transfer Date,
- (b) effective as of the Transfer Date, Tenant's Operating Expense Share shall be redetermined based on the ratio that (I) the number of square feet of Rentable Area of the Premises that remains after excluding therefrom the Recapture Space, bears to (II) the number of square feet of Rentable Area in the Office Unit.
- (c) effective as of the Transfer Date, Tenant's Tax Share shall be redetermined based on the ratio that (I) the number of square feet of Rentable Area of the Premises that remains after excluding therefrom the Recapture Space, bears to (II) the number of square feet of Rentable Area of in the Office Unit,
- (d) the Fixed Rent as set forth in Article 2 hereof at any particular time from and after the Transfer Date shall be reduced by an amount equal to the Fixed Rent that would have been due under this Lease for such calendar month for the applicable portion of the Premises that constitutes the Recapture Space, and
- (C) Tenant, on the Transfer Date, shall vacate the Recapture Space and deliver exclusive possession thereof to Landlord in accordance with the terms of this Lease that govern Tenant's obligations upon the expiration or earlier termination of the Term.

# 17.4. <u>Certain Transfer Rights.</u>

Landlord shall not unreasonably withhold, condition or delay Landlord's consent to a Permitted Party's consummating a Transfer, provided that:

- (A) Tenant has theretofore instituted the Recapture Procedure to the extent required for a particular Transfer pursuant to Section 17.3 hereof;
- (B) Landlord's right pursuant to Section 17.3 hereof to elect to consummate a Recapture Termination with respect to the proposed Transfer has lapsed (without Landlord's having exercised Landlord's rights to consummate a Recapture Termination );
- (C) the Permitted Party consummates the proposed Transfer within twelve (12) months after the date that Tenant gave the applicable Transfer Notice to Landlord (it being

understood that Tenant shall have the right to give Landlord successive Transfer Notices in respect of a proposed Transfer, in which case the period of twelve (1	12
months as described in this clause (C) shall be measured from the date of the last such Transfer Notice that Tenant gives to Landlord);	

- (D) Tenant submits to Landlord a counterpart of the documents that the Transferor intends to use to consummate the proposed Transfer, which have been executed and delivered by the proposed Transferor and the proposed Transferee, and which are subject to no conditions to the effectiveness thereof (other than Landlord's granting Landlord's consent thereto or, in the case of a Major Sublease, Landlord's delivery of a Recognition Agreement in accordance with the provisions of this Lease, or other conditions which the proposed Transferor and the proposed Transferee have agreed are to be satisfied after Landlord's granting Landlord's consent); provided, however, that if such conditions are not satisfied within one hundred twenty (120) days after the granting of Landlord's consent, then Tenant must renew its request for Landlord's consent under this Section 17.3);
- (E) the Premises (or the applicable portion thereof) has not been listed or otherwise publicly advertised at a rental rate that is less than the prevailing rental rate set by Landlord for comparable space in the Office Unit for a comparable term, or, if there is no comparable space for a comparable term, the prevailing rental rate reasonably determined by Landlord (it being agreed that nothing contained in this clause (E) shall be deemed to prohibit the applicable Permitted Party, without Landlord's consent or approval, from listing with brokers the availability of the Premises for sublet or assignment at any rental rate, and Landlord hereby acknowledges that broker's fliers or listings and such Permitted Party's marketing materials shall not be deemed to constitute public advertisements as long as the rental amounts on such fliers or listings are not specified and are noted thereon as available upon request);
  - (F) no Event of Default has occurred and is continuing;
- (G) solely with respect to a Transfer that constitutes the assignment of Tenant's interest under this Lease, the proposed Transferee has a financial standing (taking into consideration the obligations of the Transferee under the applicable Occupancy Agreement) that is reasonably satisfactory to Landlord:
- (H) the proposed Transferee is of a character, is engaged in a business, and proposes to use the Premises (or the applicable portion thereof) in a manner that in each case is in keeping with the standards of a first-class office building in the vicinity of the Building and the proposed Transferee is not a Person that violates the provisions of Section 3.3 hereof (it being agreed that Landlord shall notify Tenant from time to time upon Tenant's written request therefor, whether any proposed Transferee violates the provisions of Section 3.3 hereof);
- (I) the proposed Transferee, or any Affiliate of the proposed Transferee, does not occupy any space in the Building (if Landlord or Landlord's Affiliate has or within six (6) months thereafter reasonably expects to have space available in the Building that is comparable in size to the Premises, or the portion thereof involved in the Transfer (i.e., plus or minus 10%) for a comparable lease term (i.e., plus or minus 10%) ("Comparable Space"; provided, however, that if Landlord fails to give Tenant a notice identifying the Comparable Space within five (5)

Business Days after Tenant's request that Landlord identify such Comparable Space, then the foregoing condition shall not apply to any proposed Transfer (it being agreed that Comparable Space shall not be deemed to include any space contained in the Office Unit One unless the Office Unit One Owner is Landlord or an Affiliate of Landlord) (it being agreed that solely for purposes of this Clause (I) and the immediately following Clause (J), the definition of Control for purposes of defining an Affiliate of a proposed Transferee, shall be the same as the definition of Control contained in Section 1.6 hereof except that the percentage contained in such definition shall be deemed to be seventy-five percent (75%));

- (J) if Landlord has Comparable Space in the Building, the proposed Transferee (or an Affiliate of the proposed Transferee), is not a Person with whom Landlord is then engaged in bona fide active negotiations regarding the leasing or subleasing of space in the Building (provided, however, that if Landlord fails to give Tenant a notice identifying the Comparable Space within five (5) Business Days after Tenant's request that Landlord identify such Comparable Space, then the foregoing condition shall not apply to the subletting or assignment in question);
- (K) the Transferor and each other Permitted Party (if any) whose interest is superior to the interest of the Transferor, and the Transferee, executes and delivers to Landlord a consent to the Transfer in a form reasonably designated by Landlord;
- (L) if the Transfer constitutes an assignment of the tenant's interest under this Lease, the assignee has expressly assumed all of the obligations of Tenant hereunder to the extent accruing from and after the date that the Transfer is effective;
- (M) the space involved in the applicable Transfer does not constitute only the Mailroom, only the Storage Space, or only the Mailroom and the Storage Space (it being agreed that the Mailroom and the Storage Space may not be sublet separate and apart from any other portion of the Premises); and
- (N) if the Transfer constitutes a sublease (or a further sublease), such sublease provides expressly that (i) such sublease is subject and subordinate to the Lease (and to the terms thereof), and (ii) if this Lease terminates, then Landlord, at Landlord's option (but subject to the terms and conditions of any Recognition Agreement between Landlord and such subtenant), may take over all of the right, title and interest of the Transferor under such sublease, and the Transferee, at Landlord's option, shall attorn to Landlord pursuant to the then executory provisions of such sublease, except that Landlord shall not be:
- (1) liable for any act or omission of the Transferor under such sublease (except for any such acts or omissions that (x) continue after the date that Landlord succeeds to the interest of the Transferor under such sublease, and (y) may be remedied by the providing a service or performing a repair, replacement or alteration),
  - (2) subject to any defense or offsets which the Transferee may have against the Transferor,
- (3) bound by any previous payment that the Transferee made to the Transferor more than thirty (30) days in advance of the date that such payment was due,

- (4) bound by any obligation to make any payment to or on behalf of the Transferee that accrues prior to the date that Landlord succeeds to the interest of the Transferor under such sublease,
- (5) except to the extent required under Section 17.4(N)(1) above bound by any obligation to perform any work or to make improvements to the Premises, or the applicable portion thereof demised by such sublease (other than the obligation to perform ordinary maintenance that first becomes necessary from and after the date that Landlord succeeds to the interest of the Transferor under such sublease),
- (6) bound by any amendment or modification of such sublease made without Landlord's consent if and to the extent that such consent is required pursuant to this Article 17, or
- (7) bound to return the Transferee's security deposit, if any, until such deposit has come into Landlord's actual possession and the Transferee is entitled to such security deposit pursuant to the terms of such sublease (the requirements of a proposed sublease as set forth in this Section 17.4(N) being collectively referred to herein as the "Basic Sublease Provisions").

# 17.5. <u>Conditional Approval.</u>

Tenant shall have the right at any time (including, without limitation, simultaneously with its submission of a Transfer Notice to Landlord, either as a separate notice or as part of the Transfer Notice) to request that Landlord approve a particular Transfer by giving notice thereof to Landlord, which notice shall set forth the material economic terms under which Tenant, or such other Permitted Party, proposes to consummate such Transfer (such terms being referred to herein as the "Proposed Transfer Terms"; and any such notice that Tenant gives to Landlord being referred to herein as a "Conditional Consent Notice"). A Conditional Consent Notice shall not be effective for purposes of this Section 17.5 unless Tenant sets forth therein (i) the name of the proposed Transferee, (ii) a reasonably detailed description of the proposed Transferee's business, (iii) the sublease rental or assignment consideration (as the case may be) to be paid by such Transferee, (iv) the term of the proposed Transfer, (v) any work allowance to which such Transferee is entitled, and (vi) any work to be performed by Tenant or such other Permitted Party to prepare the Premises, or the applicable portion thereof, for occupancy by the Transferee. Landlord shall not unreasonably withhold, condition or delay Landlord's approval of the Transfer identified in the Conditional Consent Notice. If Landlord fails to respond to a Conditional Consent Notice within ten (10) Business Days after the date that Tenant gives the Conditional Consent Notice to Landlord, then Landlord shall be deemed to have approved the proposed Transfer, subject only to (x) Tenant's submission to Landlord of completed, fully executed documents consummating such Transfer (herein called "Final Transaction Documents") meeting all of the requirements set forth in this Article 17 hereof, (y) the execution of a written instrument setting forth Landlord's consent to the Transfer as required pursuant to Section 17.4(K) hereof, and (z) payment by Tenant of the amounts, if any, required to be paid to Landlord pursuant to Section 17.2 hereof; provided, however, that if Landlord fails to provide to Tenant the form of consent that Landlord proposes for a particular Transfer as provided in Section 17.4(K) with reasonable promptness after Tenant's request therefor, then clause (y) above shall not constitute a condition to Landlord's consent to the applicable Transfer becoming effective. If Landlord approves (or is deemed to have approved) a proposed Transfer, Landlord

will deliver its written consent to such Transfer within ten (10) days following submission to Landlord of the Final Transaction Documents provided that Tenant submits such Final Transaction Documents to Landlord on or prior to the first (1st) anniversary of the date on which Tenant gives the Conditional Consent Notice to Landlord and provided that such Final Transaction Documents contain terms that are substantially similar to the Proposed Transfer Terms. Nothing contained in this Section 17.5 modifies or waives the provisions of Section 17.3 hereof, it being agreed that for any proposed Transfer in respect of which Landlord has the right to exercise a Recapture Termination, Landlord shall be entitled to exercise such right within the time frames and pursuant to the provisions of Section 17.3 hereof.

# 17.6. <u>Deemed Approval.</u>

If (x) Tenant requests Landlord's approval of a proposed Transfer as provided in Section 17.4 hereof, and (y) Landlord fails to respond to Tenant's request within ten (10) Business Days after the date that Tenant gives Landlord notice thereof, then Landlord shall be deemed to have approved Tenant's aforesaid request for purposes of Section 17.4 hereof, provided that the Transferor and each other Permitted Party (if any) whose interest is superior to the interest of the Transferor, and the Transferee, executes and delivers to Landlord a consent to the Transfer in a form reasonably designated by Landlord as provided in Section 17.4(K) hereof; provided, however, that if Landlord fails to provide to Tenant the form of consent that Landlord proposes for a particular Transfer as provided in Section 17.4(K) with reasonable promptness after Tenant's request therefor, then the execution and delivery thereof shall not constitute a condition to Landlord's consent to the applicable Transfer becoming effective.

# 17.7. <u>Transfer Taxes.</u>

Tenant shall pay any transfer taxes (and other similar charges and fees) that any Governmental Authority imposes in connection with any Transfer, provided, however, that with respect to any such transfer taxes, charges or fees that a Governmental Authority imposes in connection with Landlord's exercising Landlord's rights to consummate a Recapture Termination, such amounts shall be paid by Landlord or Tenant based on whether Landlord or Tenant is primarily liable for such amounts in accordance with any applicable law, rule, regulation or policy of such Governmental Authority.

#### 17.8. <u>Transfer Profit.</u>

(A) Subject to the terms of this Section 17.8 and Section 17.9 hereof, Tenant shall pay fifty percent (50%) of Transfer Profit to Landlord as additional rent. Tenant shall make payments to Landlord on account of Transfer Profit, in arrears, on the first (1<sup>st</sup>) day of each calendar month during the Term in the same manner as Fixed Rent.

(B)

(1) The term "Transfer Profit" shall mean, with respect to any particular Transfer for any particular calendar month, the excess (if any) of (x) the Transfer Inflow for such Transfer for such calendar month, over (y) the sum of (I) the Transfer Outflow for such Transfer for such calendar month, and (II) the Transfer Expenses for such Transfer for such calendar month.

(2) The term "Transfer Inflow" shall mean, with respect to any particular Transfer for any particular calendar month, the amount that Tena
receives during such calendar month from or on behalf of its immediate Transferee in connection with the applicable Transfer provided, however, that a credit
shall be applied against the Transfer Inflow for such Transfer (until such credit is exhausted) in an aggregate amount equal to the Transfer Expenses for such
Transfer it being agreed that the Transfer Expenses are to be treated as Transfer Expenses in full, as and when incurred, and not amortized.

# (3) The term "<u>Transfer Outflow</u>" shall mean:

- (a) with respect to any Transfer that is a sublease (or a further sublease), the aggregate amount that Tenant pays during the applicable calendar month for the Premises (or the applicable portion thereof that is involved in the Transfer) to Landlord, and
- (b) with respect to any Transfer that is an assignment of Tenant's interest under this Lease or the assignment of a subtenant's interest under a sublease, the Transfer Outflow therefor shall be zero.
- (C) If Tenant (or an Affiliate thereof) receives consideration from its immediate Transferee (or an Affiliate thereof) in a transaction that occurs concurrently with the applicable Transfer, then, if and to the extent that such consideration relates to the use and enjoyment of the portion of the Premises involved in such Transfer in a real estate sense (for example, the sale of furniture and fixtures located at the Premises, but not the sale of intellectual property or client lists), the Transfer Inflow shall include (in addition to the consideration that Tenant receives for the Transfer) an amount equal to the excess of (I) such other consideration, over (II) the cost that Tenant (or such Affiliate thereof) incurs in acquiring any personal property that Tenant (or such Affiliate thereof) transferred to its immediate Transferee (or an Affiliate thereof) in such concurrent transaction (to the extent that such cost has not theretofore been amortized in accordance with generally accepted accounting principles).

# 17.9. <u>Permitted Transfers.</u>

(A) A Permitted Party shall have the right to assign such Permitted Party's entire interest under the applicable Occupancy Agreement to an Affiliate of such Permitted Party without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that in each case (i) Tenant gives to Landlord, not later than the tenth (10th) Business Day after any such assignment is consummated, an instrument, duly executed by such Permitted Party and the aforesaid Affiliate of such Permitted Party, in form reasonably satisfactory to Landlord, to the effect that such Affiliate assumes all of the obligations of such Permitted Party under such Occupancy Agreement to the extent arising from and after the date of such assignment, and (ii) Tenant, with such notice, provides Landlord with a certificate, executed by an officer of the applicable Permitted Party, stating that the Person to which such Permitted Party is so assigning such Permitted Party's interest under such Occupancy Agreement constitutes an Affiliate of such Permitted Party.

- (B) The reorganization of a Permitted Party, or the merger or consolidation of a Permitted Party into or with another Person shall be permitted without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that in each case (i) such reorganization, merger or consolidation is not principally for the purpose of transferring such Permitted Party's interest in the applicable Occupancy Agreement, (ii) Tenant gives Landlord notice of such reorganization, merger or consolidation not later than the tenth (10th) Business Day after the occurrence thereof, and (iii) Tenant, within ten (10) Business Days after such reorganization, merger or consolidation, provides Landlord with a certificate, executed by an officer of Tenant, stating that the requirement described in clause (i) above has been satisfied.
- (C) The assignment of a Permitted Party's entire interest under the applicable Occupancy Agreement in connection with the sale of all or substantially all of the assets of such Permitted Party shall be permitted without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that in each case (i) Tenant gives to Landlord, not later than the tenth (10th) Business Day after any such assignment is consummated, an instrument, duly executed by such Permitted Party and the Transferee, in form reasonably satisfactory to Landlord, to the effect that such Transferee assumes all of the obligations of such Permitted Party to the extent arising under the applicable Occupancy Agreement from and after the date of such assignment, (ii) such sale of all or substantially all of the assets of such Permitted Party is not principally for the purpose of transferring such Permitted Party's interest in such Occupancy Agreement, and (iii) Tenant, within ten (10) Business Days after such sale, provides Landlord with a certificate, executed by an officer of the applicable Permitted Party, stating that the requirement described in clause (ii) above has been satisfied.
- (D) The direct or indirect transfer of shares or equity interests in a Permitted Party (including, without limitation, the issuance of treasury stock, or the creation or issuance of a new class of stock, in either case in the context of an initial public offering or in the context of a subsequent offering of equity securities) shall be permitted without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that in each case (i) such transfer is not principally for the purpose of transferring the interest of such Permitted Party under the applicable Occupancy Agreement, (ii) Tenant gives Landlord notice of such transfer not later than the tenth (10th) Business Day after the occurrence thereof, and (iii) Tenant, within ten (10) Business Days after the date that such transfer occurs, provides Landlord with a certificate, executed by an officer of the applicable Permitted Party, stating that the requirement described in clause (i) has been satisfied (except that Tenant shall not be required to comply with this clause (iii) to the extent that such direct or indirect transfer of shares or equity interests is accomplished through the public "over-the-counter" securities market or through any recognized stock exchange).
- (E) A Permitted Party shall have the right to sublease or license (or further sublease or sublicense) the Premises, or any portion thereof, to an Affiliate of such Permitted Party, without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a

Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that in each case (i) Tenant gives to Landlord a copy of such sublease or license, not later than the tenth (10th) Business Day after any such sublease or license is consummated, (ii) Tenant, with such copy of such sublease or license, provides Landlord with a certificate, executed by an officer of the applicable Permitted Party, stating that the Person to which such Permitted Party is so subleasing or licensing the Premises or a portion thereof constitutes an Affiliate of such Permitted Party, and (iii) such sublease includes the Basic Sublease Provisions, provided, however, that notwithstanding anything to the contrary contained in this Lease or in the Basic Sublease Provisions: (x) Landlord shall be bound by any termination of such sublease or any modification or amendment of such sublease which does not delete therefrom the Basic Sublease Provisions, and (y) the requirement that the subtenant under such sublease shall, at Landlord's option, attorn to Landlord as a direct tenant pursuant to the then executory provisions of such sublease if this Lease terminates shall be inapplicable. In addition to the foregoing, Tenant shall have the right to permit its Affiliates to occupy portions of the Premises without a written sublease or license.

# 17.10. Special Occupant.

Tenant may permit portions of the Premises to be occupied, at any time and from time to time, by Persons who are not members, officers or employees of Tenant (each such Person who is permitted to occupy portions of the Premises pursuant to this Section 17.10 being referred to herein as a "Special Occupant"), without (x) Landlord's prior approval, (y) Landlord's having the right to consummate a Recapture Termination in respect thereof, and (z) Tenant's being required to pay Transfer Profit to Landlord in connection therewith, provided that, in each case, (i) no demising walls are erected in the Premises separating the space used by a Special Occupant from the remainder of the Premises, (ii) the Special Occupant uses the Premises in conformity with all applicable provisions of this Lease, (iii) the use of any portion of the Premises by any Special Occupant shall not create any real property interest of the Special Occupant in or to the Premises, (iv) the portion of the Premises used by all Special Occupants shall not exceed fifteen percent (15%) of the Rentable Area of the Premises, (v) such Person maintains a business relationship with Tenant (other than by virtue of such occupancy) and such business relationship extends during the term of such occupancy, (vi) the Special Occupant does not pay for its occupancy rights an amount greater than the Rental that is reasonably allocable to the portion of the Premises that the Special Occupant has the right to occupy (it being understood that amounts that the Special Occupant pays to Tenant to reimburse Tenant reasonably for customary office services shall not be included in the calculation of the amount that the Special Occupant pays for its occupancy rights as provided in this clause (vi)), and (vii) at least ten (10) days prior to a Special Occupant taking occupancy of a portion of the Premises, Tenant gives notice to Landlord advising Landlord of (1) the name and address of such Special Occupant, (2) the character and nature of the business to be conducted by such Special Occupant, (3) the number of square feet of Rentable Area to be occupied by such Special Occupant, (4) the duration of such occupancy, and (5) the fee, if any, to be paid by such Special Occupant for its use of the applicable portion of the Premises. Within ten (10) Business Days after request by Landlord from time to time, Tenant shall provide Landlord with a list of the names of all Special Occupants then occupying any portion of the Premises and a description of the spaces occupied thereby.

#### 17.11. Recognition Agreements.

- (A) The term "Applicable Terms" shall mean all of the terms and conditions set forth in this Lease, except that:
- (1) the aggregate of the annual Fixed Rent and Escalation Rent (the "<u>Aggregate Rental</u>") payable by the applicable subtenant at any time from and after the Recognition Effective Date shall be an amount equal to the greater of (A) the Aggregate Rental that would have been payable by the applicable subtenant under the Major Sublease at such time if the applicable Major Sublease remained in effect, and (B) the Aggregate Rental that would have been payable hereunder for the Premises at such time (or the applicable portion thereof that is demised by the Major Sublease) (assuming that this Lease remained in effect);
- (2) Tenant's Operating Expense Share shall be redetermined based on the ratio that (I) the number of square feet of Rentable Area of the Premises that constitutes the space demised by the applicable Major Sublease, bears to (II) the number of square feet of Rentable Area of the Office Unit,
- (3) Tenant's Tax Share shall be redetermined based on the ratio that (I) the number of square feet of Rentable Area of the space demised by the applicable Major Sublease, bears to (II) the number of square feet of Rentable Area of the Office Unit,
- (4) the term of the applicable subtenant's direct tenancy shall expire, with respect to each portion of Rentable Area that is demised by the applicable Major Sublease, on the date that the term of this Lease applicable to such portion of Rentable Area would have expired had this Lease not terminated; provided, however, in no event shall such subtenant have the right to extend the term of such direct tenancy by exercise of a Renewal Option;
- (5) if, on the Recognition Effective Date, the tangible net worth of the applicable subtenant or the applicable Major Sublease Guarantor, determined in accordance with generally accepted accounting principles, is less than ten (10) times the annual Rental that is reasonably expected to be payable by the subtenant in connection with such direct tenancy, then, on the Recognition Effective Date, the applicable subtenant shall deposit with Landlord cash or a letter of credit reasonably acceptable to Landlord equal to such annual Rental as of the Recognition Effective Date as security for such subtenant's obligations to Landlord in respect of such direct tenancy;
  - (6) the applicable subtenant shall have no right to exercise the Option or the Renewal Option;
- (7) for purposes of such direct tenancy, references herein to the Premises shall be deemed to be references to the portion of the Premises demised by the applicable Major Sublease;
- (8) the applicable subtenant shall not be deemed to constitute the Person that executed and delivered this Lease initially (or an Affiliate of such Person) for purposes hereof;

- (9) the applicable subtenant shall not have the right to such direct tenancy (and accordingly, the applicable subtenant, at Landlord's option, shall have no right to remain in occupancy of the applicable portion of the Premises from and after the Recognition Effective Date) if (x) this Lease is terminated by reason of an Event of Default that derives from the applicable subtenant's default under the applicable Major Sublease, or (y) the applicable subtenant is the Person, or an Affiliate of the Person, that constituted Tenant immediately prior to the Recognition Effective Date;
  - (10) Landlord shall not have any obligation to consummate Recognition Agreements with further subtenants of any such subtenant;
  - (11) the Applicable Terms shall not include any rights that Tenant did not grant to the subtenant under the applicable Major Sublease; and
  - (12) Landlord shall not be:
- (a) liable for any act or omission of such subtenant's lessor prior to the Recognition Effective Date (except for any acts or omissions that (x) continue after the Recognition Effective Date, and (y) may be remedied by providing a service or performing a repair);
  - (b) subject to any credits, defenses or offsets which the applicable subtenant may have against any prior lessor;
- (c) bound by any payment of rental which the applicable subtenant may have made to any prior lessor more than thirty (30) days in advance of the month in which such payment was due; or
- (d) bound by any of the provisions of the applicable Major Sublease except to the extent that Landlord expressly consented to such provisions.
  - (B) The term "Major Sublease" shall mean a sublease, between Tenant, as sublessor, and a third party, as sublessee, which:
  - (1) Tenant enters into as the Transferor in accordance with the provisions of this Article 17,
  - (2) demises to the sublessee not less than the entire Rentable Area in a Major Sublease Unit,
- (3) expires no earlier than the day immediately preceding the Fixed Expiration Date (or the day immediately preceding the last day of the Renewal Term, with respect to any such subleases that Tenant executes and delivers from and after the first day of the Renewal Term, if Tenant has exercised the Renewal Option),
- (4) demises to the sublessee the entire Rentable Area of each Major Sublease Unit covered by such sublease (so that such sublease does not demise only a portion of any Major Sublease Unit), and

- (5) demises the uppermost or lowermost Major Sublease Unit or Major Sublease Units in the Premises (or, if Landlord has theretofore entered into a Recognition Agreement as contemplated by this Section 17.11 with a sublessee under another Major Sublease for space in the Premises, then the condition described in this Section 17.11(B)(5) shall be deemed to be satisfied if the Major Sublease Units demised by such other Major Sublease, and the Major Sublease Units demised by the particular sublease in question, constitutes the uppermost or lowermost portion of the Premises).
- (C) The term "Major Sublease Guarantor" shall mean a Person that executes and delivers a Recognition Agreement or another agreement to guaranty (on terms that are reasonably acceptable to Landlord) the performance of the obligations of the subtenant under a Major Sublease on the Applicable Terms if such subtenant becomes the direct tenant of Landlord.
  - (D) The term "Major Sublease Unit" shall mean the entire Rentable Area on a particular floor of the Building.
- (E) The term "Recognition Effective Date" shall mean the date that Landlord becomes the direct lessor of the applicable subtenant under a Major Sublease as contemplated by a Recognition Agreement.
- (F) If Tenant enters into a Major Sublease, then, subject to the terms of this Section 17.11, Landlord, on or prior to fifteen (15) Business Days after Tenant's request, shall execute and deliver to the applicable subtenant under such Major Sublease, and Tenant shall cause the subtenant under the applicable Major Sublease to execute and deliver to Landlord, an agreement (a "Recognition Agreement"), substantially in the form annexed as Exhibit "17.11" attached hereto and made a part hereof, to the effect that if this Lease terminates during the term of the applicable Major Sublease for any reason other than by reason of the occurrence of a fire or other casualty, or a condemnation, or Tenant's exercising Tenant's right to terminate this Lease in accordance with the express terms hereof, then (i) Landlord will not evict such subtenant, disturb such subtenant's possession or terminate or disturb such subtenant's occupancy of the space that the applicable Major Sublease demises, and will recognize such subtenant as the direct tenant of Landlord on the Applicable Terms from and after the Recognition Effective Date, and (ii) such subtenant will recognize Landlord as such subtenant's direct landlord on the Applicable Terms from and after the Recognition Effective Date. Tenant shall not have the right to request a Recognition Agreement as contemplated by this Section 17.11 (w) if the subtenant under the applicable Major Sublease is Tenant or an Affiliate of Tenant, (x) if an Event of Default has occurred and is then continuing, or (y) if the financial condition of the applicable subtenant is not reasonably satisfactory to Landlord (it being understood that if the tangible net worth of such subtenant or any applicable Major Sublease Guarantor, determined in accordance with generally accepted accounting principles, is equal to or greater than ten (10) times the annual Rental that would be reasonably expected to be payable by the applicable subtenant to Landlord pursuant to the Applicable Terms, then such subtenant's fina
- (G) Tenant shall submit to Landlord, with each request for a Recognition Agreement, financial information regarding the subtenant for whose benefit such agreement is

requested, including, without limitation, documentation of such subtenant's net worth, determined in accordance with generally accepted accounting principles.

- (H) Tenant shall reimburse Landlord for the reasonable out-of-pocket costs incurred by Landlord in consummating a Recognition Agreement within thirty (30) days after Landlord's request therefor. Landlord shall include with any such request reasonable supporting documentation for the charges described therein.
- (I) Subject to the terms of this Section 17.11(I), if this Section 17.11 obligates Landlord to execute and deliver a Recognition Agreement with the subtenant under a Major Sublease, then Landlord shall use reasonable efforts to cause the Condominium Board, each Mortgagee and each Superior Lessor to agree to be bound by the Recognition Agreement (if the Condominium Board, such Mortgagee or such Superior Lessor subsequently succeeds to the interest of Landlord), promptly after Tenant's request. Nothing contained in this Section 17.11(I) obligates Landlord to incur costs (other than *de minimis* costs) or to institute any legal proceeding in so using reasonable efforts to cause the Condominium Board and each Mortgagee and each Superior Lessor to agree to be bound by the Recognition Agreement.

# Article 18 TENANT'S RIGHT OF FIRST OFFER TO LEASE

#### 18.1. Right of First Offer.

- (A) Landlord, in its capacity as owner of the Office Unit, shall not lease to any Person other than Tenant or Landlord's Affiliate that is leasing the 29<sup>th</sup> Floor Option Space for its own use and occupancy the 29<sup>th</sup> Floor Option Space (or a part thereof) at any time during the Term, without first instituting the procedure described in, and subject to the limitations set forth in, this Article 18. Landlord, in its capacity as Office Unit One Owner, shall not lease to any Person other than Tenant or Landlord's Affiliate that is leasing the Unit One Option Space for its own use and occupancy the Unit One Option Space (or a part thereof) at any time during the Term, without first instituting the procedure described in, and subject to the limitations set forth in, this Article 18. Landlord (i) represents and warrants that Landlord is the current Office Unit One Owner, and (ii) agrees that by virtue of its execution and delivery of this Lease, any successor to Landlord's interest in Office Unit One shall be bound by the provisions of this Article 18.
  - (B) The term "Option Space" shall mean, collectively, the 29th Floor Option Space and the Unit One Option Space.
- (C) The term "29th Floor Option Space" shall mean the portion of the twenty-ninth (29th) floor of the Building as shown on the applicable floor plan annexed as Exhibit "18.1" attached hereto and made a part hereof, which 29th Floor Option Space is comprised of twelve thousand six hundred eight (12,608) square feet of Rentable Area. Landlord represents that the Office Unit is comprised of one hundred ninety thousand two hundred fourteen (190,214) square feet of Rentable Area.

- (D) The term "<u>Unit One Option Space</u>" shall mean any leasable space in the Building that is located on the seventeenth (17th), eighteenth (18th), nineteenth (19th), and twentieth (20th) floors of the Building, each as shown on the applicable floor plans annexed as Exhibit "18.1" attached hereto. Landlord represents that (x) each floor that comprises the Option Space (other than the twenty-ninth (29<sup>th</sup>) floor) is comprised of twenty-two thousand (22,000) square feet of Rentable Area, and (y) the Office Unit One is comprised of six hundred ninety-six thousand seven hundred eighty-six (696,786) square feet of Rentable Area.
- (E) Landlord represents and warrants to Tenant that as of the date hereof no other Person has the right to lease any Option Space that is superior to the rights of Tenant to lease the Option Space as provided in this Article 18, other than the tenants to which the applicable Option Space has been leased as of the date hereof.

# 18.2. Option Notice.

- (A) Landlord shall institute the procedure described in this Article 18 by giving notice thereof (the "Option Notice") to Tenant, which Option Notice shall (i) describe the Option Space (or the applicable portion thereof) (the Option Space (or such portion thereof) that is described in a particular Option Notice being referred to herein as the "Applicable Option Space"), (ii) have attached thereto a floor plan depicting the Applicable Option Space, (iii) set forth the date that Landlord reasonably expects that the Applicable Option Space will be vacant and available for Tenant's occupancy (such date designated by Landlord being referred to herein as the "Scheduled Option Space Commencement Date"), and (iv) set forth Landlord's calculation of the number of square feet of Rentable Area in the Applicable Option Space. The Scheduled Option Space Commencement Date shall not be more than four hundred fifty (450) days, and not less than one hundred eighty (180) days, after the date that Landlord gives the Option Notice to Tenant. If, however, any particular Option Space becomes available earlier than the scheduled expiration date set forth in the lease or occupancy agreement in respect of such Option Space (other than by virtue of a voluntary termination of the lease or occupancy agreement for such Option Space entered into by Landlord and the tenant or occupant of such Option Space), then the Scheduled Option Space Commencement Date shall instead be no more than four hundred fifty (450) days, and not less than sixty (60) days, after the date that Landlord gives the Option Notice to Tenant.
- (B) Landlord agrees to use reasonable efforts to discuss with Tenant any Option Space that Landlord expects will become available for leasing pursuant to this Article 18, and the Rental Value therefor, within a reasonable period of time prior to the date that Landlord gives Tenant an Option Notice for such Option Space (it being agreed that neither Landlord nor Tenant shall be bound by any matters discussed with respect to such Option Space and any such discussions shall in no event be deemed to modify or waive the terms and provisions of this Article 18).

# 18.3. Option Procedure.

(A) Subject to the terms of this Section 18.3, Tenant shall have the option (the "Option") to lease the Applicable Option Space for a term (the "Option Term") commencing on the Option Space Commencement Date and expiring on the Expiration Date by giving notice

thereof (the "Option Response Notice") to Landlord not later than the thirtieth (30th) day after the date that Landlord gives the Option Notice to Tenant. Time shall be of the essence as to the date by which Tenant must give the Option Response Notice to Landlord to exercise the Option. Tenant shall have the right to elect to lease less than all of the Applicable Option Space offered to Tenant in a particular Option Notice as follows: (i) Tenant may elect to lease Applicable Option Space comprised of the twenty-ninth (29<sup>th</sup>) floor of the Building separate and apart from any other Applicable Option Space offered to Tenant; (ii) Tenant may elect to lease one (1) or more floors of any other Applicable Option Space offered to Tenant but, subject to the provisions of clause (i) above with respect to the twenty-ninth (29<sup>th</sup>) floor, must first elect to lease the Applicable Option Space that is located on the highest floor of the Applicable Option Space that is described in such Option Notice and any other Applicable Option Space that is vertically contiguous to such highest floor of the Applicable Option Space; and (iii) with respect to any Applicable Option Space offered to Tenant that is located on a particular floor in the Building, Tenant may not elect to lease less than all of the Rentable Area of such Applicable Option Space located on such floor. Tenant shall set forth in the Option Notice the Option Space that Tenant elects to lease (if Tenant elects to lease less than all of the Option Space offered to Tenant in an Option Notice), it being agreed that if Tenant does not so set forth in the Option Notice the Option Space that Tenant elects to Lease, then Tenant shall be deemed to have elected to lease all of the Option Space described in an Option Notice.

(B) If (x) Tenant does not give the Option Response Notice to Landlord on or prior to the thirtieth (30th) day after the date that Landlord gives the Option Notice to Tenant, or (y) elects not to lease certain Applicable Option Space described in an Option Notice, then Landlord shall thereafter have the right to lease the Applicable Option Space (or any part thereof) to any other Person on terms acceptable to Landlord in Landlord's sole discretion without being required to make any other offer to Tenant regarding the Applicable Option Space under this Article 18 (and, accordingly, such Applicable Option Space shall not thereafter constitute Option Space); provided, however, that if Landlord leases the Applicable Option Space to another Person (after Tenant does not elect to exercise the Option with respect thereto), then the provisions of this Article 18 shall apply (including, without limitation, the provisions of Section 18.4(B) hereof) before Landlord leases such Applicable Option Space to another Person (other than Tenant or Landlord's Affiliate) upon the expiration or earlier termination of the term of such lease with such other Person. If (x) Landlord gives Tenant an Option Notice, (y) Tenant does not give an Option Response Notice to Landlord to exercise the Option pursuant to this Section 18.3 hereof or otherwise notifies Landlord that Tenant elects not to lease any or all of the Applicable Option Space described in such Option Notice, and (z) the Applicable Option Space or a portion of the Applicable Option Space remains unleased on the date that is one (1) year after the date Landlord gave Tenant such Option Notice, then Landlord shall not thereafter lease to any Person other than Tenant or Landlord's Affiliate that is leasing the Option Space for its own use and occupancy the Option Space (or a part thereof), without first again instituting the procedure described in, and subject to the limitations set forth in, this Article 18 with respect to such unleased Applicable Option Space. Tenant shall not h

#### 18.4. <u>Certain Limitations.</u>

- (A) Tenant shall not have the right to exercise the Option with respect to any Applicable Option Space (and, accordingly (x) Landlord shall have no obligation to give an Option Notice to Tenant, and (y) Landlord shall have the right to lease the Applicable Option Space to any other Person without first offering the Applicable Option Space to Tenant as contemplated by this Article 18) if:
  - (1) the Minimum Occupancy Requirement is not satisfied,
  - (2) the Minimum Demise Requirement is not satisfied, or
  - (3) the Applicable Option Space constitutes Recapture Space with respect to which Landlord exercised its rights under Section 17.3 hereof.
- (B) Tenant shall not have the right to exercise the Option prior to Landlord's leasing the Option Space (or the applicable portion thereof) to any Person that then occupies the Option Space (or such portion thereof) (regardless of whether such leasing is pursuant to an option or right contained in such Person's lease), and, accordingly, (I) Landlord shall have no obligation to give an Option Notice to Tenant with respect to the Option Space (or such portion thereof), and (II) Landlord shall have the right to lease the Option Space (or such portion thereof) to any such Person without first offering the Option Space (or the applicable portion thereof) to Tenant as contemplated by this Article 18.
- (C) If (i) Tenant exercises the Option, and (ii) at any time prior to the Option Space Commencement Date, (i) the Minimum Demise Requirement is not satisfied, or (ii) the Minimum Occupancy Requirement is not satisfied, then, at any time prior to the Option Space Commencement Date, Landlord shall have the right to declare Tenant's exercise of the Option ineffective by giving notice thereof to Tenant, in which case Landlord shall have the right to lease the Applicable Option Space (or any portion thereof) to any other Person on terms acceptable to Landlord in Landlord's sole discretion.
- (D) Tenant shall not have the right to exercise the Option from and after the Option Cutoff Date, and, accordingly, from and after the Option Cutoff Date, (I) Landlord shall have no obligation to give an Option Notice to Tenant with respect to the Option Space (or any portion thereof), and (II) Landlord shall have the right to lease the Option Space (or such portion thereof) to any other Person without first offering the Option Space (or such portion thereof) to Tenant as contemplated by this Article 18 (subject to the provisions of Section 18.3 hereof). The term "Option Cutoff Date" shall mean the date that is five (5) years before the Fixed Expiration Date, except that (x) if Tenant exercises the Renewal Option, then the Option Cutoff Date shall be the date that is five (5) years before the last day of the Renewal Term, and (y) if at the time any particular Option Space is available there then remains less than five (5) years before the Fixed Expiration Date and Tenant's right to exercise the Renewal Option has not yet lapsed, then Landlord shall be required to give Tenant an Option Notice but Tenant must first exercise the Renewal Option in accordance with Article 19 hereof prior to Tenant's exercising the Option (if Tenant elects to exercise the Option).

# 18.5. <u>Lease Provisions Apply.</u>

If Tenant exercises the Option in accordance with the provisions of this Article 18, then, on the Option Space Commencement Date for such Applicable Option Space, the following provisions shall become effective:

- (A) If such Applicable Option Space is the 29<sup>th</sup> Floor Option Space, such Applicable Option Space shall be added to the Premises for purposes of this Lease. If such Applicable Option Space is all or a portion of the Unit One Option Space, then the Office Unit One Owner and Tenant shall enter into a lease in accordance with the provisions of Section 18.7 hereof.
- (B) If such Applicable Option Space is the entire 29<sup>th</sup> Floor Option Space, then Tenant's Tax Share shall be increased by six and five hundred eight one-thousandths percent (6.508%); if such Applicable Option Space is only a portion of the 29<sup>th</sup> Floor Option Space, Tenant's Tax Share shall be increased by the percentage derived by multiplying (x) six and five hundred eight one-thousandths percent (6.508%), by (y) a fraction, the numerator of which is the number of square feet of Rentable Area in the Applicable Option Space and the denominator of which is twelve thousand six hundred eight (12,608).
- (C) If such Applicable Option Space is all or a portion of the Unit One Option Space, the Tax Payment for each Tax Year from and after the Option Space Commencement Date with respect to such Unit One Option Space shall be an amount equal to the product obtained by multiplying (A) the quotient (expressed as a percentage) derived by dividing (I) the number of square feet of Rentable Area contained in Space by (II) the total number of square feet of Rentable Area contained in Office Unit One, by (B) the excess of (x) Taxes for Office Unit One for the applicable Tax Year, over (y) the Taxes for Office Unit One for the Tax Year in which the Option Space Commencement Date occurs.
- (D) If such Applicable Option Space is the entire 29<sup>th</sup> Floor Option Space, Tenant's Operating Expense Share shall be increased by six and sixty-three one hundredths percent (6.63%); if such Applicable Option Space is only a portion of the 29<sup>th</sup> Floor Option Space, Tenant's Operating Expense Share will be increased by the percentage arrived at by multiplying (x) six and sixty-three one hundredths percent (6.63%) by (y) a fraction, the numerator of which is the number of square feet of Rentable Area in the Applicable Option Space and the denominator of which is twelve thousand six hundred eight (12,608).
- (E) If such Applicable Option Space is all or a portion of the Unit One Option Space, the Operating Expense Payment for each Operating Expense Year from and after the Option Space Commencement Date with respect to such Unit One Option Space shall be an amount equal to the product obtained by multiplying (I) the quotient (expressed as a percentage) derived by dividing (A) the number of square feet of Rentable Area contained in such Unit One Option Space by (B) the total number of square feet of Rentable Area contained in Office Unit One, by (II) the excess of (x) Operating Expenses for Office Unit One for the applicable Operating Expense Year, over (y) the Operating Expenses for Office Unit One for the calendar year in which the Option Space Commencement Date occurs.

- (F) Landlord shall not be obligated to perform any work or make any installations in such Applicable Option Space or grant Tenant a work allowance therefor.
  - G) The Fixed Rent for such Applicable Option Space shall be an amount equal to the Fair Market Rent therefor.

# 18.6. Delivery.

Landlord shall deliver vacant and exclusive possession of the Applicable Option Space to Tenant on the later to occur of (i) the Scheduled Option Space Commencement Date therefor, and (ii) the thirtieth (30<sup>th</sup>) day after Tenant gives the Option Response Notice to Landlord for such Applicable Option Space; provided, however, that (x) if a Person remains in occupancy of the Applicable Option Space (or any portion thereof) on the later to occur of the foregoing dates, then Landlord, at Landlord's expense, shall use reasonable diligence to cause vacant and exclusive possession of such Applicable Option Space to be delivered to Tenant as promptly as reasonably practicable thereafter but in no event earlier than five (5) Business Days after the date Landlord gives Tenant notice of the delivery of such Applicable Option Space (the Scheduled Option Space Commencement Date, or such later date on which Landlord delivers vacant and exclusive possession of any Applicable Option Space as contemplated by this Section 18.6, being referred to herein as the "Option Space Commencement Date"), and (y) if such Person's right to remain in occupancy of such Applicable Option Space (or a portion thereof) terminates prior to the Scheduled Option Space Commencement Date therefor, then Landlord shall have no liability to Tenant (except as otherwise set forth in clause (x) above), and Tenant shall have no right to terminate or rescind this Lease or Tenant's exercise of the Option or reduce the Rental, in each case deriving from Landlord's failure to deliver vacant and exclusive possession of such Applicable Option Space to Tenant on the Scheduled Option Space Commencement Date therefor. Landlord and Tenant intend that this Section 18.6 constitutes an "express provision to the contrary" for purposes of Section 223-a of the New York Real Property Law. Notwithstanding anything to the contrary contained herein, if Landlord fails to deliver possession of the Applicable Option Space to Tenant on or before the date that is one hundred eighty (180) days after the Scheduled Option Space Commencement Date therefor, then Tenant shall have the right to (i) terminate this Lease solely with respect to the 29<sup>th</sup> Floor Option Space if the Applicable Option Space is the 29th Floor Option Space, or (ii) rescind Tenant's election to lease the Applicable Option Space, if the Applicable Option Space is all or a portion of the Unit One Option Space, in either case, by giving notice thereof to Landlord at any time after such one hundred eightieth (180<sup>th</sup>) day that Landlord has not delivered such Applicable Option Space to Tenant, in which event Tenant shall have no further obligations or liabilities in connection with such Applicable Option Space.

# 18.7. Special Rules.

(A) If Tenant exercises the Option for any portion of the Unit One Option Space in accordance with the terms of this Article 18, then, subject to the terms of this Section 18.7, Landlord shall cause the owner of the superior interest in such Applicable Option Space to lease such Applicable Option Space to Tenant, on the terms set forth in this Lease (including, without limitation, the terms with respect to renewal set forth in Article 19 hereof), except to the extent such terms are not applicable to the Applicable Option Space as provided in this Article

18. Landlord and Tenant each acknowledge that such other lease and this Lease shall be independent of each other, and, accordingly (I) the landlord's default under such other lease shall not constitute a default of Landlord under this Lease, and Landlord's default hereunder shall not constitute a default of such other landlord under such other lease, and (II) the occurrence of an event of default hereunder shall not constitute a corresponding default of the tenant under such other lease, and the occurrence of an event of default (after the expiration of any applicable grace period) by tenant under such other lease shall not constitute an Event of Default hereunder.

- (B) Landlord shall have no liability to Tenant whatsoever under this Lease if, for any reason, the Office Unit One Owner fails to comply with the provisions of this Article 18 with respect to the Unit One Option Space, or otherwise fails deliver to Tenant any Unit One Option Space for which Tenant has exercised the Option hereunder, notwithstanding that the Office Unit One Owner may then be Landlord or an Affiliate of Landlord, it being agreed that if the Office Unit One Owner so fails to comply with the provisions of this Article 18 with respect to the Unit One Option Space or otherwise fails to deliver to Tenant any Unit One Option Space for which Tenant has exercised the Option hereunder, then Tenant shall have the right to bring an action against the Office Unit One Owner for specific performance or to seek any other remedy against the Office Unit One Owner that may be available to Tenant at law or in equity, provided, however, that in no event shall the Office Unit One Owner be liable for consequential damages that Tenant might incur as a result of the Office Unit One Owner's failure to comply with the provisions of this Article 18 with respect to the Unit One Option Space or otherwise fail to deliver any Unit One Option Space to Tenant.
- (C) The Office Unit One Owner shall have no liability to Tenant whatsoever under this Lease if, for any reason, Landlord fails to comply with the provisions of this Article 18 with respect to the 29<sup>th</sup> Floor Option Space, or otherwise fails deliver to Tenant any 29<sup>th</sup> Floor Option Space for which Tenant has exercised the Option hereunder, notwithstanding that Landlord may then be the Office Unit One Owner or an Affiliate of the Office Unit One Owner, it being agreed that if Landlord so fails to comply with the provisions of this Article 18 with respect to the 29<sup>th</sup> Floor Option Space or otherwise fails to deliver to Tenant any 29<sup>th</sup> Floor Option Space for which Tenant has exercised the Option hereunder, then Tenant shall have the right to bring an action against Landlord for specific performance or to seek any other remedy against Landlord that may be available to Tenant at law or in equity, provided, however, that in no event shall Landlord be liable for consequential damages that Tenant might incur as a result of Landlord's failure to comply with the provisions of this Article 18 with respect to the 29<sup>th</sup> Floor Option Space or otherwise fail to deliver any 29<sup>th</sup> Floor Option Space to Tenant.

Article 19 RENEWAL

# 19.1. Renewal Option.

(A) Subject to the terms of this Article 19, Tenant shall have the option (the "Renewal Option") to extend the term of this Lease for the entire Premises for one (1) additional period of five (5) years (the "Renewal Term"), which Renewal Term shall commence on the day

immediately succeeding the Fixed Expiration Date and end on the day that is five (5) years after the Fixed Expiration Date, provided that this Lease has not been previously terminated.

(B) The Renewal Option shall be exercisable only by Tenant's delivering a notice (the "Renewal Notice") to Landlord stating that Tenant elects to exercise the Renewal Option not later than the four hundred fifty-sixth (456th) day before the Fixed Expiration Date (as to which date time shall be of the essence). Between the five hundred forty-sixth (546<sup>th</sup>) day and the four hundred fifty-sixth (456<sup>th</sup>) day prior to the Fixed Expiration Date, Landlord agrees to use reasonable efforts to discuss with Tenant Tenant's plans and potential needs with respect to the Renewal Option and the Renewal Premises, and the potential Rental Value for the Renewal Premises (it being agreed that neither Landlord nor Tenant shall be bound by any matters discussed with respect to the Renewal Option and any such discussions shall in no event be deemed to modify or waive the terms and provisions of this Article 19).

#### 19.2. <u>Partial Renewal Space.</u>

- (A) Tenant shall have the right to renew the term hereof for the Renewal Term with respect to either (x) the entire Premises demised hereby on the Fixed Expiration Date (and, at Tenant's option, the entire Rentable Area of any Option Space that has theretofore been leased by Tenant in Office Unit One pursuant to Article 18 hereof as long as Tenant renews for the entire Premises demised hereby on the Fixed Expiration Date), or (y) portions of the Premises that are described in Exhibit "19.2" attached hereto and made a part hereof (the portion of the Premises with respect to which Tenant exercises the Renewal Option in accordance with the parameters set forth in this Section 19.2 being referred to herein as the "Partial Renewal Space"; the Premises, or the Partial Renewal Space with respect to which Tenant exercises the Renewal Option, being referred to herein as the "Renewal Premises"; the portion of the Premises that does not constitute the Partial Renewal Space is referred to herein as the "Removed Space").
- (B) If (x) Tenant gives the Renewal Notice to Landlord, and (y) Tenant fails to (I) indicate therein that Tenant is exercising the Renewal Option for only the Partial Renewal Space, or (II) identify therein the Partial Renewal Space with reasonable particularity, then Tenant shall be deemed to have designated that the Renewal Premises constitutes the entire Premises demised hereby as of the Fixed Expiration Date.
- (C) If Tenant exercises the Renewal Option for only the Partial Renewal Space as contemplated by this Section 19.2, then (x) on the Fixed Expiration Date, Tenant shall surrender to Landlord vacant and exclusive possession of the Removed Space in accordance with the provisions of this Lease that govern Tenant's obligations in respect of the delivery of vacant and exclusive possession of the Premises to Landlord upon the expiration or earlier termination of the Term, and (y) on or prior to the Fixed Expiration Date, Tenant, at Tenant's expense and otherwise in accordance with the terms of Article 7 hereof, shall demise the Removed Space separately from the Partial Renewal Space.

# 19.3. <u>Lease Provisions Apply.</u>

If Tenant exercises the Renewal Option, then the leasing of the Premises during the Renewal Term shall be upon the terms set forth herein, except that:

- (A) the Fixed Rent for the Renewal Premises during the Renewal Term shall be the Rental Value thereof;
- (B) Landlord shall have no obligation to perform any work in connection with Tenant's exercise of the Renewal Option;
- (C) Landlord shall have no obligation to grant to Tenant any work allowance in connection with Tenant's exercise of the Renewal Option;
- (D) if Tenant exercises the Renewal Option for the Partial Renewal Space (and not for the entire Premises), then, effective as of the first (1st) day of the Renewal Term, Tenant's Tax Share shall be recalculated as the ratio (expressed as a percentage) that the number of square feet of Rentable Area of the Premises that constitutes the Partial Renewal Space bears to the number of square feet of Rentable Area of the Office Unit;
- (E) if Tenant exercises the Renewal Option for the Partial Renewal Space (and not for the entire Premises), then, effective as of the first (1st) day of the Renewal Term, Tenant's Operating Expense Share shall be recalculated as the ratio (expressed as a percentage) that the number of square feet of Rentable Area of the Premises that constitutes the Partial Renewal Space bears to the number of square feet of Rentable Area of the Office Unit; and
  - (F) the provisions of this Article 19 shall not be applicable to permit Tenant to further extend the Term.

## Article 20 FAIR MARKET RENT

## 20.1. <u>Certain Definitions.</u>

- (A) The term "Rental Value" shall mean, at any particular time, an amount equal to the Fair Market Rent of the Applicable Area on the Applicable Date; provided, however, that at no time shall the Rental Value for the Applicable Area be less than the Base Rental Amount.
  - (B) The term "Fair Market Rent" shall mean annual fair market rental value of the Applicable Area on the Applicable Date.
  - (C) The term "Applicable Area" shall mean:
  - (1) the Premises, in connection with the determination of the Fair Market Rent thereof,
  - (2) the Applicable Option Space, in connection with the determination of the Fair Market Rent thereof, and
  - (3) the Antennae Site, in connection with the determination of the Fair Market Rent thereof.

- (D) The term "Applicable Date" shall mean:
- (1) the Fixed Expiration Date, in connection with the determination of the Fair Market Rent of the Premises,
- (2) the Scheduled Option Space Commencement Date, in connection with the determination of the Fair Market Rent for the Applicable Option Space, and
  - (3) the Antennae Commencement Date, in connection with the determination of the Fair Market Rent therefor.
- (4) The term "Base Rental Amount" shall mean, in connection with the determination of the Rental Value of the Premises, an amount equal to the sum of (X) the Fixed Rent that is payable hereunder as of the Fixed Expiration Date for the Premises or the portion thereof that constitutes the Renewal Premises (without taking into account any abatement of Fixed Rent that is in effect as of the Fixed Expiration Date), and (Y) the product obtained by multiplying (x) Five Dollars (\$5.00), by (y) the number of square feet of Rentable Area in the Premises or the portion thereof that constitutes the Renewal Premises on the last day of the initial term of this Lease.

### 20.2. <u>Fair Market Rent Assumptions.</u>

The Fair Market Rent shall be determined assuming that the Applicable Area is free and clear of all leases and tenancies (including this Lease), that Landlord has had a reasonable time to locate a tenant, and that neither Landlord nor the prospective tenant is under any compulsion to rent, and taking into account all relevant factors.

## 20.3. Fair Market Procedure.

- (A) Landlord and Tenant shall each deliver simultaneously to the other, at Landlord's office, a notice (each, a "Rent Notice"), on a date mutually agreed upon, but in no event later than:
- (1) one hundred eighty (180) days before the Fixed Expiration Date, with respect to the Rent Notice for the determination of the Fair Market Rent for the Premises.
- (2) the later to occur of (I) three (3) months before the Scheduled Option Space Commencement Date, and (II) the thirtieth (30th) day after the date that Tenant gives the applicable Option Response Notice to Landlord, with respect to the Rent Notice for the determination of the Fair Market Rent for the Applicable Option Space, and
- (3) the date that Landlord grants to Tenant access to the Antennae Site, as the case may be, which Rent Notice shall set forth each of their respective determinations of the Fair Market Rent (Landlord's determination of the Fair Market Rent is referred to as "Landlord's Determination" and Tenant's determination of the Fair Market Rent is referred to as "Tenant's Determination"). If (i) Landlord fails to give Landlord's Determination to Tenant, and (ii) Tenant tenders Tenant's Determination to Landlord, then the Fair Market Rent for the Applicable Area shall be Tenant's Determination. If (i) Tenant fails to give Tenant's

Determination to Landlord, and (ii) Landlord tenders Landlord's Determination to Tenant, then the Fair Market Rent for the Applicable Area shall be Landlord's Determination.

- (B) If Landlord and Tenant do not agree on the Fair Market Rent for the Applicable Area within thirty (30) days after the date that Landlord gives Landlord's Determination to Tenant, and the date that Tenant gives Tenant's Determination to Landlord, then Landlord and Tenant shall select jointly an independent real estate appraiser that (i) neither Landlord nor Tenant, nor any of their respective Affiliates, has engaged during the immediately preceding period of three (3) years, and (ii) has at least ten (10) years of experience in leasing properties that are similar in character to the Building (such appraiser being referred to herein as the "Appraiser"). Landlord and Tenant shall each pay fifty percent (50%) of the Appraiser's fee. If Landlord and Tenant do not agree on the Appraiser within ten (10) days after the last day of such period of thirty (30) days, then either party shall have the right to institute an Expedited Arbitration Proceeding for the sole purpose of designating the Appraiser.
- (C) The parties shall instruct the Appraiser to (i) conduct the hearings and investigations that he or she deems appropriate, and (ii) choose either Landlord's Determination or Tenant's Determination as the better estimate of Fair Market Rent for the Applicable Area, within thirty (30) days after the date that the Appraiser is designated. The Appraiser's aforesaid choice shall be conclusive and binding upon Landlord and Tenant. Each party shall pay its own counsel fees and expenses, if any, in connection with the procedure described in this Article 20. The Appraiser shall not have the power to supplement or modify any of the provisions of this Lease.
- (D) If the final determination of the Fair Market Rent is not made on or before the Applicable Date in accordance with the provisions of this Article 20, then, pending such final determination, the Fair Market Rent shall be deemed to be an amount equal to the average of Landlord's Determination and Tenant's Determination. If, based upon the final determination hereunder of the Fair Market Rent, the payments made by Tenant on account of the Rental for the period prior to the final determination of the Fair Market Rent were less than the Rental payable for such period, then Tenant, not later than the tenth (10th) day after Landlord's demand therefor, shall pay to Landlord the amount of such deficiency, together with interest thereon at the Base Rate. If, based upon the final determination of the Fair Market Rent were more than the Rental due hereunder for such period, then Landlord, not later than the tenth (10th) day after Tenant's demand therefor, shall pay such excess to Tenant, together with interest thereon at the Base Rate; it being agreed that if Landlord fails to pay such amount to Tenant on or prior to thirty (30) days after the date of such demand by Tenant, then Tenant shall have the right to credit such amount (with interest as aforesaid) against the Rental thereafter accruing hereunder.

### Article 21 DEFAULT

### 21.1. Events of Default.

The term "Event of Default" shall mean the occurrence of any of the following events:

- (A) Tenant fails to pay any installment of Fixed Rent when due and such failure continues for ten (10) days after the date that Landlord gives written notice of such failure to Tenant;
- (B) Tenant fails to pay any installment of Rental (other than Fixed Rent) when due and such failure continues for twenty (20) days after the date that Landlord gives written notice of such failure to Tenant;
- (C) Tenant's interest under this Lease devolves upon or passes to any other Person, whether by operation of law or otherwise, except as expressly permitted under Article 17 hereof, unless reversed within ten (10) Business Days after written notice from Landlord;
- (D) A Permitted Party's interest (other than a Permitted Party that is Tenant) under this Lease devolves upon or passes to any other Person, whether by operation of law or otherwise, except as expressly permitted under Article 17 hereof, unless reversed within ten (10) Business Days after written notice from Landlord to Tenant, except that if (i) such default cannot be remedied by Tenant with reasonable diligence during such period of ten (10) Business Days, (ii) Tenant takes reasonable steps during such period of ten (10) Business Days to commence Tenant's remedying of such default, and (iii) Tenant prosecutes diligently Tenant's remedying of such default to completion, then an Event of Default shall not occur by reason of such default, provided that Tenant completes its remedying of such default within a reasonable period of time (under the circumstances in question) after the date that Landlord gives Tenant such notice.;
- (E) Tenant defaults in respect of Tenant's obligations under Section 4.15 hereof, and such default continues for more than five (5) Business Days after Landlord gives Tenant written notice thereof;
- (F) Tenant defaults in respect of Tenant's obligations under Section 7.5(A)(2) and such default continues for more than fifteen (15) days after Landlord gives Tenant written notice thereof; or
- (G) Tenant defaults in the observance or performance of any other covenant of this Lease on Tenant's part to be observed or performed and Tenant fails to remedy such default within thirty (30) days after Landlord gives Tenant written notice thereof, except that if (i) such default cannot be remedied with reasonable diligence during such period of thirty (30) days, (ii) Tenant takes reasonable steps during such period of thirty (30) days to commence Tenant's remedying of such default, and (iii) Tenant prosecutes diligently Tenant's remedying of such default to completion, then an Event of Default shall not occur by reason of such default, provided that Tenant completes its remedying of such default within a reasonable period of time (under the circumstances in question) after the date that Landlord gives Tenant such notice.

## 21.2. <u>Termination.</u>

If (1) an Event of Default occurs, and (2) Landlord, at any time thereafter, at Landlord's option, gives a notice to Tenant stating that this Lease and the Term shall expire and terminate on the fifth (5th) Business Day after the date that Landlord gives Tenant such notice, then this Lease and the Term and all rights of Tenant under this Lease shall expire and terminate as of the fifth (5th) Business Day after the date that Landlord gives Tenant such notice, and Tenant immediately shall quit and surrender the Premises, but Tenant shall nonetheless remain liable for all of its obligations hereunder, as provided in Article 23 hereof and Article 24 hereof.

## Article 22\_ TENANT'S INSOLVENCY

- 22.1. <u>Assignments pursuant to the Bankruptcy Code.</u>
  - (A) The term "Bankruptcy Code" shall mean 11 U.S.C. Section 101 et seq., or any statute of similar nature and purpose.
- (B) If Tenant, Tenant's trustee or Tenant as debtor-in-possession proposes to assign the tenant's interest hereunder pursuant to the provisions of the Bankruptcy Code to any Person that has made a bona fide offer to accept an assignment of the tenant's interest under this Lease on terms acceptable to Tenant, Tenant's trustee or Tenant as debtor-in-possession, then Tenant, Tenant's trustee or Tenant as debtor-in-possession shall give to Landlord notice of such proposed assignment no later than twenty (20) days after the date that Tenant, Tenant's trustee or Tenant as debtor-in-possession receives such offer, but in any event no later than ten (10) days before the date that Tenant, Tenant's trustee or Tenant as debtor-in-possession makes application to a court of competent jurisdiction for authority and approval to consummate such assignment. Such notice to Landlord shall (a) set forth the name and address of such Person that has made such bona fide offer, (b) set forth all of the terms and conditions of such bona fide offer, and (c) confirm that such Person will provide to Landlord adequate assurance of future performance that conforms with the terms of Section 22.1(C) hereof. Landlord shall have the right to accept an assignment of this Lease upon the same terms and conditions and for the same consideration, if any, as the bona fide offer made by such Person (less any brokerage commissions that would otherwise be payable by Tenant, Tenant's trustee or Tenant as debtor-in-possession out of the consideration to be paid by such Person in connection with such assignment of the tenant's interest under this Lease), by giving notice thereof to Tenant, Tenant's trustee or Tenant as debtor-in-possession at any time prior to the date that is ten (10) days prior to the effective date of such proposed assignment.
- (C) A Person that submits a bona fide offer to take by assignment the tenant's interest under this Lease as described in Section 22.1(B) hereof shall be deemed to have provided Landlord with adequate assurance of future performance only if such Person (a) deposits with Landlord simultaneously with such assignee's taking by assignment the tenant's interest under this Lease an amount equal to the then annual Fixed Rent, as security for the faithful performance and observance by such assignee of the tenant's obligations of this Lease (and such Person gives to Landlord, at least five (5) days prior to the date that the proposed assignment

becomes effective, information reasonably satisfactory to Landlord that indicates that such Person has the ability to post such deposit), (b) gives to Landlord, at least five (5) days prior to the date that the proposed assignment becomes effective, such Person's financial statements, audited by a certified public accountant in accordance with generally accepted accounting principles, consistently applied, for the three (3) fiscal years that immediately precede such assignment, that indicate that such Person has a tangible net worth of at least ten (10) times the then annual Fixed Rent for each of such three (3) years, and (c) gives to Landlord, at least five (5) days prior to the date that the proposed assignment becomes effective, such other information or takes such action that in either case Landlord, in its reasonable judgment, determines is necessary to provide adequate assurance of the performance by such assignee of the obligations of the tenant under this Lease; provided, however, that in no event shall such adequate assurance of future performance be less favorable to Landlord than the assurance contemplated by Section 365(b)(3) of the Bankruptcy Code (notwithstanding that this Lease may be construed as a lease of real property in a shopping center).

- (D) If Tenant's interest under this Lease is assigned to any Person pursuant to the provisions of the Bankruptcy Code, then any such assignee shall (x) be deemed without further act or deed to have assumed all the obligations of the tenant arising under this Lease from and after the date of such assignment, and (y) execute and deliver to Landlord upon demand an instrument confirming such assumption.
  - (E) Nothing contained in this Article 22 limits Landlord's rights against Tenant under Article 17 hereof.

## 22.2. Replacement Lease.

If (i) Tenant is not the Person that constituted Tenant initially, and (ii) either (I) this Lease is disaffirmed or rejected pursuant to the Bankruptcy Code, or (II) this Lease terminates by reason of occurrence of an Insolvency Event, then, subject to the terms of this Section 22.2, then any Persons that constituted Tenant hereunder previously, including, without limitation, the Person that constituted Tenant initially (any such Person that previously constituted Tenant hereunder (but does not then constitute Tenant hereunder), and with respect to which Landlord exercises Landlord's rights under this Section 22.2, being referred to herein as a "Predecessor Tenant") shall (1) pay to Landlord the aggregate Rental that is then due and owing by Tenant to Landlord under this Lease to and including the date of such disaffirmance, rejection or termination, and (2) enter into a new lease, between Landlord, as landlord, and the Predecessor Tenant, as tenant, for the Premises, and for a term commencing on the effective date of such disaffirmance, rejection or termination and ending on the Fixed Expiration Date (or the last day of the Renewal Term, if such disaffirmance, rejection or termination occurs during the Renewal Term), at the same Fixed Rent and upon the then executory terms that are contained in this Lease, except that (a) such Predecessor Tenant's rights under the new lease shall be subject to the possessory rights of Tenant under this Lease and the possessory rights of any Person claiming by, through or under Tenant or by virtue of any statute or of any order of any court, and (b) such new lease shall require all defaults existing under this Lease to be cured by such Predecessor Tenant with reasonable diligence. Landlord shall have the right to require any such Predecessor Tenant to execute and deliver such new lease on the terms set forth in this Section 22.2 only by giving notice thereof to Tenant and such Predecessor Tenant within thirty (30) days after

Landlord receives notice of any such disaffirmance or rejection (or, if this Lease terminates by reason of Landlord making an election to do so, then Landlord may exercise such right only by giving such notice to Tenant and such Predecessor Tenant within thirty (30) days after this Lease so terminates). If any such Predecessor Tenant defaults in its obligation to enter into said new lease for a period of ten (10) Business Days following Landlord's request therefor, then, in addition to all other rights and remedies by reason of such default, either at law or in equity, Landlord shall have the same rights and remedies against such Predecessor Tenant as if such Predecessor Tenant had entered into such new lease and such new lease had thereafter been terminated as of the commencement date thereof by reason of such Predecessor Tenant's default thereunder.

## 22.3. <u>Insolvency Events.</u>

This Lease shall terminate automatically upon the occurrence of any of the following events:

- (A) a Tenant Obligor commences or institutes any voluntary case, proceeding or other action (a) seeking relief on its behalf as debtor, or to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, or (b) seeking appointment of a receiver, trustee, custodian or other similar official for it or for all or any substantial part of its property; or
  - (B) a Tenant Obligor makes a general assignment for the benefit of creditors; or
- (C) any case, proceeding or other action is commenced or instituted against a Tenant Obligor (a) seeking to have an order for relief entered against it as debtor or to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, or (b) seeking appointment of a receiver, trustee, custodian or other similar official for it or for all or any substantial part of its property, which in either of such cases (i) results in any such entry of an order for relief, adjudication of bankruptcy or insolvency or such an appointment or the issuance or entry of any other order having a similar effect, and (ii) remains undismissed for a period of ninety (90) days; or
- (D) any case, proceeding or other action is commenced or instituted against a Tenant Obligor seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its property which results in the entry of an order for any such relief which is not vacated, discharged, or stayed or bonded pending appeal within ninety (90) days from the entry thereof; or
- (E) a Tenant Obligor takes any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clauses (A), (B), (C), or (D) above; or

(F) a trustee, receiver or other custodian is appointed for any substantial part of a Tenant Obligor's assets, and such appointment is not vacated or stayed within thirty (30) days, or if such event occurs without the acquiescence of Tenant, ninety (90) days (the events described in this Section 22.3 being collectively referred to herein as "Insolvency Events").

The term "Tenant Obligor" shall mean (a) Tenant, (b) any Person that comprises Tenant (if Tenant is comprised of more than one (1) Person), (c) any partner in Tenant (if Tenant is a general partnership), (d) any general partner in Tenant (if Tenant is a limited partnership), (e) any Person that has guarantied all or any part of the obligations of Tenant hereunder (if such Person is an Affiliate of Tenant), and (f) any Person that preceded Tenant as the tenant hereunder and continues to be jointly and severally liable for the performance of Tenant's obligations under this Lease (if such Person is an Affiliate of Tenant). If this Lease terminates pursuant to this Section 22.3, then (I) Tenant immediately shall quit and surrender the Premises, and (II) Tenant shall nonetheless remain liable for all of its obligations hereunder, as provided in Article 23 hereof and Article 24 hereof.

### 22.4. Effect of Stay.

Notwithstanding anything to the contrary contained herein, if (i) Landlord's right to terminate this Lease after the occurrence of an Event of Default, or the termination of this Lease upon the occurrence of an Insolvency Event, is stayed by order of any court having jurisdiction over an Insolvency Event, or by federal or state statute, (ii) the trustee appointed in connection with an Insolvency Event, or Tenant or Tenant as debtor-in-possession, fails to assume Tenant's obligations under this Lease on or prior to the earliest to occur of (a) the last day of the period prescribed therefor by law, (b) the one hundred twentieth (120th) day after entry of the order for relief, or (c) a date that is otherwise designated by the court, or (iii) said trustee, Tenant or Tenant as debtor-in-possession fails to provide adequate protection of Landlord's right, title and interest in and to the Premises or adequate assurance of the complete and continuous future performance of Tenant's obligations under this Lease as provided in Section 22.1(C) hereof, then Landlord, following the expiration of any such stay (if applicable), to the extent permitted by law or by leave of the court having jurisdiction over such proceeding, shall have the right, at its election, to terminate this Lease on ten (10) Business Days of advance notice to Tenant, Tenant as debtor-in-possession or said trustee, and, upon the expiration of said period of ten (10) Business Days, this Lease shall cease and expire as aforesaid and Tenant, Tenant as debtor-in-possession or said trustee shall immediately quit and surrender the Premises as aforesaid.

## 22.5. Rental for Bankruptcy Purposes.

Notwithstanding anything contained in this Lease to the contrary, all amounts payable by Tenant to or on behalf of Landlord under this Lease, regardless of whether such amounts are expressly denominated as Rental, shall constitute rent for the purposes of Section 502(b)(6) of the Bankruptcy Code, and Tenant's payment obligations with respect thereto shall constitute obligations to be timely performed pursuant to Section 365(d) of the Bankruptcy Code.

### Article 23\_ REMEDIES AND DAMAGES

### 23.1. <u>Certain Remedies.</u>

- (A) If (x) an Event of Default occurs and this Lease and the Term expires and comes to an end as provided in Article 21 hereof, or (y) this Lease terminates as provided in Section 22.3 hereof, then:
- (1) Tenant shall immediately quit and peacefully surrender the Premises to Landlord, and Landlord and its agents may, without prejudice to any other remedy which Landlord may have, (a) re-enter the Premises or any part thereof, without notice except as otherwise provided pursuant to applicable Requirements, either by summary proceedings, or by any other applicable action or proceeding, or otherwise as permitted pursuant to applicable Requirements (but in no event by forcible entry) without being liable to indictment, prosecution or damages therefor (except to the extent resulting from Landlord's gross negligence or willful misconduct), (b) repossess the Premises and dispossess Tenant and any other Persons from the Premises, and (c) remove any and all of their property and effects from the Premises; and
- (2) Landlord, at Landlord's option, may relet the whole or any portion or portions of the Premises from time to time, either in the name of Landlord or otherwise, to such tenant or tenants, for such term or terms ending before, on or after the Fixed Expiration Date, at such rental or rentals and upon such other conditions, which may include concessions and free rent periods, as Landlord, in its sole discretion, may determine.
- (B) Landlord shall have no obligation to relet the Premises or any part thereof and shall not be liable for refusal or failure to relet the Premises or any part thereof, or, in the event of any such reletting, for refusal or failure to collect any rent due upon any such reletting. Any such refusal or failure on Landlord's part shall not relieve Tenant of any liability under this Lease or otherwise affect any such liability. Landlord, at Landlord's option, may make such repairs, replacements, alterations, additions, improvements, decorations and other physical changes in and to the Premises as Landlord, in its sole discretion, considers advisable or necessary in connection with any such reletting or proposed reletting, without relieving Tenant of any liability under this Lease or otherwise affecting any such liability.
- (C) In the event of a breach or threatened breach by Tenant, or any Persons claiming by, through or under Tenant, of any term, covenant or condition of this Lease, Landlord shall have the right to (1) enjoin or restrain such breach, (2) invoke any other remedy allowed by law or in equity as if re-entry, summary proceedings and other special remedies were not provided in this Lease for such breach, and (3) seek any declaratory, injunctive or other equitable relief, and specifically enforce this Lease. The right to invoke the remedies hereinbefore set forth are cumulative and nonexclusive and shall not preclude Landlord from invoking any other remedy allowed at law or in equity.

### 23.2. No Redemption.

Tenant, on its own behalf and on behalf of all Persons claiming by, through or under Tenant, including all creditors, does hereby waive any and all rights which Tenant and all such

Persons might have under any present or future law to redeem the Premises, or to re-enter or repossess the Premises, or to restore the operation of this Lease, after (a) Tenant has been dispossessed by a judgment or by warrant of any court or judge or other process of law, or (b) any re-entry by Landlord, or (c) any expiration or termination of this Lease and the Term, whether such dispossess, re-entry, expiration or termination is by operation of law or pursuant to the provisions of this Lease. The words "re-enter," "re-entry" and "re-entered" as used in this Lease shall not be deemed to be restricted to their technical legal meanings.

### 23.3. <u>Calculation of Damages.</u>

- (A) If this Lease terminates by reason of the occurrence of an Event of Default or by reason of the occurrence of an Insolvency Event, then Tenant shall pay to Landlord, on demand, and Landlord shall be entitled to recover as damages:
- (1) all Rental payable under this Lease by Tenant to Landlord (x) to the date that this Lease terminates, or (y) to the date of re-entry upon the Premises by Landlord, as the case may be;
- (2) unless Landlord elects to require Tenant to pay damages calculated as set forth in clause (3) below, the excess of (a) the Rental for the period which otherwise would have constituted the unexpired portion of the Term, over (b) the net amount, if any, of rents collected under any reletting effected pursuant to the provisions of clause (2) of Section 23.1(A) hereof for any part of such period (such excess being referred to herein as a "Deficiency"), as damages (it being understood that (x) such net amount described in clause (b) above shall be calculated by deducting from the rents collected under any such reletting all of Landlord's reasonable and actual expenses in connection with the termination of this Lease, Landlord's re-entry upon the Premises and such reletting, including, but not limited to, all reasonable and actual repossession costs, customary brokerage commissions, reasonable and actual legal expenses, attorneys' fees and disbursements, customary alteration costs, customary contributions to work and other customary and reasonable expenses of preparing the Premises for such reletting, (y) any such Deficiency shall be paid in monthly installments by Tenant on the days specified in this Lease for payment of installments of such Fixed Rent or Escalation Rent, as the case may be, and (z) Landlord shall be entitled to recover from Tenant each monthly Deficiency as it arises, and no suit to collect the amount of the Deficiency for any month shall prejudice Landlord's right to collect the Deficiency for any subsequent month by a similar proceeding); and
- (3) regardless of whether Landlord has theretofore collected any monthly Deficiency as aforesaid, and in lieu of any further Deficiency, as and for liquidated and agreed final damages, an amount equal to the excess, if any, of (a) the Rental for the period which otherwise would have constituted the unexpired portion of the Term (commencing on the date immediately succeeding the last date with respect to which a Deficiency, if any, was collected), over (b) the then fair and reasonable rental value of the Premises for the same period, both discounted to present value at the Base Rate. If, before presentation of proof of such liquidated damages to any court, commission or tribunal, the Premises, or any part thereof, have been relet by Landlord for the period which otherwise would have constituted the unexpired portion of the Term, or any part thereof, then the amount of rent reserved upon such reletting shall be deemed, prima facie, to be the fair and reasonable rental value of the Premises (or the applicable part

thereof) so relet during the term of the reletting, provided that such reletting shall constitute a bona-fide arms-length third party transaction.

- (B) If the Premises, or any part thereof, are relet together with other space in the Building, then the rents collected or reserved under any such reletting and the expenses of any such reletting shall be equitably apportioned on a square foot basis for the purposes of this Section 23.3. Tenant acknowledges and agrees that in no event shall it be entitled to any rents collected or payable under any reletting, regardless of whether such rents exceed the Rental reserved in this Lease.
- (C) Nothing contained in this Article 23 shall be deemed to limit or preclude the recovery by Landlord from Tenant of the maximum amount allowed to be obtained as damages by any applicable statute or rule of law in effect at the time when, and governing the proceedings in which, such damages are to be proved, or of any sums or damages to which Landlord may lawfully be entitled in addition to the damages set forth in this Section 23.3.
  - (D) Nothing contained in this Article 23 shall be deemed to modify or limit the provisions of Section 32.17 hereof.

## Article 24 LANDLORD'S EXPENSES AND LATE CHARGES

### 24.1. Landlord's Costs After Event of Default.

Tenant shall pay to Landlord an amount equal to the reasonable out-of-pocket costs that Landlord actually pays to an unrelated third party in instituting or prosecuting any legal proceeding against Tenant (or any other Person claiming by, through or under Tenant) after the occurrence of an Event of Default, together with interest thereon calculated at the Applicable Rate from the date that Landlord incurs such costs, within thirty (30) days after Landlord gives to Tenant an invoice therefore, accompanied by reasonable supporting documentation (it being understood that Landlord shall have the right to collect such amount from Tenant as additional rent to the extent that Landlord incurs such costs during the Term and as damages to the extent that Landlord incurs such costs after the Expiration Date).

## 24.2. <u>Interest on Late Payments.</u>

If Tenant fails to pay any item of Rental on or prior to the fifth (5<sup>th</sup>) day after the date that such payment is due, then Tenant shall pay to Landlord, in addition to such item of Rental, as a late charge and as additional rent, an amount equal to the interest at the Applicable Rate on the amount unpaid, computed from the date such payment was due to and including the date of payment, except that, not more than one time in any twelve (12) month period occurring during the Term, such interest shall not accrue until Landlord shall have given Tenant notice of its failure to pay the item of Rental and such failure shall have continued for five (5) days after the date that such notice is given to Tenant. Nothing contained in this Section 24.2 limits Landlord's rights and remedies, by operation of law or otherwise, after the occurrence of an Event of Default.

### Article 25 END OF TERM

### 25.1. End of Term.

On the Expiration Date, Tenant shall quit and surrender to Landlord the Premises, vacant, broom-clean, in good order and condition, ordinary wear and tear and damage for which Tenant is not responsible under the terms of this Lease excepted, and otherwise in compliance with the provisions hereof. Tenant expressly waives, for itself and for any Person claiming by, through or under Tenant, any rights which Tenant or any such Person may have under the provisions of Section 2201 of the New York Civil Practice Law and Rules and of any successor law of like import then in force in connection with any holdover summary proceedings that Landlord institutes to enforce the provisions of this Article 25.

### 25.2. Holdover.

If vacant and exclusive possession of the Premises is not surrendered to Landlord on the Expiration Date, then Tenant shall pay to Landlord on account of use and occupancy of the Premises, for each month (or any portion thereof) during which Tenant (or a Person claiming by, through or under Tenant) holds over in the Premises after the Expiration Date, an amount equal to the greater of (I) the product obtained by multiplying the aggregate Rental that was payable under this Lease during the last full month of the Term, by (x) one hundred twenty-five percent (125%) for the first forty-five (45) days of such holdover, and (y) one hundred fifty percent (150%) for any days of such holdover thereafter, and (II) the then fair market rental value of the Premises. Landlord's right to collect such amount from Tenant for use and occupancy shall be in addition to any other rights or remedies that Landlord may have hereunder or at law or in equity (including, without limitation, Landlord's right to recover Landlord's damages from Tenant that derive from vacant and exclusive possession of the Premises not being surrendered to Landlord on the Expiration Date), but only if Tenant holds over in the Premises more than one hundred twenty (120) days after the Expiration Date and subject to the provisions of 32.17 hereof. Nothing contained in this Section 25.2 shall permit Tenant to retain possession of the Premises after the Expiration Date or limit in any manner Landlord's right to regain possession of the Premises, through summary proceedings or otherwise. Landlord's acceptance of any payments from Tenant after the Expiration Date shall be deemed to be on account of the amount to be paid by Tenant in accordance with the provisions of this Article 25.

### Article 26 NO WAIVER

### 26.1. No Surrender.

- (A) Landlord shall be deemed to have accepted a surrender of the Premises only if Landlord executes and delivers to Tenant a written instrument providing expressly therefor.
- (B) No employee of Landlord or of Landlord's agents shall have any power to accept the keys to the Premises prior to the Expiration Date. The delivery of such keys to any

employee of Landlord or of Landlord's agents shall not operate as a termination of this Lease or a surrender of the Premises. If Tenant at any time desires to have Landlord sublet the Premises on Tenant's account, then Landlord or Landlord's agents are authorized to receive said keys for such purpose without releasing Tenant from any of Tenant's obligations under this Lease.

### 26.2. No Waiver by Landlord.

- (1) Landlord's failure to seek redress for violation of, or to insist upon the strict performance of, any covenant or condition of this Lease, or any of the Rules, shall not be deemed to be a waiver thereof. The receipt by Landlord of Rental with knowledge of the breach of any covenant of this Lease by Tenant shall not be deemed a waiver of such breach.
- (2) No payment by Tenant or receipt by Landlord of a lesser amount than the monthly Fixed Rent or other item of Rental herein stipulated shall be deemed to be other than on account of the earliest stipulated Fixed Rent or other item of Rental, or as Landlord may elect to apply such payment. No endorsement or statement on any check or any letter accompanying any check or payment as Fixed Rent or other item of Rental shall be deemed to be an accord and satisfaction. Landlord may accept such check or payment without prejudice to Landlord's right to recover the balance of such Fixed Rent or other item of Rental or to pursue any other remedy provided in this Lease or otherwise available to Landlord at law or in equity.
- (B) Landlord's failure during the Term to prepare and deliver any invoices, and Landlord's failure to make a demand for payment under any of the provisions of this Lease, shall not in any way be deemed to be a waiver of, or cause Landlord to forfeit or surrender, its rights to collect any item of Rental which may have become due during the Term (except to the extent otherwise expressly set forth herein). Tenant's liability for such amounts shall survive the expiration or earlier termination of this Lease, except to the extent otherwise expressly set forth herein.
  - (C) No provision of this Lease shall be deemed to have been waived by Landlord, unless such waiver is in writing signed by Landlord.

## 26.3. No Waiver by Tenant.

Tenant's failure to seek redress for violation of, or to insist upon the strict performance of, any covenant or condition of this Lease on Landlord's part to be performed, shall not be deemed to be a waiver. The payment by Tenant of any item of Rental or performance of any obligation of Tenant hereunder with knowledge of any breach by Landlord of any covenant of this Lease shall not be deemed a waiver of such breach, nor shall it prejudice Tenant's right to pursue any remedy against Landlord in this Lease provided or otherwise available to Tenant in law or in equity. Tenant's failure during the Term to make a demand for any payment to be made to it by Landlord under any of the provisions of this Lease shall not in any way be deemed to be a waiver of, or cause Tenant to forfeit or surrender, its rights to collect any payment which may have become due to it during the Term (except to the extent otherwise expressly set forth herein). Landlord's liability for such amounts shall survive the expiration or earlier termination of this Lease. No provision of this Lease shall be deemed to have been waived by Tenant, unless such waiver is in writing signed by Tenant.

# Article 27 JURISDICTION

### 27.1. Governing Law.

This Lease shall be construed and enforced in accordance with the laws of the State of New York.

### 27.2. Submission to Jurisdiction.

Tenant hereby (a) irrevocably consents and submits to the jurisdiction of any federal, state, county or municipal court sitting in the State of New York for purposes of any action or proceeding brought therein by Landlord against Tenant concerning any matters relating to this Lease, (b) irrevocably waives all objections as to venue and any and all rights it may have to seek a change of venue with respect to any such action or proceedings, (c) agrees that the laws of the State of New York shall govern in any such action or proceeding and waives any defense to any action or proceeding granted by the laws of any other country or jurisdiction unless such defense is also allowed by the laws of the State of New York, and (d) agrees that any final unappealable judgment rendered against it in any such action or proceeding shall be conclusive and may be enforced in any other jurisdiction by suit on the judgment or in any other manner provided by law. Tenant further agrees that any action or proceeding by Tenant against Landlord concerning any matters arising out of or in any way relating to this Lease shall be brought only in the State of New York, County of New York.

## 27.3. Waiver of Trial by Jury.

Landlord and Tenant hereby waive trial by jury in any action, proceeding or counterclaim brought by either of the parties hereto against the other on any matters whatsoever arising out of or in any way connected with this Lease, the relationship of Landlord and Tenant, Tenant's use or occupancy of the Premises, or for the enforcement of any remedy under any statute, emergency or otherwise. If Landlord commences any summary proceeding against Tenant, then Tenant shall not interpose any counterclaim of whatever nature or description in any such proceeding (unless such counterclaim is mandatory or compulsory), and shall not seek to consolidate such proceeding with any other action which may have been or will be brought in any other court by Tenant, provided that nothing herein shall be deemed to preclude Tenant from bringing a separate action for any claim that Tenant may have hereunder.

Article 28 NOTICES

### 28.1. <u>Addresses; Manner of Delivery.</u>

Except as otherwise expressly provided in this Lease, any bills, statements, consents, notices, demands, requests or other communications given or required to be given under this Lease shall (1) be in writing, (2) be deemed sufficiently given if (a) delivered by hand (against a signed receipt), (b) sent by registered or certified mail (return receipt requested) posted in a United States post office station or letter box in the continental United States, or (c) sent by a

nationally-recognized overnight courier with verification of delivery, and (3) be addressed in each case:

if to Tenant, at:

333 West 34th Street

New York, New York 10001

Attn.: Leslie Heifetz

with a copy to:

Paul Hastings Janofsky & Walker LLP 75 East 55th Street New York, New York 10022

Attn.: Dean Stiffle, Esq.

if to Landlord, at:

c/o Alexander's Inc. 888 Seventh Avenue

New York, New York 10019

Attn.: David R. Greenbaum

with a copy to:

c/o Vornado Realty Trust 210 Route 4 East

Paramus, New Jersey 07652

Attn: Joseph Macnow

and with a copy to:

Proskauer Rose LLP 1585 Broadway

New York, New York 10036

Attn.: Lawrence J. Lipson, Esq.

or to such other address or addresses as Landlord or Tenant may designate from time to time on at least ten (10) Business Days of advance notice given to the other in accordance with the provisions of this Article 28. Any such bill, statement, consent, notice, demand, request, or other communication shall be deemed to have been given (x) on the date that it is hand delivered, as aforesaid, or on the date that such delivery is refused, if applicable, or (y) three (3) Business Days after the date that it is mailed, as aforesaid, or (z) on the first (1st) Business Day after the date that it is sent by a nationally-recognized courier, as aforesaid. Any such bills, statements,

consents, notices, demands, requests or other communications that the Person that is the property manager for the Building gives to Tenant in accordance with the terms of this Article 28 shall be deemed to have been given by Landlord (except that Landlord, at any time and from time to time, shall have the right to terminate or suspend such property manager's right to give such bills, statements, consents, notices, demands, requests or other communications to Tenant by giving not less than five (5) days of advance notice thereof to Tenant). Any such bills, statements, consents, notices, demands, requests or other communications that Tenant's attorney or Landlord's attorney, as the case may be, gives to the other party's attorney in accordance with the terms of this Article 28 shall be deemed to have been given by Tenant or Landlord, as the case may be. All notices given by Landlord pursuant to Section 21.1 and Section 21.2 hereof shall contain a statement in bold type and capital letters stating "THIS IS A DEFAULT NOTICE" as a condition to the effectiveness thereof.

## Article 29 BROKERAGE

### 29.1. <u>Broker.</u>

Landlord and Tenant each represent to the other that it has not dealt with any broker, finder or salesperson in connection with this Lease other than Cushman & Wakefield, Inc. and Newmark and Company Real Estate, Inc. (collectively, the "Brokers"). Landlord shall pay the Brokers any commission or other compensation to which the Brokers may be entitled in connection with this Lease pursuant to separate agreements between Landlord and the Brokers.

## Article 30 INDEMNITY

## 30.1. <u>Tenant's Indemnification of the Landlord Indemnitees.</u>

- (A) Subject to the terms of this Section 30.1, Tenant shall indemnify the Landlord Indemnitees, and hold the Landlord Indemnitees harmless, from and against, all losses, damages, liabilities, costs and expenses (including, without limitation, reasonable attorneys' fees and expenses) that are incurred by a Landlord Indemnitee and that derive from a claim (a "Tenant Liability Claim") made by a third party against such Landlord Indemnitee arising from or alleged to arise from:
- (1) a wrongful act or wrongful omission of any Tenant Indemnitee during the Term (it being understood that Tenant shall not have responsibility under this clause (1) for any wrongful act or wrongful omission of a Recapture Subtenant);
- (2) an event or circumstance that occurs during the Term in the Premises or in another portion of the Building with respect to which Tenant has exclusive use pursuant to the terms hereof (subject, however, to Landlord's rights of access under Article 9 hereof) (it being understood that Tenant's liability under this clause (2) shall not apply to the extent that Landlord exercises Landlord's rights under Section 17.3 hereof with respect to the Recapture Space);

- (3) the breach of any covenant to be performed by Tenant hereunder;
- (4) a misrepresentation made by Tenant hereunder (including, without limitation, a misrepresentation of Tenant under Section 29.1 hereof);
- (5) a Compliance Challenge (or Tenant's delaying Tenant's compliance with a Requirement during the pendency of a Compliance Challenge);
  - (6) Landlord's cooperating with Tenant as contemplated by Section 7.4(A) hereof;
- (7) a claim by a broker, finder or salesperson in connection with a Permitted Party's making, or proposing to make, a Transfer in accordance with the terms hereof; or
  - (8) Tenant's not complying with a Requirement pursuant to Section 11.4 hereof.

Tenant shall not be required to indemnify the Landlord Indemnitees, and hold the Landlord Indemnitees harmless, in either case as aforesaid, to the extent that it is finally determined that the act or omission of a Landlord Indemnitee contributed to the loss or damage sustained by the Person making the Tenant Liability Claim. Nothing contained in this Section 30.1 limits the provisions of Section 32.17 hereof.

- (B) The term "<u>Landlord Indemnitees</u>" shall mean, collectively, Landlord, each Lessor, each Mortgagee, the Condominium Board, any subcommittee of the Condominium Board and their respective partners, members, managers, shareholders, officers, directors, employees, trustees and agents.
- (C) The term "Tenant Indemnitees" shall mean, collectively, Tenant and each Permitted Party and their respective partners, members, managers, shareholders, officers, directors, employees, trustees and agents.
  - (D) The parties intend that the Landlord Indemnitees (other than Landlord) shall be third-party beneficiaries of this Section 30.1.
- (E) Nothing contained in this Section 30.1 shall be deemed to relieve Landlord or the Condominium Board, as the case may be, of the obligation to maintain insurance in respect of the Building in accordance with the terms of Article 14 hereof.

## 30.2. <u>Landlord's Indemnification of the Tenant Indemnitees.</u>

(A) Subject to the terms of this Section 30.2, Landlord shall indemnify the Tenant Indemnitees, and hold the Tenant Indemnitees harmless, from and against, all losses, damages, liabilities, costs and expenses (including, without limitation, reasonable attorneys' fees and expenses) that are incurred by a Tenant Indemnitee and that derive from a claim (a "Landlord Liability Claim") made by a third party against such Tenant Indemnitee arising from or alleged to arise from:

- (1) the breach of any covenant to be performed by Landlord hereunder;
- (2) a misrepresentation made by Landlord hereunder (including, without limitation, a misrepresentation of Landlord under Section 29.1 hereof);
  - (3) Landlord's failure to pay the Brokers a commission or other compensation in connection herewith;
- (4) a wrongful act or wrongful omission of any Landlord Indemnitee (including, without limitation, a wrongful act or wrongful omission of the Person that has the right to occupy the Recapture Space by virtue of Landlord's exercising Landlord's rights under Section 17.3 hereof); or
  - (5) any adjacent excavation performed by Landlord or its employees or contractors pursuant to Section 32.12 hereof.

Landlord shall not be required to indemnify the Tenant Indemnitees, and hold the Tenant Indemnitees harmless, in either case as aforesaid, to the extent that it is finally determined that the act or omission of a Tenant Indemnitee contributed to the loss or damage sustained by the Person making the Landlord Liability Claim.

(B) The parties intend that the Tenant Indemnitees (other than Tenant) shall constitute third-party beneficiaries of this Section 30.2.

## 30.3. <u>Indemnification Procedure.</u>

- (A) If at any time a Landlord Liability Claim is made or threatened against a Tenant Indemnitee, or a Tenant Liability Claim is made or threatened against a Landlord Indemnitee, then the Person entitled to indemnity under this Article 30 (the "Indemnitee") shall give to the other party (the "Indemnitor") notice of such Landlord Liability Claim or such Tenant Liability Claim, as the case may be (the "Claim"); provided, however, that the Indemnitee's failure to provide such notice shall not impair the Indemnitee's rights to indemnity as provided in this Article 30 except to the extent that the Indemnitor is prejudiced materially thereby. Such notice shall state the basis for the Claim and the amount thereof (to the extent such amount is determinable at the time that such notice is given).
- (B) The Indemnitor shall have the right to defend against the Claim using attorneys that the Indemnitor reasonably selects and that are approved by the Indemnitee, which approval shall not be unreasonably withheld, conditioned or delayed. If the Indemnitee does not respond to Indemnitor's written request for the Indemnitee's approval to the Indemnitor's proposed attorneys on or prior to ten (10) Business Days after written notice from the Indemnitor to the Indemnitee requesting such approval, then such proposed attorneys shall be deemed approved by the Indemnitee (it being understood that the attorneys designated by the Indemnitor's insurer shall be deemed approved by the Indemnitee for purposes hereof). The Indemnitor's failure to notify the Indemnitee of the Indemnitor's election to defend against the Claim within thirty (30) days after the Indemnitee gives such notice to the Indemnitor shall be deemed a waiver by the Indemnitor of its aforesaid right to defend against the Claim.

- (C) Subject to the terms of this Section 30.3(C), if the Indemnitor elects to defend against the Claim pursuant to Section 30.3(B) hereof, then the Indemnitee may participate, at the Indemnitee's expense, in defending against the Claim. The Indemnitor shall have the right to control the defense against the Claim (and, accordingly, the Indemnitee shall cause its counsel to act accordingly). If there exists a conflict between the interests of the Indemnitor and the interests of the Indemnitee, then the Indemnitor shall pay the reasonable out-of-pocket fees and disbursements of any counsel that the Indemnitee retains in so participating in the defense against the Claim. Except as set forth in the immediately preceding sentence, if the Indemnitor elects to defined against the Claim pursuant to Section 30.3(B) hereof, and provided that the Indemnitor complies with the requirements of this Section 30.3 with respect to any Claim, the Indemnitor shall not be liable for the costs of any separate counsel employed by the Indemnitee with respect thereto.
- (D) If the Claim is a Tenant Liability Claim, then Landlord shall cooperate reasonably with Tenant in connection therewith. If the Claim is a Landlord Liability Claim, then Tenant shall cooperate reasonably with Landlord in connection therewith.
- Subject to the terms of this clause (E), the Indemnitor shall not consent to the entry of any judgment or award regarding the Claim, or enter into any settlement regarding the Claim, except in either case with the prior approval of the Indemnitee (any such entry of any judgment or award regarding a Claim to which the Indemnitor consents, or any such settlement regarding a claim to which the Indemnitor agrees, being referred to herein as a "Settlement"). The Indemnitee shall not unreasonably withhold, condition or delay the Indemnitee's approval of a proposed Settlement, provided that the Indemnitor pays, in cash, to the Person making the Claim, the entire amount of the Settlement contemporaneously with the Indemnitee's approval thereof (so that neither the Indemnitor nor the Indemnitee have any material obligations regarding the applicable Claim that remain executory from and after the consummation of the Settlement). The Indemnitee's consent shall not be required with respect to a Settlement entered into by the Indemnitor pursuant to which the Indemnitor does not pay the entire amount of the Settlement in case to the Person making the claim, as long as such Person releases the Indemnitee from all liability pursuant to such Claim so that the Indemnitee has no obligations whatsoever with respect to the Claim that remain executory from and after the consummation of the Settlement. If (x) the terms of the Settlement do not provide for the Indemnitor's making payment, in cash, to the Person making the Claim, the entire amount of the Settlement contemporaneously with the Indemnitee's approval thereof and the Person making the Claim does not release Indemnitee as aforesaid, and (v) the Indemnitee does not approve the proposed Settlement, then the Indemnitor's aggregate liability under this Article 30 for the Claim (including, without limitation, the costs incurred by the Indemnitor for legal costs and other costs of defense) shall not exceed an amount equal to the sum of (i) the aggregate legal costs and defense costs that the Indemnitor incurred to the date that the Indemnitor proposes such Settlement, (ii) the amount that the Indemnitor would have otherwise paid to the Person making the applicable Claim under the terms of the proposed Settlement, and (iii) the aggregate legal costs and defense costs that the Indemnitor would have reasonably expected to incur in consummating the proposed Settlement.
- (F) If the Indemnitor does not elect to defend against the Claim as contemplated by this Section 30.3, then the Indemnitee may defend against, or settle, such claim,

action or proceeding in any manner that the Indemnitee deems appropriate, and the Indemnitor shall be liable for the Claim to the extent provided in this Article 30.

# Article 31 LANDLORD'S CONSENTS; ARBITRATION

### 31.1. <u>Certain Limitations.</u>

(A) Subject to the terms of Section 31.2 hereof, Tenant hereby waives any claim against Landlord for Landlord's unreasonably withholding, unreasonably conditioning or unreasonably delaying any consent or approval requested by Tenant in cases where Landlord expressly agreed herein not to unreasonably withhold, unreasonably condition or unreasonably delay such consent or approval. If there is a determination that such consent or approval has been unreasonably withheld, unreasonably conditioned or unreasonably delayed, then (1) the requested consent or approval shall be deemed to have been granted, and (2) Landlord shall have no liability to Tenant for its refusal or failure to give such consent or approval, except to the extent set forth in Section 31.2(C) hereof. Tenant's sole remedy for Landlord's unreasonably withholding, conditioning or delaying consent or approval shall be as provided in this Article 31.

## 31.2. <u>Expedited Arbitration; Tenant's Remedies.</u>

- (A) If (I) (i) this Lease obligates Landlord to not unreasonably withhold, condition or delay Landlord's consent or approval for a particular matter or requires Landlord to exercise Landlord's reasonable judgment with respect to a certain matter, (ii) Landlord withholds, delays or conditions its consent or approval for such matter or makes a determination with respect to a certain matter, and (iii) Tenant believes that Landlord did so unreasonably, or (II) this Lease expressly states that a given matter be subjected to an Expedited Arbitration Proceeding, then, in the case of clause (I) above, Tenant shall have the right to submit the issue of whether Landlord unreasonably withheld, delayed or conditioned such consent or approval or whether Landlord exercised its judgment unreasonably in making such determination, as the case may be, to an Expedited Arbitration Proceeding only by giving notice thereof to Landlord on or prior to the thirtieth (30th) day after the date that Landlord denied or conditioned such consent or approval or made such determination, as the case may be, or the thirtieth (30th) day after the date that Tenant claims that Landlord's delaying such consent or approval first became unreasonable, as the case may be, and in the case of clause (II) above, either party shall have the right to submit the applicable dispute to an Expedited Arbitration Proceeding only by giving notice thereof to the other party.
- (B) With respect to clause (I) of Section 31.2(A) hereof, the sole decision to be made in the Expedited Arbitration Proceeding shall be whether Landlord unreasonably withheld, delayed or conditioned its consent or whether Landlord exercised Landlord's judgment unreasonably in making its determination, as the case may be, with respect to the particular matter being arbitrated. If the decision in the Expedited Arbitration Proceeding is that Landlord unreasonably withheld, conditioned, or delayed consent or that Landlord exercised its judgment unreasonably in making such determination, as the case may be, with respect to such matter, then (i) Landlord shall be deemed to have consented to such matter or to have determined such matter

in a reasonable manner in accordance with the arbitrator's determination, as the case may be, and (ii) Landlord shall execute and deliver documentation that is reasonably requested by Tenant to evidence such consent or such determination (provided that Landlord's failure to execute and deliver any such documentation shall not be deemed to modify the provisions of clause (i) of this sentence).

- (C) With respect to clause (I) of Section 31.2(A) hereof, Tenant shall have the right to institute legal proceedings in a court of competent jurisdiction against Landlord to recover the actual damages that Tenant sustains by reason of Landlord's unreasonably withholding, conditioning or delaying Landlord's consent or approval or exercising its judgment unreasonably in making a determination with respect to a particular matter (in cases where Landlord agreed not to, or applicable Requirements obligated Landlord not to, unreasonably withhold, condition or delay such consent or approval or where Landlord agreed to, or applicable Requirements obligated Landlord to, exercise its judgment reasonably, as the case may be). Tenant shall not have the right to make any such recovery unless such court makes a final determination that Landlord unreasonably withheld, unreasonably delayed or unreasonably conditioned such consent or approval or exercised its judgment unreasonably, as the case may be, capriciously and arbitrarily or that in so withholding, delaying or conditioning such consent or approval or in so exercising its judgment, as the case may be, Landlord acted maliciously or in bad faith.
- (D) The term "Expedited Arbitration Proceeding" shall mean a binding arbitration proceeding conducted in The City of New York under the Commercial Arbitration Rules of the American Arbitration Association (or its successor) and administered pursuant to the Expedited Procedures provisions thereof; provided, however, that with respect to any such arbitration, (i) the list of arbitrators referred to in Section E-5(b) shall be returned within five (5) Business Days from the date of mailing; (ii) the parties shall notify the American Arbitration Association (or its successor) by telephone, within four (4) Business Days, of any objections to the arbitrator appointed and, subject to clause (vii) below, shall have no right to object if the arbitrator so appointed was on the list submitted by the American Arbitration Association (or its successor) and was not objected to in accordance with Section E-5(b) as modified by clause (i) above; (iii) the notification of the hearing referred to in Section E-8 shall be four (4) Business Days in advance of the hearing; (iv) the hearing shall be held within seven (7) Business Days after the appointment of the arbitrator; (v) the arbitrator shall have no right to award damages or vary, modify or waive any provision of this Lease; (vi) the decision of the arbitrator shall be final and binding on the parties; and (vii) the arbitrator shall not have been employed by either party (or their respective Affiliates) during the period of three (3) years prior to the date of the Expedited Arbitration Proceeding. The arbitrator shall determine the extent to which each party is successful in such Expedited Arbitration Proceeding in addition to rendering a decision on the dispute submitted. If the arbitrator determines that one (1) party is entirely unsuccessful, then such party shall pay all of the fees of such arbitrator. If the arbitrator determines that both parties are partially successful and Tenant is twenty percent (20%) successful, then Landlord shall be responsible for twenty percent (

### Article 32 ADDITIONAL PROVISIONS

### 32.1. <u>Tenant's Property Delivered to Building Employees.</u>

Any Building employee to whom any property is entrusted by or on behalf of Tenant shall be deemed to be acting as Tenant's agent with respect to such property, provided, however, that subject to the provisions of Section 32.17 hereof, nothing contained in this Section 32.1 shall be construed to exculpate Landlord for loss, injury or damage to any such property to the extent caused by or resulting from the negligence of Landlord, its agents, servants, employees and contractors in the operation or maintenance of the Premises or the Real Property.

### 32.2. <u>Not Binding Until Execution.</u>

This Lease shall not be binding upon Landlord or Tenant unless and until Landlord and Tenant have executed and unconditionally delivered a fully executed original counterpart of this Lease to each other.

### 32.3. No Third Party Beneficiaries.

Landlord and Tenant hereby acknowledge that they do not intend for any other Person to constitute a third-party beneficiary hereof, except to the extent otherwise set forth herein.

### 32.4. Extent of Landlord's Liability.

- (A) The obligations of Landlord under this Lease shall not be binding upon the Person that constitutes Landlord initially after the sale, conveyance, assignment or transfer by such Person of its interest in the Building or the Real Property or the Office Unit (or the Office Unit One with respect to Landlord's obligation to enter into a lease for any Option Space located in Office Unit One pursuant to Section 18.7 hereof), as the case may be (or upon any other Person that constitutes Landlord after the sale, conveyance, assignment or transfer by such Person of its interest in the Building or the Real Property or the Office Unit (or the Office Unit One as aforesaid), as the case may be), to the extent such obligations accrue from and after the date of such sale, conveyance, assignment or transfer; provided, however, that the transferee has assumed and agreed in writing (or is required by a Nondisturbance Agreement between such transferee and Tenant or by operation of law) to perform and observe all obligations of Landlord herein during the period it is the holder of Landlord's interest under this Lease (or the obligations of Landlord with respect to any Option Space located in Office Unit One as aforesaid).
- (B) The members, managers, partners, shareholders, directors, officers and principals, direct and indirect, comprising Landlord shall not be liable for the performance of Landlord's obligations under this Lease. Tenant shall look solely to Landlord to enforce Landlord's obligations hereunder.
- (C) The liability of Landlord for Landlord's obligations under this Lease shall be limited to Landlord's interest in the Real Property and the Office Unit (or the Office Unit One with respect to Landlord's obligation to enter into a lease for any Option Space located in Office Unit One pursuant to Section 18.7 hereof) and the proceeds thereof (including, without limitation, any sales, insurance, condemnation or refinancing proceeds thereof). Tenant shall not look to any property or assets of Landlord (other than Landlord's interest in the Real Property

and the Office Unit and the proceeds thereof as aforesaid (or the Office Unit One with respect to Landlord's obligation to enter into a lease for any Option Space located in Office Unit One pursuant to Section 18.7 hereof)) in seeking either to enforce Landlord's obligations under this Lease or to satisfy a judgment for Landlord's failure to perform such obligations.

### 32.5. Survival.

Tenant's liability for all amounts that are due and payable to Landlord hereunder shall survive the Expiration Date except as otherwise provided in this Lease.

### 32.6. Recording.

Landlord and Tenant shall execute, acknowledge and deliver a memorandum of this Lease that is in reasonable form and that otherwise complies with the provisions of Section 291-C of the Real Property Law of The State of New York, which Landlord shall then submit promptly to the appropriate recorder's office. Tenant shall reimburse Landlord for the reasonable out-of-pocket costs that Landlord incurs in so preparing and recording such memorandum, within thirty (30) days after Landlord submits to Tenant an invoice therefor.

### 32.7. <u>Entire Agreement.</u>

This Lease contains the entire agreement between the parties and supersedes all prior understandings, if any, with respect thereto. This Lease shall not be modified, changed, or supplemented, except by a written instrument executed by both parties.

## 32.8. Exhibits.

If any inconsistency exists between the terms and provisions of this Lease and the terms and provisions of the Exhibits hereto, then the terms and provisions of this Lease shall prevail.

## 32.9. Gender; Plural.

Wherever appropriate in this Lease, personal pronouns shall be deemed to include the other gender and the singular to include the plural.

### 32.10. <u>Divisibility.</u>

If any term of this Lease, or the application thereof to any Person or circumstance, is held to be invalid or unenforceable, then the remainder of this Lease or the application of such term to any other Person or any other circumstance shall not be thereby affected, and each term shall remain valid and enforceable to the fullest extent permitted by law.

### 32.11. Vault Space.

If (i) Tenant uses or occupies any vaults, vault space or other space outside the boundaries of the Real Property that in each case is located below grade, and (ii) such space is diminished by any Governmental Authority or by any utility company, then such diminution shall not constitute an actual or constructive eviction, in whole or in part, or entitle Tenant to any abatement or diminution of Rental, or relieve Tenant from any of its obligations under this Lease, or impose any liability upon Landlord.

### 32.12. Adjacent Excavation.

If an excavation is made upon land adjacent to the Building, or is authorized to be made, then, subject to any applicable provisions of Article 9 hereof, Tenant, upon reasonable advance notice, shall grant to the Person causing or authorized to cause such excavation a license to enter upon the Premises for the purpose of doing such work as said Person deems necessary to preserve the Building from injury or damage and to support the same by proper foundations, without any claim for damages or indemnity against Landlord, or diminution or abatement of Rental, provided that Landlord shall use reasonable efforts to cause the foregoing to be performed in such a manner as to minimize any interference with Tenant's operation of its business in the Premises.

### 32.13. Captions.

The captions are inserted only for convenience and for reference and in no way define, limit or describe the scope of this Lease nor the intent of any provision thereof.

### 32.14. Parties Bound.

The covenants, conditions and agreements contained in this Lease shall bind and inure to the benefit of Landlord and Tenant and their respective legal representatives, successors, and, except as otherwise provided in this Lease, their assigns.

### 32.15. Authority.

- (A) Tenant hereby represents and warrants to Landlord that (i) Tenant is duly organized and validly existing under the laws of United States of America, and possesses all licenses and authorizations necessary to carry on its business, (ii) Tenant has full power and authority to carry on its business, enter into this Lease and consummate the transaction contemplated hereby, (iii) the individual executing and delivering this Lease on Tenant's behalf has been duly authorized to do so, (iv) this Lease has been duly executed and delivered by Tenant, (v) this Lease constitutes a valid, legal, binding and enforceable obligation of Tenant (subject to bankruptcy, insolvency or creditor rights laws generally, and principles of equity generally), (vi) the execution, delivery and performance of this Lease by Tenant will not cause or constitute a default under, or conflict with, the organizational documents of Tenant or any agreement to which Tenant is a party, (vii) the execution, delivery and performance of this Lease by Tenant will not violate any Requirement, (viii) all consents, approvals, authorizations, orders or filings of or with any court or governmental agency or body, if any, required on the part of Tenant for the execution, delivery and performance of this Lease have been obtained or made, and (ix) the Office of Foreign Assets Control of the United States Department of the Treasury has not listed Tenant or any of Tenant's Affiliates, or any Person that Controls, is Controlled by, or is under common Control with Tenant, on its list of Specially Designated Nationals and Blocked Persons.
- (B) Landlord hereby represents and warrants to Tenant that (i) Landlord is duly organized and validly existing in good standing under the laws of Delaware, and possesses all licenses and authorizations necessary to carry on its business, (ii) Landlord has full power and authority to carry on its business, enter into this Lease and consummate the transaction contemplated hereby, (iii) the individual executing and delivering this Lease on Landlord's

behalf has been duly authorized to do so, (iv) this Lease has been duly executed and delivered by Landlord, (v) this Lease constitutes a valid, legal, binding and enforceable obligation of Landlord (subject to bankruptcy, insolvency or creditor rights laws generally, and principles of equity generally), (vi) the execution, delivery and performance of this Lease by Landlord will not cause or constitute a default under, or conflict with, the organizational documents of Landlord or any agreement to which Landlord is a party, (vii) the execution, delivery and performance of this Lease by Landlord does not violate any Requirement, (viii) all consents, approvals, authorizations, orders or filings of or with any court or governmental agency or body, if any, required on the part of Landlord for the execution, delivery and performance of this Lease have been obtained or made, and (ix) the Office of Foreign Assets Control of the United States Department of the Treasury has not listed Landlord or any of Landlord's Affiliates, or any Person that Controls, is Controlled by, or is under common Control with Landlord, on its list of Specially Designated Nationals and Blocked Persons.

### 32.16. Rent Control.

If at the commencement of, or at any time or times during, the Term, the Rental reserved in this Lease is not fully collectible by reason of any Requirement, then Tenant shall enter into such agreements and take such other steps (without additional expense to Tenant) as Landlord may reasonably request and as may be legally permissible to allow Landlord to collect the maximum rents which may from time to time during the continuance of such legal rent restriction be legally permissible (and not in excess of the amounts reserved therefor under this Lease). Upon the termination of such legal rent restriction prior to the expiration of the Term, (a) the Rental shall become and thereafter be payable hereunder in accordance with the amounts reserved in this Lease for the periods following such termination, and (b) Tenant shall pay to Landlord, if legally permissible, an amount equal to the excess of (i) the items of Rental which would have been paid pursuant to this Lease but for such legal rent restriction, over (ii) the rents paid by Tenant to Landlord during the period or periods such legal rent restriction was in effect.

## 32.17. Consequential Damages.

Tenant shall have no liability for any consequential, indirect or punitive damages that are suffered by Landlord or any Person claiming by, through or under Landlord. Landlord shall have no liability for any consequential, indirect or punitive damages that are suffered by Tenant or any Person claiming by, through or under Tenant.

### 32.18. <u>Tenant's Advertising.</u>

Tenant shall not use a picture, photograph or drawing of the Building (or a silhouette thereof) in Tenant's letterhead or promotional materials without Landlord's prior approval.

# Article 33 TENANT'S BENEFIT PROGRAMS

Upon the request of Tenant, Landlord shall join in and promptly execute any applications for any applications or other documentation that Tenant submits to an applicable Governmental Authority in connection with Tenant's seeking to obtain benefits under a particular program run by such Governmental Authority related to its occupancy as a tenant in the Building (provided

that applicable program requires Landlord to join in any such application or documentation) and shall otherwise provide Tenant with reasonable cooperation in connection with any such programs. Tenant shall reimburse Landlord for Landlord's actual, out-of-pocket costs paid to an unrelated thirty party in connection with such cooperation on or prior to thirty (30) days after Landlord's rendition of a statement therefor, which statement shall have annexed thereto documentation that reasonably substantiates the charges set forth thereon.

IN WITNESS WHEREOF, Landlord and Tenant have duly executed and delivered this Lease as of the date first above written.

By:	731 O	ffice One Holding LLC, member
	By:	Alexander's Inc., member
	Bv·	
	<i></i>	Name:
		Title:
CITIB	ANK, N.	A.
Ву:		
	Name:	
	Title	

731 OFFICE ONE LLC

Exhibit "A"

<u>Premises</u>

See Attached

140

# Exhibit "3.3-1"

# Bloomberg Primary Competitors

Bridge Reuters Group PLC Thomson Corp.

### Exhibit "3.3-2"

### Modification of List of Bloomberg Primary Competitors

In no event may Bloomberg replace any Person on the list of Bloomberg Primary Competitors or place any other Person thereon, except that if any Person succeeds to the interest of a Person on the list of Bloomberg Primary Competitors as a result of a merger or consolidation or the sale of all or substantially all of the assets of such Bloomberg Primary Competitor, then the list of Bloomberg Primary Competitors shall be updated to remove such Person therefrom and place the Person surviving such merger or consolidation or such sale thereon if such Person surviving such merger or consolidation or such sale derives, in such Person's most recently ended fiscal year, more than one-half (1/2) of its revenues from a business or businesses in either case in competition with Bloomberg's Core Business. The term "Core Business" shall mean the business of providing to the general public (on a subscription basis or otherwise) financial information such as news, data and analysis of financial markets and businesses via any medium, including, without limitation, internet, dedicated communication network, television, radio and print, and the business of operating an electronic communications network that matches buyers and sellers of securities and that is registered with the Securities and Exchange Commission as an electronic communications network (it being understood that the business of serving as a securities broker for the transfer of securities using an organized securities exchange shall not constitute Bloomberg's Core Business for purposes hereof).

## Exhibit "3.4-1"

### Rules

- 1. Tenant shall not obstruct the common areas of the Building. Tenant shall not use the common areas of the Building for any purpose other than for the purpose that the applicable common area is used ordinarily.
- 2. Tenant shall not use any plumbing fixtures that are connected to Building Systems for any purpose other than the ordinary purpose for which such plumbing fixtures are installed.
- 3. Tenant shall not use the Premises in any manner that materially and unreasonably interferes with the use of any other portion of the Building for ordinary business purposes.
- 4. Tenant shall not at any time keep in the Premises any flammable, combustible or explosive substance, except for any such substances that are incidental to the use of the Premises for ordinary office purposes.
- 5. Tenant shall not bring any bicycles, vehicles or animals of any kind (except for service animals) into the Premises or the Building.
- 6. Tenant shall comply with the security procedures that Landlord reasonably adopts from time to time for the Building. Tenant acknowledges that Landlord's security procedures may include, without limitation, (i) Landlord's denying entry to the Building by any person who does not present a Building pass or who does not comply with Landlord's procedures regarding the registration of visitors to the Building, and (ii) procedures governing the inspection of freight that arrives at the loading facilities for the Building.
- 7. Landlord shall have the right to require Tenant to (x) direct Persons who are delivering packages to the Premises to make delivery to an office in the Building that Landlord designates (in which case Landlord shall make arrangements for such packages to be delivered to Tenant using other personnel that Landlord engages), or (y) arrange for such Persons to be escorted by a representative of Tenant while such Person makes delivery to the Premises.
- 8. Tenant shall subject to inspection by Landlord or Landlord's designee all items being brought into the Building by or on behalf of Tenant (including, without limitation, packages, boxes, bags, handbags, attache cases, and suitcases). Landlord may refuse entry into the Building to any Person who refuses to cooperate with such inspection or who is carrying any item which has a reasonable likelihood of being dangerous to persons or property.
- Tenant, at Tenant's expense, shall operate its interior lights for the employees of Landlord while such employees make repairs in the Premises or perform cleaning services in accordance with the terms of this Lease.

- 10. Tenant shall not canvass or solicit the other occupants of the Building. Tenant shall cooperate reasonably with Landlord in connection with Landlord's efforts to prevent any Person from canvassing, soliciting or peddling in the Building.
- 11. Tenant shall use in the Building only hand trucks and hand carts that in either case are equipped with rubber tires and side guards.

Exhibit "3.4-2"

# Alterations Rules and Regulations

[See Attached]

Exhibit "3.6"

Risers

[See Attached]

Exhibit "3.7-1"

<u>Lobby Sign</u>

[See Attached]

Exhibit "3.7-2"

Exterior Sign

[See Attached]

Exhibit "3.11"

# Telephone/Electric Closets

[See Attached]

Exhibit "3.12"

Access Agreement

[See Attached]

4366/75979-NYLIB1/1797569v18 100 Exhibit "3.13"

## **Lobby Turnstiles Location**

[See Attached]

4366/75979-NYLIB1/1797569v18 100

# Exhibit "3.14"

# Fitness Center Plans and Specifications

[See Attached]

 $\frac{4366/75979\text{-NYLIB1}/1797569v18}{100}$ 

Exhibit "3.15"

Floor Loads

[See Attached]

 $\frac{4366/75979\text{-NYLIB1}/1797569v18}{100}$ 

Exhibit "4.2-1"

## Passenger Elevator Locations

Exhibit "4.2-2"

# Passenger Elevator Specifications

Exhibit "4.3"

Freight Elevator

Exhibit "4.4"

## **HVAC Specifications**

#### Exhibit "4.5"

#### **Cleaning Specifications**

#### NIGHTLY (ON BUSINESS DAYS)

- Sweep hard-surfaced flooring in general office space using a dust-down preparation.
- Carpet sweep carpets in general office areas without moving heavy furniture (such as desks, file cabinets, computer stands, and sofas).
- Hand dust and wipe clean all furniture, fixtures and window sills in the general office areas that are
  within reach of the cleaning staff without ladders.
- Empty and clean ashtrays and screen all sand urns.
- Empty and clean waste receptacles in the general office areas and remove wastepaper.
- Dust the interior of waste receptacles in the general office areas.
- Wash clean water fountains and coolers in the general office areas.
- Sweep private stairways within the premises.
- Sweep and wash (using disinfectant) all floors in the base building lavatories that are located in the Building core.
- Wash and polish mirrors, shelves, bright work and enameled surfaces in the base building lavatories that
  are located in the Building core.
- Wash and disinfect basins, bowls and urinals in the base building lavatories that are located in the Building core.
- Wash toilet seats in the base building lavatories that are located in the Building core.
- Hand dust and clean all partitions, tile walls, dispensers and receptacles in the base building lavatories that are located in the Building core.
- Empty paper receptacles and remove wastepaper in the base building lavatories that are located in the Building core.
- Fill toilet tissue holders in the base building lavatories that are located in the Building core.
- Empty and clean sanitary disposal receptacles in the base building lavatories that are located in the Building core.

#### WEEKLY

- Vacuum clean carpeting and rugs in the general office areas without moving heavy furniture (such as
  desks, file cabinets, computer stands, and sofas).
- Dust door louvres and other ventilating louvres that are within reach of the cleaning staff without ladders.
- Wipe clean bright work.

#### QUARTERLY

- High dust the Premises, including the following:
  - Dust pictures, frames, charts, graphs and similar wall hangings that are not reached in nightly or weekly cleaning.
  - Dust clean vertical surfaces, such as walls, partitions, doors and door bucks and other surfaces not reached in nightly or weekly cleaning.
  - Dust pipes, ventilating and air-conditioning louvers, ducts, high moldings and other high areas not reached in nightly or weekly cleaning.
  - Dust Venetian blinds.

#### ADDITIONAL SERVICES

• Wash the exterior of windows periodically, subject to weather conditions and Requirements.

Exhibit "4.9"

## Class "E" System Connection Points

## Exhibit "6.1(B)"

## Landlord's Work

Removal of the construction shanties that are currently located on Floors 22 and 23 of the Premises, repair any damage caused by such removal, and deliver such floors in broom-clean condition.

## Exhibit "7.3(F)"

## Alterations Approved in Concept

- 1. The installation of interconnecting stairways in the Premises.
- 2. The installation of a security system in the Premises, including the installation of proximity card readers at locations on each floor of the Premises including in the portion of the Fire Stairs to which Tenant has access pursuant to the Lease.

Exhibit "7.13(A)"

# Standby Generator Specifications

## Exhibit "7.15"

# Location of Supplemental HVAC System

## Exhibit "8.2"

## <u>Distribution Portion of Building Systems in Premises</u>

## Exhibit "13.1"

## Condominium Non-Disturbance Agreement

## Exhibit "13.9"

# Mortgages and Superior Leases

Amended, Restated and Consolidated Mortgage, Security Agreement, Financing Statement and Assignment of Leases, Rents and Security Deposits, dated as of February 13, 2004, between 731 Office One LLC, as borrower, and German American Capital Corporation, as lender.

Exhibit "17.11"

Recognition Agreement

Exhibit "18.1"

Option Space

#### Exhibit "19.2"

#### Partial Renewal Space

The Partial Renewal Space shall consist of:

- (I) the entire Rentable Area of three (3), four (4), or five (5) floors of the Building leased by Tenant in the Office Unit, provided that (w) Tenant leases the entire Rentable Area of each such floor, (x) such floors are vertically contiguous to one another; (y) such floors constitute the uppermost or lowermost floors of the Premises leased by Tenant in the Office Unit, and (z) if one of such floors constitutes the twenty-eighth (28<sup>th</sup>) floor of the Building, then the Partial Renewal Space must include the entire Rentable Area of any portion of the twenty-ninth (29<sup>th</sup>) floor of the Building leased by Tenant;
- (II) if (x) Tenant renews any portion of the Premises pursuant to clause (I) above, and (y) Tenant elects to renew the entire Option Space included in Office Unit One that has theretofore been leased by Tenant, the entire Option Space theretofore leased in Office Unit One; and
- (III) if (x) Tenant renews any portion of the Premises pursuant to clause (I) above, and (y) Tenant elects to renew the Storage Space and/or the Mailroom Space, the entire Rentable Area of the Storage Space and/or the Mailroom Space.

#### ASSIGNMENT AND ASSUMPTION AND CONSENT AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AND CONSENT AGREEMENT (this "<u>Agreement</u>") is dated as of the 25th day of March, 2009, among 731 OFFICE ONE LLC, a Delaware limited liability company, having an address c/o Alexander's Inc., 888 Seventh Avenue, New York, New York 10019 ("<u>Landlord</u>"), CITICORP NORTH AMERICA, INC., a Delaware corporation, having an address c/o City Realty Services, Two Court Square – 4th Floor, Long Island City, New York 11120 ("<u>Assignor</u>"), and BLOOMBERG L.P., a Delaware limited partnership, having an address at 731 Lexington Avenue, New York, New York 10022 ("<u>Assignee</u>").

#### WITNESSETH:

WHEREAS, pursuant to an Agreement of Lease (the "Original Lease"), dated as of February 7, 2005, between Landlord, as landlord, and Citibank, N.A. ("Citibank"), as tenant, Landlord did demise and let to Citibank, and Citibank did hire and take from Landlord, certain premises in the building that is known by the street address of 731 Lexington Avenue, New York, New York 10022, on the terms and subject to the conditions set forth therein;

WHEREAS, the Original Lease was amended pursuant to (i) a side letter agreement (the "<u>First Side Letter Agreement</u>"), dated February 7, 2005, between Landlord and Citibank, and (ii) a side letter agreement (the "<u>Second Side Letter Agreement</u>"), dated February 7, 2005, between Landlord, Citibank and Assignee (the Original Lease, as so amended by the First Side Letter and the Second Side Letter Agreement, being referred to herein as the "<u>Lease</u>");

WHEREAS, pursuant to the Assignment and Assumption Agreement (the "<u>Citibank Assignment Agreement</u>"), dated as of January 21, 2009, between Citibank, as assignor, and Assignor, as assignee, Citibank assigned all of Citibank's right, title and interest in and to the Lease to Assignor, and Assignor accepted such assignment and assumed all of the obligations of Citibank as tenant under the Lease:

WHEREAS, Assignor and Assignee have entered into an Amended and Restated Transfer and Escrow Agreement (the "<u>Transfer Agreement</u>"), dated as of the date hereof;

WHEREAS, Assignor desires to assign to Assignee all of Assignor's right, title and interest in and to the Lease and the documents (the "Other Lease Documents") set forth on Exhibit "A" attached hereto and made a part hereof, and Assignee desires to accept such assignment and assume all of Assignor's obligations under the Lease and the Other Lease Documents, in each case, to the extent accruing from and after the date hereof;

WHEREAS, the Lease requires Assignor to obtain the consent of Landlord to any such assignment and assumption of the Lease; and

WHEREAS, Landlord desires to so consent to such assignment and assumption of the Lease, on the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the premises and other good and valuable consideration, the mutual receipt and legal sufficiency of which the parties hereto hereby acknowledge, Landlord, Assignor and Assignee hereby agree as follows:

- 1. Definitions. All capitalized terms used herein without definition shall have the meanings ascribed thereto in the Lease.
- Assignment and Assumption.
- (a) Effective as of the date hereof, (i) Assignor hereby grants, assigns, transfers and conveys and sets over to Assignee all of Assignor's right, title and interest, as tenant, in, to and under the Lease, and (ii) Assignee hereby accepts such grant, assignment, transfer and conveyance from Assignor, and agrees to perform and fulfill all of the terms, covenants, and obligations that are imposed upon the tenant under the Lease, in each case, to the extent accruing from and after the date hereof.
- (b) Effective as of the date hereof, (i) Assignor hereby grants, assigns, transfers and conveys and sets over to Assignee all of Assignor's right, title and interest, in, to and under the Other Lease Documents, and (ii) Assignee hereby accepts such grant, assignment, transfer and conveyance from Assignor, and agrees to perform and fulfill all of the terms, covenants, and obligations that are imposed upon Assignor under the Other Lease Documents, in each case, to the extent accruing from and after the date hereof.
- 3. <u>Consent</u>. In reliance upon the agreements and representations contained in this Agreement, Landlord hereby consents to the aforementioned assignment and assumption of the Lease (the aforementioned assignment and assumption of the Lease being referred to herein as the "<u>Bloomberg Assignment</u>").
- 4. Representations and Warranties. Assignor and Assignee represent and warrant to Landlord that (i) no rent or other consideration is being paid or is payable to Assignor by Assignee for the right to use or occupy the space demised under the Lease (the "Premises") or for the use, sale or rental of Assignor's fixtures, leasehold improvements, equipment, furniture or other personal property in excess of the Fixed Rent and any additional rent payable pursuant to the Lease, (ii) a true, correct and complete copy of the Transfer Agreement is attached hereto as Exhibit "B" and made a part hereof, and (iii) the Transfer Agreement and this Agreement constitute the entire agreement of Assignor and Assignee with respect to the Bloomberg Assignment.
- 5. <u>No Waiver, Modification, Etc.</u> Neither Landlord's execution and delivery of this Agreement, nor any acceptance of rent or other consideration from Assignor or Assignee by Landlord or Landlord's agent, shall operate to or be construed to (a) modify, waive, impair, release or in any manner affect any of the provisions,

covenants, agreements, terms or conditions contained in the Lease or Assignor's or Assignee's obligations or liability thereunder, (b) waive any of Assignor's or Assignee's breach or violation of any provision of the Lease or any rights or remedies of Landlord against any person, firm, association or corporation liable or responsible for the performance thereof, (c) enlarge or increase Landlord's obligations or diminish Landlord's rights under the Lease or otherwise, or (d) enlarge or increase any of Assignor's or Assignee's obligations or rights or diminish Assignor's or Assignee's obligations or rights under the Lease or otherwise; and all provisions, covenants, agreements, terms, rights and conditions of the Lease are hereby ratified and affirmed. Except as provided in Paragraph 3 hereof, Landlord's consent to the Bloomberg Assignment shall not constitute a consent to any sale, assignment, transfer or disposition of Assignor's or Assignee's interests or rights under the Lease, including, without limitation, any assignment of the tenant's interest under the Lease from Assignee to Assignor, or a subleasing of the Premises (or any portion thereof) thereunder, and, except as expressly permitted under the Lease, no such sale, assignment, transfer, disposition, or subleasing shall be made without the prior written approval of Landlord pursuant to and in accordance with the provisions of the Lease.

- 6. <u>No Approval of Transfer Agreement</u>. Nothing contained herein shall be construed as a consent to, or approval of, or ratification by Landlord of any of the terms of the Transfer Agreement, or as a representation or warranty by Landlord (except as expressly provided herein). Landlord has not, and shall not, review or pass upon any of the specific terms of the Transfer Agreement and shall not be bound or estopped in any way by the specific terms of the Transfer Agreement. Landlord's consent to the Bloomberg Assignment shall not constitute consent to the performance of Alterations or to any other matter that may be referred to or contemplated by the Transfer Agreement.
- 7. <u>Continuing Liability.</u> Nothing contained in this Agreement shall be construed as relieving or releasing the Assignor from any of its obligations under the Lease, and it is expressly understood that Assignor shall remain liable for such obligations notwithstanding subsequent assignment(s), sublease(s) or transfer(s) of the interest of the Assignee under the Lease, as provided in Section 17.1(G) of the Lease.
- 8. <u>Costs For Consent</u>. Assignor acknowledges and agrees that in connection with the Bloomberg Assignment Landlord has incurred costs in an amount (the "<u>Assignment Amount</u>") equal to Seventy-Nine Thousand Five Hundred Eighty-Six and 21/100 Dollars (\$79,586.21). Assignor agrees to pay the Assignment Amount to Landlord within ten (10) days after the date hereof.
- 9. <u>Defaults</u>. Assignor and Assignee acknowledge and agree that a material default of Assignor or Assignee hereunder that continues for twenty (20) days after Landlord has notified Assignor or Assignee, as the case may be, of such material default shall constitute an Event of Default under the Lease and Landlord shall have the right and may purse all of the rights, powers and remedies provided for in the Lease or at law or in equity or by statute or otherwise with respect to defaults, provided, however, if such material default cannot be cured within such twenty (20) day period, and Assignor or Assignee, as the case may be, notifies Landlord that it wishes to extend its cure period, then Landlord shall allow Assignor or Assignee, as the case may be, such

additional reasonable period of time as is required to cure the same so long as such cure has been commenced within such twenty (20) day period and is being diligently pursued to completion.

- 10. <u>Entire Agreement</u>. This Agreement contains the entire agreement of Landlord, on the one hand, and Assignor and Assignee, on the other hand, with respect to the matters contained herein and may not be modified, amended or otherwise changed except by a written instrument signed by the parties sought to be bound.
- 11. <u>Governing Law</u>. This Agreement shall for all purposes be construed in accordance with, and governed by, the laws of the State of New York.
- 12. Broker. Assignor and Assignee each represent and warrant to Landlord that it has not dealt with any broker in connection with this Agreement, other than CB Richard Ellis, Inc. ("Broker"). The execution and delivery of this Agreement by Landlord shall be conclusive evidence that Landlord has relied upon the foregoing representation and warranty of Assignor and Assignee. Assignor shall indemnify and hold Landlord harmless from and against any and all claims for commission, fee or other compensation by any person who shall claim to have dealt with Assignor in connection with this Agreement (including Broker), and for any and all costs incurred by Landlord in connection with such claims, including, without limitation, reasonable attorneys' fees and disbursements. Assignor shall be entitled, at its sole cost and expense, to defend any such claim with counsel reasonably approved by Landlord, and to settle any such claim at Assignor's sole cost and expense. Assignee shall indemnify and hold Landlord harmless from and against any and all claims for commission, fee or other compensation by any person who shall claim to have dealt with Assignee in connection with this Agreement (including Broker), and for any and all costs incurred by Landlord in connection with such claims, including, without limitation, reasonable attorneys' fees and disbursements. Assignee shall be entitled, at its sole cost and expense, to defend any such claim with counsel reasonably approved by Landlord, and to settle any such claim at Assignee's sole cost and expense. The provisions of this Paragraph 12 shall survive the cancellation or expiration of the Lease, and this Agreement.

#### 13. Miscellaneous.

- (a) Subject to the limitations set forth in the Lease, each right and remedy of Landlord provided for in this Agreement or in the Lease shall be cumulative and shall be in addition to every other right and remedy provided for herein and therein or now or hereafter existing at law or in equity or by statute or otherwise, and the exercise or commencing of the exercise by Landlord of any one or more of the rights or remedies so provided for or existing shall not preclude the simultaneous or subsequent exercise by Landlord of any or all other rights or remedies so provided for or so existing.
- (b) The terms and provisions of this Agreement shall bind and inure to the benefit of the parties hereto and their respective successors and assigns except that no violation of the provisions of Paragraph 5 hereof shall operate to vest any rights in any successor or assignee of Assignor or Assignee.

(c) If any one or more of the provisions contained in this Agreement shall be invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby.
(d) The captions contained in this Agreement are for convenience only and shall in no way define, limit or extend the scope or intent of this Agreement or any particular paragraph hereof, nor shall such captions affect the construction hereof.
(e) This Agreement may be executed in several counterparts and transmitted by facsimile or e-mail, each of which shall be deemed an original but all of which shall constitute one and the same agreement.
(f) Assigner and Assigned each represents and warrants that each has full right, nower and authority to enter into this

- (f) Assignor and Assignee each represents and warrants that each has full right, power and authority to enter into this Agreement and that the person or persons executing this Agreement on behalf of Assignor or Assignee, as the case may be, are duly authorized to do so.
- (g) Landlord represents and warrants to Assignor and Assignee that (i) it is the owner of a fee simple interest in the condominium unit of which the Premises is a part, (ii) it has full right, power and authority to enter into this Agreement and that the person or persons executing this Agreement on behalf of Landlord are duly authorized to do so, and (iii) to the actual knowledge of Landlord, neither Assignor nor Citibank is in default under the Lease, and no event exists which, but for the giving of notice or the passage of time, would constitute an Event of Default by Citibank or Assignor under the Lease.
- (h) Assignor and Assignee represent and warrant to Landlord that to the actual knowledge of Assignor and Assignee, Landlord is not in default under the Lease and no event exists which, but for the giving of notice or the passage of time, would constitute an Event of Default by Landlord under the Lease.
- (i) This Agreement is offered for signature by Assignor and Assignee and it is understood that this Agreement shall not be binding upon Landlord unless and until Landlord shall have executed and delivered a copy of this Agreement to both Assignor and Assignee.
- (j) In accordance with Section 28.1 of the Original Lease, any notice, demands, requests or other communications given or required to be given under the Lease to Tenant by Landlord shall be delivered to Tenant at the following address: Bloomberg L.P., 731 Lexington Avenue, New York, New York 10022, Attention: Peter Smith, Director of Global Leasing.
- (k) Landlord acknowledges that on the date hereof (or promptly thereafter) Assignor and Assignee shall execute and deliver that certain Memorandum of Assignment in the form attached hereto as <a href="Exhibit" "C"</a> and duly submit same (or cause same to be duly submitted) for recording with the Office of the New York City

Register, New York County.

IN WITNESS WHEREOF, Landlord, Assignor and Assignee have respectively executed this Agreement as of the 25th day of March, 2009.

LANDLORD:
731 OFFICE ONE LLC
By: 731 Office One Holding LLC, member
By: Alexander's Inc., member
Ву:
Name: Title:
ASSIGNOR:
CITICORP NORTH AMERICA, INC.
Ву:
Name: Title:
The state of the s
ASSIGNEE:
BLOOMBERG L.P.
By: Bloomberg Inc., its general partner
Ву:
Name: Title:

#### Exhibit "A"

#### Other Lease Documents

- Condominium Recognition and Attornment Agreement with The Board of Managers of the Beacon Court Condominium dated as of February 7, 2005
- 2. Subordination, Nondisturbance and Attornment Agreement with Wells Fargo Bank National Association as trustee dated as of February 7,
- 3. Letter regarding parking between 150 East 58<sup>th</sup> Street LLC and Citibank dated as of February 7, 2005
- 4. Memorandum of Lease dated as of February 7, 2005
- 5. Assignment and Assumption Agreement by Citibank, N.A., as assignor, and Citicorp North America, Inc., as assignee
- 6. Memorandum of Assignment by Citibank, N.A., as assignor, and Citicorp North America, Inc., as assignee

Exhibit "B"

Transfer Agreement

(See attached)

## Exhibit "C"

# FORM OF MEMORANDUM OF ASSIGNMENT OF LEASE

#### MEMORANDUM OF ASSIGNMENT OF LEASE

THIS MEMORANDUM OF ASSIGNMENT OF LEASE (this "Memorandum") is made and entered into as of 25th, 2009, by and between CITICORP NORTH AMERICA, INC., a national banking association ("Assignor"), and BLOOMBERG L.P., a Delaware limited partnership ("Assignee").

#### WITNESSETH:

WHEREAS, 731 Office One LLC, a Delaware limited liability company ("<u>Landlord</u>") and Assignor entered into that certain Lease dated as of February 7, 2005 (as the same may be modified, amended or supplemented from time to time, the "Lease");

WHEREAS, the Lease pertains to certain premises located at Block 1313, Lot 1003 in the City of New York, County of New York, State of New York (the "Premises");

WHEREAS, pursuant to that certain Assignment and Assumption and Consent Agreement dated as of 25th, 2009 between Assignor, Assignee and Landlord (the "<u>Assignment</u>"), Assignor has assigned to Assignee, and Assignee has assumed from Assignor, all of Assignor's rights and obligations under the Lease;

WHEREAS, pursuant to the Assignment Landlord has consented to the Assignment per the terms of the Lease; and

WHEREAS, Assignor and Assignee desire to submit this Memorandum for recording with the Office of the New York City Register, New York County to evidence the Assignment by the recitations contained in this Memorandum.

NOW, THEREFORE, Assignor and Assignee affirm that:

- 1. Capitalized items not otherwise defined herein shall have the meaning given such terms in the Assignment.
- 2. Assignee is the Tenant (as defined in the Lease) under the Lease and all notices required to be sent to Tenant under the Lease shall be sent to Assignee at the following address:

BLOOMBERG L.P. 731 Lexington Avenue New York, New York 10022 Attention: Peter Smith, Director of Global Leasing

3. Assignor has certain rights of reassignment with respect to the Lease, which are (i) more particularly set forth in that certain Amended and Restated Transfer and Escrow Agreement between Assignor and Assignee, dated as of March 25th, 2009 and are (ii) subject, in all respects, to, among other things, the terms of the Lease and Landlord's prior written consent thereunder.

4.	This Memorandum is subject to all conditions, terms and provisions of the Lease and the Assignment, which agreements are hereby		
adopted and made a part hereof by reference in the same manner as if all the provisions thereof were copied herein in full.			

5. In the event of a conflict between the terms of the Lease or the Assignment, on the one hand, and this Memorandum, on the other hand, the Lease or the Assignment, as the case may be, shall prevail. Reference should be made to the Lease and the Assignment for a more detailed description of all matters contained in this Memorandum.

[SIGNATURES TO FOLLOW ON THE NEXT PAGE]

IN WITNESS WHEREOF, Assignor and Assignee have respectively executed and delivered this Memorandum of Assignment of Lease as o date first above written.		
CITICORP NORTH AMERICA, INC., Assignor		

Ву:		
Name:		
Title:		
BLOOMBERG L.P., Assignee		
y: Bloomberg Inc., its general partner		
Ву:		
Name:		
Title:		

STATE OF NEW YORK COUNTY OF NEW YORK	) ) ss.: )
name is subscribed to the within	
	Notary public
STATE OF NEW YORK COUNTY OF NEW YORK	) ) ss.: )
name is subscribed to the within	
	Notary public